## The International Political Economy of Electricity Markets:

#### **Cross-Border Electricity Trade & Interconnections in East Asia**

Clare Garrigus Richardson-Barlow

Submitted in accordance with the requirements for the degree of Doctor of Philosophy

The University of Leeds East Asian Studies / Sustainability Research Institute School of Languages, Cultures & Societies / School of Earth & Environment

June 2020

The candidate confirms that the work submitted is his/her own and that appropriate credit has been given where reference has been made to the work of others.

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

The right of Clare Garrigus Richardson-Barlow to be identified as Author of this work has been asserted by her in accordance with the Copyright, Designs and Patents Act 1988.

© 2020 The University of Leeds and Clare Garrigus Richardson-Barlow

#### Acknowledgements

I would like to acknowledge and thank the University of Leeds for providing the necessary funding for this research to be carried out: The Leeds International Research Scholarship, Energy Leeds, and East Asian Studies at the University of Leeds all provided generous funding. Other institutions provided travel support and opportunities to receive feedback, to which I am forever grateful: The National Bureau of Asian Research, the Center for Strategic & International Studies, the Renewable Energy Network for the 21<sup>st</sup> Century, the European Research School Network of Contemporary East Asian Studies, and the ASEAN Centre for Energy.

I am indebted to the great many experts that assisted and motivated this work: My supervisors, Dr. Caroline Rose, Dr. Stephen Hall, and Dr. James Van Alstine have provided encouragement and knowledge that cannot be equalled. This Ph.D. dissertation would not be possible without the interview subjects that volunteered their time to speak with me; I am so thankful for their insight, expertise, and patience. Many academics across the Department of East Asian Studies in the School of Languages, Cultures & Societies (LCS) and the Sustainability Research Institute (SRI) in the School of Earth & Environment helped along the way: Dr. Kweku Ampiah provided support early on in the research design, academics in LCS provided advice regarding research themes post-fieldwork, and the SRI Energy & Climate Change Mitigation research group led by Dr. Katy Roelich provided constructive criticism during the data analysis. Diane Yates in LCS provided a healthy and safe space for this research to take place. Christine Richardson-Barlow offered her English expertise and heartfelt support when I was lost. And Myrna gave me a welcome distraction when it was needed most. Additional, unnamed academics and experts inspired, supported, and guided this work directly and indirectly from inception to completion.

Fellow students, both PhD colleagues and undergraduate pupils, have motivated me throughout this process. Their enthusiasm has been contagious.

My husband, William Schaffer, was a cheerleader and voice for this research when I was unable to be. Without him I would not still be standing. The completion of this work is his achievement as much as my own.

Dr. Jeffrey G. Barlow encouraged this work before it began and he left before it was completed. His impact on my interest in East Asian Studies cannot be overstated. He is forever a part of this dissertation, written into the following pages and everything that comes next.

#### Abstract

This research examines the extent to which climate change has created imperatives for cross-border electricity trade and power sector reform in East Asia; this will include assessing the role of the dominant sub-regional governance structure, the Association of Southeast Asian Nations (ASEAN), in progressing energy market integration (EMI) efforts among member economies. This research argues there are gaps in current literature on East Asian studies development paradigm, developmental statism (DS), doesn't accurately account for neoliberal market engagement; technically, the current literature argues for fully liberalised power sectors in order to engage in cross-border interconnections, which is not reflected in market realities. This research will address these gaps using a combination of elite interviews and market analysis, resulting in an updated, East Asian studies-based interpretation of the international political economy (IPE) of Southeast Asian cross-border electricity trade.

This research includes: analysis of the background and development patterns in East Asian countries and the Southeast Asian sub-region; focus on the evolution of DS and creation of neo-developmental statism as an explanatory tool of subregional IPE; an explanation of the role for electricity trade given global and subregional climate goals and contradictions between these goals and national policy. This research expounds on limits to ASEAN EMI using a case study analysis of the Brunei-Indonesia-Malaysia-Philippines East Asia Growth Area (BIMP-EAGA). Three major themes emerged in data analysis: importance of subregional market factors, primacy of national market factors in progressing subregional reforms, and governance challenges to EMI and increased interconnections. These themes reveal findings unique to this research: ASEAN EMI targets don't align with national efforts; national EMI efforts appear driven by economic not climate incentives; engagement in global climate commitments remains performative, reflecting sub-regional priority to engage in the global, neoliberal market system; finally, neo-developmental statism offers an evolved development paradigm for understanding the IPE of cross-border electricity trade.

**Key Words:** ASEAN; ASEAN Power Grid; Cross-border electricity trade; crossborder interconnections; Developmental Statism; Economic development; Electricity Market Integration; Neoliberalism;

Acknowledgements	iii
Abstract	iv
Table of Contents	v
List of Abbreviations	ix
List of Tables	xii
List of Figures	.xiii
Preface: Notes on Style and Referencing	. xiv
Chapter 1 Introduction: Global Climate Challenges & Regional Solutions	1
1.1 Introduction	1
1.2 Research Rationale	6
1.2.a Economic Development & Climate Change Imperatives in East Asia	6
1.2.b Green Growth	9
1.2.c Clean Energy	11
1.2.d Electricity Markets & Regional Energy Mix	15
1.3 Research Aims	21
1.3.a Theoretical Framework	22
1.4 Study Contributions	25
1.4.a Empirical Contributions	26
1.4.b Methodological Contribution	27
1.4.c Theoretical Contributions	28
1.5 Thesis Structure	29
1.5.a Objectives & Questions	29
1.5.b Chapter Overviews	31
1.6 Conclusions	34
Chapter 2: Cross-Border Electricity Trade: Conceptual Approach & Review of Literature	35
2.1 Introduction	35
2.2 East Asia's Growth-Environment Nexus	36
2.2.a Sub-Regional Imperatives for Cross-Border Electricity Trade	42
2.3 Cross-Border Electricity Trade in Southeast Asia	49
2.3.a Electricity Market Integration	49

## **Table of Contents**

2.3.b Governance of Cross-Border Electricity Trade	53
2.3.c ASEAN Governance	56
2.3.d Cross-border Electricity Trade Solutions in ASEAN	60
2.4 International Political Economy & Foundations of Trade	64
2.4.a IPE Foundations: Neoliberalism & the World Economic Order	65
2.5 International Political Economy in East Asia	69
2.5.a IPE Foundations: Developmental Statism	71
2.5.b Criticisms: DS Policy	74
2.5.c Neo-Developmental Statism	79
2.6 Conclusions	85
Chapter 3 Research Design & Plan of Enquiry	87
3.1 Introduction	87
3.2 Methodological Approach (WC: 2641)	89
3.2.a Critical Realism	93
3.3 Research Design	97
3.3.a Case Study Design	98
3.3.b Elite & Stakeholder Interviews	103
3.3.c Reliability & Validity	105
3.3.d Ethical Considerations	108
3.3.e Data Protection	110
3.4 Methods: Data Collection & Analysis	111
3.4.a Preparing for Data Collection	111
3.4.b Documentary Data	112
3.4.c Interview Data	113
3.4.d Data Analysis: Interview Coding	117
3.4.e Data Analysis: Thematic Analysis	118
3.5 Positionality of Research	120
3.6 Conclusions	122
Chapter 4 Sub-Regional Market Factors	126
4.1 Introduction	126
4.2 Sub-Regional Interconnections: ASEAN	127
4.3 Subsystem Interconnections: BIMP-EAGA	136
4.4 Discussion	141
4.4.a National vs. Sub-Regional	142

	4.5 Conclusions	145
Chaj	pter 5 National Market Factors	.147
	5.1 Introduction	147
	5.2 Development of National Electricity Markets	148
	5.2.a Current Market Structures: Overview	149
	5.2.b BIMP-EAGA	150
	Brunei Darussalam	. 150
	Indonesia	. 152
	Malaysia	. 154
	Philippines	. 155
	5.3 Oseni & Pollitt Criteria	157
	5.4 Reform & Assessment	170
	5.4.a Market Liberalisation	173
	5.4.b Coordination, Harmonisation & Standardisation	177
	5.4.c Subsidies	180
	5.5 Discussion	185
	5.5.a National Priorities	186
	5.5.b Neo-Developmental Statism & National Market Factors	190
	5.6 Conclusions	195
Chaj	pter 6 Governance Challenges	.197
	6.1 Introduction	197
	6.2 Sub-Regional versus National Interests	198
	6.2.a Sub-Regional Interests	199
	6.2.b National Interests	204
	6.3 Trust Among ASEAN Members	208
	6.3.a Mistrust Moving Actors to Trust	213
	6.3.b Bilateral vs. Multilateral Cooperation	215
	Existing Interconnections	. 216
	6.4 Trust of Outsiders	219
	6.4.a Chinese Investment through the Belt & Road Initiative (BRI)	220
	6.4.b ASEAN-China Bilateral Relations	226
	6.5 ASEAN Power Grid (APG): Status	229
	6.6 Discussion	231

	232
6.6.b Rural vs Urban	233
6.6.c A Comprehensive, Neo-Developmental Statism	
Explanation2	236
6.7 Conclusions	239
Chapter 7 Conclusions2	241
7.1 Introduction	241
7.2 Summary of the Thesis	241
7.3 Original Contributions	246
7.4 Research Limitations	248
7.5 Suggestions for Further Study	249
7.5.a Water Policy in the GMSR	249
7.5.b China & BRI	250
7.5.c Moral IPE Considerations	251
7.5.d Grid-Connected Distributed Generation	253
7.6 Policy Implications	254
7.7 Concluding Remarks	256
Bibliography2	259
Bibliography2 Appendix A Interview Codes	259 310
Bibliography2 Appendix A Interview Codes	259 810 811
Bibliography	259 810 811 813
Bibliography	259 810 811 813 813
Bibliography       2         Appendix A Interview Codes       3         Appendix B Ethical Approval       3         Appendix C Interview Protocol       3         Contextual Protocol: Research Objectives       3         Introductory Protocol       3	<ul> <li>259</li> <li>810</li> <li>811</li> <li>813</li> <li>313</li> <li>313</li> </ul>
Bibliography       2         Appendix A Interview Codes       3         Appendix B Ethical Approval       3         Appendix C Interview Protocol       3         Contextual Protocol: Research Objectives       3         Introductory Protocol       3         Background Protocol       3	<ul> <li>259</li> <li>810</li> <li>811</li> <li>813</li> <li>313</li> <li>313</li> <li>314</li> </ul>
Bibliography       2         Appendix A Interview Codes       3         Appendix B Ethical Approval       3         Appendix C Interview Protocol       3         Contextual Protocol: Research Objectives       3         Introductory Protocol       3         Background Protocol       3         Conversational or Transition Questions: (Pick and choose based on situation)       3	<ul> <li>259</li> <li>310</li> <li>313</li> <li>313</li> <li>314</li> <li>315</li> </ul>
Bibliography       2         Appendix A Interview Codes       3         Appendix B Ethical Approval       3         Appendix C Interview Protocol       3         Contextual Protocol: Research Objectives       3         Introductory Protocol       3         Background Protocol       3         Conversational or Transition Questions: (Pick and choose based on situation)       3         Key Questions: Electricity Markets:       3	<ul> <li>259</li> <li>310</li> <li>311</li> <li>313</li> <li>313</li> <li>314</li> <li>315</li> <li>315</li> <li>315</li> </ul>
Bibliography       2         Appendix A Interview Codes       3         Appendix B Ethical Approval       3         Appendix C Interview Protocol       3         Contextual Protocol: Research Objectives       3         Introductory Protocol       3         Background Protocol       3         Conversational or Transition Questions: (Pick and choose based on situation)       3         Key Questions: Electricity Markets:       3         Key Questions: Electricity Markets & East Asia / Europe:       3	<ul> <li>259</li> <li>310</li> <li>311</li> <li>313</li> <li>313</li> <li>314</li> <li>315</li> <li>315</li> <li>316</li> </ul>
Bibliography       2         Appendix A Interview Codes       3         Appendix B Ethical Approval       3         Appendix C Interview Protocol       3         Contextual Protocol: Research Objectives       3         Introductory Protocol       3         Background Protocol       3         Conversational or Transition Questions: (Pick and choose based on situation)       3         Key Questions: Electricity Markets:       3         Key Questions: Electricity Markets & East Asia / Europe:       3         Key Questions: Assessment       3	<ul> <li>259</li> <li>310</li> <li>311</li> <li>313</li> <li>313</li> <li>314</li> <li>315</li> <li>315</li> <li>316</li> <li>316</li> <li>316</li> </ul>
Bibliography       2         Appendix A Interview Codes.       3         Appendix B Ethical Approval.       3         Appendix C Interview Protocol.       3         Contextual Protocol: Research Objectives       3         Introductory Protocol.       3         Background Protocol.       3         Conversational or Transition Questions: (Pick and choose based on situation).       3         Key Questions: Electricity Markets:       3         Key Questions: Electricity Markets & East Asia / Europe:       3         Key Questions: Assessment.       3         Key Questions: Institution/Department/Discipline       3	<ul> <li>259</li> <li>310</li> <li>311</li> <li>313</li> <li>313</li> <li>314</li> <li>315</li> <li>316</li> <li>316</li> <li>317</li> </ul>
Bibliography       2         Appendix A Interview Codes       3         Appendix B Ethical Approval       3         Appendix C Interview Protocol       3         Contextual Protocol: Research Objectives       3         Introductory Protocol       3         Background Protocol       3         Conversational or Transition Questions: (Pick and choose based on situation)       3         Key Questions: Electricity Markets:       3         Key Questions: Electricity Markets & East Asia / Europe:       3         Key Questions: Institution/Department/Discipline       3         Key Questions: Institution/Department/Discipline       3         Concluding Questions / Post Interview Comments/Considerations       3	<ul> <li>259</li> <li>310</li> <li>311</li> <li>313</li> <li>313</li> <li>314</li> <li>315</li> <li>316</li> <li>316</li> <li>317</li> <li>317</li> </ul>

# List of Abbreviations

AAG	Australian Asian [Power] Grid
ACE	ASEAN Centre for Energy
ADB	Asian Development Bank
ADBI	Asian Development Bank Institute
AFC	Asian Financial Crisis
AIMS	ASEAN Interconnection Master Plan Study
APAEC	ASEAN Plan of Action for Energy Cooperation
APEC	Asia Pacific Economic Cooperation
APG	ASEAN Power Grid
ASEAN	Association of Southeast Asian Nations
BIMP-EAGA	Brunei Darussalam-Indonesia-Malaysia-Philippines East Asia Growth Area
BRI	Belt & Road Initiative
CBDR	Common but Differentiated Responsibilities
CCD	Climate Compatible Development
CO <sub>2</sub>	Carbon Dioxide
DOE	Department of Energy (USA)
DS	Developmental Statism
EGS	Environmental Goods & Services
EIA	Energy Information Administration (USA)
EIU	Economist Intelligence Unit
EMI	Energy Market Integration
ЕМТ	Ecological Modernisation Theory
GDP	Gross Domestic Product

GE	Growth/Environment (Dilemma)
GEI	Global Energy Interconnection
GEIDCO	Global Energy Interconnection Development & Cooperation Organization
GMSR	Greater Mekong Sub-region (Mekong Sub-region)
HAPUA	Heads of ASEAN Power Utilities/Authorities
HPAEs	Highest Performing Asian Economies
ICTSD	International Centre for Trade & Sustainable Development
IEA	International Energy Agency
IMF	International Monetary Fund
IPE	International Political Economy
IPP	Independent Power Producers
IRENA	International Renewable Energy Agency
METI	Ministry of Economy, Trade & Industry
MPAC	Master Plan on ASEAN Connectivity
NAPOCOR	National Power Corporation (Philippines)
NDRC	National Development and Reform Commission (PRC)
NIE	Newly Industrialised Economies
NTS	Non-traditional Security
OECD	Organisation for Economic Cooperation & Development
REN21	Renewable Energy Network for the 21 <sup>st</sup> Century
ROK	Republic of Korea
RQ	Research Question(s)
SAPP	South African Power Pool

SDG Sustainable Development Goal(s)

- TSO Transmission System Operator
- UN United Nations
- UNCTAD United Nations Conference on Trade & Development
- UNDSD United Nations Division for Sustainable Development
- UNESCAP United Nations Economic & Social Commission for Asia & the Pacific
- UNFCCC United Nations Framework Convention on Climate Change
- UNIPCC United Nations Intergovernmental Panel on Climate Change
- VRE Variable Renewable Energy
- WCED World Commission on Environment & Development
- WHO World Health Organisation
- WTO World Trade Organisation

# List of Tables

Table 1.1 Percentage of electrification in urban and rural areas in         Southeast Asia (2017)	16
Table 1 2 Objectives Needs & Research Questions	20
Table 2.1 Pagic Sub Pagional Environmental Dicture	12
	F3
Table 2.2 Sub-Regional Electricity & Renewables Picture	15
Table 3.1 Research Plan   9	<b>}0</b>
Table 3.2 Objectives & Analysis Needs         9	<del>)</del> 2
Table 3.3 Case Study Design10	)0
Table 3.4 Sub-Questions12	24
Table 4.1 Sub-Regional Snapshot: BIMP-EAGA Renewable Potential13	38
Table 5.1 O&P EMI Criteria15	59
Table 5.2: BIMP-EAGA Scorecard16	52
Table 5.3: O&PCRB EMI Criteria         16	55
Table 5.4 O&PCRB BIMP-EAGA Scorecard16	57
Table 5.5: BIMP-EAGA Total Subsidies, 2020	32
Table 5.6 O&PCRB BIMP-EAGA Scorecard with Reform Level	36
Table 7.1 Consolidation of Research Aims	ł2

# List of Figures

Figure 1.1 Geographical Context	5
Figure 2.1 ASEAN Interconnection Subsystems	63
Figure 3.1 Interconnected Case Study Methods	97
Figure 3.2 Embedded Case Study	99
Figure 3.3 Interview Protocol Overview1	.16
Figure 3.4 Resulting Mind Map1	.20
Figure 4.1: An integrated approach to EMI in Southeast Asia1	.29
Figure 4.2 ASEAN Interconnections Summary1	.30
Figure 4.3 Forecast ASEAN Interconnections1	.31
Figure 4.4 Current ASEAN Interconnections1	.31
Figure 4.5 Summary of Key Sub-Regional Barriers1	.34
Figure 4.6 BIMP-EAGA Planned interconnections1	.37
Figure 4.7 BIMP-EAGA Actual Interconnections1	.39
Figure 5.1 Brunei Darussalam Electricity Industry Structure1	.52
Figure 5.2 Indonesia Electricity Industry Structure1	.54
Figure 5.3 Malaysia Electricity Industry Structure1	.55
Figure 5.4 Philippines Electricity Industry Structure1	.57
Figure 7.1 Policy Implications2	255

### **Preface: Notes on Style and Referencing**

This dissertation adheres to Leeds Harvard referencing style throughout. In-text citations consist of the author's last name followed by the year of publication. For works with more than three authors, the first author's name followed by "et al." is provided in text, with all authors listed in the bibliography. Leeds Harvard referencing style is also used in the citation of policy documents and reports issued by governments and international organisations. Since these institutions often have long names, full names followed by abbreviations have been used for the first in-text citation, followed by abbreviations in all subsequent uses. Abbreviations have been used in the bibliography. A full list of abbreviations referenced is included in *List of Abbreviations*.

In addition, this dissertation uses distinct codes to reference interview subjects and conferences or workshops where opinions or feedback was received. Codes are used in order to identify the expert's field, which is relevant to their expertise and viewpoints, as well as to simultaneously protect the identities of interview subjects. For interviews this code is organisation title followed by interview number, ranging from 1 to 32. For example: *Government Official 22.* For conferences and workshops where feedback was solicited and attendees agreed to anonymous reference, this code is event type followed by event number. For example: *Government Workshop 2.* All codes are listed in numeric order in text. A full list of interview codes is available in Appendix A.

# Chapter 1 Introduction: Global Climate Challenges & Regional Solutions

## **1.1 Introduction**

In 2020 there is an escalating global health crisis prompted by the novel coronavirus pandemic (Campbell and Doshi, 2020), while at the same time the United Kingdom wrangles over its post-Brexit economic and trade relationship with the European Union (Boffey, 2020). Economic growth in the global south has challenged the traditional global order previously dominated by western powers. The rise of nationalist and protectionist regimes across the world (Snyder, 2019) suggest the Washington Consensus has failed and a series of new development models have been emerging for many years amid a changing global order.

Where the Washington consensus was defined by trade liberalisation, openness to foreign direct investment, privatisation, fiscal discipline and market fundamentalism (Serra & Stiglitz, 2008), capitalism with Asian values (Robison, 1996) has tended towards much higher levels of state investment in infrastructure and government directed policymaking with a partial rejection of market competition prescriptions for all sectors of the economy (Leftwhich, 1995).

The financial and economic institutions of globalisation and trade have to navigate a more complex development terrain, where the conditionality of past international loans from the International Monetary Fund (IMF)/World Bank and recently, regional institutions such as the Asia Infrastructure Investment Bank (AIIB), are no longer legitimate (Koeberle et al., 2005). Trade policy must navigate the historical institutional development of large sunk investments, capital, and cultural attachments to institutions and place. This is certainly the case for the energy sector, in which many years of trying to apply liberalised trade policies and competition and market conditions have produced mixed results, partial privatisations, and a confusing tapestry of political economies of national energy and power systems (United Nations [UN], 2006; Owen et al., 2017).

In parallel, building scientific evidence on the urgency of climate action (UN Intergovernmental Panel on Climate Change [UNIPCC], 2007) and subsequent international agreement on national decarbonisation targets via the Paris

Agreement (EU, 2017) place pressure on both developed and developing economies to direct development in climate compatible ways. The United Nations (UN) Sustainable Development Goal (SDG) 7 prioritises sustainable energy for all (UN, 2015). SDG 7 recognises that reliable and affordable access to energy, particularly electricity, has consistent and proven positive developmental outcomes (UN, 2015). Access to clean and sustainable energy can deliver these developmental benefits whilst simultaneously removing harmful local pollutants and mitigating global greenhouse gas emissions and the destructive environmental impacts they create.

The East Asian region includes many of the dominant actors in the organised, global responses to climate change and economic development challenges, such as the SDGs (UN et al., 2017). East Asian economies have contributed significantly to the root causes of global climate change concerns, including increased air pollution and carbon dioxide (CO<sub>2</sub>) emissions; but they are also major global players in clean energy technology development, investment, and policy mechanisms aimed at green growth and sustainable development solutions (Renewable Energy Network for the 21<sup>st</sup> Century [REN21], 2019a). The economic development story that made East Asia one of the largest contributors to global emissions also made the region one of the dominant global contributors to policy responses and solutions.

The significance of addressing decarbonisation and renewable energy integration challenges, and of incorporating an evaluation of East Asian governance and economic policies as they relate to regional electricity markets, lies in the pressing need for coordinated, multi-level (global, subregional, regional, national) responses to climate change. Global economic growth has resulted in significant environmental and climate damage. This is due in part to a heavy reliance on fossil fuel energy sources and industrial and export driven development. East Asia is not alone in its obligation to address global climate challenges. However, as a major contributor to global emissions and emerging leader in world economic growth, technology innovation, and international trade, East Asia is in a unique position to contribute significantly to the global response. One way to do this is by strengthening renewable energy integration and electricity capacity through sub-regional and regional electricity trade and the buying and selling of excess capacity depending on energy needs. However, problems arise in policy coordination and coherence, particularly related to national policy measures that impact sub-regional and regional clean energy utilisation, and in turn have an impact on global markets, policy, and trade. Understanding the national and sub-regional linkages, and addressing any arising challenges, can positively impact East Asian clean energy development strategies and policy responses to climate change.

This research grew out of an initial desire to better understand the relationship between climate change mitigation, economic development, and trade practices in East Asia, with a focus on cross-border electricity trade. The topic of crossborder electricity trade is alive and well in East Asia, with much research interest in the physical and policy requirements to expand electricity interconnections<sup>1</sup> among neighbouring countries; The Asian Development Bank (ADB), Association of Southeast Asian Nations (ASEAN), the ASEAN Centre for Energy (ACE), Global Energy Interconnection Development and Cooperation Organization (GEIDCO), International Energy Agency (IEA), International Renewable Energy Agency (IRENA), REN21, and many other international research or policy organisations have research projects, initiatives, or whole centres devoted to the expansion of this topic. However, no research to date combines the issue of cross-border electricity trade with a discussion of the political economy of electricity markets in the East Asian economic development context. This research seeks to close that gap, adding context to energy market integration (EMI) discussions by explaining the nuances and challenges of electricity market reforms in a national economic and political environment that is dominated by state-directed policy, yet operating in a global system that is dominated by Bretton Woods notions of free markets and competition. Specifically, this research seeks to explain how East Asian nations, particularly those in the sub-region of Southeast Asia, view the imperative to construct and operate new power sector integration projects, and how the tapestry of competing interests (state, quasi state, private and corporate utilities) that form individual energy markets position themselves with regard to interconnection between markets. The aim to understand these challenges is achieved by exploring the following primary research questions (RQs):

1. How do governments understand the international political economy (IPE) of sub-regional cross-border interconnections in Southeast Asia?

<sup>&</sup>lt;sup>1</sup> Electricity Interconnections refer to the physical connections that allow the transfer of electricity across borders (IEA, 2019a,d). Interconnection enable cross-border electricity trade and the two are used frequently throughout this research to reference the act of transferring or trading electricity.

- 2. How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?
- 3. What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian development?

Further, in this research a thorough examination and critique of the dominant neoliberal political economy frameworks and their proposed application to East Asian economic & political structures will provide context to the development of sub-regional interconnectors that impact clean energy utilisation and subregional EMI. Significant attention will be paid to trade and industrial policy development in an East Asian context. Policies that make the best case for optimising clean energy's contribution in addressing global climate goals, and examples from the East Asian sub-region of Southeast Asia and the Brunei-Indonesia-Malaysia-Philippines East Asia Growth Area (BIMP-EAGA), will be utilised. An analysis of the political-economy of current electricity markets will identify drivers and barriers to cross-border electricity trade utilisation regionally, sub-regionally, and nationally.

The geographical focus of this research lies in the East Asian<sup>2</sup> sub-region of Southeast Asia, due to Southeast Asia's major cross-border electricity trade initiatives and broader, sub-region-wide goals of deeper energy market integration.<sup>3</sup> The collection of 10 ASEAN member states makes up the fastest growing sub- region in world, with the associated economic and environmental issues that come with rapid growth, making it a highly relevant place to examine

<sup>&</sup>lt;sup>2</sup> The definition of East Asia used in this context encompasses the two sub-regions of Northeast Asia (which is made up of the economies of Japan, The Democratic People's Republic of Korea or DPRK, the People's Republic of China—hereby referred to as China—The Republic of Korean or ROK, and Mongolia) and Southeast Asia (Brunei Darussalam, Cambodia, Indonesia, Lao PDR, Malaysia, Myanmar, Philippines, Singapore, Thailand, and Vietnam). For similar definitions see work by Dent (2008, 2014, 2016). This definition was chosen under the guidance of senior scholars in East Asian-oriented research and in line with definitions used in East Asian studies at the University of Leeds.

<sup>&</sup>lt;sup>3</sup> In this research 'region' is used to refer to the region of East Asia (see above definition, footnote 1) and 'sub-region' is used to refer to the sub-region of Southeast Asia. In addition, 'subsystem' is used to refer to the smaller, sub-regional groupings of countries in Southeast Asia, such as the Brunei-Indonesia-Malaysia-Philippines Growth Area (BIMP-EAGA) as it is defined and used by the Association of Southeast Asian Nations (ASEAN), the sub-regional governing body in Southeast Asia (ASEAN, 2004). ASEAN has 10 Southeast Asian member states, including: Brunei Darussalam, Cambodia, Indonesia, Lao PDR, Malaysia, Myanmar, Philippines, Singapore, Thailand, and Vietnam. The countries of East Timor and Papua New Guinea are ASEAN observer states and therefore not included despite their relevance in broader Southeast Asian energy security and EMI discussions.

EMI and the IPE of cross-border electricity trade. Over the period of 1997 to 2005 ASEAN experienced an average annual GDP growth rate of 7 percent (ASEAN, 2012b). The region is expected to continue to experience economic growth and prosperity, with a projection of 5.2% average real GDP growth from 2019-2023 (OECD, 2018b). ASEAN is the 7th largest global economy, and the IEA projects its economy will double in size by 2040, reaching roughly the size of China's economy in 2017 (IEA, 2017a). As a result of rapid economic and population growth, the region has experienced high growth in energy consumption and electricity access (Vithayasrichareon et al., 2012).

Unique to this research is the examination of EMI and cross-border electricity trade in East Asia at both the regional and sub-regional level—this has, unexpectedly resulted in additional analysis at the national level. Based on interviews done in Southeast Asia and among relevant parties, the focus of the case study selection will be only one of the identified subsystems and its relevant EMI initiative, BIMP-EAGA (This will be discussed in more detail in Chapter 3, *Research Design & Plan of Enquiry*). This geographical context can be visualised as follows:





A number of factors play a part in how this research has taken shape: the symbiotic relationship between economic development and climate change; the role of East Asia in global responses to climate change; and the significance of electricity markets in any coordinated policy response to the economic and political challenges associated with climate change. In the following sections

Source: Authors own creation

each of these issues will be explained, and their relevance to this research demonstrated.

### **1.2 Research Rationale**

This research is situated between two intersecting issue areas: (1) responses to the global fight against climate change and (2) the political economy of East Asia's regional economic development path. These two areas are interrelated via the energy policy solutions to hydrocarbon intensive economic development and its impact on the global climate. In this section the research rationale will be discussed in more detail, illustrating how the concept of cross-border electricity trade is conveniently situated between the twin challenges of climate change and economic development that are posed to the region of East Asia and the subregion of Southeast Asia. An overview of the link between economic development and climate change imperatives in East Asia, the concept of green growth, clean energy solutions, and electricity markets are helpful to understand for this research as they set the scene for a deeper discussion in the data chapters regarding the IPE of cross-border electricity trade and the market realities in East Asia. In this section these issues will be introduced in order to provide the necessary background for later analysis.

# 1.2.a Economic Development & Climate Change Imperatives in East Asia

East Asia has a complicated relationship with climate change—the region is one of the world's ascendant actors in the global fight against climate change. East Asia has significantly contributed to the deterioration of the global environment, climate, and air quality (REN21, 2019a); but it is also the site of increased alternative energy investment and development, as well as innovative clean energy policy mechanisms (ASEAN Centre for Energy [ACE], 2017). The same phenomenon that contributed to the global challenge of climate change—fossil fuelled economic development and industrialisation—has also made East Asia an important geographical locus of the mitigation of global emissions for preventing further degradation of the global climate.

The growing consumption of traditional hydrocarbon energy fuels (coal, gas, oil) as a result of global increases in electricity consumption has been linked to climate change as a result of the greenhouse effect, whereby greenhouse gas emissions, in particular CO<sub>2</sub>, are trapped in the earth's atmosphere, heating the surface of the earth at a temperature higher than normal and trapping the

warmth (International Energy Agency [IEA], 2012). It is widely accepted "that the global climate is warming, and we are to blame" (Evans 2012, p. 2). According to the Intergovernmental Panel on Climate Change (UN Intergovernmental Panel on Climate Change [UNIPCC], 2019), global greenhouse gas emissions have increased as a result of human activity, and the resulting changes in the earth's temperature are already having devastating effects on various regions of the globe.

ASEAN'S energy related CO<sub>2</sub> emissions have consistently grown over the period of 1971-2016, despite efficiency gains undertaken across the region (Sandu, et al., 2019, p. 11). As of 2018 ASEAN contributed 3.5% to global emissions; the IEA predicts that contribution could grow to 5% (IEA, 2019a). In addition, average temperatures have risen every year in Southeast Asia, and of the 20 countries predicted by the World Bank and the Global Climate Risk Index to be the most impacted by climate change, five of them are found in ASEAN (Prakash, 2018, p. 2). Climate change impacts by the sub-region and in the sub-region are serious.

The global and regional impacts of rising climate risks and growth in CO<sub>2</sub> emissions include altered ecosystems, financial repercussions, and development challenges—all of which are often the most dramatically felt among less developed communities (UNIPCC, 2007). Additionally, hydrocarbon extraction is also linked to environmental degradation from increased transportation, freshwater shortages, and waste as a result of the extraction and utilisation process. Reducing the use of hydrocarbons as a primary source of energy and as a dominant resource for secondary energy sources (i.e. electricity) holds potential answers to a variety of important climate and environmental problems. Clean energies are a means to realize very necessary changes in global energy use and electricity generation.

Recognition of the disastrous effects of climate change (Stern et al., 2007) has been driven by the nations of the European Union (pre-British exit from the European Union), and the primary signatories to the Kyoto Protocol – an agreement that commits signatories to a reduction in greenhouse gas emissions based on acknowledgment that global warming is real and is a result of human carbon emission activity (UNFCC, 2014). Built under the UN Framework Convention on Climate Change (UNFCCC), the internationally binding treaty places the majority of responsibility for  $CO_2$  emissions on developed economies, operating under the premise of common but differentiated responsibilities (CBDR) based on their history of industrialisation, economics, and capacity (Bauer et al., 2014). The 2015 COP21 Paris Agreement brought signatories together within the UNFCCC framework to reach climate goals, including emissions, mitigation, transparency and finance, from 2020-2050. This agreement is considered the largest, most comprehensive global climate agreement, with more than 190 signatories to date. The success of the COP 21 Paris Agreement depends on the ability of the signatory countries to not only meet emissions reduction targets, but also balance the challenges of economic growth, climate responsibility, and sustainable development (Stern, 2014).

The unprecedented economic growth and integration of East Asian economies that was viewed with awe from the 1970s onward has come with significant costs. Increased air pollution and a growing public health crisis have caught public attention and are having an important effect on public policy (Corfee-Morlot et al., 2009). Similar to other regions in the world, fossil fuel resources dominate the electricity generation of East Asian countries. The ADB estimates that under the current energy model CO<sub>2</sub> emissions will double by 2035 (ADB, 2013). The World Health Organisation (WHO) estimates that in 2012 around 7 million deaths were caused by air pollution, with roughly 2.7 million of those deaths in the Asia-Pacific, 1.3 million in China alone (WHO, 2014). WHO experts in East Asia believe that reductions in air pollution levels to WHO guidelines could prevent "37,000 early deaths in 27 cities" (WHO, 2014, p. 8). Air quality in the Asia-Pacific is increasingly becoming a global issue; the global community is well aware that air quality is a trans-boundary as well as local issue, and one that not just the largest emitters but their neighbours, allies, and economic partners must address.

The deterioration of air quality, environmental degradation as a result of over reliance on fossil fuels for electricity generation, industrialisation, and heavy industries, rising CO<sub>2</sub> emissions, and a growing range of energy security fears top the list of energy and environmental concerns in East Asia (REN21, 2019a). Public policy and research organisations throughout the Asia-Pacific are increasing their focus on synthesizing public policy to address CO<sub>2</sub> emissions in a variety of industries and policy areas (ACE, 2017). Governments are responding to such demands by increasing focus on alternative energy development, encouraging diversification of energy resources used for electricity generation, strengthening national electricity markets to account for higher shares of renewable energy integration, funding research and development into indigenous energy sources as a means to meet energy demand, circumventing supply disruptions, and tackling further environmental degradation (IRENA & ACE, 2016; REN21, 2019b).

The following sub-section describes East Asian green growth strategies as policy responses to global climate challenges, connecting the use of renewable energy to continued domestic economic development. In East Asia green growth has become an integral part of regional and sub-regional responses to climate change and is reflected in the policy initiatives within ASEAN, including those that utilise cross-border electricity trade. Following the discussion on green growth the author will connect three disparate areas of study that are relevant to this research—international political economy (IPE), sustainable development, and East Asian studies—to the larger research topic of electricity markets and cross-border electricity trade.

#### 1.2.b Green Growth

Green growth is a concept closely tied to global arguments for sustainable development and East Asian economic growth strategies in the age of climate change and environmental concerns. Understanding green growth and sustainable development arguments provides context to justification of this research, setting the scene for the examination of policy responses and necessities for climate change and power sector reform in East Asia and the sub-region of Southeast Asia.

Popularised in the 1970s, 'sustainable development' is a product of growing global awareness of environmental concerns and a hesitance to cede economic development to environmentalism<sup>4</sup> (Evans, 2012) in the age of the Anthropocene, or age of human development.<sup>5</sup> In the case of climate change and

<sup>&</sup>lt;sup>4</sup> The 1987 publication of the Brundtland Commission (formally known as *Our Common Future, From One Earth to One World: An Overview by the World Commission on Environment and Development* [WCED]) brought sustainable development to the environmentalist community as a discourse for describing the balance between economic development and the societal exploitation of resources (WCED, 1987). Like the other theoretical frameworks and discourses discussed later in this paper, sustainable development has adapted over the years, and it is used differently depending on the organisation and context. There are a variety of views on how sustainable development can or should be used and what policies and tools would be most helpful in reaching a long-term, sustainable development goals. For example, see: Agyeman, et al., 2003; Dryzek 2013; Heinberg, 2007; Huber 2000; Springett, 2013; World Bank, 2012; Zaccai, 2012. For the case of this research, sustainable development is taken as a widely understood discourse, and one that is used by the global policy community as a means to describe policy solutions to global climate and environmental concerns that allow for continued economic growth.

<sup>&</sup>lt;sup>5</sup> The Anthropocene refers to the age of humans, whereby the acceleration of human development has impacted nature (Roncancio et al., 2019). For further debates surrounding this epoch refer also to: Arias-Maldonado (2013); Decuypere, et al. (2019); Houston (2013); Johnson & Morehouse (2014); Knight (2015); Knight & Harrison (2014); among many others.

environmental policy, sustainable development is often used in reference to energy consumption, but it can also be used in reference to wildlife, minerals, land, etc. (Ayers et al., 2001). Sustainable development aims to guarantee "nondeclining per capita wealth by replacing or conserving the sources of that wealth, namely produced, human, social and natural capital" (Statistics Netherlands, 2013, p. 6). Sustainable development broadly conceived aims to achieve stability between opposing needs juxtaposed with an awareness of the economic, environmental, and societal limitations (Agyeman, et al., 2003). Within sustainability studies there are a number of issues that can be broken down, including the level of sustainability possible, ecological modernisation theory (EMT),<sup>6</sup> green growth, green economy, and others.

Green growth represents a variant of the same idea (United Nations Division for Sustainable Development [UNDSD], 2012), but positions the balance between environmental stewardship and economic growth solely within the framework of capital and economics, often within East Asia and the Pacific region, with emphasis on 'significant' environmental protection (Jacobs, 2013). Green growth is not an alternative to sustainable development, but a method for achieving it (Jacobs, 2013). Green growth seeks a similar environmental and economic balance as sustainable development, but also seeks to move beyond rhetoric and into specific economic policies that take advantages of and create synergies between environmental and economic concerns in order to eradicate poverty (UNDSD, 2012). Similar to sustainable development green growth is built into the economic development and poverty reduction discussions among many global multilateral institutions, including the ADB, UN, World Bank, and also the Korean-based Global Green Growth Institute, of which former UN Secretary General Ban Ki-Moon is in his second term as president and chair. Climate compatible development (CCD) has become an overarching goal that has been similarly written into global economic development strategies and institution mandates (Whitley, 2013a,b).

Green growth is a useful discourse in the context of clean energy use as it provides an alternative to traditional, finite hydrocarbon resources that power most economies and electricity markets (Owusu & Asumadu-Sarkodie, 2016).

<sup>&</sup>lt;sup>6</sup> EMT is discussed further in Chapter 2, *Conceptual Approach & Review of Literature*. Further discussion on sustainable development arguments and approaches is limited as they are out of the scope of this research, which aims to focus on the imperative for power sector reform in East Asia specifically as opposed to the variety of government and policy solutions available within sustainability approaches.

Whether or not green growth can actually allow economies to continue economic development at their current level and with a continued focus on industrialisation and improvements to quality of life (World Bank, 2012), but without damaging the environment through hydrocarbon use, is arguable (Vezzoli et al., 2018). However, it is this exact expectation that makes green growth an attractive policy consideration, as reflected by the incorporation of green growth strategies into many East Asian economic policies and energy systems; Clean energy development plans, clean industry strategies, and revised green growth driven development agendas are evident in a variety of East Asian economies (Organisation for Economic Cooperation & Development [OECD], 2017; REN21, 2019a). Unfortunately, many of the industrial structures and electricity markets of these economies still rely heavily on hydrocarbons that receive generous government subsidies (Phoumin & Kimura, 2015).<sup>7</sup> The challenge of sustainable development targets versus green growth realities at the national and sub-regional level is discussed at great length in this research in relation to renewable energy integration and sub-regional EMI, highlighting the challenge of realising these targets given conflicting priorities at multiple levels of governance. The following subsection will introduce clean energy use as a green growth policy response, leading into a primer on the role cross-border electricity trade can play in helping countries reach these clean energy goals.

### 1.2.c Clean Energy

Clean energy, energy with neutral emissions, often (but not always) from a renewable energy source, holds the potential to address two critical challenges facing East Asia (and the world)— (1) the dual eradication of energy poverty and quest for energy security, and (2) environmental and climate repercussions that occur as a result of traditional, fossil fuel-based energy consumption (REN21, 2019b). Clean energy provides multiple solutions to these two issues: clean energy technologies have a much lower environmental impact than fossil fuels; they offer longer term answers to energy needs than traditional, depletable fossil fuels; and clean energy technologies are not beholden to the same energy market price and supply fluctuations as are oil, gas, and coal (IRENA et al., 2018).

As a result of an increasing global focus on  $CO_2$  emissions and sustainable development programmes, the global market for clean energy has grown rapidly

<sup>&</sup>lt;sup>7</sup> The contradictions between sustainable development and modernist industrialism in East Asia and Southeast Asia have not been adequately addressed through policy mechanisms but will be discussed in greater length in chapters 4-6 of this research.

in the last decade, with new investment in renewable power and fuels reaching US\$2,378 billion in 2018 (REN21, 2019b). Globally, in 2018 the power sector experienced the growth in renewable energy capacity accounting for 75% of net additions to worldwide power generating capacity (IEA, 2019b). Global investment in new renewable power capacity is estimated to increase by 50% by 2024 (IEA, 2019b). Chinese investment in renewables has remained consistently high in recent years, with China regularly ranking as the regional and global leader (REN21, 2019, p. 93). In 2015, China contributed 36% of global investment in renewable energy, it invested more money in renewable energy than the entire continent of Europe, and China contributed 43% of the total growth in global renewable energy jobs (REN21, 2016).

Southeast Asia, on the other hand, has seen uneven investment trends, with renewable investments reflecting the resource capabilities of particular countries (ex: Thailand higher shares of solar PV investment, Cambodia more hydropower, etc.) and varying depending on the year (REN21, 2019, pp. 93-94). Barriers to investment in Southeast Asia are in part to blame for year-on-year investment discrepancies, with only a few regional financial institutions and development banks able to finance renewable projects (REN21, 2019, p. 93). Investment is not low because of lack of need or lack of targets, but a reflection of the market capabilities (p. 95). China's contribution to global and regional renewable energy development may trump all other nations in East Asia, however, opportunity in developing Southeast Asian markets is growing (REN21, 2019).

In Southeast Asia demand and emissions are expected to simultaneously grow in coming years, both as a result of economic and demographic growth (ACE, 2017, p.98): Power demand in ASEAN is projected to triple (to 640 GW) by 2040 (ACE, 2017) and greenhouse gas emissions double (2.3 billion metric tonnes) in the same time frame (IEA, 2017). Enhancing renewable energy use can augment both rises—increase electricity access and help reduce sub-regional emissions (ACE, 2017, p. 22).

Asia's region-wide contribution to global greenhouse gas emissions and the effect of the environment on its economic success and public health have, in part, driven the surge in clean energy investment and sustainable development policy solutions in the region, including the topic of this research, electricity market integration. East Asia, which has an abundant supply of renewable and clean energy resources spread throughout a variety of countries, suffers from uneven distribution, inflexible power sectors, and economic concerns that have

prevented full development and utilisation of these resources for electricity generation (Chang & Li, 2015). Policy measures in developed economies have largely encouraged the adoption of renewable technology through financial support and incentives, regulation and national mandates, and market mechanisms (REN21, 2019b). However, countries in general perceive (sometimes incorrectly) the costs of renewable energy growth as a near insurmountable barrier, necessitating the domestic development of cheaper technology and national policies that promote government investment to overcome existing investment barriers, and entrenched political and business interests.

Integrating clean energy sources into global power grids and markets is challenging for three main reasons: variability, uncertainty, and flexibility (Jones, 2014). This includes variability, uncertainty, and flexibility of resource supply, financial support, physical infrastructure, and policy mechanisms (IRENA, 2018a, pp. 11-13). Variability is the hour-by-hour, minute-by-minute availability of intermittent resources (Impram et al., 2020, p. 2); uncertainty is the longer-term questions regarding intermittency and difficulty predicting future weather and solar and wind power (p. 2). The existing solution to this is to build more flexibility into current systems (National Renewable Energy Laboratory [NREL], 2014, p. 2-3). Variability and uncertainty create the need for flexibility and the need for management through flexibility (NREL, 2014).

At the region-wide level, cross border electricity trading would aid in managing diversity in loads and resources, it would increase supply security and efficiency, could potentially reduce electricity prices, and also encourage further deployment of clean energy technologies (UN, 2006; Kunstýř & Mano, 2013; Chang & Li, 2015). In the case of traditional renewable energy fuels, such as solar and wind power, intermittency in supply poses challenges for larger increases in a country's energy mix. For example, wind power, which is created via the movement of wind turbines, is only available during times when the wind is blowing; Cross-border electricity trade would allow countries to trade excess power across interconnectors to countries that are experiencing power deficits due to intermittent resources, like in this example, those that have larger solar energy resources, etc. Energy Market Integration (EMI) through power connectivity is one-way electricity trading could be encouraged. East Asia is a prime place for interconnectivity to be encouraged given the diversity of clean energy resources available, the presence of pre-existing multilateral institutions,

and governance structures that operate there, plus the economic and physical geography of countries within the region.

Very little work has been done in ASEAN countries on cost comparisons of integrating renewables (General comparisons: UN, 2006; ASEAN specific comparisons: Chang & Li, 2012; Wu et al., 2012; Li & Chang, 2014); however, work by the U.K. Energy Research Council [UKERC] (Gross et al., 2017) suggests that dealing with variability of renewables up to a 30% penetration costs on average £5/mwh; in addition, the costs of retaining sufficient capacity at 30% penetration of renewables is on average between £4-7/mwh. In ASEAN 2016 renewable penetration was roughly 9% of primary energy, with a target of 23% by 2025 (IRENA, 2016, p. 15). This suggests it is not primarily the integration of renewables that will drive any attempt to integrate the ASEAN power grid using electricity interconnectors, however, these costs will be a factor in the future. What is more likely to drive interconnection projects is inefficient, outdated, or unreliable power systems (UN, 2006). In this regard the option of interconnection provides another solution for system operators.

Alternative routes to renewable energy integration solutions exist outside of interconnections—this thesis does not argue that interconnections are the only solution. Rather, this thesis analyses one of ASEAN's preferred solutions, crossborder interconnections. Battery storage of distributed energy systems (DES) in remote areas offers a different solution in rural communities where grid connection is not an option, including mini-grids and nano-grids (Energy System Management Assistance Program [ESMAP] & International Finance Corporation [IFC], 2017, p. 8). Alternatively, grid scale battery storage, or battery energy storage systems (BESS), allow for the storage of energy from a grid or power plant for use later (United States Agency for International Development [USAID] & NREL, 2019, p.1). Battery storage systems allow for increased power system flexibility—whereby power systems can respond to intermittency and variability of renewable energy resources (Jones, 2014). Regulatory, market, and capability barriers prevent the scalability of these systems depending on their location (USAID & NREL, 2019, p.3) and despite continually falling costs the price of large-scale battery storage remains prohibitive (ESMAP & IFC, 2017, p. 1). This thesis does not dismiss the capability of battery storage or the certain evolution of battery storage, it simply looks at the current, present preference of ASEAN and ASEAN member states towards cross-border interconnections.

This brings us to the next topic, an explanation of the relevance of electricity markets and their relationship to the energy mix in East Asia. This primer will

present ways in which electricity markets can be better expanded for renewable energy integration and how challenges to expansion are currently being explored in East Asia and the sub-region of Southeast Asia.

#### 1.2.d Electricity Markets & Regional Energy Mix

Electricity is defined as a secondary energy source produced by transforming primary sources of energy, such as oil or solar energy, into electrical power (Energy Information Administration [EIA], 2019). In addition, electricity is an energy carrier, whereby it can also be converted into other forms of energy. Primary energy is both renewable and non-renewable, whereas electricity is neither renewable nor non-renewable (EIA, 2019). Traditionally, electricity is produced via hydrocarbon resources;<sup>8</sup> the industrialisation of many countries has been based, in fact, on energy intensive industries that utilised coal, oil and later gas (Harris & Lang, 2015). However, the global energy transition towards cleaner energy resources with less environmental impact has created a number of challenges associated with renewable energy integration in electricity markets, mainly flexibility challenges associated with renewable energy intermittency (IEA, 2019b). In response to these variable renewable energy (VRE) challenges governments have increasingly been focused on the ability of existing markets and regulatory frameworks to continue providing a reliable, affordable, and efficient supply of electricity (IEA, 2019b).

In the case of electricity production, regardless of the source, the power created provides improvements to quality of life, aids responses to global development needs, and is recognised as a necessary requirement for modern life (IEA, 2019d). Globally, roughly 860 million people still live without access to electricity (IEA, 2019c), and the power sector continues to be the primary focus of clean energy policy support among governments (REN21, 2019b). In East Asia the growth of electricity consumption has put pressure on supply side demand for both power generation and transmission capacities (Chang & Li, 2017a) but national electrification rates are extremely stratified depending on rural versus urban electrification (Naimoli & Nakano, 2018). National estimates of

<sup>&</sup>lt;sup>8</sup> There are exceptions, of course, including a history of hydropower development in the Greater Mekong Sub-region (GMSR) in China, Myanmar, and Thailand (Simpson, 2007). However, due to its negative environmental and social impacts there are concerns regarding the sub-regional deployment of hydropower (Rosa et al., 2004; Huang et al., 2019). Section 7.5, *Suggestions for Further Study*, explores lessons learned in GMSR water policy and how future cross-border electricity trade research might incorporate the GMSR hydropower experiences.

electrification rates across the sub-region range from 57% (Myanmar) to 100% (Thailand), with a variety of national, sub-regional, and regional mandates regarding reliable, affordable access (IEA, 2019). These statistics are even more stratified at the urban versus rural level, with many states seeing a wide variety of electrification depending on location:





Source: Cravioto et al., 2020, p. 3.

Energy poverty arguments are part of the rational for ASEAN's own connectivity agenda, and the efforts of national governments to engage in increased crossborder electricity trade. Whether the climate imperatives are fully operationalised via policy measures and economic changes is to be seen but will be explored further in the data chapters of this research (chapters 4-6).

There is a growing recognition of the potential for cross-border electricity trade to in part respond to the variability, uncertainty and flexibility challenges posed by increased clean energy use in electrification. In East Asia, where geographical, policy, and financial limitations differ from country to country, cross-border electricity trade is viewed with growing interest and hope for contributing to the attainment of future renewable energy and climate targets (Chang & Li, 2015, 2013). In Southeast Asia, under the sub-regional organisational umbrella of ASEAN, the Association of Southeast Asian Nations, has set EMI targets associated with expanding cross-border electricity trade and growing renewable energy's share in the sub-regional and national power sectors by 23% (ACE, 2013a). The global challenge of climate change has forced governments around the world to reconsider their energy mix and approach to electricity production—transitioning from carbon intensive energy resources towards more renewable sources in an effort to reduce global CO<sub>2</sub> emissions. This is illustrative of part of the climate imperative for increased cross-border electricity trade—higher shares of clean energy in a country's energy mix has a wide variety of advantages.

According to a variety of studies these advantages are as follows: In addition to augmenting a country's reliance on hydrocarbons the transition to increased shares of renewable energy in electricity production has the added benefit of (1) providing increased access to electricity in regions that have not yet reached 100% electrification (IEA, 2014), (2) of improvements to quality of life via welfare impacts (ADB, 2012, 2014) and (3) of boosting national economic development effects at the same time (Shi & Widodo, 2014). Additional benefits to national economies and welfare via increased trade liberalisation and economic integration are varied, but overall positive (Bhattacharya & Kojima, 2010); investment liberalisation would have positive regional impacts on gross domestic product (GDP) and on growth in East Asia (Bhattacharya & Kojima, 2010), with potential to reduce regional economic disparities and reduce the development gaps between the richest and poorest countries in the region (Watcharejyothin & Shrestha, 2009; Sheng & Shi, 2013). Reductions in total energy system costs (Chang & Li, 2012) would also coincide with CO<sub>2</sub> emissions reductions in sub-regional groupings throughout Southeast Asia and represent additional climate and energy system benefits (Watcharejyothin & Shrestha, 2009; Shi & Widodo, 2014).

Additional cost benefits of an interconnected sub-region also abound: Blakers et al. (2012) found that in addition to Southeast Asian electricity demand being satisfied by interconnected Australian electricity, locally produced renewable and conventional electricity could supplement gaps and fulfil the sub-region's goal of higher renewable energy integration. Similarly, research by Taggart et al (2012) found that region-wide interconnection would decrease both electricity costs and emissions related costs region-wide. Taggart (2013) found similar electricity cost-reductions based on an interconnected pan-Asian region, including Australia; and a 2009 ADB & Asian Development Bank Institute (ADBI) study found a net cost benefit of roughly \$3.5 billion USD from increasing cross-border infrastructure in Southeast and Northeast Asia (p. 75). A 2004 study by the Asia Pacific Energy Research Centre found net savings across the three proposed ASEAN subsystems, resulting in roughly \$662 million in net savings (p. 41-43), a figure that further supported ASEAN study group estimations of net

savings in 2003 (AIMS Study Group, 2003, p. 29) and ACE estimations of cost benefits in 2013 (ACE, 2013a). According to a 2017 study on cost optimization of sustainable power systems in Asia, Gulagi et al. (2017) found that grid integration at a large scale would reduce total electricity costs in Southeast Asia and the Pacific rim, with ASEAN benefiting from increasing interconnection infrastructure and development of sub-regional grids. Owen et al. (2017) argue that, when using cost benefit analysis to select project, its equally important to balance the environmental, security and socio-economic impacts, which is as important as cost reductions in ASEAN EMI initiatives (p. 149)

Overall, the variety of cost analyses that have been done on sub-regional and regional interconnection all pointing to economic benefits from increasing regional interconnections and higher shares of renewables—facts that ASEAN has employed to its advantage. This research will further explore how the IPE of cross-border electricity trade encourages these opportunities, examining the political and economic dynamics that influence individual markets and implications for the development of an integrated sub-regional electricity market.

Meeting growing electricity demand in ASEAN will also require significant development of the sub-regional power system, including changes at the national level for smooth sub-regional integration (IEA, 2017a), which can be achieved via increased cross-border electricity trade. According to the IEA (2017b), power generation capacity will double by 2040, outpacing the global average and feeding sub-regional economic growth. The diversity of resources, varying levels of energy access, complementary systems, and different levels of economic development mean ASEAN is well placed to benefit from increasing its share of renewable energy and increasing the adoption of clean energy technologies as a means to meet projected power generation (Huang et al., 2019). According to Huang et al. (2019) this can include 'leapfrogging' reliance on fossil fuel, shared renewable energy integration practices across complimentary grids, and information sharing among cross-border trading partners and ASEAN members involved in connectivity initiatives and projects (p. 712-714). Interconnections between resource-rich and resource poor economies can also help to meet growing demand, as well as mitigate resource fluctuations and reduce cost of sub-regional electricity demand (Chang & Li, 2013).

In addition to poverty alleviation and developmental impacts described above, ASEAN has also identified that cross-border electricity trade can provide power sector improvements, too: these include reducing energy prices (Chang & Li, 2012), mitigating against shortages and power shocks (Pollitt, 2008; Shi, 2014), incentivising deeper market integration (Wu et al., 2012), managing regional and sub-regional resource endowment differences (Shi, 2014), and facilitating sustainable development in the power sector (Ahmed et al., 2017). However, electricity accounted for roughly 4 percent of total global exports in 2015 and just 3 percent of total production (IEA, 2015). These numbers are in stark contrast to hydrocarbon energy resources that are widely traded globally, and that illustrate the presence of barriers to cross-border electricity trade in the sub-region.

Commonly identified barriers to increased cross-border electricity trade are often technical, finance or investment-related and have been studied at great length via European examples of EMI and cross-border trade (See for example: Amundsen & Bergman 2006, 2007; Conlon, 2009; Wu, 2012; Wu et al., 2012; Bahar & Sauvage 2013; Mundaca et al., 2013; Oseni & Pollitt 2014; among many others). Nord Pool is one of the most successful and the largest example of an internationally integrated power market. Nord Pool utilises resource complementarities in Denmark, Finland, Norway and Sweden to provide electricity to 17 countries via an energy mix of hydro, nuclear, fossil, wind and biomass resources (Integrated Research and Action for Development, 2016). The Southern Africa Power Pool (SAPP) has similar aims as Nord Pool and is made up of 12 member countries. It is the only functioning energy market in sub-Saharan Africa, with plans for similar power pool projects elsewhere across the continent but not yet realised. Nord Pool and SAPP represent successful, multicountry power pooling projects with reach across many member economies (Barker et al., 1997), with additional, smaller power pooling and trade arrangements present in North America, the United Kingdom, Australia, and across Europe (Wu, 2012; IEA, 2019); The United Kingdom alone offers many examples of successful bilateral and small multilateral interconnection projects, with 17 operating, contracted or planned interconnectors between the United Kingdom and its European neighbours, facilitating successful market coupling (or efficient trading) between partners (Froggart et al., 2017, pp. 15-16). Research into the potential for developing a multi-country power system in South Asia and Southeast Asia has been conducted. However, no markets of the same level as Nord Pool or SAPP currently operates in the East Asian region (Kimura et al., 2013; Mundaca et al., 2013). In Southeast Asia, underneath the umbrella of ASEAN, a number of sub-regional projects have come to fruition although all are still in the development phases (Wu et al., 2012; Shi, 2016).

According to research by Chang and Li (2015) EMI (energy market integration) in ASEAN is the connecting of national power grids in order to trade electricity across interconnectors. EMI is shown to "significantly promote the adoption of renewable energy" (p. 159), contributing to global, regional and subregional sustainable development goals and moderate emissions reductions. However, despite disjointed interconnection on a bilateral basis, a Southeast Asian sub-regional power grid has received little progress via multilateral trading (IEA, 2017a). Despite the abundance of renewable energy resources available within ASEAN, over 70% of electricity demand is supplied by fossil fuels, and this number is expected to grow based on current projections of electricity consumption (IEA, 2017a). The International Energy Agency has emphasised the need for ASEAN to strengthen sub-regional institutions if regional EMI is to be fully realised and electricity from fossil fuels decreased. This includes expanding staff and resources as well as the development and harmonisation of policies and regulations that do not yet formally exist. Wu et al. (2012) make similar recommendations, and caution that the link between institutional strength and market liberalisation cannot be ignored if subregional EMI is to move forward. Mundaca et al. (2013) firmly insist that ambitious clean energy goals and policy support mechanisms are also a necessary component of EMI and diversification in East Asia.

Additional governance considerations related to EMI broadly, and directly relevant to cross-border electricity trade, have been identified by Oseni and Pollitt (2014) in a multi-country study for the World Bank that identifies the criteria necessary for EMI to take place. These criteria are summed up in the following general requirements: commitment to free trade; efficient market design; governance support; jurisdictional considerations & management; transmission capacity; operational national electricity transmission systems; and management of distributional effects. In addition, modern power systems are efficient, flexible, and reliable in order to meet fluctuations in demand and provide quick response times as supply and demand change rapidly depending on intermittent renewable sources; additionally, physical integration (or connections) between two systems is necessary in order for exchange of power to occur (Shi et al., 2019) and alignment between system processes, standards, and regulations is required between trade partners in order for long-term multilateral trading to occur (Owen et al., 2017). In order to integrate multiple national power systems for the trade of electricity across borders these criteria must be met (the Oseni and Pollitt criteria will be more seriously analysed in the context of Southeast Asia in Chapter 5).

Overall, the market logic for increasing cross-border electricity trade and EMI aligns with climate and development goals, and as such governments use these arguments to justify the economic, policy, and structural changes that are required. One aspect this research explores is whether or not these imperatives are truly reflected in the needs of individual markets. This research seeks to explain the limits to sub-regional power market development and the IPE of these limits, while examining the competing interests at play in their development. In explaining the IPE of cross-border electricity trade in East Asia and the climate change imperatives that have made it relevant, this research also seeks to position the IPE of cross-border interconnections among individual governments in Southeast Asia and their opportunities to engage in cross-border trade.

This section has brought together three areas that are important to understanding this research—(1) links between economic development and climate change via green growth, and the resulting imperatives for sustainability studies that include energy use and diversification of a country's energy mix; (2) East Asia's role in responding to global and regional climate concerns via clean energy utilisation; and (3) the importance of cross-border electricity trade in responding to both of these issues and the resulting economic and political challenges that may arise in sub-regional energy markets if cross-border electricity trade is to become a reality in Southeast Asia. The following section will explain how this research will address these factors and will discuss the gaps in research this project will fill.

### **1.3 Research Aims**

The aim of this research is to explore the extent to which climate change has created imperatives for energy market integration (EMI) and power sector reform in East Asia, with a specific focus on cross-border electricity trade and an IPE explanation of the competing interests at play. The aim to understand these challenges is achieved by exploring the following primary research questions (RQs):

- 1. How do governments understand the international political economy (IPE) of sub-regional cross-border interconnections in Southeast Asia?
- 2. How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?

3. What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian development?

This research will generate a detailed overview of the political economy of crossborder electricity trade and the national markets required for sub-regional energy market integration in Southeast Asia. The variability of electricity produced from clean energy resources provides an opportunity for regional and international electricity market expansion and the efficient disposal and purchase of power capacity. Based on the resulting potential for crossborder electricity trade, this research examines the expansion of these markets in East Asia and opportunities for their further development. A number of issues play a role in this research: including national and sub-regional electricity market development, energy market integration (EMI), sub-regional governance, interconnectors, and the clash of public and private interests. This political economy analysis will include a case study examination of a potential growth market in Southeast Asia's Brunei-Indonesia-Malaysia-Philippines East Asia Growth Area (BIMP-EAGA) sub-region following informative expert interviews that will help shape the view of this subsystem, the sub-region of Southeast Asia, and the broader region. A mix of quantitative and qualitative methods will be used to address these issues in relation to the regional political economy of EMI.

### 1.3.a Theoretical Framework

The history of economic development in East Asia, and its continued presence within the global, neoliberally-driven economic system has shaped the theoretical framing of this research. First, shared development stories and comparable histories of colonisation and subsequent independence in East Asia have contributed to related government and economic structures, as well as similar policy choices across the region. Secondly, East Asian economies were not afraid of government economic intervention, foreign technology, and inward and outward facing economic policies (Wade, 1988; White and Wade, 1988). Third, the economies of East Asia engaged fully in the global system of free trade as a means of increasing economic development, benefiting from global value chains and the expansion and globalisation of historical trade routes into modern systems of economic exchange. The combination of these three methods formulated the state-led economic development practices utilised by the majority of East Asian countries in their quest towards greater economic and societal advancement.
This model of state-led growth was first utilised successfully by Japan in the 1970s, and subsequent employment of similar practices occurred in South Korea, Taiwan, and Southeast Asia from the 1980s onwards. Commonly referred to as the "flying geese" model, this pattern of development utilised a model of industrial development that is "transmitted from the lead goose (Japan) to follower geese (Newly industrialised Economies (NIEs), ASEAN, etc.)" (Kojima, 2000, p. 376). Industrialisation is passed from one country to another via the transplant of "…comparatively disadvantageous production" from a host country (Japan) to NIEs, boosting the host's comparative advantage while facilitating pro-trade FDI within the NIEs (Kojima, 2000, p. 376). The combination of state-directed economic policies is commonly identified as developmental statism (DS) and represents a combination of state directed policies that includes a focus on industrialisation, industry targeting, and mercantilist trade policies (Leftwich, 1995).

Much of the modern, western economic theory rests on the premise that the best trade is free trade—when prices and industry growth are determined by supply and demand in international markets, not government policy. However, the electricity sector makes a strong case for temporary government policies that support local industry development. This does not, however, mean that free trade is abandoned, advocates say; simply that short-term government protection will aid in industry development and thus economic success. The nature of clean energy is such that international organisations and governments encourage their expansion and use, but the high cost and high level of innovation required for electricity market integration and variability responses requires policy support that promotes innovation and development locally. The pressing need for clean energy utilisation further points to potential changes to the theoretical thinking behind many of the world's most important trade policy decisions and the policy makers who support and implement them. Here neodevelopmental statism has a very specific role to play, providing an explanation for a simultaneous reliance on state-directed economic policy that can support strategic industries (like the power sector) and also allowing for integration into global and regional markets via neoliberal trade practices.

The prevalence of this East Asian model of economic development policy (DS, referenced above) within a global, free trade and capitalism values-driven economic system is interesting in light of the competing economic values within the global economic system. Neoliberalism has, it is argued in this research, shaped much of the global system of international trade and exchange of goods

and services. Being a member of this global economic system requires a commitment to market values and liberalisation. Indeed, suggested electricity market reforms deemed necessary for cross-border electricity trade and EMI require liberalisation and a continued commitment to neoliberal economic values. Yet the East Asian DS model persists nationally and regionally. These two seemingly contradictory theoretical frameworks are combined to explain the political economy of current electricity markets in Southeast Asia, forming what the author refers to as *neo-developmental statism*. Neo-developmental statism is rooted in the concepts of liberal developmentalism and neo-developmentalism but, in this context, the combination of these two concepts is applied to Southeast Asian countries and the electricity markets of ASEAN members for the first time. Neo-developmentalism, while an evolution of traditional DS paradigms, represents a theoretical approach to explaining cross-border electricity trade and barriers to EMI in ASEAN that have not been explored in other theoretical iterations of DS paradigms. The author's premise of neo-developmental statism is further elaborated throughout the thesis and used to explain and frame the IPE of cross-border electricity trade in sub-region of Southeast Asia.

DS is commonly applied to the economic development practices utilised across East Asia, offering an explanation for the economic policymaking of these nations predominantly from the 1970s onwards. There are vestiges of these economic practices that are still relevant throughout East Asia and the sub-region of Southeast Asia; however, these concepts have not been applied to EMI questions that dominate energy policy initiatives in ASEAN. Neo-developmental statism offers an IPE explanation for the drivers behind cross-border electricity trade in a complex neoliberal world order, but within subsystems where collectivist, Asian value approaches to policymaking dominate sub-regional and national politics. In the context of this research, neo-developmental statism is used to explain the IPE of common policy and market reforms that are applied to subregional power markets in order to achieve increased EMI and cross-border electricity trade. In this sense, neo-developmental statism is filling a gap in explaining the plausibility, difficulty and reality of applying neoliberal market reforms in a collectivist and Asian value-driven political and economic system.

Arguments in support of interconnections are relatively similar across the literature— efficiency and welfare gains which result in lower prices (Turvey, 2006), wider economic growth (Sheng & Shi, 2013), diversification of national supply (Shi, 2014), achievement of energy and sustainability targets (Andrews-Speed & Hezri, 2013), potential for modernisation and infrastructure

improvements (Wu, 2012, 2019), and potential for increased energy access and reliability (Ahmed, et al., 2017). Incentives for cross-border electricity trade are widely recognised by academics and institutions (See: ACE, 2013a; ADB & ADBI, 2009; ADB, 2017; IEA, 2014, 2015, 2017a; IRENA, 2018; REN21, 2019a; UN, 2006; World Bank, 2008). However, conflicts and contradictions at the state and national level prevent full utilisation of cross-border electricity trade, subregional EMI, and power sector reform. These conflicts, including difficulty in coordination between state, local, and national/sub-regional interests make progress slow. In addition, states are not just developmental—tensions between neoliberal and developmental interests result in contradictory and slow-moving policy across ASEAN member states. The tension between these interests is in part what makes neo-developmental statism a useful tool of analysis. Unfortunately, there is no single, planned rationale for how to approach crossborder interconnections, and while the rational for cross-border electricity trade exists, it doesn't necessarily outweigh performative state and regional actions and entrenched business, economic, and political interests.

Ultimately, the overarching aim of this research (to explore the extent to which climate change has created imperatives for EMI and power sector reform in East Asia and explain how East Asian economies view the climate imperatives to construct and operate these power sector infrastructure projects amid competing interests) is framed by the concept of neo-developmental statism. Common policy recommendations for the realisation of energy market integration and increased cross-border electricity trade are fundamentally arranged around neoliberal market ideals, yet being applied to traditionally DS economies, where the free market is not the only force and government directed policy remains equally important. Neo-developmental statism offers an explanation for this seemingly contradictory dichotomy of state-vs-market, aiding the political economy understanding this research aims to bring to the multidisciplinary topic of power sector development and renewable energy integration in ASEAN.

# **1.4 Study Contributions**

This research makes a number of contributions across the fields of IPE, Asian studies, and sustainability research. Broadly, this research contributes a unique way of approaching cross-border electricity trade by combining these three fields in the examination of sub-regional EMI. As stated earlier, this approach has not been taken in the same manner, with a consideration of East Asian economic

development practices in the analysis. Overall empirical, methodological and theoretical contributions have revealed the difficulty in addressing EMI in systems in which national and sub-regional interests prioritise neoliberal market factors, and traditional East Asian studies development paradigms struggle to account for the political and economic dynamics at play in power sector integration and cross-border interconnections. In the following subsections these findings and original contributions are discussed in more detail.

## **1.4.a Empirical Contributions**

While there are some academic studies of the BIMP-EAGA sub-region and its potential for EMI, this thesis is the first to analyse the IPE of individual electricity markets in the BIMP-EAGA subsystem while utilising an analysis of economic development patterns via DS. This analysis resulted in three distinct themes, each of which are discussed across the three data chapters, 4-6. These themes include sub-regional market factors (Chapter 4), national market factors (Chapter 5) and governance challenges (Chapter 6).

In Chapter 4 an analysis of the market structures in BIMP-EAGA frames crossborder electricity trade and EMI more fully within the subsystem as well as the wider sub-region. The analysis conducted includes not only mapping existing electricity interconnectors, but also identifying shortcomings in current markets for increased EMI and electricity trade. Most of this interconnection data is publicly available; however, it is not updated across sources and includes some outdated information that required author confirmation utilising a variety of resources coupled with interview subject feedback-resulting in unique updating of the sub-regional interconnection landscape. The energy and economic data analysed is publicly available, however, here it is uniquely compared and contrasted with research into the market structure of the subsystem, resulting in a more detailed image of sub-regional interconnections. This data collection informed the theme of Chapter 4, sub-regional market factors, explaining the collective BIMP-EAGA market landscape. Analysis of subregional market factors resulted in the research findings that hydrocarbon resources remain dominant in the sub-regional energy mix across the BIMP-EAGA subsystem despite sub-regional climate commitments and opportunities for renewable energy growth, and also that there is a need for national-level reforms in order for sub-regional EMI to flourish.

Chapter 5 introduces the second theme that emerged from data collection, national market factors. This chapter also covers the additional data analysis

that was conducted via an examination of EMI criteria coined by Oseni and Pollitt (2014), uniquely juxtaposed against sub-regional market factors and BIMP-EAGA data previously introduced in Chapter 4. In addition, the author expands on Oseni and Pollitt's 2014 criteria with information gathered via interviews and documentary analysis, creating what the author terms 'O&PCRB EMI Criteria", which is a unique contribution to research on the necessary market reforms for increased EMI and cross-border electricity trade in Southeast Asia. This data reveals two main findings: (1) the common identified national-level reforms for EMI are neoliberal in nature and reflect the popular needs of the global, neoliberal economic system and (2) the hydrocarbon dominance identified in Chapter 4, coupled with the national electricity needs of BIMP-EAGA and ASEAN identified in Chapter 5, signal that the sub-region is not uniformly committed to decarbonisation of power systems. As a result, sub-regional commitments to climate change appear to be performative in nature and are not reflective of national-level efforts—or needs—given the surplus of electricity in each BIMP-EAGA country.

Elite interviews also largely informed the third theme that emerged from this research, governance challenges, discussed in Chapter 6. Via interviews it became clear that trust and national interests are major barriers to EMI and cross-border electricity trade—not just trust of outsiders but also trust within ASEAN's membership and the prioritisation of national interests over sub-regional commitments. In this chapter the author makes the link between trust and outside actors to China's own power connectivity ambitions in Southeast Asia. The main findings of this chapter are that sub-region and national interests complicate the ability of EMI to address renewable energy integration challenges, and that APG progress is complicated by uncertainty at both sub-regional and national levels and various power dynamics at play.

### 1.4.b Methodological Contribution

Two issue areas relevant to this research—(1) responses to the global fight against climate change and (2) the political economy of East Asia's regional economic development path—are interrelated via the energy policy solutions to hydrocarbon intensive economic development and its impact on the global climate. These issue areas are situated across three fields: IPE, East Asian studies, and sustainability research. Using a mixed methods approach this crossdisciplinary research has combined conceptual, theoretical and methods choices across these three fields, resulting in unique data and analysis. Detailed examination of these choices and the resulting methodology is carried out in Chapter 3, *Research Design & Plan of Enquiry*. However, a relevant overview is provided here in relation to the methodological contributions of this research.

First, qualitative analysis of current policies and examples from Southeast Asia are used to fully depict the sub-regional, regional and global electricity trade and energy policy picture. In tandem with the collection of data, elite interviews were conducted to inform this study, increase understanding of the relevant issues, and advancing analysis. Extensive, high level interviews with current and past government, business, academic and technical leaders throughout East Asia provides insight into the current state of cross-border electricity trade in East Asia, governance options for sub-regional expansion of electricity trade, options for sub-regional power market integration, and shortcomings of current proposals for deeper Southeast Asian energy market integration. Each of the three dominant themes identified and analysed in chapter 4-6 are supported via coding of elite interviews for key terms and patterns.

Data collection of national electricity market structures in Southeast Asia was conducted via documentary analysis and simultaneously informed by interviews. This data collection included analysis of existing and proposed interconnections in Southeast Asia and a thorough look at the electricity market structure of the individual BIMP-EAGA economies. The combination of these two methods also resulted in the O&PCRB EMI Criteria that is introduced in Chapter 5.

# **1.4.c Theoretical Contributions**

Various iterations of DS exist across East Asian studies. However, the application of neo-developmental statism to electricity markets is novel in its interpretation and application. The author's conceptualisation of neo-developmental statism provides a unique lens from which to understand the BIMP-EAGA economies and their potential for EMI, highlighting the challenge of applying neo-liberal economic reforms in state-dominated markets. Neo-developmental statism also provides an updated theoretical evolution of the traditional DS paradigm from which to explain the challenge of centrally planned economies participating in neoliberal markets and the impact these challenges have on the political economy of cross-border interconnections in Southeast Asia. This approach draws attention to the economic and political drivers behind EMI in the first place, and the global structure of climate change challenges, performative responses, and sub-regional governance institutions within an East Asian context. The application of neo-developmental statism to sub-regional electricity markets in Southeast Asia also provides a unique response to the author's prior analysis of sub-regional EMI: it is a mistake to reform on the trajectory towards a fully liberalised power sector—instead, like neo-developmentalism, a hybrid approach<sup>9</sup> to reform is the legitimate form of East Asian developmentalism.

# **1.5 Thesis Structure**

So far this chapter has framed the basis of this research, explaining the rationale behind the research aims that makeup this body of work. This chapter has also introduced the original empirical, methodological and theoretical contributions of this research, which will be further expanded upon in the data chapters 4-6 and the concluding chapter. The following sections outline the structure of this thesis, beginning with the presentation of research objectives and questions throughout data chapters, followed by a detailed overview of each chapter and its contribution to the entire project.

## 1.5.a Objectives & Questions

Previous sections in this chapter have introduced the different dynamics at play in this research, including the intersection of East Asia's economic development history and climate change, East Asian and global responses to climate change, and the significance of electricity markets in policy responses to the economic and political challenges connected with climate change in the sub-region of Southeast Asia. Ultimately, these considerations developed into the following research objectives and research questions, which frame the subsequent chapters and results of this research. In the following table research questions identified in the introduction section of this chapter have been paired with corresponding research objectives and the resulting themes that have emerged throughout the research process—each of these themes and the corresponding research questions and objectives are discussed in a separate data chapter (chapters 4-6). For example, Chapter 4 is focused on the first theme to emerge

<sup>&</sup>lt;sup>9</sup> Hybrid approaches to electricity sector reform are not new or novel—examples of hybrid approaches exist throughout sub-Saharan Africa and Europe (Gratwick & Eberhard, 2008). However, these hybrid examples are not prescriptive (Gratwick & Eberhard, 2008, p. 3959)— they are adaptive and based on the experiences and intricacies of individual markets and the economic and business characteristics present. As such, the hybrid approaches mentioned in reference to ASEAN are specific to their characteristics. In addition, in the literature on ASEAN hybrid approaches are mentioned few and far between; the common recommendations reflect the standard model of neoliberal market approaches to power sector reform (Sen, 2008, pp. 1-4) and the standard neoliberal model of development (Kasahar, 2013, p. 1). For more debate on neoliberal approaches to development, and East Asia, see also: Brohman, 1996.

from data collection and analysis, and answers Research Question 1 and addresses (research) Objective 1; Chapter 5 is focused on the second theme to emerge from data collection and analysis, while answering Research Question 2 and addressing Objective 2. Chapter 6 is focused around the third and final theme to emerge, while answering answers Research Question 3, and addressing Objective 3. These pairings are presented together in the following table with the corresponding research objectives and data themes, illustrating the flow of ideas and concepts as they are revealed in chapters 4-6:

## **Table 1.2** Objectives, Needs, & Research Questions

## **Objective 1 (Chapter 4 – Sub-Regional Market Factors)**

Objective: Provide overview of current status and needs of cross-border interconnections in East Asia, specifically the sub-region of Southeast Asia:

- Assess the role of cross-border electricity trade in Southeast Asia subregional EMI;
- Review and analyse current and projected interconnections;
- Review and analyse ASEAN interconnection projects;
- Synthesise lessons learned among selected projects.

### Research Question

RQ1: How do governments understand the IPE of cross-border interconnections in Southeast Asia?

# **Objective 2 (Chapter 5 – National Market Factors)**

Objective 2: Assess market factors in sub-region of Southeast Asia and case study selection and identify areas of individual and collective reform needed to reach deeper levels of EMI among ASEAN member states:

- Synthesis overview of current market structures in national and subregional electricity markets;
- Assess national needs for sub-regional EMI;
- Review and analyse any links to the integration of renewables into national energy mixes, review sub-regional energy poverty reductions, and identify any links between national energy goals and EMI initiatives;

### **Research Question**

RQ2: How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?

### **Objective 3 (Chapter 6 – Governance Challenges)**

Objective 3: Identify and analyse governance challenges to deeper EMI and increased electricity interconnections among ASEAN member states and case study selection:

- Identify trade and economic development policies used in response to these issues;
- Provide realistic, relevant, IPE based policy recommendations for further sub-regional renewable energy integration;

#### **Research Question**

RQ3: What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian Development?

Central to the research objectives and research questions is the consideration of East Asian economic development paths, highlighting the contradictions between traditional state-led development practices and global neoliberal, capitalist-driven market reforms. These considerations drive the theoretical framework that shapes this research from within, but also the issues themselves—both within national governments and within sub-regional governance structures.

### **1.5.b Chapter Overviews**

Chapter 2, Conceptual Approach & Review of Literature, provides context to the intersection of three areas of study (East Asian studies, IPE, and sustainability research) while examining prior cross-border electricity trade research in East Asia and identifying existing analytical gaps. This chapter sets the scene for answering research questions 1-3 and later identified sub-questions. This includes an examination of the literature surrounding energy market integration as a response to climate change, foundations of IPE, functions of IPE in East Asia, and IPE explanations of cross-border electricity trade. In Chapter 2 the theoretical framework that supports this research, neo-developmental statism, is also explained in more detail including an overview of the two theories that make up this explanatory tool, neoliberalism and DS. The common global reliance on neoliberal market economics is contrasted against the DS policies most prevalent in East Asia, providing scope for eventual examination of the IPE of electricity markets in Southeast Asia, and the compatibility of reforms needed in order for EMI and cross-border electricity trade to thrive. As a result of the coexistence of seemingly contradictory economic practices, existing developmental state models used to explain regional and sub-regional economic and political choices do not fully encompass the competing economic and political dynamics in Southeast Asia. The author's conceptualisation of neodevelopmental statism is introduced and provides an explanation of the hybrid policy approaches utilised to realise a fully integrated ASEAN electricity market. Chapter 3, *Research Design & Plan of Enquiry*, explains the methodological approaches and methods utilised in this research. This includes an introduction to the theoretical underpinnings of this research, critical realism, and how this epistemological approach informs the collection and analysis of data. Research design, including case study design and elite interviews are explained in detail, with reference to reliability and validity concerns, ethical considerations, and data protection measures that were employed. This chapter will explore the methods used to reach three themes identified by the author via data collection, setting up the further introduction of these data themes in subsequent chapters (chapters 4-6).

Chapters 4-6 represent the data analysis chapters that resulted from interviews and data collection. The order of these chapters was chosen specifically to paint the picture of cross-border electricity trade in Southeast Asia and the IPE considerations ASEAN and ASEAN member states must make. First, Chapter 4 presents a sub-region-wide picture of the current state of cross-border electricity trade. Next, Chapter 5 drills down into the IPE of specific power systems in ASEAN, with a focus on BIMP-EAGA power systems at the national level. Finally, Chapter 6 presents the shared governance challenges that threaten regional and sub-regional progression of EMI and cross-border electricity trade. Together these three chapters represent the multi-level intricacies and interests of greater ASEAN EMI.

Chapter 4 is the first of these three data analysis chapters, each of which are framed around themes that emerged in the research process. Chapter 4 is structured around the theme of *Sub-Regional Market Factors*. This chapter explores ASEAN-specific EMI projects and the nature of EMI and cross-border electricity trade in ASEAN at the moment, in an effort to address RQ1, *How do governments understand the IPE of cross-border interconnections in Southeast Asia?* This includes an introduction to the case study, BIMP-EAGA.

The findings of Chapter 4 set the scene for the subsequent data presented in Chapter 5. Chapter 5, the second of three data analysis chapters, is framed around the theme of *National Market Factors* and addresses RQ2, *how does the IPE of cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?* Chapter 5 covers the additional data analysis that was conducted via an examination of EMI criteria coined by Oseni and Pollitt (2014) and juxtaposed against ASEAN market structure and national market factors of BIMP-EAGA economies. Oseni and Pollitt's criteria are expanded by the author to include additional market requirements identified in research interviews and data analysis and the resulting necessary economic, physical and political reforms that are necessary for sub-region EMI to flourish; this new criteria is called the O&PCRB EMI Criteria and provides a thorough view of the IPE of BIMP-EAGA markets and the potential for cross-border electricity trade growth.

Chapter 6, the third of three data analysis chapters, is framed around the theme of *Governance Challenges* and addresses RQ3, *what can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian development?* This chapter introduces the author's findings that there are three additional barriers to deeper EMI and enhanced cross-border electricity trade that fall under the umbrella of governance challenges, including (1) sub-regional versus national-level interests, (2) trust among ASEAN members, and (3) ASEAN trust of outsiders, particularly China and its role in sub-regional power sector development. This chapter expands on the original two themes of *Sub-Regional Market Factors* and *National Market Factors* to include outside factors that may influence the progress of EMI and cross-border electricity trade under the umbrella of governance challenges.

Finally, Chapter 7, Conclusions, synthesises and summarises the main points of this research by further pulling together the threads identified in each of the data chapters and summarising the answers to the primary research questions: Governments understand the IPE of cross-border interconnections (RQ1) as opportunities—but not just opportunities for integration and utilisation of renewables, also as economic opportunities. Similarly, the IPE of cross-border electricity trade provides opportunities for renewable energy participation in sub-regional markets (RQ2), but national policies do not signal the same level of commitment and interest in renewable integration via sub-regional EMI. Instead, many nations are incorporating renewables and upping shares of hydrocarbons in their energy mix while failing to make the necessary national power sector reforms for further EMI. And finally, cross-border electricity trade and renewable energy integration tell us that the IPE of East Asian development (RQ3) generally and Southeast Asia specifically is largely driven by neoliberal markets and economic development ambitions, not climate considerations. Participation in sub-regional and global climate targets is performative and expectations of fully liberalised power sectors in order to reach integrated ASEAN electricity markets are unrealistic in the face of hybrid developmental models explained by neo-developmental statism. At the end of Chapter 7 the author will propose a series of policy changes and adaptations that will suggest

an alternative approach for improving cross-border electricity trade in East Asia along with a series of recommendations for continued research and evaluation of this field.

## **1.6 Conclusions**

The inclusion of higher shares of clean energy in a county's energy mix places pressure on electrical power systems as they are forced to move away from historically fossil fuel-based electricity production. Entrenched, fossil fuel power systems, while polluting, do conveniently allow for ramping up and down of electricity production based on demand; intermittent renewables, on the other hand, have flexibility challenges based on their non-controllable generation patterns. Where previously it was possible to build power sectors within national boundaries, increased pressure to source a substantial and growing percentage of a nation's load from renewable resources means national systems become more variable as more intermittent resources are introduced. The prospect of electricity interconnections, where nations can buy surplus power during low generation hours from a neighbouring market which may not be experiencing scarcity (and vice versa) becomes a more pressing national concern and more attractive international investment than has previously been the case. Put simply, electricity is good for development, but old models of fossilbased centralised systems are not compatible with global climate commitments. Clean energy investment leads to a better case for investment in interconnectors and more interconnectors mean higher shares of renewables and cleaner electricity production with lower emissions. The opportunities for cross-border electricity trade in East Asia broadly, and Southeast Asia specifically, are abundant and well justified.

In order to address the challenges of renewable energy integration, climate change, and cross-border electricity trade development, in the following pages this research will provide a thorough examination and critique of the dominant neoliberal political economy frameworks and their proposed application to East Asian economic & political structures. This examination will provide context to the development of sub-regional interconnectors that impact clean energy utilisation and sub-regional EMI in Southeast Asia. Ultimately the overarching thread described through this research is that climate change adaptation is messy, policy has the potential to be flawed in its incentives, and there is a delicate balance between national, sub-regional and global goals and the necessary means to achieve them (if at all).

# Chapter 2: Cross-Border Electricity Trade: Conceptual Approach & Review of Literature

## **2.1 Introduction**

This chapter explores the basics of this complex, multi-dimensional topic by examining the foundational IPE concepts in East Asian and ASEAN specific crossborder electricity trade. Two dominant IPE concepts were chosen from which to build the conceptual understanding of cross-border electricity trade that is framed in this chapter—neoliberalism and developmental statism (DS). Chapter 2 will provide the necessary background needed in order to address the three primary research questions that frame this project:

- 1. How do governments understand the IPE of cross-border interconnections in Southeast Asia?
- 2. How does the IPE of cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?
- 3. What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian development?

This chapter will make the case for which concepts and empirical theories are relevant to this study, setting the scene for further analysis of cross-border electricity trade and EMI in East Asia via the primary research questions. While none of these research questions will be explicitly answered in this chapter, the literature review and theoretical framework discussed in the following sections will identify what current academic explanations are missing and how the author's own framing can fill these gaps. In the following analysis a major point is that, ultimately, the dominant East Asian IPE theory, developmental statism, is outdated in its application. In addition, the traditional neoliberal, western oriented IPE analysis fails to account for East Asian experiences and is not applied to electricity markets. As such, a combined theoretical framework that includes neoliberal ideals of market access, free trade, liberalisation and open markets with developmental statism ideals of state directed economic policy is necessary in order to frame explanations for increased cross-border electricity trade. The major divergence from previous iterations of alternative developmental statisms is the application to electricity markets and energy sectors in the sub-region of Southeast Asia and within the context of a need to increase cross-border electricity trade in response to renewable energy integration challenges.

This chapter will explore three distinct areas of research and how they relate to this cross-disciplinary study: East Asian studies, IPE, and sustainability research. This chapter will be structured as follows: First, in Section 2.2, East Asia's Growth Environment Nexus, East Asia's growth story will be told via the intersection of economic development and environmental challenges, explaining the climate, economic and environmental imperatives for cross-border electricity trade and the relevant sustainable development and green growth themes; this section will provide the necessary background and literature for eventually answering RQ1 in data Chapter 4. Next, in Section 2.3, Cross-Border Electricity Trade, the literature on cross-border electricity trade and Southeast Asian EMI will be presented, making a case for the governance role of ASEAN and its sub-regional interconnection initiatives and limits; this section will provide the necessary background and literature for eventually answering RQ2 in data Chapter 5. Following, in Section 2.4, International Political Economy & Foundations of Trade, neoliberalism will be introduced, explaining its relevance to global economic structures and international trade; then, in Section 2.5, International Political Economy in East Asia, the common East Asian studies IPE theme of DS will be introduced, including an examination of the limits to this area of research and the creation of the author's own conceptualisation of neo-developmental statism used to explain the IPE of cross-border electricity trade and EMI in Southeast Asia. Section 2.4 and 2.5 will, together, provide the necessary literature for answering RQ3 in data Chapter 6. Finally, Section 2.6, Conclusion, will bring the main points regarding climate change, renewable energy integration, power sector reform, and cross-border electricity trade together, tying the theoretical framework of neo-developmental statism into these issues prior to the subsequent chapter, Research Design & Plan of Enquiry.

# 2.2 East Asia's Growth-Environment Nexus

This section will frame the intersection of the three areas of study relevant to this research and the 3 primary research questions (East Asian studies, international political economy and sustainability research) by explaining the impact East Asia's economic development has had on the sub-regional environment and sub-regional EMI policy making. This section will provide the necessary context for understanding the climate, economic, and environmental imperatives for sub-regional cross-border interconnections, providing the literature needed to eventually address RQ1, *how do governments understand the IPE of cross-border interconnections in Southeast Asia*?

East Asia's modern economic development has been largely rooted in industrial policy and high-tech development. Government policies that promote infant industries, protect domestic producers, and encourage domestic innovation have played a significant role in the economic development and subsequent modernisation of Japan, South Korea, Singapore, Malaysia and others. Modern societies around the world are based upon common notions of growth (economic and cognitive) and the expectation of continued development and advancement (Charlton & Andras, 2003). Modernisation posits that scientific and technological advancements aid in the progress of humanity and society, often incorporated with aspects of equitability or inclusiveness (Hoselitz & Moore, 1963; Bernstein, 1971; Arat, 1988; Tominaga, 1991; Charlton & Andras, 2003; Chatterjee, 2005; McGregor, 2008). Modernisation and economic advancement largely hinge on access to modern forms of energy, and access is a requirement in order to:

overcome poverty, promote economic growth and employment opportunities, support the provision of social services, and, in general, promote sustainable human development (Johansson et al., 2012, pp. 153).

Access to electricity is a quantifiable economic advancement that can be monitored for the purpose of strategic development plans, and such quantifiable indicators have greatly shaped the policy design of East Asian nations' strategic and economic development (Naimoli & Nakano, 2018). It is common for stateled development practices to include industrial policies, and a number of East Asian economies have utilised or still utilise these practices today (Cai, 2008). Exploitation of the region's human capital and environmental and natural resources aided Northeast Asia's economic giants in their own economic modernisation, and NIE of Southeast Asia have also adopted such practices (McGregor, 2008). Improvements to quality of life as a result of modernisation has made East Asia one of the fastest developing regions of the world, but also one of the key players in innovative technology developments, including those in electricity production and transmission.

Modernisation, while bringing about technological, social, and economic advancement, does have a cost. In East Asia, where modernisation has occurred at a rapid rate, human, environmental and socio-cultural issues have also occurred with great intensity (Broadbent, 1998; McGregor, 2008). The damage of extracting resources for energy production, polluting the air with industrialisation practices, overreliance on fossil fuels for electricity creation, and stressing cities and rural communities with ever-growing populations and

greater electricity needs, juxtaposed with the global dependence on improved quality of life and economic growth, has created what Broadbent (1998) refers to as the Growth/Environment (GE) dilemma:

If we grow jobs and profits, it seems, we further destroy the environment. But if we protect the environment we slow down the economic growth that makes increasing profits and jobs possible, thereby threatening both (p. xiii).

The GE dilemma impacts most societies and challenges leaders to find a happy medium between (1) consumption and waste and (2) long-term economic and societal survival (Broadbent, 1998). Northeast Asia's largest economies have very publicly battled with the GE dilemma, and in some cases have managed to balance responses to pollution (both the natural and social intensity of pollution) with modern growth strategies (Broadbent, 1998; Jacobs, 2013). Green growth practices, as responses to the GE dilemma, are also built into East Asia's national and regional discussions on climate adaptation, energy system transformation, economic development policies, and clean energy development strategies (UN et al., 2017).

Nowhere is the struggle with the GE dilemma more apparent than Japan, where in a period of 50 years it has undergone a variety of responses: it countered postwar poverty through economic practices that resulted in the Japanese "economic miracle," which in turn led to industrialisation and export expansion, causing the "pollution debacle" and "urban debacle" of the 1970s and 1980s, which were eventually somewhat addressed through investment, efficiency measures, national regulations, and the export of pollution to developing economies (Broadbent 1998, pp. 18). Similar investment, regulatory, and market practices have been emulated in China and the Republic of Korea, with comparable environmental challenges and subsequent green growth strategies (Asian Development Bank Institute [ADBI], 2020). In Southeast Asia the GE dilemma has influenced green growth strategies via integration into the common stateled economic development model, influencing the development practices of NICs as well as industrialised economies, and reflected in ASEAN economic development initiatives (OECD, 2017).

Ecological Modernisation Theory (EMT) offers an evolved form of modernisation in an attempt to respond to the social, health and environmental problems associated with modernisation in East Asia and the GE dilemma, and is also situated within many sustainable development and green growth debates, originally introduced in Chapter 1, *Introduction* (Dryzek, 1993; Mol, 1995; Buttel, 2000; Mol & Spaargaren 2000; UNDSD, 2012; Jacobs, 2013; OECD, 2017). EMT posits that the promises of traditional modernisation can be realised while also considering ecological and environmental concerns but framing the issues as solvable within existing structures (Murphy & Gouldson, 2000; Bailey et al., 2011). Contrary to more dramatic views on economic development and environmental degradation, ecological modernisation was first introduced in the 1980s by Huber (1982) and Janicke (1984) but the concept has evolved, and the debates have changed along with developments in industrialisation and modernisation, resulting in a variety of arguments for and against EMT (See: Langhelle, 2000; Mol & Spaargaren, 2000; York & Rosa, 2003; Bailey et al., 2011; Bowen & Franhauser, 2011; Dryzek, 2013; Jacobs, 2013; among others). Varieties of EMT fail to account for socio-cultural challenges associated with inequality and instead focus primarily on neoliberal structures and processes, further encouraging market driven, neoliberal-based solutions (Bailey et al., 2011).

In the context of East Asia, Dent (2014) argues EMT is deeply integrated into East Asian development practices and is reflected in policies directed at renewable energy industry growth specifically. Dent's (2014) *new developmentalism* places a high importance on EMT in East Asian policy making, particularly in relation to targeting renewable energy industries for increased low carbon development (p.63). Here Dent does not specify beyond types of renewable energies, nor does his examination extend to integration and power sectors; instead, the focus is on low carbon development generally, with emphasis placed on EMT as a theoretical driver. The stated goal of Dent's new developmentalism is to transform economic development, driven by strategic industries,<sup>10</sup> energy security, and environmental imperatives (p. 63); the role of private-public relationships is also stressed, although altruistic, environmentally driven

<sup>&</sup>lt;sup>10</sup> Protectionist trade and industrial policies such as strategic trade, infant industry protections, and targeted industrial policy have been intertwined for several centuries. The first notable example is Alexander Hamilton's (1791) "Report on the Subject of Manufactures". German scholar Friedrich List (1909) also argued early on in favour of targeted, protectionist government policies as a means of building up infant industries and expanding trade. In the late 1980s targeted industrial policy underwent a period of increasing recognition, due in part to the previously mentioned strategic trade policy research by Brander & Spencer (1985, 2008) and further popularised by U.S. Economist Paul Krugman's (1983, 1986) own work on the theoretical balance between strategic trade and industrial policy. See also: Grossman & Richardson (1985); Stegemann (1985); Carliner (1986); Grossman (1986); Shafaeddin (2000); Katz & Lee (2011); Tawney (2012); and others. In this research these arguments will not be explored further except in relation to developmental statism in East Asia broadly, and where relevant to the power sector specifically.

incentives receive the most emphasis. Dent takes a broad approach to developmentalism, highlighting the importance of state capacity theories broadly rather than developmental statism specifically. In addition, new developmentalism assumes a high level of state capacity and a reliance on industrial targeting and socio-environmental pressures that this author does not believe are reflected in the market drivers among ASEAN economies. In addition, based on a reading of the green growth literature, this author finds that EMT, while relevant, is not the basis for policy responses to climate change; instead, it appears that in Southeast Asia specifically, economic development and neoliberal economic gains make up most of the incentives at the national level. This assertion is supported in chapters 4-6, and relates directly to RQ1 (How do governments understand the IPE of cross-border interconnections in Southeast Asia?), RQ2 (How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?), and RQ3 (What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian development?), and directly supports the application of the author's own theoretical framework, neodevelopmental statism, discussed further in section 2.4.

The prevalence of a variety of green growth and sustainable development strategies and practices throughout East Asia (and elsewhere) are due in large part to the economic interests that support the opportunity environmental protection offers in the form of industrial development and economic growth (Bowen & Frankhauser, 2011; Jacobs, 2013). Green growth and similar sustainable development discourses are directly related to the 'wicked problem' of global climate change—whereby no single, straightforward answer exists for a complicated, transboundary problem (Sun & Yang, 2016).<sup>11</sup> As a result, many conflicting solutions and outcomes persist, each with a different set of issues and sub-issues that crop up (Conklin, 2005). The problem, in a sense, is so complicated that a solution remains evasive, and to answer only part of the solution is to act morally unjust (Churchman, 1967, p. B-141). According to Churchman:

<sup>&</sup>lt;sup>11</sup> This research will not delve into a debate about the seriousness of climate change; instead, this research is based on the widely accepted premise that global climate change is a pressing issue that requires policy action at a wide variety of levels. For more on the challenge of climate change please see IPCC (2019); Stern (2014); Stern et al. (2006). For scholarship on the challenge of climate change in East Asia see IRENA & ACE (2016); IRENA (2018); IEA (2019); REN21 (2019a); among others.

The adjective 'wicked' is supposed to describe the mischievous and even evil quality of these problems, where proposed 'solutions' often turn out to be worse than the symptoms (1967, p. B-141).

Sustainable development and green growth are solutions that solve *part* of the problem (environmental impact) associated with climate change, but not the entire problem (industrial development and environmental impacts of economic development), thereby leaving solution makers to "exclude potentially promising alternatives" (Cook, 2019, p. 36). The wicked problem<sup>12</sup> of climate change cannot necessarily be solved by a single solution, such as green growth or cross-border electricity trade; however, a holistic approach that includes multiple solutions and options but recognises the need for continued adaptation and response is a more realistic way forward. As such this research does not seek to advocate for cross-border electricity trade as the only solution, just one of many.

One such area where an all of the above approach can be quite effective is in East Asian energy markets, where a variety of energy resources are being integrated into the energy mix of economies as a means to reduce reliance on fossil fuels and reduce harmful emissions. As was discussed at the outset of this research, the inclusion of renewable energy in an economy's energy mix creates variability challenges that fluctuate with the availability of renewable resources (Huang et al., 2019). For example, in the case of wind power, which is created via the movement of wind turbines, energy is only produced when the wind is blowing. Economies in Europe have responded to this variability by purchasing power from neighbours when they need it and selling their excess power when they don't. Cross-border electricity trade has enabled this exchange and offers the ability to overcome clean energy limitations in electricity production. In Southeast Asia, ASEAN has acknowledged that cross-border electricity trade has the potential to respond to, and improve, a number of challenges associated with the impact economic development and industrialisation have had on the region (ASEAN 2004, 2007, 2010, 2011, 2012, among many others). ASEAN leads the

<sup>&</sup>lt;sup>12</sup> Definitions of wicked problems do vary depending on the scholar and problem explored. Additional classification of wicked problems, super wicked problems, and their application to climate change abound. However, the author has reserved engaging with these debates based on space limitations, and therefore will use the general application of wicked problems described above. However, additional arguments and accounts of wicked problems and the application in climate change and sustainability studies, can be found in the work of Whelton & Ballard (2002); Head (2008); Skaburskis (2008); Lazarus (2009); Levin et al. (2007, 2012); Incropera (2015); among others.

way in region-wide interconnections, not only aiming to increase renewable energy integration among ASEAN member nations, but also respond to energy poverty and climate change concerns simultaneously. The following subsection explains this approach within ASEAN, as well as discusses ASEAN's governance abilities in relation to EMI and increased cross-border interconnections.

## 2.2.a Sub-Regional Imperatives for Cross-Border Electricity Trade

Established in August 1967, the original five members of ASEAN included Malaysia, Singapore, Philippines, Indonesia and Thailand. The five founding members of ASEAN sought a community that would encourage political and social stability among post-colonial East Asian states (Zhao, 2016). In the 1980s Brunei Darussalam joined the original five, and later in the 1990s Cambodia, Lao PDR, Myanmar and Vietnam further joined. Overtime ASEAN adjusted its focus to an economic and sub-regional integration agenda that is reflected in its progress reducing tariff barriers to trade among member economies, the development of the ASEAN Free Trade Area (AFTA) and growth of the ASEAN Economic Community ([AEC] ADB, 2008a).<sup>13</sup>

Chief among ASEAN economic integration initiatives is the role of energy, and as a result connectivity and electricity market access. It is ASEAN's own need for increased electrification, sub-regional and regional initiatives focused on climate change and renewable energy integration, as well as shared climate challenges, that make Southeast Asia an interesting and unique sub-region of the world to study energy and climate change issues. The irony of needing to study these issues in the first place is not lost on researchers—the need for increased crossborder electricity trade and increased renewable energy integration stems from ASEAN's own poverty alleviation, sub-regional economic development, and the success of the AEC. Linkages between electricity consumption and economic growth demonstrate the positive relationship between electricity access and the development process (Apergis & Payne, 2011). Additional benefits when energy poverty alleviation intersects with climate change mitigation can be realised via policy efforts on efficiency and energy mix diversification (Ürge-Vorsatz &

<sup>&</sup>lt;sup>13</sup> The AEC is one of ASEAN's flagship initiatives, focused on increasing economic prosperity, developing "rules-based, competitive, resilient" and globally integrated regional market and stronger regional economic community by a 2015 due date (Shi, 2014, p. 116). The AEC was largely achieved, in part due to the broad and generalised goals of improving development and governance across the sub-region. Many of ASEAN's subsequent initiatives involve aspects of the AEC, which remains an important cornerstone of ASEAN identity and achievement (ASEAN, 2015) and is often referenced as a successful ASEAN initiative.

Herrero, 2012). In ASEAN targeting the economic benefits of electricity access have been built into sub-regional programmes and remain a major goal of AEC initiatives. Forecasting of future energy demand in the sub-region shows a continued rise in access and the need for capacity, infrastructure and policies that support this growing demand (Chang & Li, 2013). In addition, rapid economic development requires increased energy access, which puts pressures on the climate and environment and requires energy solutions that respond to global greenhouse gas emissions while incorporating responses to climate change concerns (Gulagi et al., 2017). ASEAN's energy poverty reductions, economic growth success, and environmental challenges are all interlinked in the development and growth story of the sub-region.

Despite progress, the energy security and access picture in ASEAN is predicted to remain mixed. Population, per capita CO2 emissions, and renewable capacity is varied across the sub-region.

Country	Population (2017)	Per capita CO2 Emissions from Fuel Combustion (2016)	Total Renewables Capacity (2019)
Brunei	0.4 million	14.9 metric tonnes	1 MWh
Indonesia	264.0 million	1.7 metric tonnes	9,471 MWh
Malaysia	31.6 million	7.0 metric tonnes	8,157 MWh
Philippines	104.9 million	1.1 metric tonnes	6,482 MWh
Cambodia	16.0 million	0.6 metric tonnes	1,438 MWh
Lao PDR	6.9 million	No data	5,118 MWh
Myanmar	53.4 million	0.4 metric tonnes	3,315 MWh
Singapore	5.7 million	8.1 metric tonnes	279 MWh
Thailand	69.0 million	3.6 metric tonnes	10,441 MWh
Vietnam	95.5 million	2.0 metric tonnes	18,523 MWh

 Table 2.1 Basic Sub-Regional Environmental Picture

Sources: UNESCAP, 2019, pp. 2, 40; IRENA, 2019b, pp. 2-4.

Since 2002 energy demand in the region has grown 60%; however, in 2020 10% of the AEAN population still remains without access to electricity (Suryadi, 2020). While the IEA predicts that ASEAN will reach 100% electrification by

2030, it still forecasts that 175 million people in the region will remain without access to electricity for clean cooking in 2040, with a prevailing reliance on solid biomass for cooking fuel<sup>14</sup>; as of 2016 only 56.3% of the population had access to clean cooking fuels and technology (UNESCAP, 2017). The population of ASEAN is estimated to increase to 770 million by 2040 (IEA, 2019a), with universal electricity access reached (ACE, 2017) and energy demand skyrocketing (Huber et al., 2015; Shi & Widodo, 2014). The majority of this demand is projected to come from fossil fuels; coal and gas will remain the majority providers of electricity generation (IEA, 2019a), with a variety of potential increases for renewables depending on sub-regional policy making, investments, and infrastructure growth (Huber et al., 2015). Cross-border electricity trade, EMI, and power system reform provide management and investment options that could counter this growth and aid in renewable energy integration and utilisation (IEA, 2019a; Huber et al., 2015).

Electricity access is a critical focus of ASEAN as an institution (ASEAN 2004, 2006, 2010, 2011), as Southeast Asia remains home to electricity access limitations, particularly in the least developed and island nations of Southeast Asia where energy services remain stranded among non-urban populations (Vithayasrichareon et al., 2012). Since 2000 electricity access has grown 28%, rising to 90% electrification rate across the region; today electricity makes up 52% of the region's primary energy demand (IEA, 2019a).

<sup>&</sup>lt;sup>14</sup> Solid biomass is responsible for the majority of indoor air pollution world-wide (IEA, 2019a) and indoor air pollution results in 3.8 million premature deaths a year and is a serious health hazard in developing countries and among the world's poor, according to the WHO (IEA, 2017a; WHO, 2018).

Country	% of Population with electricity access (2016)	Electrification Targets (2019)	% Renewable Electricity Generation of Total Output (2016)	Renewable Targets (2019)
Brunei	100.0%	n/a	0.05%	10% RE in power generation by 2035
Indonesia	98.0%	99.7% by 2025	12.8%	23% RE use by 2025, and 31% by 2030
Malaysia	100.0%	n/a	13.5%	RE installed capacity of 2,080 MW (excluding large hydro) by 2020; 20% clean energy use
Philippines	91.0%	100.0% by 2022	24.2%	40% by 2020 100% by 2050; 15.2 GW of RE installed capacity by 2030
Cambodia	50.0%	70.0% by 2030	47.6%	No specific RE target, but establishment of large hydro of 2,241 MW by 2020

Table 2.2 Sub-Regional	Electricity &	& Renewahles	Picture
Table Li Jub Regional	LICCULLY C	x nenewables	Incluic

Country	% of Population with electricity access (2016)	Electrification Targets (2019)	% Renewable Electricity Generation of Total Output (2016)	Renewable Targets (2019)
Lao PDR	87.0%	98% by 2025	No data	30% RE share of total energy consumption by 2025
Myanmar	57.0%	80% by 2030	54.4%	30% RE sources for electricity generation
Singapore	100.0%	n/a	1.9%	350 MWp of solar power installation by 2020
Thailand	100.0%	n/a	15.2%	30% renewable in total energy consumption by 2036
Vietnam	99.6%	Grid access for most rural house by 2020	39.0%	21% RE of 60 GW installed capacity in 2020

Sources: UNESCAP, 2019, pp. 4, 22, 31; IEA, 2019, P. 52.

Sub-regional access and generation targets highlight the need importance of electrification and renewable contributions, but also show a critical weakness--based on current renewable projections and sub-regional targets ASEAN will still fall short of its 23% renewables by 2025, with just 17% total share of renewables instead (IRENA, 2016, p. 10).

- 47 -

access areas, makes continued, rapid expansion more difficult than many other regions (IEA, 2019a). In addition, upgrading national grids will need to occur, including increased grid capacity, modernisation and standardisation. institutional and regulatory adjustments, and the expansion of national interconnections, among other variables (Porter & Situmeang, 2005). However, according to a multi-country study on preparedness for grid integration of renewables by Huang et al. (2019), ASEAN is well positioned to improve grid flexibility<sup>15</sup> to reach previously discussed 23% renewable energy goals—but only if the sub-region addresses power system flexibility challenges (p. 722).

In addition to the climate imperatives and energy poverty reduction goals of ASEAN member economies, the ability to diversify the sub-regions energy mix via EMI and cross-border electricity trade provides additional incentives for member state support of ASEAN initiatives, primarily within the framework of the APG and associated integration projects. Unfortunately for ASEAN one of its greatest assets (diversity of the region) is also one of its greatest limitations when it comes to growth in renewables (Ahmed et al., 2017). Distribution of renewable energy resources among Southeast Asian states is unbalanced, limiting options for countries with highly variable resource endowments (Ahmed et al., 2017). The unbalanced level of economic development within the region also poses investment and development complications in order to meet growing electricity demand (Atchatavivan, 2006; Chang & Li, 2013). A more integrated sub-regional electricity market and reform of current power sectors would augment cross-border electricity trade from resource rich to resource poor economies, contribute to energy poverty reduction efforts, as well as encourage increased renewable energy investment among ASEAN's least developed members (Ahmed et al., 2017).

Reaching the ASEAN goal of 23% renewables by 2025 requires national commitments in addition to the sub-regional goal, aimed at helping each

<sup>&</sup>lt;sup>15</sup> Traditionally grid flexibility refers to the capability of centralised electricity grids to respond to changes in supply and demand from a variety of different sources. Today flexibility also includes the ability to balance intermittent renewable energy sources and accommodate energy storage and new technologies while also responding to supply and demand (Martinot, 2016). Further references to gird flexibility will be made in Chapter 5, National Market Factors, when assessing the national electricity markets of individual BIMP-EAGA economies; these references will continue to be based on a broad definition of flexibility and do not reference flexibility challenges associated with distributed generation. However, distributed generation is referenced briefly in Section 7.5, Suggestions for Further Study. See also: Huang et al. (2019); Martinot (2016); Passey et al. (2011); Quiggin & Froggatt (2018).

member economy make the transition based upon their own particular economic and policy mechanisms (Huang et al., 2019; *Former Government—Informant 1; Multilateral Organisation—Informant 2).* In the case of Brunei Darussalam, for example, this means increasing renewables to 10% share of energy by 2035; for Indonesia, that number goes up to 23% by 2025 (ACE, 2017). According to the ASEAN Energy Outlook, produced by ACE, ASEAN member economies will not reach this goal unless significant policy and infrastructure changes are made; this currently looks unlikely to happen in the time frame given (ACE, 2017). Similar global renewable energy assessments have the same findings (See: IEA, 2017a, 2019; IRENA, 2016), and multiple experts interviewed for this research also echoed this sentiment (*Former Government—Informant 1; Multilateral Organisation—Informant 2; Multilateral Organisation—Informant 2; Multilateral Organisation 18; Academic 21; Academic 25*).

In addition, while power capacity is expected to increase three-fold to accommodate sub-regional electricity demand, coal is estimated to provide 42% of new capacity by 2040; renewables maintain a lower share, estimated to provide 29% if all proposed policy and structural adjustments are made to accommodate increased variable capacity (ACE, 2017). These numbers do not bode well for renewable targets, however, that does not mean progress won't be made in increasing electricity access and renewable grid integration, a point ASEAN is eager to make (Li & Chang, 2014; Navarro & Sambodo, 2013;. Wu, 2019). Chapter 4 and Chapter 5 of this research explore the sub-regional and national market factors making up this data, offering an IPE explanation for why ASEAN's own targets (both connectivity and climate targets) don't match the national level outcomes, and what this means for answering the first two research questions identified in this research:

- 1. How do governments understand the IPE of cross-border interconnections in Southeast Asia?
- 2. How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?

As a multilateral, sub-regional organisation, ASEAN has, by its very existence, made the case for governing cross-border electricity trade in the sub-region. It is this author's position that ASEAN's dominance in East Asia, the membership of sub-regional economies, its focus on processes and shared challenges, and its own creation of cross-border electricity trade initiatives makes it a reasonable choice for exploration of cross-border electricity trade expansion and the governance required to see increased sub-regional EMI. In addition, ASEAN's

combined policy focus on both climate change mitigation and electricity access shifts the focus from one or the other to a multi-faceted policy structure that utilises the political drive behind poverty alleviation to simultaneously address environmental and climate concerns that are linked to sub-regional economic and development challenges (Ürge-Vorsatz & Herrero, 2012).<sup>16</sup> Whether national governments similarly see the IPE of cross-border electricity trade will be answered in Chapter 4, Research Question 1, *How do governments understand the IPE of cross-border interconnections in Southeast Asia?* 

This section has examined the growth-environment nexus in East Asia, explaining the climate, economic, and energy imperatives for cross-border electricity trade generally. The following section will drill down further into the sub-region of Southeast Asia by examining the growth of cross-border electricity trade as a response to climate, economic and energy concerns previously introduced.

# 2.3 Cross-Border Electricity Trade in Southeast Asia

This section will combine literature across sustainability research and East Asian studies by examining cross-border electricity trade in Southeast Asia. This section will do four things: (1) introduce EMI broadly as well as in Southeast Asia specifically; (2) briefly touch upon the governance of EMI globally, making the case for ASEAN as the most relevant institution in the case of cross-border electricity trade in Southeast Asia; (3) discuss ASEAN's governance shortcomings as they relate to EMI and cross-border electricity trade; (4) introduce ASEAN's own cross-border electricity trade solutions and its premier EMI project, the ASEAN Power Grid, which is the subject of data collection and interviews presented in chapters 4-6. Coupled with the literature reviewed in the previous section, this section will provide an introduction to the literature necessary for answering RQ2, *how does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets*?

### 2.3.a Electricity Market Integration

Three primary models of electricity market integration are employed globally— (1) consolidation (of markets and system operators) and (2) coordination (of system operators), or (3) a hybrid of both 1 and 2 (IEA, 2014). Consolidation is

<sup>&</sup>lt;sup>16</sup> This issue will be discussed throughout data chapters 4-6 and in Chapter 7, *Conclusions*.

more feasible in already integrated or similar markets, coordination ideal for different market structures and areas with geographical variability (p.9). In the case of ASEAN, the IEA (2014) recommends a hybrid approach—coordination and consolidation at the ASEAN and sub-ASEAN level, utilising ASEAN's own incremental approach.

This ASEAN approach to incremental integration is being done via the progression of ASEAN subsystems, which are made up of smaller groupings of ASEAN member countries based on geographical location and power systems. These systems are: the upper west system, which consists of the Greater Mekong Sub-Region (GMSR, established in 1992), including Cambodia, Lao PDR, Myanmar, Thailand and Vietnam; the lower west system, consisting of Indonesia, Malaysia and Thailand-Growth Triangle (referred to as IMT-GT, established in 1994) and Indonesia, Malaysia and Singapore (referred to as IMS Growth Triangle sub-region); and the east system, or BIMP-EAGA (established in 1994), the Brunei Darussalam-Indonesia-Malaysia-Philippines East Asia Growth Area (ASEAN, 2010, p. 28). The case study selection for this research was chosen from these subsystems, resulting in the examination of the BIMP-EAGA subsystem in Chapter 4; GMSR was originally also chosen for a mini-case study, but later abandoned in the case study design (discussed in Chapter 3, section 3.3.a). To date the IEA hybrid approach recommendation still makes sense for ASEAN economies, and is complemented by the variety of market structures represented in the sub-region.

It is this author's distinction that cross-border electricity trade is the physical action of trading electricity across borders, whereas EMI is the structural and procedural process required in order to physically trade. The action of EMI facilitates cross-border electricity trade, whereas cross-border electricity trade cannot exist without some level of EMI. The farthest progression of EMI would be power pools, however, in the case of this research the focus is simply on any increase in cross-border electricity trade and movement towards the creation of the APG, a type of power pool based around sub-regional cooperation and multilateral electricity trade.

Conventional IPE analysis of interconnections proceeds from a strong market and competition-logic, where the ideal is establishment of a strong and efficient national electricity market as a necessary prerequisite in order to engage in cross-border electricity trade (World Bank, 2008; Pollitt, 2004; Oseni & Pollitt, 2014). The development of national electricity markets in industrialised economies has predominantly occurred first locally and progressed to interconnection between different service areas, via either vertical or horizontal integration depending on the economy and partners involved. National electricity systems that centrally dispatch energy from different stations to provide electricity are referred to as power pools (Oseni and Pollitt, 2014). Regional and international power pooling is one mechanism that allows for the trade of electricity across borders. Regional power pooling is the combining of government:

efforts to create more robust regional power grids with the potential of lowering capital investment requirements across time and reducing operational costs (World Bank, 2008).

Bilateral pooling agreements between two economies are also conceivable, and additional participants can be added over time, expanding the scope of regional power pools as they develop. The development of regional power pools requires significant upfront costs and investments, and so the involvement of multilateral, independent institutions is encouraged as a means to facilitate and manage capital and human costs.

Overall ASEAN electricity markets are structured around a few key factors: markets are predominantly state driven, with state oversight even in liberalised electricity markets; limited competition with few exceptions (Singapore, Thailand, and the Philippines); presence of supply and domestic electricity subsidies; a range of development levels within national electricity markets; and varying levels of electrification—Cambodia, Lao PDR, and Myanmar with the lowest electrification rates in the region (Wu, 2019). There is no one-size-fits all recommendation for EMI and electricity market development in ASEAN, however, there are common—as well as unique—solutions for each independent market and the sub-region as a whole.

Junlakarn and Wangiraniran (2016) assessed market attractiveness of each ASEAN nation using a modified framework adopted from the Global Infrastructure Investment Index. This assessment sought to provide recommendations for improvements that would increase investment in the sub-regional power sector as a response to economic growth and energy demand. The authors selected 20 criteria from the Index to "assess an overall situation of power generation investment in the region", aiming to judge the attractiveness of ASEAN markets for power generation investments (p. 496). Based on their framework Junlakarn and Wangiraniran found that three countries in particular are attractive for power sector investment, Singapore, Malaysia and Thailand; Malaysia, one of the BIMP-EAGA countries examined in this research, is

attractive due to its "mature infrastructure, low risk environment and ease of doing business" (p. 477). The Philippines, Brunei and Indonesia fell within the middle of the ranking; notably the largest weakness among these countries was infrastructure, followed by economic, business and financial factors.

Ultimately, the link Junlakarn and Wangiraniran seek to make is that growth in the sub-regional power sector is dependent on infrastructure investment, which is limited by weaknesses in individual economies. These weaknesses can be easily improved with ASEAN cooperation, addressing business and legal concerns for investors, increasing individual market strength through transmission systems like the APG, and developing common regulatory guidelines (p. 478). Each of these issues is, ultimately, an individual market issue and requires reform of current systems and processes at the national and sub-regional level across ASEAN.

The hybrid and common market structures currently in place among ASEAN member economies are, based on a review of the literature, not entirely conducive to the increased integration of renewable energy and growth in cross-border electricity trade as a response to common global energy challenges. The introduction of clean energy technologies for electricity production creates grid and market concerns, including physical pressures, flexibility concerns, and power quality (Zhang et al. 2008; Wu et al. 2012; Oseni & Pollitt 2014; Chang & Li 2015; Huang et al. 2019). Huang, et al (2019) found in their multi-country study of grid flexibility that while flexibility varies across the sub-region and the technical opportunity exists, as a whole ASEAN nations are not currently in a favourable position for increased renewable energy integration without sub-regional cooperation and multi-country engagement (p. 719).

This does not mean, however, that market structures need to be completely revamped. Countries that are better prepared cluster together, as do countries that are least prepared and moderately prepared (Huang et al., p. 719). In the case of these clusters predominantly similar adjustments are required improved infrastructure among least prepared economies (Cambodia, Myanmar, Vietnam) and improved transmission access to neighbouring electricity markets among moderately prepared economies (Indonesia and Malaysia). Among the better prepared nations needs cluster primarily in pairs—increasing interconnection capacity (Singapore and Thailand), improve forecasting practices (Brunei and Singapore), and increased interconnections among the one outlier (Philippines, also the most flexible economy according to the research). Ultimately Huang et al. find that grid flexibility increases with increased market integration, creating an environment where the problems and solutions are intertwined in the progress of EMI.

Here an argument can be made for sub-regional interaction and cooperation in effect, APG efforts will likely address many of the market concerns found within ASEAN electricity markets. Chang and Li (2015) argue that increased EMI can improve existing grid and market limitations, including the mobilisation of stranded renewable resources, coordination of policies and standards, and even mechanisms for trade (p. 39). In addition, knowledge transfer, the sub-regional promotion of renewable energy use vis-à-vis coordinated EMI efforts, and finance and investment region-wide, enabling poorer countries to take part in the necessary physical development.

## 2.3.b Governance of Cross-Border Electricity Trade

There is overwhelming consensus that institutional oversight is a requirement for significant increases in the practice of cross border electricity trade (Antweiller et al., 2001; Pritchard, 2003; Oseni & Pollitt, 2014; IEA, 2015; Andrews-Speed, 2016; Li & Kimura, 2016; Polit, Yang & Chen, 2017; Grossi et al., 2018; among many others). What does effective and active oversight and management of cross-border electricity trade look like? First, we must consider the issue of governance—the management and guidance of processes to address societal issues and transitions towards collective responses to global problems (Evans, 2012). Governance is the process of decision making undertaken by socio-political groups, states, and even institutions, as a means of addressing problems on a local, national or global scale.

Governance specific to cross-border electricity trade and regional energy market integration is a complex issue when juxtaposed with global climate initiatives and trade regulations. Many of the necessary monitoring and implementation of EMI projects must first occur at the national and subnational level (Li and Chang, 2014); however, global climate and trade regulation is largely overseen by two dominant global bodies, the UN Framework Convention on Climate Change (UNFCCC) and the World Trade Organisation (WTO).

When it comes to climate change governance the prime actor is the UNFCCC. The UNFCCC oversees global environmental and climate change governance through the coordination of state-level action in the form of treaties (UNFCCC, 2020). The UNFCCC, which was adopted in 1992, has been largely ineffective in past endeavours (See: Kyoto Protocol). Under the UNFCCC, the recent COP 21 Paris Agreement is the largest global initiative to address changes to the earth's

climate (Bultheel, et al., 2016), but accomplishment of signatory targets has been mixed. UNFCCC governance has fallen short in providing a single, universal treaty or comprehensive climate institution that incorporates a series of institutions and regimes with multiple policy elements and multiple actors that spans the globe and successfully incorporates regional and national climate considerations (Keohane & Victor, 2011). Keohane and Victor (2011) argue that, given the 40+ years spent trying to address climate challenges with no single regulatory body to speak of, a regime complex remains a more flexible and adaptable choice; they even go as far to suggest increased focus on a climate regime that includes regional trading systems, WTO incorporation, and the strengthening of the UNFCCC to act as a convenor and negotiating forum. ASEAN, in practice, meets many of these requirements, operating as a sub-regional regime, utilising WTO rules, and adopting global climate goals and initiatives. Based on this, as well as other criticisms<sup>17</sup> of the scope of the UNFCCC to govern the intersection of global climate and trade governance, the UNFCCC does not meet the institutional requirements needed to foster increased cross-border electricity trade and power sector responses to climate change.

While the WTO has been largely successful in reducing tariff barriers to trade and increasing transparency it has played little role in global energy markets and impacting climate change responses involving higher shares of VREs. Today the WTO does not have any energy-specific trade measures,<sup>18</sup> and it only address clean energy technologies through general rules, which largely target industrial policies that encourage unfair trade competition. Electricity plays no significant role in WTO rules as a result of the fact that it is both a good and a service; production and transmission of energy requires technology, which falls under the purview of intellectual property rights (WTO, 2009). In addition, "the [energy] sector shows a high level of governmental involvement which calls for

<sup>&</sup>lt;sup>17</sup> Notably, for example, the COP21 Paris Agreement does not mention trade at all, even though trade and economic development are a major catalyst for the changing global climate (Dent & Richardson-Barlow, 2016). In addition to Keohane & Victor (2011) see also: Epps & Green (2010); Whally (2011).

<sup>&</sup>lt;sup>18</sup> The Environmental Goods and Services (EGS) liberalisation trade negotiations were initiated under the Doha Development Round in 2001. The EGS will, it is hoped, clarify boundaries of environmental and clean energy trade and help further progress in environmental protection among trading partners. However, the progress of the negotiations has been held up over disputes about product classification and the delayed progress of Doha. A number of Asia-Pacific regional trade agreements have attempted to further clarify EGS trade, but progress is slow and will still not result in WTO clean energy specific rules (Dent & Richardson-Barlow, 2016; WTO, 2016a).

coherent rules on competition and government procurement" (WTO, 2009, no pagination). At the time of this writing the WTO has failed to adequately provide a platform for governance of cross-border electricity trade in-line with the specificity and cohesion of other WTO trade governance initiatives.

Therefore, given the paucity of impact the UNFCCC and WTO have, to date, had in EMI and cross-border electricity trade, respectively, the issues of governance moves beyond the two dominant global governance bodies in trade and climate change. Here, the author believes, ASEAN governance has a role to play; this assertion is supported by expert interviews (*Former Government—Informant 1, Multilateral Organisation—Informant 2, Multilateral Organisation 4, Think Tank 5, Academic 10, Academic 21, Business Leader 22, Academic 26, Former Government 27, Former Government 28*), a thorough review of the literature, and analysis of cross-border electricity trade progress (Chapter 4, *Sub-Regional Market Factors*). The multiple dynamics at play in ASEAN EMI (global, regional, sub-regional and national) also suggests that a regional or sub-regional organisation may be better equipped to accompany regional or sub-regional goals.

Governance considerations relevant to cross-border electricity trade and EMI have been identified by Oseni and Pollitt (2014) as necessary in order for market integration to take place. These criteria include national commitments to free trade; efficient market design; governance support; jurisdictional considerations & management; transmission capacity; operational national electricity transmission systems; and management of distributional effects. Deeper analysis of this criteria will take place in Chapter 5, aiding the answering of the three primary research questions:

- 1. How do governments understand the IPE of cross-border interconnections in Southeast Asia?
- 2. How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?
- 3. What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian development?

Where this subsection has introduced the governance of cross-border electricity trade, making the case for the role of ASEAN over other governance institutions adjacent to this research space, the following subsection will discuss ASEAN governance of cross-border electricity trade as it relates to the sub-regional focus of this East Asian oriented research. The following subsection will introduce the organisation of ASEAN, examine relevant institutional limitations,

and introduce its EMI initiatives and power sector connectivity goals in the context of cross-border electricity trade expansion and sub-regional climate and economic goals.

### 2.3.c ASEAN Governance

Among the many regional governance organisations in East Asia, ASEAN remains the "primary shaping mechanism in these bodies" (Foot, 2011, p. 1).<sup>19</sup> As a result, the majority of regional institutions in the broader East Asian region (the ASEAN Regional Forum, aspects of the Asia Pacific Economic Cooperation (APEC) Forum, the ASEAN Plus Three, or APT, and other ASEAN member organisation such as ASEAN Plus Five and ASEAN Plus Eight, as well as the East Asian Summit, EAS) are modelled after the original formation of ASEAN (Foot, 2011). In addition, many of ASEAN's regulatory norms are highly reflective in subsequent organisations—non-interference, respect and acknowledgment of the importance of territorial boundaries, and the supremacy of sovereignty (Foot, 2011). Many of the alternative regional and sub-regional organisations also follow in the footsteps of ASEAN's own focus on equality among members, mutual prosperity, and growth of a regional community, setting the stage for recognition of the importance multilateralisation plays in driving both positive international relations and the agenda of many regional initiatives (Sudo, 2003; Acharya, 2009; Foot, 2011).

Additional aspects of ASEAN's organisational structure have carried over into other regional and sub-regional groupings. This includes an institutional focus on cooperative efforts, transmission of ideas, and the use of social and institutional pressures as opposed to binding legal constraints (Foot, 2011). A number of the institutional practices within ASEAN, including a preponderance of meetings, internal membership groupings, issue focused initiatives, and a rotating leadership or "hosting" duties, have also carried over from ASEAN into other regional and sub-regional organisations. As a result of ASEAN's primacy in the creation and operation of many sub-regional groupings (Acharya, 2018), ASEAN is a natural multilateral governance institution to explore when it comes to questions of sub-regional or regional initiatives and responses to shared challenges.

<sup>&</sup>lt;sup>19</sup> In this research the author refers to ASEAN when referencing the organisational structure. When discussing individual ASEAN economies the author references 'ASEAN member states/economies' or references individual states, groupings of states, or subsystems by name.

In the case of cross-border electricity trade and EMI in Southeast Asia, ASEAN is the most active governance structure. However, there are governance limitations that ASEAN has experienced consistently since its inception. Many of the positive commonalities of ASEAN's governance structure are not the only aspects to carry over into other regional governance institutions—the shortcomings are as well. Supremacy of state sovereignty has resulted in a region-wide focus on non-intervention and its subsequent normalisation (Foot, 2011). As a result, non-binding resolutions and lack of legality are a norm in East Asian regional institutions, and often associated by critics with slowly progressing initiatives, frequently moving deadlines, and a tendency towards bilateral not multilateral cooperation (ADB, 2008a; United Nations Conference on Trade & Development [UNCTAD], 2017). These limitations are a common theme in ASEAN relations and are reflected in initiatives across a wide span of issue areas.

One of ASEAN's greatest strength—its diversity—is also its greatest weakness. While ASEAN member states share some commonalities previously referenced, such as culture, history, and language, within each overarching similarity there are many differences (Albert, 2017), in some cases hundreds. For example, ASEAN member states have shared colonial histories. However, few of them share a similar controlling colonial power. Where culture within ASEAN is similar in that it is East Asian, within ASEAN member states there are hundreds of cultures and languages represented. The presence of different religions (Islam in Brunei and Indonesia, for example), different government systems (a sultanate in Brunei versus a parliamentary representative system in Singapore), and different levels of democracy (Myanmar and Cambodia versus Singapore) all complicate national priorities when juxtaposed with regional, ASEAN goals. Conflicting national concerns cause a lack of movement on some issues within ASEAN, and electricity trade and energy policy are no strangers to this stagnation.

Further arguments against ASEAN's effectiveness include a lack of sub-regional identity and limited unifying message (Hutt, 2017). When ASEAN was first formed more than 50 years ago the member economies were unified around a message of anti-communism amid the cold war (Beeson, 2009). Today that message no longer exists, and instead the organisation is arranged around a variety of issues and challenges, for which each member economy has its own problem, perception, and goal—not the least of which is energy security.

As with many analyses of ASEAN projects, a major limitation to progress in ASEAN initiatives is the bureaucracy of the organisation itself. Intersecting committees, overlapping organisational groupings, and a plethora of subinitiatives litter the progress of institutional development. Multiple feasibility studies with different outcomes, long wait times on program progress assessments, and competing interests within ASEAN sponsored institutions have resulted in difficulty assessing the long-term feasibility of the APG and other EMI efforts. Interview subjects from within relevant government organisations are positive, but very vague as to the eventual completion or adjustments to, the APG. Just as there is ambiguity within ASEAN's overarching goals (the ASEAN Economic Community and sub-regional interconnection and integration), there is also ambiguity within the APG and ASEAN EMI (Leu, 2011). Indeed, ASEAN itself has ambiguous organisational governance mechanisms, as the operation of the organisation relies on a practice of non-interference, fraught mutual trust, and non-binding resolutions.

In addition, within ASEAN an aversion to legalisation has resulted in "…voluntary compliance and polite non-interference with each other's actions even in the face of violations of commitments…" (Habito et al., 2004). As it currently stands, the ASEAN legalisation processes falls into the category of "soft legalisation", identified by Abbott et al. (2000, p. 404) to primarily include non-legal norms, as opposed to binding rules; vague principles, as opposed to precise, highly elaborate rules; and diplomacy, as opposed to international courts and domestic legal application.<sup>20</sup> To address this shortcoming, both in ASEAN and in other regional institutions, policy recommendations often include strengthening peer pressure processes and improving (or developing) legal ramifications.

Additional ASEAN limits coincide with global movements towards more highly integrated, neoliberal market economies. While community building has been a staple of ASEAN policy for 20 plus years it has been met with opposition from national level actors for just as long. According to Chandra (2016) "The rise of protectionism, as an expression of economic nationalism, in particular, has been seen by many experts and practitioners alike as a key hindrance to ASEAN's effort to deepen its economic integration project" (p. 1). Chandra goes further to point out that protectionism and economic nationalism within ASEAN are rooted

<sup>&</sup>lt;sup>20</sup> Kahler (2000) takes this delineation one step further, articulating that while ASEAN does lean more towards soft legalisation than hard, there are exceptions to every rule; however, to date, few examples of legalisation processes and enforcement in ASEAN exist.
in the national economic policies of member economies and perceptions of free trade and liberalisation. For example, while ASEAN pledges sub-regional reductions in non-tariff barriers to trade, developing and emerging economies within ASEAN have historically expressed their scepticism that increased competition would benefit their national economies at this stage in their development (p. 2). This scepticism has carried over into other economic issues, giving rise to economic nationalism throughout the sub-region (Jones, 2016).

Clearly, these shortcomings have the potential to be augmented by other international, multilateral institutions. In fact, the role of the WTO may become increasingly important as cross-border electricity trade grows and dispute settlements become relevant.<sup>21</sup> Given the difficulty in ceding to international, multilateral organisation and the value-set of ASEAN (Kahler, 2010) when it comes to sub-regional growth in EMI and cross-border electricity trade, ASEAN is currently the supreme sub-regional governance actor. Development of EMI will no doubt precipitate the need for inclusion of rules and regulations specific to cross-border electricity trade; however, likely not until disputes arise and/or:

require settlement that ASEAN's [practice of] non-interference and social pressures... cannot address (Business 22).<sup>22</sup>

These exact issues remain relevant in cross-border electricity trade, EMI, and sub-regional climate responses, and will remain present throughout chapters 4-6 as they are repeatedly raised among expert interviews.

The previous subsections have established a number of issues relevant to this research: the need for cross-border electricity trade, the role of governance, or management, of cross-border electricity trade generally, and the role of ASEAN as the governor of cross-border electricity trade in the East Asian sub-region of Southeast Asia. The following sub-section will introduce ASEAN's cross-border electricity trade energy and climate challenges among member states.

<sup>&</sup>lt;sup>21</sup> The Asia Pacific Economic Cooperation (APEC), for example, flirted with WTO involvement in dispute settlement practices in the 1990s, although ultimately compliance—not legalisation— was the result (Kahler, 2000).

<sup>&</sup>lt;sup>22</sup> In the long term a climate regime that incorporates multiple organisations, as laid out earlier in this chapter (the regime complex envisioned by Keohane & Victor, 2011) may be a logical development; however, at this point in time no regime like this exists.

#### 2.3.d Cross-border Electricity Trade Solutions in ASEAN

ASEAN began including energy within its sub-regional integration efforts early on in the global fight against climate change, first committing to the Agreement on ASEAN Energy Cooperation in 1987, where ASEAN member states agreed to cooperate on shared energy and environmental challenges. This was followed by the establishment of the ASEAN Centre for Energy (ACE) in 1999 (ACE, 2019). ACE acts as an intergovernmental organisation underneath the umbrella of ASEAN, meant to represent the interests of the 10 member states in energy and related sectors; today this includes sustainable development, sub-regional environmental stability, and climate change responses, as well as coordination of energy policy strategy within the region (ACE, 2019). Working with the relevant energy ministries, ASEAN related networks (ex: sub-sector networks within ASEAN, Specialised Energy Bodies, etc.) and the ASEAN secretariat, ACE facilitates the implementation of the ASEAN Plan of Action for Energy Cooperation (APAEC) within the wider ASEAN community and integration goals (ACE, 2019).

Outlined in APAEC 2016-2020, ASEAN set an aspirational 23% renewables target that followed on ASEAN's previous renewable integration targets and includes an additional goal to achieve a 20% reduction in energy intensity by 2020 and 30% by 2025 (ACE, 2015). Increased EMI and sub-regional electricity interconnection are at the centre of these ASEAN energy goals, reflected in a variety of programmes, projects, and ministerial meetings, as well as a key goal of the APAEC since its inception in 1998 (Andrews-Speed, 2016). ASEAN has committed to deeper renewable integration on paper, but there is some scepticism about the viability of such targets given a regional and sub-regional reliance on coal and the dominance of hydrocarbon-oriented companies in the regional political economy. This clash of interests—ASEAN goals versus national reliance on hydrocarbons—remains a theme that is present throughout this research (and is reflected in IPE analysis at a later point, in chapters 4-6).

ASEAN energy and electricity access related projects run the gambit of broad, cooperative programs and commitments that span a variety of energy resources and environmental concerns (Andrews-Speed & Hezri, 2013). The ASEAN Power Grid Project (APG)<sup>23</sup> was announced in 1997 based on recognition that

<sup>&</sup>lt;sup>23</sup> The APG technically falls under the management of the Heads of ASEAN Power Utilities/Authorities (HAPUA), tasked with promoting energy security and energy cooperation across the region (ASEAN, 2013) in tandem with regional energy goals and additional

integrating electricity systems across the region would benefit ASEAN Vision 2020 goals, including greater energy access, reduced energy poverty, and increased sub-regional connectivity (ASEAN, 1997). The Greater Mekong Sub-region (GMSR) and the BIMP-EAGA are included in current power grid expansion proposals utilising ASEAN member states in the APG, and the Australian-Asian [Power] Grid (AAG) proposal also incorporates the ASEAN governance structure in the implementation of deeper power systems integration.<sup>24</sup>

Via the APG ASEAN member economies primarily trade electricity between transmission system operators (TSO). At this point in time an ASEAN merchant model is extremely unlikely, however, each case within the sub-region is different, as is each proposed interconnector (IEA, 2015). While cross-border interconnectors are desired, the performative nature of neo-developmental statism means the region isn't there yet, and interconnections have been implemented primarily on a bilateral basis (ACE, 2018b); ASEAN as a whole is nowhere near internal market coordination or a uniform standard that everyone works to, so a piecemeal approach has been taken across the sub-region (see also the subsystem approach illustrated in Figure 2.1). The current approach includes TSO-TSO interconnections, as well as "joint venture agreements between state-owned companies", where differing national regulations and lack of uniformity creates delays and complications (IEA, 2015, p. 15). The establishment of "an APG transmission system operators' institution and an APG generation and transmission system planning institution" would solve some of the harmonisation and standardisation issues (IEA, 2015, p. 16), but the development of these institutions is still underway within ASEAN (ACE, 2018b).

Today the APG complements ASEAN's Paris Agreement commitments, and ASEAN targets and the APG are further justified via the potential for carbon emissions reductions via the decoupling of economic growth and energy

coordination among ASEAN members and the relevant energy ministries and national level government bodies. HAPUA has no APG enforcement capabilities. HAPUA was established in 1967 with the aim of increasing electricity interconnection and collaboration, however, it remained relatively dormant until the 1997 announcement of the APG and its sister project, the Trans ASEAN Gas Pipeline (TAGP), as part of the ASEAN Vision 2020; an MOU for the APG was signed by ASEAN member states in 2007 (ASEAN, 2007).

<sup>&</sup>lt;sup>24</sup> The AAPG will not be focused on due to the large scope of the project, limitations in project development, and interview insight into the lagging nature of such a large, multilateral project within the scope of governance, societal and cultural pressures. Further insight into the AAPG can be found in Halawa et al. (2018) or Gulagi et al. (2017).

demand. The 10 ASEAN member states were among the early signatories of the Paris Climate Accord, signing the agreement in April 2016 (EU, 2017). ASEAN member states have also agreed to a sub-regional goal, increasing renewable energy up to 23 percent of the region's primary energy mix by 2025 (ACE, 2017). The ASEAN Energy Outlook (AEO) 2015-2040 provides three policy scenarios, similar to those used by the IEA in their own global outlook, providing policy and structural recommendations for decreasing energy intensity, increasing the share of renewables, and addressing energy and electricity concerns. ASEAN believes all three scenarios complement the Paris Climate Accord and has incorporated recognition of man-made climate change into the majority of its energy and environment policy documents (ACE, 2017).

Analysis of the motives behind these targets and commitments, and how they are reflected in the IPE of current and projected cross-border interconnections as well as sub-regional power sectors emerges in chapters 4-6, reflected in both the market factors, reform needs, and governance challenges outlined in these later data chapters.

In 1997 ASEAN member economies prioritised energy security and access by establishing the APG blueprint (ASEAN, 2016). The APG Blueprint seeks to respond to energy security concerns by increasing connectivity and developing interconnections among member economies all while balancing concerns related to energy poverty, variability, service access, and infrastructure restrictions (ASEAN, 2016). This blueprint also follows the common ASEAN model of facilitating knowledge exchange and increasing economic and social connectivity in line with the wider AEC goals. A further vision for the APG and ASEAN connectivity is addressed in the Master Plan on ASEAN Connectivity (MPAC) 2025, overseen by the Heads of ASEAN Power Utilities Authority (HAPUA) (ASEAN, 2017a). This specific policy focus within ASEAN has resulted in pledges for increased integration of renewables, diversification of the subregion's energy mix, increased connectivity between members, national and subregional electricity market expansion, and financial pledges at the national and sub-regional level. This includes incorporation of the energy goals within the ASEAN Vision 2020, an agreed upon vision of ASEAN as a "concert of nations" committed to shared prosperity and peaceful resolutions to common problems, including territorial, economic, societal, and environmental (ASEAN, 1997a).

Development of the APG has followed recommendations outlined in the ASEAN Interconnection Master Plan Studies (AIMS), of which three have been completed and updated since 2003 (AIMS 2003, 2010, 2018). Each AIMS is

carried out by regional experts with oversight and management from HAPUA. The first AIMS studies (AIMS I) was published in 2003 and actually found that it was economically prohibitive to construct a sub-regional grid, instead advocating for bilateral connections across the region; specifically, AIMS I proposed eleven interconnections be made by 2019 (not all of these were achieved). AIMS II was published in 2010, following the reorganisation of HAPUA, and found that an ASEAN power grid was, in fact, economically viable but should be approached incrementally, via subsystems and then linkages developed between and among subsystems, eventually forming the wider APG (this approach is discussed in more detail in Chapter 4, *Sub-Regional Market Factors).* 





Source: Author's creation from description in Li & Kimura, 2015, p. 42.

Section 2.3 has discussed the cross-border electricity trade imperatives and solutions present in East Asia, including an introduction to: EMI broadly and in Southeast Asia specifically, energy and climate imperatives for increased cross-border electricity trade in Southeast Asia, the governance of cross-border electricity trade generally and in Southeast Asia specifically, and the ASEAN specific EMI projects that are relevant to this research and questions regarding the IPE of cross-border electricity trade in the Southeast Asian sub-region. The

following section, 2.4, will transition to the foundational IPE theories relevant to this research, providing context for later IPE analysis and identifying gaps in the current IPE literature as it can be applied to cross-border electricity trade in Southeast Asia and EMI initiatives among ASEAN members.

## 2.4 International Political Economy & Foundations of Trade

One of three bodies of literature relevant to this research falls into the broad category of IPE. The literature covered in this section, coupled with Section 2.5, will provide context to understanding the IPE of cross-border electricity trade and answering RQ3, *what can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian Development?* 

IPE is a common theme in many forms of economic analysis, often referred to as the study of production, distribution and consumption of resources and the government institutions that interact with these forms of economic activity (Krugman & Obstfeld, 2009). It is a main tenet of IPE "that individual and sectional interests not only do compete for the possession of resources but that this competition is healthy and should be encouraged" (Atkinson, 1991, p. 3). The OECD defines political economy analysis as being:

Concerned with the interaction of political and economic processes in a society; including the distribution of power and wealth between groups and individuals, and the processes that create, sustain and transform these relationships over time (OECD, 2010, pp.2).

IPE is, broadly, the combination of political science, international relations and economics to explain global events and economic interactions, often among nation states and relating to the management of political and economic affairs (Strange, 1988, p. 8). As IPE is a large, multidisciplinary field of study with a vast array of literature across disciplines, this research will focus on a narrow selection of literature chosen for its direct relevance to EMI and role in answering the previously posed research questions. Initial reviews of IPE literature included an examination of trade policy broadly, seeking to identify the specific role of trade within cross-border electricity trading. This included exploration of trade policy limitations, the history of trade policy development sub-regionally, regionally and globally, modern trade governance via the WTO, and even trade policy linkages within climate governance structures and global energy transitions. However, as the research expanded and cross-border electricity trade was conceptualised within East Asia and eventually Southeast

Asia, this literature became less relevant.<sup>25</sup> Instead, market structures emerged as the necessary focus for creating an environment conducive to increased crossborder electricity trade and integrated electricity systems in national markets and the creation of a fully integrated sub-regional electricity market.

Here the IPE of cross-border electricity trade emerged more clearly in the literature—examination of regional and international market structures revealed the dominant role of neoliberalism globally and, ultimately, the basis for much of the world's economic systems of exchange. Neoliberalism emerged as one of the areas within IPE literature most relevant to improving EMI and cross-border electricity trade within national, regional and global markets. In addition, in an effort to identify how governments understand the IPE of cross-border electricity trade affects opportunities for renewable energy to participate in national markets (RQ2), political and economic structures emerged as central to progress from an IPE perspective. The following subsection will introduce neoliberalism as a foundation of IPE analysis. This examination will provide context to the spread of neoliberal economic policies in a region and sub-region historically dominated by state-led economic policy making.

## 2.4.a IPE Foundations: Neoliberalism & the World Economic Order

Neoliberalism, referred to in this research in relation to the existing, dominant global economic ideology, is the first step to understanding current international trade and global economics. Neoliberal ideology is rooted in classical liberalism but has evolved past a primary focus on markets and the free exchange of goods and services to now include political and social factors of the global system as well as power<sup>26</sup> dynamics across actors (Hill, 2007; Strange, 1998). The global

<sup>&</sup>lt;sup>25</sup> See for example: Aaron (2011) for work on renewable energy subsidies in the WTO; Antweiler et al. (2001) on free trade and the environment; Araya (2016) on the environmental goods agreement and sustainable development goals; Bhagwati (1995) and Panezi (2016) on the global proliferation of FTAs; Epps and Green (2010) on the WTO and climate change; Mundaca and Richter (2015) on economic policies targeting renewables; Urpelainen (2013) on trade and international environmental cooperation; and the International Centre for Trade and Sustainable Development (ICTSD) for a wide variety of research on the intersection of trade and climate change, among many other resources.

<sup>&</sup>lt;sup>26</sup> Power comes into play repeatedly throughout this research via the challenges of working with actors across global, regional, sub-regional and national markets and the incentives behind actor actions. The author has explored various definitions of power that could be utilised in understanding power dynamics in energy markets, including: political science-based methods of power via Foucault's (1991) notion of knowledge and discipline; international relations-based theories of power via Luke's (2005) three faces of power—decision making, non-decision

spread of neoliberalism is largely a result of western adoption, beginning with the United States and United Kingdom and spreading from there (Hill, 2007, p. 1); as such much of neoliberal thought is conceptualised differently from place to place but does, ultimately, include the "commodification of resources" (Radice, 2008, p. 1155). Like liberalism before it, neoliberalism is able to flexibly meet the needs of those using it; similar to many political ideologies, neoliberalism is flexible and is a conceptual redefining of the root concepts of liberalism based on the needs and changes of the global economic system (Radice, 2008).

Today neoliberal economic thought is centred on "free markets in land, labor and money; models the state on the enterprise, the citizen on the consumer, and governance on business management" (Radice, 2008, p. 1). While Neoliberalism may vary in its implementation from economy to economy or state to state, there are a few root neoliberal beliefs that can be found across different economies. These include "a belief in the self-regulating capacity of the market" (Radice, 2008, p. 1155), commitment to free trade, deregulation of financial markets and "privatisation of state enterprises" (Hill, 2007, p. 2) and instructions to leave the market alone (Weiss, 2012). In their analysis of neoliberal climate policy MacNeil and Paterson (2012) argue for a nuanced conceptualisation of neoliberalism whereby neoliberalism is understood to be made up of various parts depending on where it is utilised:

All neoliberalisms are distinct; all neoliberalisms have competing logics and interests that fight for expression in the policy process; and all neoliberalisms exist within pathdependent institutional contexts that inform policy responses and the specific forms they take (p. 231).

Neoliberalism has experienced setbacks in relation to the ideology in theory versus in practice. Electricity market stakeholders, including those on the consumer and supply side, have made a concerted effort to deregulate—or restructure—global electricity markets in favour of more competition (Anderson, 2009, p. 70). This has been justified in an effort to reduce prices and encourage investment and innovation based on assumptions about consumer

making, and ideological; and IPE notions of power via Strange's (1988) characterization of structural and relational power. All of these definitions play a role in the author's understanding of power, given the critical realism epistemological approach the author has chosen to take (see Chapter 3, Research Design & Plan of Enquiry for more on epistemology). However, this author is particular to Strange's characterisation of power given the structural and relational challenges the author has identified in Southeast Asian EMI. However, focus is not given to power in this research as its relevance emerged in the data analysis process. Future research could focus more specifically on power, a choice that is discussed further in Chapter 7, Section 7.5.

choices and market forces resulting in "efficient supply and demand" (Hall & Nguyen, 2017, p. 100). As a result global institutions and leaders have collectively attempted to unbundle markets for the sake of prices, recommendations that have been repeated at various levels of governance and research (see, for example: World Bank, 2008; IEA, 2014, 2015; IRENA, 2018a). Unfortunately, efforts at deregulation and restructuring have not collectively fostered more efficient markets or price decreases as a result of consumer choices (Anderson, 2009, p. 71). Instead, prices have increased as many markets have become more inefficient, reflecting generator and supplier ignorance of consumer concerns, power prevalence in market structure (whereby stronger powers influence structure regardless of price or efficiency goals), lack of technological innovation, difficulty in contract negotiating, resource inadequacy, lack of consumer protection, and inadequate transparency and policing (p. 83). In short, international experiences across Europe, the Americas and Asia shows that unbundling and deregulation has proven different in practice than in theory (Anderson, 2009; Hall & Nguyen, 2017; Aris et al., 2020).

Regardless of failures in implementation of neoliberalism in practice versus theory, some level of competition and restructuring is still recommended across the literature; however, the level at which this takes place will vary across markets (Aris et al., 2020), as seen in Southeast Asia's restructured markets. Chapter 5 discussed how the Philippines' markets liberalisation, while more than its BIMP-EAGA neighbours, is still limited in implementation though it is the farthest along in the subsystem, followed by Indonesia and Malaysia, where state owned companies still dominate the sector. Hall & Nguyen found in their 2017 study that reform and liberalisation efforts across 20 developing countries over a 20 year period did not, in fact, create new markets, restructure markets entirely, or even increase competition across the majority of countries (Hall & Nguyen, 2017, p. 112). Instead, many developing markets have frozen or reversed their liberalisation efforts and in the process are increasing state involvement in the power sector in order to continue driving state investment and also state support of private investors (p. 113). Tensions between neoliberal market ideology and practical implementation in developing—and developed economies continue across power sectors and integrated markets (p. 114).

The adaptive ability of neoliberal economic policy to be conceptualised in different ways from country to country is similar to other political and economic ideologies that may differ slightly from case to case. Here neoliberalism intersects with the concept of globalisation, whereby economic, political and social processes "transcend national borders" and connect the world in increasingly complex ways (Beeson, 2014, p. 20). A large part of this interconnection is economic in nature, via trade relationships, global value chains, and international production networks (Radice, 2008; Beeson, 2014; Kimura, 2015; Taguchi & Murofushi, 2014). The process of globalisation has spread neoliberal economic policies, creating a standard global expectation about liberalisation, free trade, and open markets (Wong, 2004).

Notably, in this context globalisation and neoliberalism also intersect with ASEAN and the global challenge of climate change. The continued success of the current, interconnected global system is dependent, in many cases, on finding solutions to transnational issues; many of these global solutions require monitoring and enforcement through a system of governance. The interconnectedness of ASEAN member economies via trade and economic cooperation, and the building of a sub-regional economic community centred around shared goals and values, is one of many global examples where the theory of interdependence (and predecessor of globalisation) established by Keohane and Nye in 1970 is alive and well. Also related to this research, globalisation and neoliberalism have both exposed power imbalances between national and subnational governments and governance.<sup>27</sup>

The elaboration of neo-developmental statism is one way of furthering this analysis and can be applied in a variety of East Asian economic systems where energy and environmental concerns lead economic development challenges while simultaneously creating opportunities. In the next subsection, focus will be given to IPE in East Asia and the common economic practices employed in these centrally planned economies. This information sets the scene for the examination of the common economic practice of DS, explaining how this government directed policy making is realised in the context of EMI and crossborder electricity trade in Southeast Asia.

<sup>&</sup>lt;sup>27</sup> Chapter 6, Governance Challenges, includes an examination of the issue of trust, one of the governance challenges identified across elite interviews and documentary research. This analysis includes an examination of regional versus national policy considerations and perceptions, which are linked to similarities described here regarding neoliberalism at the global, regional, national, and sub-national level and the power dynamics that emerge.

#### 2.5 International Political Economy in East Asia

Whereas the previous section examined IPE theory and the theoretical foundation of the global, market economic system, the following section will specifically examine IPE measures as they are applied, adapted and utilised in East Asia via developmental statism. This section will situate the economic development of East Asia and regional growth in energy and environmental concerns within the framework of IPE and neoliberalism. The literature presented in Section 2.4 and this section, 2.5, will aid in answering RQ3, *what can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian development?* While all the research questions will be answered later in chapters 4-6, the basics of the IPE theory that will assist in this later analysis will be explored here.

Just as there are common arguments in support of cross-border interconnection (discussed in Section 2.3, including efficiency and welfare gains, diversification of supply, sustainability and energy goals, among others), there are also a number of common explanations for limited progress in the development of cross-border interconnections globally and approaches to understanding these limits. These include: business arguments that take into account both economic and business model approaches to designing efficient market environments (Poudineh & Rubino, 2016; Hall & Roelich, 2016); neo-realist approaches to state sovereignty, which can cross-over with nationalism and challenges of trust among cross-border trade partners (Andrews-Speed & Len, 2013); neoliberal arguments related to liberalisation and market reform, closely linked with the 'standard model'<sup>28</sup> of electricity reform (Sen, 2014); political approaches related to governance and policy limitations (Puka & Szulecki, 2014); disparities in government ideology and leadership capacity (Erdogdu, 2014); and even the 'dual market model of dual firms'<sup>29</sup> (Victor & Heller, 2007), a transitionary phase found somewhere between pre-reform and the standard model of reform.

<sup>&</sup>lt;sup>28</sup> The standard model of electricity market reform includes liberalisation, increased competition, corporatisation, divesture or privatisation of competitive sectors, and unbundling of markets (Sen, 2008, p. 1). The standard model is, in fact, in line with many of the recommendations reflected among interview subjects in this research. This model is also reflected across a variety of popular studies and has previously been recommended across global markets (Gratwick & Eberhard, 2008).

<sup>&</sup>lt;sup>29</sup> The 'dual market model of dual' firms was coined by Victor & Heller (2007) and encompasses the process whereby a firm is able to simultaneously (1) utilise resources via political connections and (2) avoid the pressure of taking on uneconomic projects via government

While the author could have used any of these approaches, individually, to analyse cross-border electricity trade among ASEAN member states, the choice to focus on developmental statism and the elaboration of neo-developmental statism reflects a merging of all these issues. The neo-developmental statism lens allows an inclusion of economic, political and governance explanations under the same umbrella—and in a geographical area where developmental statism has historically been applied and recognised. The research findings spread across chapters 4-6 highlight many of these issues, including national incentives and nationalist arguments against power sector cooperation, regional governance challenges associated with ASEAN's capacity as an institution with no legally binding power and where issues of trust persist, limits to market structure and national governance capacity, and a propensity among ASEAN member states to make performative agreements but act in their own national interest. The prior application of developmental statism (the predecessor to neo-developmental statism) in Southeast Asia also makes the use of neo-developmental statism a natural choice for the author in this context. <sup>30</sup>In addition, the contradiction between neoliberalism and legacy elements of developmental statism make the use of neo-developmental statism an interesting application in electricity market reform, where the standard reform model relies heavily on liberalisation.

It should be noted that upon further inspection in the sub-region (as discussed in chapters 4-6), competing factions within individual states also contribute to the existence of and arguments for neo-developmental statism. For example, in Indonesia and Brunei there is the existing energy regime, which includes international policy makers and ministers at the head of national departments. In addition, there are also local departments that actually run the system—the local ministries and their staff, for example—and they don't necessarily want to change, so policy moves slowly or in contradictory manners. Thus, the prevalence of national policies that don't reflect larger sub-regional targets.

pressure—all while operating in a market that is both neither entirely liberal or entirely state protected and driven. This model is described as a transitionary phase by critics (Bartle, 2007; Hope, 2008).

<sup>&</sup>lt;sup>30</sup> The author does not presume that application of neo-developmental statism elsewhere in the world, including Africa and South America (where vestiges of developmental statism have been adopted both unsuccessfully and successfully), would offer similar useful methods for analysis of power sector reform and cross-border electricity trade. Prior research on developmental statism by Singh & Ovadia (2018) offers some insight into application of developmental statism elsewhere in the world. However, further examination of this area would benefit expansion of neo-developmental statism, power sector reform, and the political economy of cross-border electricity trade.

Here, the performative nature of liberalism in ASEAN is uniting and agreeing before coming up against entrenched interests at the state level. The tension identified in interviews and data analysis lies here, in-between the different interests; there is no single, planned rationale for how to approach cross-border electricity trade, EMI, or even climate targets. Instead, there are competing notions with seemingly no resolution, and this tension can best be described as neo-developmental statism.

As a result of the conflict between neoliberalism and developmental statism within ASEAN member economies, the author has chosen to examine the tenets of neoliberalism and developmental statism in this chapter, drilling down into the two foundational theories behind neo-developmental statism in section 2.4 (previous section) and section 2.5, to follow. In the following section the author also explores how neo-developmental statism combines some elements of neoliberal policy with legacy elements of the developmental statism paradigm.

First, an explanation of developmental statism in relation to the development practices of East Asia will be provided, explaining the dominant development paradigm in East Asia; second, a description of the limits to neoliberal economic explanations of East Asian economic development practices, given state-led economic policy, will be presented, making room for the theoretical framework of this research. The completion of this section will lead into a description of how the dominant, neoliberal IPE frameworks can be combined with the East Asian IPE practice of developmental statism to create neo-developmental statism in East Asia. This application fills the gaps left when applying neoliberal economic policy to cross-border electricity trade in DS-dominated East Asian economies. This combined approach will be used as the dominant theoretical conceptualisation applied in this research to describe the IPE processes related to electricity market expansion and cross-border integration in East Asia.

## 2.5.a IPE Foundations: Developmental Statism

There are a variety of theoretical approaches<sup>31</sup> to understanding state-directed economic development and the various tools advocated by governments and international organisations as a means to achieving economic growth, including

<sup>&</sup>lt;sup>31</sup> Dent (2014) groups the variety of developmental policies utilised by East Asian states under the title state capacity theory; however, as this is not widely used by other DS theorists these theories will be described individually for the purpose of this research.

targeted industrial policy, mercantilism, and, in the case of East Asia, DS. DS is the paradigm used to explain the political and economic process whereby states:

Shape, pursue and encourage the achievement of explicit development objectives, whether by establishing and promoting the conditions and directions of economic growth, or by organising it directly, or a varying combination of both (Leftwich, 1995, 401).

In the 1980s Chalmers Johnson first identified Asian developmental states as having a close relationship with the private sector and a heavy hand in setting and reaching economic goals, what he called developmental or plan-rational states (Johnson 1982, 1987, 1999). Johnson also attributed the creation of the industrial policy of the developmental state as a Japanese invention. In practice, developmental states typically encourage the utilisation of strategic policies and developmental partnerships, "encouraging the emergence and growth of private economic institutions" (Leftwich, 1995, p.417) while also allowing government agencies administrative guidance over the economy (Johnson, 1982; Beeson & Pham, 2012) and privileging industrial governance and growth over other processes (Thurbon, 2014, p. VIII). Developmental states are believed to aid the development process by lessening the human cost and decreasing the amount of time economic transitions require, allowing emerging markets the opportunity to quickly catch up with industrialised economies (Leftwich, 1995). In addition, there is often a strong link to targeting specific industries as well as a dominant role for political elites in directing policy priorities where DS is realised (Johnson, 1982; Thurbon, 2014).

When mapping the economic geography of East Asia, the developmental state plays an important role in the history of the region. Few, if any, discussions on political economy in East Asia fail to mention the developmental state paradigm and its role in the economic miracle of Japan (the most common developmental state success story), Malaysia, the Republic of Korea, Taiwan and Thailand.<sup>32</sup> Developmental statists attribute the success of these countries to efficient and flexible policy choices which "relate economic performance to institutional arrangements centred on the state" (Moon & Prasad, 1998, p. 9). As a result, DS theory is focused on the role of the state as a strong unitary actor, and its cooperation with the private sector as a means to "minimize the risk of opportunism" by individual actors (Moon & Prasad, 1998, p. 9). Thurbon (2014)

<sup>&</sup>lt;sup>32</sup> See Singh and Ovadia, eds. (2018a) for a thorough summary of developmental state literature; or Gârdu (2015) for an overview of early statist literature in relation to East Asia

further describes DS as an important theoretical, IPE tool used to explain state goals and national interests (p. VI). Chan et al. (1998, p. 2) emphasise this point:

the main premise of the statist perspective is that, compared to other competing variables, the state offers the greatest explanatory power in accounting for the East Asian experience.

In the case of East Asia, state intervention and market interference have been built into the core of macroeconomic and industrial development practices used across East Asia and Southeast Asia (Johnson, 1982; Chan et al., 1998). The practice of developmental statism that was common to East Asian economies in the 20th centuries includes neomercantilism, strategic trade practices and targeted industrial policy still present today. Among HPAEs (highest performing Asian economies) continued utilisation of targeted industrial policy and strategic industries have transitioned industrial structures towards modern sectors as a means of increasing national welfare (World Bank, 1993).<sup>33</sup> Japan has the longest regional history with state-led growth, but China's renewable energy policy, South Korea's Green Growth Development Plan, and Taiwan, Thailand, Indonesia and Malaysia's industrial development have made use of the policy practice as well (World Bank, 1993).

Following the 2008-2009 Asian financial crisis (AFC), many emerging economies in the region more firmly integrated targeted industrial policy into their stateled macroeconomic development plans as a means to confront the multiple challenges of energy security, climate change, economic growth, and low carbon development (Dent, 2014). Asia's approach to addressing these challenges and further incorporating clean energy sector development into the domestic economic engine has been to create "strategic master plans specifically for renewables," and is a unique feature of Asia's policy strategies (Dent 2014, pp. 68). In the case of East Asia, strategic trade and targeted industrial policies have been used to target a variety of energy industries, including solar panel development in China and Thailand, oil in Indonesia, Nuclear and power sectors in Japan, and others (Dent, 2014).

The key for many of East Asia's developmental states was a combination of state governance and market forces, utilised alongside targeted industrial policy. Indeed, industrial promotion, while commonly unpopular among mainstream,

<sup>&</sup>lt;sup>33</sup> Japan is a good example of this: following the economic and security failures of the early 1900s, the Japanese Ministry of Economy, Trade and Industry (METI) took long term economic views in order to coordinate and enact creative industrial policies that targeted areas of future growth, and eventually grew into the standard of the region, state-led industrial growth (Johnson, 1982).

neoliberal economists<sup>34</sup> does expand economic options when coupled with: strong government oversight, adaption to geopolitical changes, incorporated with national competitiveness, and policies that encourage education and advancement (Wade, 1992). This argument supports one of the premises this author's conceptualisation of neo-developmental statism: that an all-of-theabove economic approach (whereby the tools of both neoliberal and statedirected economic policy making are utilised in order to achieve goals) is necessary in order to fully realize the potential of trade as a means to respond to intermittency concerns in regional and international electricity markets due to clean energy expansion in electricity generation.

In East Asia targeted industry policy making is a noted part of economic development via DS. Subsidies for hydrocarbon energy resources are an illustrator of this, and relevant to East Asia and Southeast Asia, where national energy companies continue to receive economic advantages despite conflicting with national and regional climate goals.<sup>35</sup> As with many economic policies, short term benefits need to be balanced with long-term phase outs and alternative options as industries grow. In the power sector, where reliable, affordable energy is a priority for national governments, questions about local content vs foreign content can arise. This challenge arises further in chapters 4-6, when market factors are contrasted with sub-regional interconnection goals, and indeed this emerges as a major part of national versus sub-regional contradictions in goals and outcomes of EMI.

## 2.5.b Criticisms: DS Policy

Different levels of success have been experienced in utilising the DS paradigm across East Asia. In the case of emerging economies, the traditional DS model required incentives for elites to engage in the necessary institutional reforms for long term success, a factor that has not been fully reflected in Southeast Asian states that have grappled with what comes after the DS industrialisation and techno-upgrading, or post-developmentalism (Sen, 2018, p. 6). The case of China

<sup>&</sup>lt;sup>34</sup> During the latter half of the 20th century industrial policies gained support from a variety of international organisations, including the UN, World Bank, and ADB. These organisations advocated for limited, short-term government support of local industries as a means of aiding economic growth and encouraging international trade, however, much controversy surrounds the utilisation of industrial targeting in a predominantly free market, globalised economic system, where the common economic structure focuses on less market intervention, not more (Wade & Mkandawire, 2017).

<sup>&</sup>lt;sup>35</sup> See prior introduction to Oseni and Pollitt (2014), which is further elaborated in Chapter 5, *National Market* Factors.

is similarly explored in relation to the developmental state paradigm, with East Asian economists positing that authoritarianism is both a requirement of DS success as well as a burden for continued growth (Breslin, 2018; Öniş, 2018). In the same report Yeung (2018) identified ways in which targeted industrialisation and niche-industry growth coupled with global production and value chains has contributed to the evolution of DS and its role in regional economic development in East Asia (pp. 20-22).

While they do not deny the importance of the developmental state in Asia's development experience, many political economists (Wade, 1992; Chan et al., 1998; Gereffi, 1998; Moon & Prasad, 1998; McGregor 2008; among others) argue that the developmental state is not the only source of economic success in East Asia, nor should it be the primary tool used for continued economic growth. Instead, an approach that utilizes state, "society, culture [and] market" fully encompasses the key hallmark of the Asian "economic miracle" (Chan et al., 1998, p. 8)—variety, and remains a foundational characteristic of IPE analysis in East Asia. Moon and Prasad (1998) posit that it is the reductionist assumptions, misleading links, and "doubts about the positive correlation between the state and economic performance" that have led many East Asian political economists to advocate for economic policies that utilize both developmental state practices and free market approaches, while recognizing the influence of societal pressures and international competitiveness on economic prosperity among NICs in East Asia (p. 10).

McGregor (2008) has also noted that many Southeast Asian developmental states saw reductions in economic development and increases in domestic unrest compared to their neighbours during the 1980s. As a result, a number of these Southeast Asian nations (referred to at the time as transitional economies and today as newly industrialised economies, NIEs) adopted a combination of policies that included state-driven and market-led approaches to development which resulted in subsequent economic gains (McGregor, 2008, p. 54). Wade (1992) also argues that the role of ideas and culture need to be properly attributed to the East Asian miracle as much as DS, a place where traditional DS theory falls short. This is reflected in the evolution of DS theory in East Asia, particularly in the work of Thurbon (2014). Moving forward, any DS definition must similarly reflect this point, as the developmental state paradigm of modern society is not just an economic tool but a political and social one that is not static but evolved based on economic, geopolitical and social cultural changes (Thurbon, 2014; Gao, 1996). This becomes a particularly important

consideration when examining Research Question 3 (*What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian development?*) as an examination of cross-border electricity trade growth in Southeast Asia reveals winners and losers in societies where EMI may be encouraged for a variety of rational economic and energy reasons, but instead results in less reliable or affordable power 'products' and furthers urban and rural divides—issues which are discussed in more detail in Chapter 6 (*Governance Challenges*).

An additional criticism of developmental statism is directed at a perception of the prevalence of corruption and rent-seeking<sup>36</sup> among developmental states (Doig & McIvor, 1999). However, similar to other criticisms of developmental statism there are multiple viewpoints and much debate on the linkages between rent-seeking and developmental statism (See Singh & Ovadia, 2018a, for a broad overview) and the various iterations the presence of both can take around the world.<sup>37</sup> This author agrees with the supposition by Khan & Jomo (2012) that corruption and rent-seeking have been present across the variety of economies in the East Asian region; however, development success elsewhere is not a result of the lack of rent seeking. Rather, while the effects of corruption differ across countries the presence of corruption does not mean economic success will be significantly limited or that a standard response to or outcome of rent-seeking and corruption is predictable (Khan & Jomo, 2012, p. 5). Similarly, Moudud & Botchway (2008) argue that it is not the presence of state intervention that results in corruption, but disempowerment and lack of professional development and capacity that explains corruption among state authorities.

Corruption and rent-seeking operate and exist in developmental states similar to how they operate and exist in neoliberal, market economies. In addition, relationships between political and business elites are often discussed as traits of developmental states, however, that does not mean these relationships are always corrupt (Moudud & Botchway, 2008). Corruption and rent-seeking are

<sup>&</sup>lt;sup>36</sup> Rent-seeking refers to incomes which are commonly higher than they would be otherwise and the way in which the presence of high incomes results in the formation of "incentives to create and maintain these rents" via legal and illegal business, institutional and societal practices (Khan & Jomo, 2012, p. 6). Corruption, which often overlaps with rent-seeking (Khan & Jomo, 2012), is defined by Khan (1996) as deviations from social norms and formal rules of conduct over the actions of individuals in positions of public authority as a result of power, status and wealth motives (p. 12). See also: Doig & McIvor (1999).

<sup>&</sup>lt;sup>37</sup> See for example Booth (2011) or Kelsall (2011) on developmental patrimonialism in Africa.

not a result of developmental state or interventionist practices, nor are they a result of a lack of democracy or monarchy, for example.

Arguments regarding corruption and rent-seeking are relevant in the case of power sector governance and reform in Southeast Asia due to the nature of institutional change that is required for reform to take place and the prevalence of corruption in a variety of different sub-regional economies (Sen et al., 2016). According to Khan:

...all institutional change involves creating or destroying rents and almost all distributive conflicts can be described as conflicts where one or both sides are seeking rents. (p. 6).

Numerous examples of corruption and rent-seeking in Southeast Asian (and indeed other) economies exist, for example in the case of entrenched fossil fuel subsidies in the Philippines' energy market or the dominance of oil in Brunei's economic and political history. Rent-seeking is an economic and political practice that is present across economies and generalities about the result of rent-seeking and corruption as a result of certain economic practices does not serve political economy analysis, but instead reflects the bias and/or ideological approach of the analysis (Khan & Jomo, 2012). In addition to value-negative impacts commonly identified in traditional economic analysis of rent-seeking, rent-seeking can have value-positive impacts as well; for example, in the redistribution of wealth—which Khan & Jomo (2012) believe can have both negative and positive impacts in developing economies. Corruption is similarly intertwined with political economy analysis of economic development (See: Bardhan, 1997; Leff, 1964; Myrdal, 1968, among others), having both valuepositive and value-negative impacts depending on how corruption is controlled and within which populations it is present (Doig & McIvor, 1999; Mungiu-Pippidi & Hartmann, 2019). In the case of electricity market reform, reductions of rentseeking policies and increased transparency in an effort to reduce corruption are common recommendations (See: Sen, 2015; Sen et al., 2016, 2017), and this was also reflected among interview subjects (explored in data chapters 4-6).

Instead of positing whether corruption and rent-seeking are or are not a result of developmental statist approaches (See: Buchanan, 1980 for example), this research posits that corruption and rent-seeking are present across most societies, they are present in different and similar ways across the Southeast Asian sub-regions' economies and energy sectors, and they have primarily valuenegative impacts depending on the actors, factors, and situations in which they are present (Khan, p. 6-9). Corruption and rent-seeking are not the focus of this research. While they exist in the economic and political structures of some of the economies studied here, their presence is not based on the type of economic or political structures examined and so they will not be explored in relation to developmental statism—except to posit that developmental states and neoliberal market economies both (Moudud & Botchway, 2008) face the challenge of managing the positive and negative impacts of corruption and rent-seeking in power sectors.

It is worth noting another impact on the perception of the success of DS theories in East Asia's specific history of economic development, the AFC of 1997/98. The Southeast Asian economies most negatively impacted by the AFC also represented previously thriving examples of DS in practice (Gârdu, 2015). The AFC was the result of a dramatic drop in the value of Thai Baht, which resulted in widespread currency devaluation throughout Southeast Asia, eventually spreading to Northeast Asia (Arndt & Hill, 1999; Tan, 1999; Tan, 2000). This region-wide weakening of financial and economic systems exposed weaknesses in the national and regional mechanism meant to prevent financial uncertainty, exposing structural weaknesses in once strong economic systems. As a result, the East Asian miracle slowed dramatically (Kim, 1999) and criticisms of DS as a viable, replicable option for other developing economies emerged (Pang, 2000; Beeson, 2006; Thurbon, 2014; Gârdu, 2015; Wade & Mkandawire, 2017; and Bishop et al., 2018c provide thorough overviews of the criticism along with mong many others).

As a result of this fall out, other common criticisms of DS refer specifically to the inability of the DS paradigm to be applied to other emerging economies, replicating the East Asia experience (Thurbon, 2014; Bishop et al., 2018). These criticisms specifically point to: changing roles of the state, in particular emerging regulatory roles as opposed to developmental, therefore managing and engaging in markets as opposed to strategic intervention (Jayasuriya, 2005); embracing neoliberal market values and benefiting from market competition (Pirie, 2005, 2008; Bishop et al., 2018), which some experts argue is in fact commonplace and actually compatible with standard neoliberal economic policies and therefore cannot even be considered DS in nature (Cerny, 2005).

Thurbon (2014) responds to this criticism in the context of Korea specifically, however, it is this author's belief the response has application to the wider East Asian region and the sub-region of Southeast Asia. Thurbon's argument is that DS practices are not bound by single policy instruments—indeed, DS is a combination of state directed economic policies; the absence of a common policy

tool among DS (say, targeted industrial policy, as is the case with Taiwan) does not mean other state-directed policies were not utilised (p. IV). In many cases the DS paradigm has evolved to include new economic policy choices and new industries, such as clean energy industries in Southeast Asia. Narrow conceptualisations of DS models do not serve the actual model, which is marked by state adaptation and the directing of policy choices based on economic goals (Thurbon, 2014, p. V). Wong (2004) similarly points to the adaptive nature of the developmental state, a by-product of the nature of globalisation and quickly changing national, regional, and international markets.

This section has presented the main tenets of DS theory, application of DS models in East Asia, as well as common arguments for and against DS models of economic development and policy making. This has set the scene for understanding what cross-border electricity trade and renewable energy integration can tell us about the IPE of East Asian development. The following subsection will take this formulation of DS policies and practices and apply it to a neoliberal, market-oriented power sector and sub-regional electricity market, resulting in the articulation and conceptualisation of a neo-developmental statism model applied throughout this research and subsequent data chapters.

## 2.5.c Neo-Developmental Statism

Neo-developmental statism represents an evolution of the developmental state paradigm that has previously been used to explain state led economic development in East Asia. Where Johnson's original conceptualisation (1982, 1987) of the developmental state lost favour in the late 80s and early 90s for a neoliberal explanation of world economic order (McGreggor, 2008), the reemergence of the developmental state today is representative of a combination of these theoretical explanations-neo-developmental statism represents a degree change and evolution of traditional DS paradigms and neoliberal ways of explaining global economics that were previously separate. Neo-developmental statism is an evolution of the developmental state idea that people like Wade (1992), Gao (1996), Carlos-Bresser-Pereira (2007a, 2007b, 2008, 2009, 2010a, 2010b), Liow (2011), Ban (2013), Dent (2014), Thurbon (2014) Bishop et al. (2018), and Bishop & Payne (2018a), have explored. In addition, neodevelopmental statism is used to explain challenges in cross-border electricity trade and EMI specifically, with application limited to power sectors and energy policy, a niche application of the latest iteration of the developmental statist theory as a result of specific characteristics among ASEAN member states. This sub-section will explore the evolution of the traditional DS paradigm, the nuance and intricacies of this evolution, and the theoretical conceptualisation of neodevelopmental statism that is used in this analysis.

While the world has changed dramatically since DS was first introduced, the developmental state remains alive and well in East Asia (Wade & Mkandawire, 2017). Indeed, the developmental state is different today than when it was first imagined; strong interventionist state policies still exist in many East Asian countries, materialised in strategic industrial policy and market regulation as described previously in this chapter. Dr. R. Wade described the evolution of the DS today as being one that:

manoeuvre[s] within investment and trade rules to promote domestic industries, building on their existing role in the global supply chains as a means to move into higher value add products, and fighting corruption (Wade & Mkandawaire, 2017).

The developmental state of previous years cannot be perfectly replicated, that is true; however, economies across East Asia are using a combination of DS practices and engaging in neoliberal trade and economic policy making, just with limits compared to the "free markets" of the west. It is important to note here that the original DS literature is quite dated and used to primarily explain the economic development and industrialisation practices used in the 1980s and 1990s; the theories behind the paradigm have not, for the most part, been updated to reflect changes post-global financial crisis, the emergence of climate change as a serious international crisis, or changes to global energy markets based on supply constraints and environmental challenges. There are vestiges of developmental statism in the way economic policy is made in East Asia, but its application to specific modern contexts is limited. An updated conceptualisation of the DS paradigm is appropriate in this research as it offers an explanatory tool for the IPE of seemingly contradictory economic development practices of East Asian economies amid dynamic changes to sub-regional and regional power systems in response to climate and environmental concerns.

Neo-developmental statism<sup>38</sup> is the utilisation of common DS economic policy practices, including privileging industrial development and the targeting and state directives of markets, while also utilising free market tools typically associated with neoliberal market structures. The neo-developmental state is not entirely neoliberal in nature and still relies heavily on state directions and

<sup>&</sup>lt;sup>38</sup> Not to be confused with statism, neo-developmentalism or neo-statism. See Chowdhury (1999) or Cammack (1990) for overviews of statist literature, whereby focus is given to the primacy and autonomy of the state.

targeting. However, it also benefits from engaging in a free market, whereby the state regulates interaction and impacts (or attempts to: see 97/98 AFC, for example). The neo-developmental state theoretical framework includes not just economic policy planning, but social and ideational policy planning as well; the green developmental practices of many East Asian economies (Tan, 1999) could fall within this range, quite obviously, via the promotion of clean energy development and power sector integration and adaptation for regional and sub-regional CCD . In the case of Southeast Asia, the entire EMI directive of ASEAN is being played out via developmental states, whereby member economies are supposedly directing investment in power sectors while also supposedly<sup>39</sup> advocating for the liberalisation of electricity markets in order to increase competition and naturally upgrade systems (whether or not national policies reflect these ASEAN targets and goals is a recurring theme, discussed in chapters 4-6).

It is worth mentioning a parallel concept, neo-developmentalism, in order to avoid confusion with a similar yet different IPE evolution of the original developmental state paradigm. Neo-developmentalism is a term coined by Brazilian economist Luiz Carlos-Bresser-Pereira (2007a, 2007b, 2008, 2009, 2010a, 2010b, among many others) to describe an alternative economic framework to the neoliberal, Washington Consensus policy recommendations for post crisis developing economies (Ban, 2013). In this theoretical model, neodevelopmentalism exists on the same spectrum as neoliberal policy making, but less open than neoliberalism and less controlled than DS (Liow, 2011; Ban, 2013). The neo-developmentalism of Carlos-Bresser-Pereira is standard in development discourse, and commonly applied to Latin American countries,<sup>40</sup> primarily Brazil (Carlos-Bresser-Pereira, 2009). Similar to the DS model in East Asia, neo-developmentalism stresses industrial growth using targeted industrial policy, the use of development strategies, and engagement in global trade (Ban, 2013). Neo-developmentalism also stresses wage policies (raises, government guarantees and cash transfers) and avoidance of variabilities in exchange rates

<sup>&</sup>lt;sup>39</sup> Chapters 4-6 will further address the 'supposed' part of this statement, as some national policies do not fully reflect these ASEAN specific goals and directives. This is more a consequence of translating large, sub-regional goals to national ones and the incentives behind making large, structural and potentially economically powerful changes, such as the redirection of subsidies for oil and coal to renewable sources in national energy markets.

<sup>&</sup>lt;sup>40</sup> See Gereffi (1989) for an examination of development theory contrasted in East Asia and Latin America.

and market sentiment via financing from domestic savings (Ban, 2013, p. 302). Ban (2013) analyses the progress of neo-developmentalism in Brazil, where neodevelopment first began in Latin America, and concludes that Brazil does not fit the perfect neo-development paradigm but has, in fact, adopted some of the liberalisation policies of the Washington Consensus. Therefore, the neodevelopmentalist model applied to Brazil has, similar to DS, evolved past original definitions; it is not a perfect definition, but instead a hybrid that includes more focus on liberalisation than state directed policies, with priority given to financial controls in avoidance of common financial failures found in the economies of Latin American states.

In the case of neo-developmental *statism*, this author's application differs from Carlos-Bresser-Pereira's neo-developmentalism, in that it is a hybrid of neoliberalism and DS with a focus given to targeted industries and state management of the liberalisation process. The author's neo-developmental statist framework is not directed specifically at post crisis economies (although some may be) and emphasis is not placed on countering the Washington Consensus or similar neoliberal economic policies. In this case, neodevelopmental statism is a mix of state directed and market liberalisation policies which are used to continue to direct development across Southeast Asia. This can particularly be illustrated in the power sector in Southeast Asia, where a mix of public and private markets exist and operate, and governments aspire towards further liberalisation while also directing financial and economic measures that suit national economic development goals and state-business relationships. Chapters 4-6 will call further attention to these structures, and the application of Oseni and Pollitt's (2014) EMI Criteria to economies in the BIMP-EAGA subsystem will also highlight these relationships and market factors specific to power sector integration.

Bishop et al. (2018) delve into the re-emergence of the developmental state, and its relationship to neoliberalism in recent years. In Bishop et al. (2018) Bishop and Payne identify the goal of developmental statists as building a market but distorting that market in an effort to feed national interests via financial and policy tools (p. 2). This is an important point, given the different characteristics of neoliberal economists and developmental statists, explored further by Bishop and Payne (2018a, p. 2):

Where liberals believe countries should accept and exploit their 'comparative advantage' almost as if it is naturally determined and therefore static, theorists of the developmental state see comparative advantage as dynamic and changeable, and Pointing to the vast literature on DS, Bishop et al posit that the DS has always been intertwined with neoliberal economics and is in fact a part of the developmental story of many western countries, just not named as such arguments that help justify this author's neo-developmental state model. In their position, Bishop et al. (2018b) conclude their recent DS research and analysis by positing that the DS model is still relevant today, just not exactly replicable, characterised by operating in a more competitive international system, and also evolved from the traditional definition made famous during and after Japan's economic rise by being applied differently from economy to economy (p.25).

The weaknesses of Dent's (2014) new developmentalism were noted when discussing EMT in section 2.2 of this chapter. These are minor differences, however, in addition, new developmentalism's analytical framework has not previously been applied in ASEAN power sectors or to EMI policies; based on the combination of these factors new developmentalism has thus not been adopted by this author. Another relevant, although still not replicable, theoretic framing this author has identified lies in the work of Liow (2011) on the theoretical framework of the neoliberal-developmental state in Singapore. In Liow's work focus is given to neoliberalism as not just a tool of economic policy making but also a tool of state political rationality in Singapore, an outlier of liberalisation and neoliberal economic policy making among Southeast Asia's economies (Huff, 1995; Gainsborough, 2009; Low & Johnston, 2001). Following on the work of Foucault (2003) and Brown (2005), Liow rationalises the impetus for focusing on decision making and political rationality when examining state economic actions, which are often underpinned by similar political values that reflect the governance ideology of the state (Liow, 2011, p. 2). Liow argues that while stateled industrialisation is present in his theoretical conceptualisation of the neoliberal-developmental state, the desire to liberalise has become so engrained in policy making that it is now reflected in the political ideology of the ruling elite, and therefore has become more entrenched in the governance actions of the state and thus more dominant than the state-led policy making rationale of previous years (Liow, 2011, p.3). In addition, according to Liow (2011, pp. 3-4) this liberalisation ideology has also been incorporated into all aspects of the Singaporean state, including:

...social, political and economic life of Singapore society around the ideology of pragmatism from its early days of development, of which the overarching purpose is to enable the quicker and more efficient accumulation of capital through high levels of economic growth.

The political and economic history of Singapore, and its unique development as the economic centre of ASEAN with a focus on trade and openness to foreign businesses, make it stand apart from most other ASEAN examples (Hill, 1994; Low & Johnston, 2001). The country did utilise industrialisation in a similar DS manner, but with the widespread integration of global financial norms into the political and economic structure of the country, still present today (Huff, 1995). In addition, unique calibre of leadership, planning for development and stability of economic progress have all contributed to the successful integration of Singapore into the global economy (Huff, 1995, pp. 1430-1434). In this regard the Singaporean model of neoliberal developmentalism is not replicable, nor standard across the sub-region.

The adaptation of Liow's neoliberal-developmental state theory to this research includes more focus on state intervention and less integration of neoliberal ideology into the state political system. Similarities include a persistence of power dynamics and competing interests (Liow, 2011, p. 8), which in the case of power sectors is very evident at the national versus sub-regional level (discussed in more detail in chapters 4-6). Additional differences between neo-developmental state include a wider regional focus and the experience of unique, developing and emerging markets with quite a way to go until achieving the liberalisation and integration of the Singaporean economy.

While IPE can be used to explain a variety of global economic and political interactions and the theories that underpin them, the application of IPE in East Asia can vary slightly as a result of the unique political, economic, and social contexts within the broader region (where there is already a lot of diversity within political and economic systems just based on geography, history and culture), but often leads back to DS practices as the basis for economic growth in the region (Beeson, 2014). This is relevant here as the dominant neoliberal economic explanation for economic and political interactions are not directly applicable given different approaches to economic policy making in centrally planned economies versus free market economics. A state-directed economy will not, by very definition, freely give up economic development to the movements of the global market or to liberalisation agendas, despite the many incentives for doing so (in this case climate change adaptation, energy poverty reductions, or increases in energy security, for example). Instead, the economies of Southeast Asia appear to be taking a combined approach to economic policy making in the

energy and power sectors, which can be explained using the theoretical conceptualisation of neo-developmental statism. While Neo-developmental statism is aligned with some notions of liberal developmentalism, in this context it is applied to Southeast Asian countries and the electricity markets of the BIMP-EAGA subsystem for the first time.

Throughout the thesis, neo-developmental statism will be used to explain the plausibility, difficulty, and reality of applying largely neoliberal reforms in national and sub-region power markets. The application of neo-developmental statism will be used to explain a variety of EMI and interconnection challenges across Southeast Asia: Chapter 4 will examine the sub-regional market factors that make up interconnections in the sub-region, highlighting the limitations of current and planned interconnections; In Chapter 5 the market factors of national markets will be examined. Here the author will use neo-developmental statism to analyse the application of Oseni and Pollitt's (2014) EMI criteria to BIMP-EAGA economies, contrasting the governance, liberalisation, and management recommendations embedded in these criteria with the reality of state-directed policy making and national interests. In Chapter 6 additional governance challenges will be discussed, using the neo-developmental statism framework to explain the (1) contradictions between national versus sub-regional incentives and (2) trust in policy and practice within the sub-region.

## **2.6 Conclusions**

This chapter has traced the IPE based development path of East Asia to provide context for (1) the challenges development has exposed, particularly in relation to energy intensive development and industrialisation practices; (2) the standard IPE explanations for this process both generally and in Southeast Asia specifically; (3) and the models employed to provide context to this process in relation to EMI in Southeast Asia. The two seemingly contradictory theoretical approaches of (1) neoliberal market practices and (2) directed state policy making via DS have been combined to form the neo-developmental statism.

In this research neo-developmental statism will be applied to Southeast Asian countries and the electricity markets of the BIMP-EAGA subsystem for the first time. The application of neo-developmental statism will assist in answering the primary research questions and provide context to the study of EMI in East Asia: (RQ1) *How do governments understand the IPE of cross-border interconnections in Southeast Asia?* (RQ2) *How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national* 

*markets*? (RQ3) *What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian development*? This application will illustrate the difficulties Southeast Asian countries face in expanding EMI and cross-border electricity trade, discussed around the issues of sub-regional versus national market factors, reform, and governance.

The following chapter will outline the methodology used to explain the processes of EMI and cross-border electricity trade expansion in Southeast Asia. A focus on data collection and elite interviews will be given, along with the conceptualisation of this research and the processes utilised. These processes will frame the subsequent thematic chapters, using data and information that was extracted during the data collection and interview process. These methodological considerations will also structure the outcomes of this research, offering context to the actors involved and the difficulty of examining this interdisciplinary topic with an IPE approach. Neo-developmental statism will continue to play a role as the historic and current need for energy poverty reductions and increased electricity access will be contrasted with market structure and proposed neoliberal-oriented reforms within Southeast Asian subregional electricity markets.

# Chapter 3 Research Design & Plan of Enquiry

## **3.1 Introduction**

Chapter 3 introduces the methodological approaches of this research, presenting an overview of how these approaches have developed and shaped the design of this research project. This chapter includes mention of the required elements of the thesis (reliability & validity, risk assessment, data collection methods, documentary data used, data protection plan, case study design, methods of analysis, etc.) as well as a deeper breakdown of the case study research design and two methods of data collection employed—documentary analysis and elite interviews. In short, this chapter will provide the necessary understanding to see the links between "the research problem, the method, and the results" (Rudestam & Newton, 2007).

In this chapter an explanation of interview subjects, how they were chosen, and how they were approached will be explained, as well as reference to the interview protocol and methods used for coding interviews, resulting in thematic analysis. The choice of case study will be also be introduced, recounting the natural progression from East Asia to Southeast Asia via interviews, followed by interview feedback on narrowing down the case study to BIMP-EAGA. The conclusion section of this chapter will provide a link to the subsequent sections, which focus on specific themes identified in interviews and data analysis.

This study brings together two disparate areas of research under the umbrella of IPE—East Asian studies and sustainability research. These two distinct areas of study provide insight into the multidisciplinary research of climate change policy responses, bringing focus to this research from different, complementary viewpoints. Two data collection approaches were used in the case study design: the first, documentary analysis, aimed to explore the current realities of crossborder electricity trade in East Asia using publicly available policy documentation, stakeholder reports and grey literature; the second, elite and stakeholder interviews, was used to provide critical context to policy decisions that have contributed to the current state of energy policy and will impact future progress in electricity trade.

The three primary research questions these methods are designed to answer are:

- 1. How do governments understand the IPE of cross-border interconnections in Southeast Asia?
- 2. How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?
- 3. What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian development?

These research questions were derived from the gaps identified in the literature review both in the academic debate on IPE and poor stakeholder understanding of the development of electricity interconnection in East Asia as a geopolitical issue of importance and a practical problem for regional energy ministries (See White, 2009 for more on policy context and formulating research questions that reflect social problems).

These research questions are theory confirming; for example, question 3 produces and fine tunes a theory (White, 2009, p. 25)—the theory of neodevelopmental statism described in chapter two is interrogated in light of the particular technical and political-economic challenges of electricity market interconnection. In addition, the iterative methodological choices made in this research reflect the nature of the research questions chosen and their ability to address complex social problems (energy security, sustainability, and access challenges) and the relevant contemporary policy linkages.

This chapter is structured in six sections: In the second section, *Methodological Approach*, the epistemological reasoning of the author is introduced, including a brief overview of critical realism and how it frames the author's research design and choices regarding methods. The methods used in this research are grounded in critical realism, and as such the epistemological and philosophical context will be explored first. In section 3, *Research Design*, the case study design of this research is explored, examining the author's approach to overall research design, elite and stakeholder interviews, reliability & validity, ethical considerations, the necessary risk assessment, data protection, and how the author has ensured a quality research design. This also includes explaining how the case study was narrowed down to a single subsystem within ASEAN, the BIMP-EAGA subsystem.

In section 4, *Methods: Data Collection & Analysis*, the varieties of data collection and resulting analysis of data is presented, focusing primarily on documentary analysis and elite interviews. In section 5, *Positionality of Research*, the author's research is framed in the context of the author's pre-existing viewpoints and

internal bias. Finally, in section 6, *Conclusions*, all the methods will be summarised, explaining their relevance to this research, highlighting their complementary research contributions, and propelling this research forward from its initial research design through to subsequent analysis chapters.

# 3.2 Methodological Approach (WC: 2641)

The approach to use multiple methods—both data analysis and elite interviews—was intended to reveal patterns between Oseni and Pollitt's 2014 actual cross-border interconnections or potential EMI criteria and interconnections in Southeast Asia (Oseni and Pollitt's EMI criteria outline general requirements necessary for the integration of multiple national electricity systems and electricity trade across borders, including management, policy, physical, and system requirements.). This methodological approach is an exercise in theoretical triangulation, whereby comparing, evaluating and/or testing of multiple theories or concepts is used (Rudestam & Newton, 2007; Sovacool et al., 2018). In the context of this research project this methodology revealed relationships between elite opinions and political and economic realities in the electricity and energy market space, shedding light on what is needed in order to expand cross-border electricity trade and meet sub-regional EMI goals.

The overarching aim (A1) of this research is to explore the extent to which climate change has created imperatives for energy market integration (EMI) and power sector reform in East Asia, shedding light on what trade and economic development policies are used in response to these issues. From this aim the following research plan was identified, combining research objectives with the corresponding research questions (RQs), data needed, data collection methods required for said data, data analysis methods, and potential links to the analytical framework identified in Chapter 2 (*Cross-Border Electricity Trade: Conceptual Approach & Review of Literature*), neo-developmental statism. The research plan shown in table 3.1 is organised around the three core research questions (Andrews, 2003, p. 51-62).

<b>Objective 1 – Provide an overview of cross-border interconnections and assess market factors in Southeast Asia;</b>			
Research Question: (1) How do governments understand the IPE of cross-border interconnections in Southeast Asia?			
<ul> <li>Electricity data for countries involved</li> <li>Relevant elite's opinions</li> <li>EMI data</li> <li>Energy poverty data, sub-regional (and national)</li> <li>Climate and energy related policy goals at sub-regional and national level</li> </ul>			
Data Collection Methods• Review of literature • Interviews • Data set analysis			
<ul> <li>Data Analysis Methods</li> <li>Quantitative and thematic analysis via coding in NVIVO</li> <li>Explanation building</li> <li>Abstraction of interview data</li> </ul>			
<ul> <li>History of development in the region (state-led, predominantly)</li> <li>Role of IPE in development practices</li> <li>Structure of sub-regional climate/energy related initiatives (non-binding, led by ASEAN or another multinational organisation)</li> </ul>			
Objective 2 – Assess market factors of Southeast Asian case study selection and identify areas of individual and collective reform needed to reach deeper levels of EMI among BIMP-EAGA economies;			
Research Question: (2) How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?			
<ul> <li>Regional electricity trade stats; sub-regional (BIMP-EAGA) stats</li> <li>Relevant elite's opinions</li> <li>ASEAN data for APG, bilateral and national initiatives;</li> <li>Electricity trade and increased renewable integration data from other regions?</li> <li>Oseni and Pollitt (2014) EMI recommendations</li> </ul>			
<ul> <li>Data bases, government statistics, and ASEAN material research;</li> <li>Review of Literature</li> <li>1-on-1 Interviews</li> <li>Conference, workshop discussions or relevant meetings</li> </ul>			

Data Analysis Methods	{	<ul> <li>Case study analysis of single case (embedded design)—develop a descriptive framework for organising case study?</li> <li>Explanation building</li> <li>Thematic analysis via coding in NVIVO</li> </ul>	
Links to Analytical Framework	{	• IPE of sub-regional and national markets—structure	
Objective 3 – Identify and analyse additional governance challenges to deeper EMI and increased electricity interconnections among ASEAN member states and case study selection;			
Research Question: (3) What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian Development?			
Data Needed	{	<ul> <li>Current interconnector data</li> <li>Electricity data for countries involved</li> <li>Relevant elite's opinions</li> <li>EMI data</li> <li>Data from other regions?</li> <li>IPE theory</li> </ul>	
Data Collection Methods	{	<ul><li> IPE theory research</li><li> Review of literature</li><li> Interviews</li></ul>	
Data Analysis Methods	{	<ul> <li>Case study analysis</li> <li>Thematic analysis</li> <li>Theoretical analysis and analytical generalisation</li> <li>Explanation building</li> <li>Some historical analysis</li> </ul>	
Links to Analytical Framework	{	<ul> <li>Theory underpinning commonly proposed liberalisation/integration methods</li> <li>Contrasted with common sub-regional / national market structure</li> <li>Linked with commonly identified reform impediments</li> <li>Contrast of specific IPE (trade &amp; economic) policies and development practices regionally (national issues build into sub-regional, based on original questions)</li> </ul>	

This research plan is reflective of the recommendations of Robert K. Yin (1981a, 1984, 2003) and Rudestam and Newton (2007) on case study design and research plans, respectively. To answer these research questions a thorough examination and critique of the dominant IPE frameworks and their proposed application to East Asian economic & political structures will provide context in this research to the development of sub-regional interconnectors that impact

clean energy utilisation and sub-regional EMI. Significant attention is paid to documents covering trade and industrial policy development in an East Asian context. Policy documents that make the best case for optimizing clean energy's contribution in addressing global climate goals, and examples from Southeast Asia and the Brunei-Indonesia-Malaysia-Philippines East Asia Growth Area (BIMP-EAGA), will be utilised to address these research questions. An analysis of the political-economy of current electricity markets based on semi structured elite interviews is also used to identify drivers and barriers to cross-border electricity trade utilisation regionally, sub-regionally, and nationally. The historic and current need for energy poverty reductions and increased electricity access is also contrasted with market structure and proposed neoliberaloriented reforms within Southeast Asia's sub-regional and national electricity markets. The East Asian practices of state-led economic development and targeted industrial policies is contrasted against commonly proposed liberalisation reforms of national electricity markets.

The three primary research questions are presented across three thematic chapters, each with a different objective and different analytical needs. These are demonstrated in table 3.2 as follows:

#### **Table 3.2** Objectives & Analysis Needs

Objective 1 (Chapter 4 – Sub-Regional Market Factors)
Objective: Provide overview of cross-border interconnections in Southeast Asia; Assess market factors in the sub-region of Southeast Asia:

- Assess the role of cross-border electricity trade in sub-regional EMI;
- Review and analyse current and projected interconnections;
- Review and analyse ASEAN interconnection projects;
- Synthesise lessons learned among selected projects.

## Research Question

RQ1: How do governments understand the IPE of cross-border interconnections in Southeast Asia?

## **Objective 2 (Chapter 5 – National Market Factors)**

Objective 2: Assess market factors of Southeast Asian case study selection and identify areas of individual and collective reform needed to reach deeper levels of EMI among BIMP-EAGA economies:

- Synthesis overview of current market structures in national and subregional electricity markets;
- Assess national needs for sub-regional EMI;
- Review and analyse any links to the integration of renewables into national energy mixes, review sub-regional energy poverty

reductions, and identify any links between national energy goals and EMI initiatives;

• Identify areas of individual and collective reform needed to reach deeper levels of EMI among ASEAN member states.

#### **Research Question**

RQ2: How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?

#### **Objective 3 (Chapter 6 – Governance Challenges)**

Objective 3: Identify and analyse governance challenges to deeper EMI and increased electricity interconnections among ASEAN member states and case study selection:

- Identify trade and economic development policies used in response to these issues;
- Provide realistic, relevant, IPE based policy recommendations for further renewable energy integration sub-regionally;

Research Question

RQ3: What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian Development?

The inclusion of interviews and documentary analysis as data collection methods is supported by the "ontological and epistemological underpinnings" of critical realism, which frames the philosophical approach of the author and this research. In the following subsection critical realism will be introduced and its application to this research presented, "identifying, explaining and justifying the epistemological stance" (Crotty, 2003, p.8) that has been taken in forming this research and the plan of inquiry.

## 3.2.a Critical Realism

While the theoretical framing of the data collected is rooted in IPE theory, the paradigm through which cross-border electricity trade is examined is critical realism. The inquiry paradigm (Punch, 2014) of this research informs the ontological questions (what is the reality of this research), the epistemological questions (what is the relationship between the researcher and this research reality), and the methodological questions (what methods will be used in this research) highlighted in this chapter (p. 14-15).

Critical realism is used here because it is framework of understanding that allows researchers to excavate the 'real' causal factors driving objective phenomenon like trade negotiating strategies or motivations behind prioritising specific infrastructure projects. While the creating of empirical data may uncritically point to a hegemonic narrative (such as free trade and renewables interconnection to tackle climate change) the critical realist will interpret this data in light of more critical theory. Thus, the critical element of realism goes beyond observable objective phenomena, to interpret this 'real' through the lens of wider theory (Yeung, 2003).

This allows for the identification of multiple causal structures in the development of cross-border electricity trade in East Asia, as well as for understanding IPE in both sub-regional and national electricity markets. These multiple causal structures may well be explicit and observable in empirical data, but they may also be occluded by it, obscured by ideological bias or official narratives. The work of critical realism is to determine which casual factors are driving the observed phenomena by a mix of observation and theoretically informed inference.

Critical realism as an inquiry paradigm (Punch, 2014) enables the author to explain and consider the international relations, national and sub-regional political structures, and economics as interrelated and equally impactful on EMI and cross-border electricity trade in East Asia; no one factor of IPE plays a singularly dominant role.

When used in a methodological context critical realism refrains from reducing ontology (nature of reality) to epistemology (knowledge of reality) and is instead rooted in the belief that "human knowledge captures only a small part of a deeper and vaster reality" (Fletcher, 2017, p. 182) and a systematic critique of processes, structures, and powers is necessary for understanding (Sayer, 1992; Yeung, 1997). In the critical realist perspective knowledge is gained using theories:

that help us identify causal mechanism driving social events, activities, or phenomena, [and] are selected and formed using rational judgment of these social events (Fletcher, 2017, p. 182).

Critical realism recognises that theory is based on the study of a specific set of events where not all possible events are included, making exploration of social and political issues a natural fit (Easton, 2010). As it pertains to this research critical realism is a natural fit, given the multifaceted topic, inductive approach, and breadth of situations in which the IPE of cross-border electricity trade could differ.
Alternative epistemological approaches can of course be utilised across the breadth of cross-disciplinary social science research.<sup>41</sup> Crotty (2003) identifies common epistemological approaches across the social realism sciences, these include: objectivism, whereby meaning and reality exist apart from consciousness (p. 8), and truth and meaning can be acquired through objective research (p. 5); constructivism, whereby meaning and reality exist within engagement with the world (p. 9), i.e. meaning is constructed by people based on their individual interaction with the world (p. 43); and subjectivism, whereby meaning is imposed from a subject onto an object (p. 9). In addition, different approaches can be employed alongside different epistemological types, including positivism (relying on statistics and data to answer societal problems), interpretive, and post-positivism (a balance of positivist and interpretive approaches to analysis) (Crotty, 2003; Denzin & Lincoln, 2011; Yin, 2011). Here, however, critical realism was chosen and is reflected in the theory construction and case study design of this research as ultimately this IPE research is based on critical inquiry into economic and political structures that play a role in sustainability and economic development. Critical realism also allows for abstraction of complex phenomenon, particularly in the methods necessary for interviews (Yeung, 2003, p. 57), which is also applied here via analysis and abstraction of elite interviews (See Section 3, Research Design and Section 4, Methods: Data Collection & Analysis).

Critical theory, generally, encompasses many different approaches (Yeung, 2003), including Marxism and critical feminist theory, but have in common a critical approach to scientific inquiry (Kincheloe et al., 2011, p. 163). By and large critical inquiry is research that challenges (Crotty, 2003, p. 113). Critical researchers recognise the difficult dynamics at play between values, power, social relations, subjectivity, privilege and oppression (Kincheloe et al., 2011, p. 164) when exploring social challenges. Here the linkages to this research become clear, as green growth and energy access both largely hinge on power dynamics (economic and political), social values (green development versus heavy industry development with no regard for environmental concerns), and equity

<sup>&</sup>lt;sup>41</sup> Additional analysis of critical realism and its application in IPE, case study dominant, research could be included. However, as Rudestam and Newton (2007) point out, the philosophy of science behind PhD research is relevant only in so far as it explains the research strategy and data generation within the confines of a particular project (p. 106). As such no further debate on epistemological approaches will be covered here beyond a brief overview; however, in further research this could be explored and alternative epistemological and ontological viewpoints employed.

of access (access to electricity vs access to affordable, reliable and sustainable electricity) (Lockwood, 2015c), among other paradoxical challenges identified within this research.

Bhaskar (1979, 1997), credited with the early development of realist philosophy (Yeung, 2003), posited that researchers must not commit to a single theory initially, and instead build upon potential theories as research progresses, avoiding the conditionality of a single theory and fostering deeper investigation (p. 6). Tendencies, or demi-regularities, can be identified in data analysis, resulting in themes or trends that can occur, but are not taken to be law by critical realists—the reality of the situation could change and as a result a different trend or outcome could occur (Archer et al., 1999; Danermark et al., 2002; Easton, 2010; Gorski, 2013; Fletcher, 2016; among others). This flexible approach is a common theme in the methodology of this research, contributing to both the research design, data analysis, and potential for future research (discussed in more detail in Chapter 7, *Conclusions*).

This critical, explanation-building research is well positioned to be informed by epistemological approaches of critical realism, where knowledge is reflected in the context of action building and critical awareness of societal issues and the power dynamics at play within them (Crotty, 2003). In the context of this research critical realism provides the ability to identify the drivers behind cross-border electricity trade in a neoliberal world order, but within subsystems where Asian values and collectivist approaches to policy making dominate regional and national politics. Critical realism shapes the philosophy of this research, whereas IPE shapes the theoretical work, thus forming a balanced approach to analysis and abstraction (Yeung, 2003, p. 53). Critical realism allows the utilisation of multiple avenues of explanation, across a wide variety of fields and issues, in order for us to understand the IPE of cross-border interconnections in an East Asian context.

The author's choice of case study research is a natural fit for the epistemological underpinnings of critical realism, because it reflects the changing and interpretive reality of the complex economic, political, and social structures within sub-regional and national electricity markets. In the next section, a case study design compatible with this critical realist perspective is developed.

# 3.3 Research Design

The research design of this study was approached with the work of Robert K. Yin (2003, 2011, 2013) informing case study design and qualitative analysis of the chosen case study; the work of Sovacool et al. (2018) was used to formulate a case study research design specific to qualitative, energy social science research. The work of Braun and Clarke (2006) on thematic analysis in social science research was used in analysis of interview data and coding related to the identified case study. Ultimately, however, this research design includes a variety of techniques common in energy social science research, what is understood as methodological triangulation (Sovacool et al., 2018). This design used a combination of methods to answer the posed, cross-disciplinary research questions. The approach used here includes 4 interconnected means to answering the proposed research questions:

Figure 3.1 Interconnected Case Study Methods



Source: Author's adaptation of Yin (1984, 2003) and Sovacool, et al. (2017).

This design is inductive in approach, with no hypotheses to prove but rather investigating in detail the applicability of a neo-developmental statism to the ongoing development of electricity market interconnection. This theory driven inductive approach requires interpretative analysis and meaning-building (Trafford & Leshem, 2008, p. 98). In the following section the case study design

will be explained, followed by description of the data collection and methods chosen, providing a "philosophical and technical foundation" (Trafford & Leshem, 2008, p. 89) from which to understand the progression of this thesis research and the methodological considerations and methods utilised.

#### 3.3.a Case Study Design

The case study of this research design provides a detailed examination of EMI and cross-border electricity trade within a specific selection of countries, Brunei, Indonesia, Malaysia and the Philippines. At the outset of this research it was clear that the whole of East Asia was too large for the amount of time and resources devoted to this research project; it was the author's intention to narrow down the focus within East Asia via feedback from 4 background interviews (which also served to test suitability of initial questions); two of these interviews (one former government official and one current multilateral organisation representative) subsequently became informants, offering a variety of background and case study specific resources throughout the research process. This included professional reports, meeting minutes, and personal notes used to frame this author's knowledge and subsequent case study development. In this instance documents were identified by subject experts as being relevant to this study—while these documents were explored and utilised to direct the overall research, they are unique tools in that the author did not chose them independent of interview subjects or based on prior epistemological framing. Interviews are referred to, in this research, by their field (i.e., relevant job affiliation) and a unique number reference; for example, the two informants are referred to in the text as Former Government—Informant 1 and Multilateral organisation—Informant 2.

Yin (2013) demonstrates the importance of case studies in a wide variety of social science research whereby "a desire to understand complex social phenomena" exists (p. 14). In this research approach the author chose to use a selection of countries to analyse cross-border electricity trade based on Yin's characterisation of a case study:

...[as the investigation of] a phenomenon within its real-life context, especially when the boundaries between the phenomenon and context are not clearly evident (2003, p. 13).

The justification for cross-border electricity trade is in part the impact of changes to the global climate, the need to manage variability of renewable energy in electricity markets, and the difficulty in multilateral cross-border electricity trade based on EMI needs and requirements; the roles of trust, market

structure, and political, technical, and financial barriers all provide the context for both the success and failure of cross-border electricity trade in East Asia.

Focusing on these issues within a group of similar countries in East Asia provides a snap shot of cross-border electricity trade that includes a variety of political and economic factors. This was also more realistic in terms of time, access, and author capacity. Odell (2001) similarly reflects the benefits of qualitative case study design in IPE focused research, lending credibility to the author's case study choices and use of qualitative methods.

An embedded case study design (Yin, 2003, p. 38-53) was constructed prior to data collection, guiding the development of relevant interviews, data analysis, and the write-up process. This design involves a single case (ASEAN) with two embedded units of analysis (BIMP-EAGA, the primary unit of analysis and the Greater Mekong Sub-region, or GMSR, the secondary unit of analysis, which was later abandoned). An example of Yin's (1984, 2003, p. 40) design can be seen below:





A single,<sup>42</sup> embedded<sup>43</sup> case study was decided upon. Justification for this design choice was made based upon a number of case study design issues identified by Yin (2003): (1) time limitations and access challenges (p. 45) identified with mapping electricity markets in Southeast Asia; (2) the critical nature (p. 40) of a single case study, where the circumstances that have made BIMP-EAGA the least successful EMI subsystem in ASEAN make it a unique case among other subsystems and where the GMSR offers lessons to learn from; and (3) replication (p. 45-49) of circumstances across the BIMP-EAGA subsystem to the other ASEAN subsystems would be difficult based on unique national power markets, leaving doubt as to whether or not replication would be possible within the amount of time available for analysis and without knowing what data would be attainable. Taking into account these considerations, and others, the initial<sup>44</sup> case study was designed is as follows:

#### Table 3.3 Case Study Design

This is designed as an explanatory, single-case study, that could be complemented by an "exploratory" and "descriptive" study as well if time allows:

- Case 1: BIMP (Brunei, Indonesia, Manila and the Philippines)

Initial case study	How does the current system of electricity trade (in GMSR and BIMP) work?
questions	<ul> <li>Why is it successful or failing?</li> <li>What does it need to be successful?</li> <li>What is the difference among partner countries?</li> <li>What role does ASEAN play?</li> </ul>

<sup>&</sup>lt;sup>42</sup> An additional, exploratory case study, GMSR (Greater Mekong Sub-Region), was included in the initial case study design; this was abandoned due to time and relevance as a result of interview feedback. As a result, the GMSR case study is not included in data chapters 4-6. However, the author does identify opportunities for expansion of this research with an examination of GMSR water policy and potential replication in BIMP-EAGA. This is discussed in Section 7.5, *Suggestions for Further Study*.

<sup>&</sup>lt;sup>43</sup> Here 'embedded' refers to multiple units of analysis being embedded within the case study, in this case the BIMP-EAGA member countries (Yin, 2003, p. 42-43).

<sup>&</sup>lt;sup>44</sup> Yin (2003) advocates for continuing to revise case studies as they develop (p. 24, 55), an approach the author employed throughout the research and data collection process and reflected in the abandonment of one case study (GMSR).

	What other governance		
Potential exploratory questions	<ul> <li>If the goal is to develop pertinent hypotheses / propositions for further inquiry:</li> <li>What do these two cases have in common?</li> <li>What do these two different markets have in common?</li> <li>What is the difference in ASEAN's role within each case?</li> </ul>		
Proposition of case study questions	<ul> <li>Successful cross-border trade is in place, at some level</li> <li>Each country in the subsystem being studied (BIMP vs GMSR) is taking part in cross-border electricity trade in some form</li> <li>Cross-border electricity trade is facilitating increases in clean energy utilisation (at some level)</li> <li>ASEAN is relevant in both cases</li> <li>Both cases are relevant to ASEAN's Energy Market Integration (EMI) efforts</li> <li>Expansion of cross-border electricity trade is desired by each country</li> <li>Propositions will be linked to data</li> </ul>		
Units of analysis	<ul> <li>Geographical limits: BIMP, GMSR</li> <li>Time limits: this will be developed based on interviews, and likely represent the initial project development through to current day (i.e., February 2014 through January 2019)</li> <li>Potential for inclusion of energy / electricity data points related to consumption and production</li> </ul>		
Case study method	<ul> <li>Systematic interviewing:         <ul> <li>Researcher will follow interview protocol as well as incorporate new questions based on prior interviews or respondent recommendations</li> <li>Including 2-3 key informants, 2 of which have been identified</li> <li>Potential for inclusion of direct observation based on interview development.</li> </ul> </li> <li>Document analysis</li> <li>Data analysis</li> <li>Time series analysis (if time):         <ul> <li>Including the analysis of chronological events (sometimes considered a special form of time-series analysis); not limited to a single independent or dependent variable, but will need to identify specific indicator(s)</li> <li>Identify any embedded units for additional analysis if relevant</li> </ul> </li> </ul>		

Embedded Design	<ul> <li>Literal Replication: Researcher will state the conditions under which a particular phenomenon is likely to be found—if at all</li> <li>Theoretical Replication: Conditions when it is not likely to be found; generalisable to theoretical propositions identified earlier</li> </ul>
Analysis: use of documents, interviews	<ul> <li>Dominant Modes of Analysis: pattern matching, explanation building, time series analysis</li> <li>Secondary mode of analysis: embedded units</li> </ul>
Issues to consider Issues to consider (cont.)	<ul> <li>Construct Validity: to strengthen, the researcher will:         <ul> <li>Use multiple sources of evidence, particularly encouraging convergent lines of inquiry</li> <li>Establish a chain of evidence</li> <li>Have draft case study be reviewed by key informants</li> </ul> </li> <li>Internal Validity: Yin cautions internal validity is a concern only for causal or explanatory studies, where an investigator is trying to determine whether event x led to event y.</li> <li>External Validity: In case study analysis theory will be used as a vehicle for examining case studies and explaining key themes.</li> </ul>
Data collection	<ul> <li>Interviews</li> <li>Electricity market data</li> <li>Review of the literature</li> </ul>

Final Report will contain separate chapter narratives about each of the chosen cases singly as well as a discussion section on cross-case analysis and results.

The case study design that developed this research included identification of the number of case studies (single case design versus multiple case design), units of analysis to be employed, replication that will occur (if any), type of analysis that will be used, issues to consider (construct/internal/external validity) and data collection methods, as we well narrative style and writing process to be employed during the write-up of data. A number of reliability and validity concerns are addressed within the design, as well as a grounding of the research

in propositions of the original study and methods and methodology that were planned in advance (Yin, 2003, p. 33),

Based on Yin's analysis of case study types, the author designed an explanatory, embedded case study, that would be complemented with exploratory data collection and descriptive analysis based on data collection via interviews, data evaluation, and documentary scrutiny; this case study design also included theoretical replication by utilising Oseni and Pollitt (2014) EMI criteria among embedded units of analysis (member countries), as well as replication in the descriptive analysis of electricity markets within individual BIMP-EAGA member countries;<sup>45</sup> this research seeks to build an explanation of the occurrence of cross-border electricity trade in a particular subsystem of Southeast Asia. The IPE theories of neoliberalism and DS were identified as "vehicles" for explaining key themes identified (Vaismoradi, 2013) within this case study, resulting in analysis of deeper IPE implications and the progressive elaboration of neodevelopmental statism in energy and sustainability-based IPE analysis.

#### 3.3.b Elite & Stakeholder Interviews

When interviews were first initiated the topical focus was cross-border electricity trade in East Asia broadly. However, initial background interviews with Former Government—Informant 1, Multilateral Organisation—Informant 2, Academic 20, and Academic 26 led the author to explore Southeast Asia more specifically. Feedback pointed to Southeast Asia as a central location for increased clean energy investment and development, as well as a key player in the global response to climate change; concurrently it was also demonstrated that the sub-region is under extreme energy pressure, both for supplies to power the expanding regional economy and for growing electricity demand as a result of increased access (IEA, 2019a). The reality of the sub-regional climate and energy situation pointed to Southeast Asia as a logical location for deeper analysis. An additional background interview with Academic 26 and subsequent snowball sampling of further interviews led the author to narrow down the case study focus to BIMP-EAGA, which data analysis demonstrated was an interesting practice to undertake due to the role of cross-border electricity trade in sub-

<sup>&</sup>lt;sup>45</sup> Not to be confused with literal replication of multiple case studies, which Yin (2003) discusses at great length in case study design (pp. 39-42). This case study is not representative of all crossborder electricity trading relationships, it is representative only of the situation between the economies examined; this also supports the use of Critical Realism in the epistemological framing of this research, whereby a variety of experiences can be true in the same setting but differentiated by only a few changing variables or actors involved.

regional energy and electricity access, as well as the role of ASEAN in advancing EMI in the subsystem.

Elites and experts (both of which make up the interview subjects of this research) have been identified by Sovacool et al. (2018) as being exceptional groups that contribute to empirical novelty in research and extracting data from them "can provide an important, often missing contribution to a given literature" (p. 20). In this case, being able to combine interview data from subjects that are members of 7 different fields (Academic, Business, Development Bank, Former Government, Multilateral Organisation, Journalist, Think Tank), offered increased novelty and insight into this specific topic that is not available in the same way elsewhere.

Prior to the commencement of interviews the author created a mind map using NVIVO Plus 12. This map was meant to connect the disparate areas of study, various perceived actors to be involved, and potential issues that may arise in relation to the research topic, cross-border electricity trade in East Asia. The author attempted a mind map as an organisation tool (see figure 3.7 in Section 3.4), finding this also helped inform the development of the research project aims, research questions, and research objectives as interviews progressed. Revision of these items based on clarity of issues and interview feedback was helpful in the development of the research project aim, research questions, and research approach, and fundamental in the progression of the research methodology. The subsequent research project aim, research questions, and research objectives are summarised in Table 1: Research Plan, previously introduced in Section 1, *Methodological Approach*.

This method of matching research objectives and questions with the appropriate means of data collection and analysis is identified by Sovacool et al. (2018) as an important factor in producing rigorous qualitative research. This author chose to use a content analysis of interviews via thematic analysis of interview responses, identifying commonalities among interviews and matching these commonalities with themes that emerged and axial codes that stem from each theme.

In addition, the author chose to identify a case study for in-depth examination of a particular set of countries in East Asia and their engagement in cross-border electricity trade. Yin (2009) points to case studies as using multiple sources of evidence:

...with data needing to converge in a triangulating fashion, and as another result, benefits from the prior development of

Yin argues that, as such, a case study design is not just one method, but a collection of methods and designs used to create a research strategy that is used to answer clearly defined questions. In this context, the defining and refining of research questions based on interviews and data collection allowed the author to assess the groups, organisations, policies and countries (Sovacool et al., 2018) involved in cross-border electricity trade in East Asia and their contribution and role in the expansion of EMI in the region.

The author's approach to data analysis developed into a combination of both a narrative and systemic review—structured, including the creation of research questions, careful choice of interview subjects, thematic analysis of interview results, and also synthesising collected data within identified themes (Denzin & Lincoln, 2011). Sovacool et al. (2018) identify systemic literature reviews as being time intensive and subject to bias; this is true in the author's experience. Ultimately this inductive approach seeks to generate ideas and hypotheses (Denzin & Lincoln, 2011) related to the success and failure of cross-border electricity trade and EMI initiatives in East Asia, rather than prove one policy approach is better or worse.

# 3.3.c Reliability & Validity

Arguments for and against validity in research are often divided across quantitative and qualitative fields. In the case of this research the author does not believe that full validity can be achieved, or that reliability of interview subjects and data can be 100% guaranteed.<sup>46</sup> However, the author did take certain steps to ensure measures taken were consistent and promoted trustworthiness in research design and methodological rigor (Rudestam & Newton, 2007). These include conducting multiple interviews via the snowball sampling method, refraining from giving opinions on interview subject feedback, and the creation of a thorough interview protocol (see section 3.4 and Appendix C for more details) with open ended questions, question bridges, probe notes, and predetermined questions to focus on if time were to run out (Berry, 2002, p. 681). Efforts to ensure internal validity included utilising multiple data sources (Rudestam & Newton, 2007), consisting of stakeholder documents provided by informants, publicly provided data, and organisational data sets. The audit trail

<sup>&</sup>lt;sup>46</sup> See also Lincoln and Guba (1985) or Rudestam and Newton (2007) for more debate on the presence and presentation of validity and reliability in qualitative research.

(Rudestam & Newton, 2007) of this research was also detailed and organised, allowing for quick reference to interview transcripts, data about interview subjects, recommendations, and author notes on interviews, enhancing the trustworthiness of data gathered. Triangulation was another trustworthiness tactic that was employed, whereby the author used multiple methods and sources to illuminate common themes (Rudestam & Newton, 2007, p. 114), which is also in line with methodological choices outlined previously in this chapter. In addition, the author thoroughly read up on each interview subject prior to meeting, offered conversational questions about subjects' roles and organisations to begin, and was flexible where interviews took unexpected turns (Berry, 2002, p. 682).

The choice of interview subjects was decided via two avenues—first, a selection of interviews were identified and confirmed prior to commencing. Then using the snowball sampling method additional interviews were explored. While snowball sampling was useful in expanding the interview sample size, and also exposing the author to increasingly narrow expertise relevant to this research, there are critiques of the method. Notably in social sciences the critiques centre around sample diversity and do in some cases call into question the validity of research where all interview subjects come from the same or similar views, or when the 'grounding' of an interview sample comes from the original interview that snowball sampling originated from (Biernacki & Waldorf, 1981).

In their work on snowball sampling Kirchherr and Charles (2018) provide a broad overview of scholarly arguments for and against sample diversity, arguing that ultimately validity itself is a contested idea but sample diversity can be improved across qualitative research areas. Kirchherr and Charles (2018) define sample diversity as "the diversity of a sample as a measure of the range of viewpoints that have been gathered on a central phenomenon" (p. 3). Providing sample diversity can be achieved in a number of ways, which Kirchherr and Charles expand upon from previous literature, providing a quantitative analysis of effectiveness of particular methods. This includes using a diverse sample 'seed'—the initial interview subjects must be as diverse as possible. This was done in this research, where a variety of organisations and subjects were targeted, with additional subjects from the same organisation only being used if recommended by the original seed. Additional methods used to ensure diversity in the snowball sample included face-to-face interviews where possible and persistence (to a certain extent, as suggested by Kirchherr & Charles, 2018) in soliciting interviews from particular subjects.

One particular argument in support of snowball sampling that resonates with this author is that generalisations about complex issues do not serve the contextualisation of complex ideas or issues (Creswell, 2005, p. 203). This is particularly true in area studies, where generalisations of populations or regions are avoided (Huotari et al., 2014). Other arguments against validity concerns in snowball sampling include support for the value of unique knowledge that is obtained via interviews—no interviews like this, for example, have been conducted at this exact time or with these exact people in history (Noy, 2009).

Snowball sampling was limited to a certain number of interviews as a result of sample saturation-whereby the collection of additional interviews did not reveal new or novel results (Glaser & Strauss, 1967; Morse et al., 2002; Rudestam & Newton, 2007). Among qualitative PhD research examined by Mason (2010), the average sample size is in the low 30s, reflected similarly in this author's own work. A thorough review of literature on sample saturation by Mason found that the recommended sample size for reaching saturation in qualitative research ranges from 6 to 60+, as is similarly reflected in PhD research that was surveyed (p. 4). This presents an interesting quandary, as qualitative research requires a large enough sample size that important or relevant positions are uncovered (see earlier point re: generalisations), but small enough that the data collected does not become repetitive (Mason, 2010, p. 2). While sample saturation (or adequacy, as referenced by Morse, 1995) is a common tool employed by researchers across fields (and particularly among PhD researchers in qualitative research projects) sample saturation is a contested topic and can lead to research bias and flaws in sample selection and data analysis (Mason, 2010). Despite this contestation there are a few steps that can be taken to strengthen validity—this includes exploring context of saturation further if time allows, following the data and results as opposed to aiming for an arbitrary sample number, recognition of the limits of qualitative research, and understanding that saturation is a degree not a finite number that is dependent on the research and tools available at that particular time (Mason, 2010). In addition, this research plan did not have a final number goal for interviews and aimed to avoid directing interview sampling based on data but instead on snowball sampling.

A significant limit to the validity of this research is the sub-populations sampled. The sub-populations chosen for this research included public and private sector leaders in areas of energy, electricity, trade, and climate policy as well as some academics. A significant sub-population missing from this sample is civil society. The author chose to omit civil society subjects due to access, potential language limitations, and already existing contacts among potential interview subjects. This limitation, and how it can be addressed in future exploration of this topic, will be discussed in the final chapter of this research Chapter 7, *Conclusions*.

An additional limit to the validity of this research can be found in the recommendations—the majority of interview subjects had similar liberalisation recommendations for national electricity markets. This recommendation is also reflected in the majority of literature, and in the author's view, reflects a common global and societal tendency towards capital driven economic systems and practices. This is not to say these recommendations are inherently incorrect, simply that the positionality of interview subjects is reflected in their interview responses.

#### 3.3.d Ethical Considerations

In order to ensure reliability of data collection and research outcomes the author completed an ethical review prior to the commencement of interviews. This was completed in line with University of Leeds standards and practices and was reviewed by the University of Leeds Faculty of Arts, Humanities, and Cultures Research Ethics Committee. In addition to responding to questions related to the ethical considerations of the research planned, the author was also required to submit a risk assessment in which cultural and geographical considerations were taken into account in the research design and interview protocol. This ethical review application was approved with no corrections (See Appendix B for ethical approval response). The following subsection will explore these considerations in detail, situating the author's ethical considerations within the greater research design and methodological choices.

Elite interviews, while effective, do pose ethical challenges for researchers. In this case, the author was seeking to obtain the views, opinions, and insight of interview subjects rather than official views from the organisations they work through or represent in a professional capacity. As such, the author did not encounter ethical issues related to conflict of interest any or culturally/politically sensitive issues. The information looked for was largely based on the experience of interview subjects working publicly or researching in an energy, trade or governance capacity, therefore limited politically sensitive information was exchanged.

In one instance the interview subject (*Think Tank 11*) asked for certain experiences related to national policy making in East Asia to be omitted. In that case the questions and answers were redacted and not included in the

subsequent interview coding and thematic analysis. In another instance *Former Government—Informant 1*, who began as an interview subject but became an informant, providing multiple interviews and a variety of resources, asked for some internal organisational materials to be omitted from final presentation, but suggested they would be helpful in providing insight into the dynamics of trust within ASEAN. This was fundamental in improving the author's approach to trust, and guiding the author's exploration of the theme, but naturally influenced the author's positionality.

Information on the purpose of the research, the researcher's responsibility, and the subjects' rights to remain anonymous were introduced to the subject before he/she agreed to participate. The interviewer's identity as a non-judgemental, politically neutral researcher was especially clarified before the interview to minimise potential misunderstanding in the conversation. Information was provided via email and in person prior to the commencement of the interview, offering the interview subject opportunity to withdraw, general expectations, and insight into topics to be discussed. The author aimed for each interview subject to enter into the interview exchange fully informed with no surprises.

The researcher was also attentive to different cultural contexts and aimed to employ cultural competencies as a means to build trust with participants and effectively carry out qualitative interviews (Suh et al., 2009). Suh et al. (2009) adapt the work of Campinha-Bacote (1994, 2002) in health care industries to interviews with Asian immigrants. In this context cultural competencies are identified as cultural awareness, knowledge, skills, encounters and desires employed with interview subjects from different cultural backgrounds or in different cultural settings (p. 195). For the use of this research the author employed different cultural competencies with different interview subjects; With Japanese subjects honorific speech was spoken throughout the interview; with Chinese subjects social norms were also abided to show the interviewer's respect for subjects; with Southeast Asian interview subjects consideration of shared knowledge, experiences, and reverence for expertise were demonstrated; with American interview subjects shared connections were utilised to demonstrate utility of building relations and 'networking'. Postinterview contact (e.g., thank you letter, gift<sup>47</sup>—as is common in Asian cultures

<sup>&</sup>lt;sup>47</sup> For more on gift-giving for relationship building in Asia see D'Souza (2003). This practice was also referenced in the ethical review and approved given author's careful considerations of implications. See Appendix B.

when engaging in exchanges with senior officials—information about research progress, etc.) was made with all interview subjects in order to encourage a reciprocal relationship and aid in follow up post interview. The work of Huotari et al. (2014) was very helpful in placing regional studies within the research methodology and cultural considerations necessary for doing research in Southeast Asian contexts.

Data derived from the interviews was stored in accordance with University of Leeds standards and practices, as approved via ethical reviews. This is discussed in the following subsection.

#### 3.3.e Data Protection

During the research process, data was stored on the University of Leeds M: drive and University of Leeds OneDrive portal for temporary storage, analysis, and reference during travel and the write up process. Interviews were transcribed as soon as possible, immediately following the completion of interviews so that timely follow up could be managed if necessary. These interview transcripts were stored electronically on the M: drive and on OneDrive. Once an interview was completely transcribed, the written notes of the interview were disposed of. During the course of the research, electronic data was stored in both nonproprietary and proprietary formats (such as Microsoft Word) for the convenience of the researcher. OneDrive was also utilised while travelling for ease of access and through secure servers only (i.e., no internet cafes, etc.).

After the conclusion of research, some data will be available on the University of Leeds' Repository. Data will also be saved in open-source formats to ensure long-term accessibility. Types of data to be shared include interview transcripts, statistics, and case study data. All anonymity of subjects will be preserved.

Additional data protection measures have been carried out to ensure protection of the interview subjects, who agreed to participate based upon anonymity. Data collected during the interviews was entered into the research deliverable with no information allowing the identification of the source included. Direct quotes were edited with subject approval so as to remove any identifying features. Interview notes did include the name, position, work place, home country or any other personal data of each respondent. The interview notes, when stored temporarily, were stored in password protected computer files. Coded identification data and pseudonyms was stored in separate password protected files that only the researcher maintains access to. The identity of a participant's workplace was identified only if it was important for the research's purpose, and this practice was done sparingly. For example, a participant who works for a research organisation may choose not to disclose the identity of his or her workplace but may allow inclusion in the research that he or she is a "development bank representative" or "official in the Ministry of Trade and Industry in Japan," etc. However, participants have remained anonymous unless they explicitly wish to be named in the research. No one but the researcher has access to this research data.

# 3.4 Methods: Data Collection & Analysis

The following section describes the interconnected data collection process from start to finish, identifying processes used and methods combined. This mixed method approach is reflective of the cross-disciplinary nature of cross-border electricity trade and the need for a multi-level perspective of EMI issues given the local, sub-regional, regional and global actors and institutions involved.

#### 3.4.a Preparing for Data Collection

In preparation for data collection and based on the critical realism epistemological starting point described previously, this research was initiated with an exploratory approach (with the intention of ultimately building an explanation for this unique case) focused on identified research questions. This was done with the aim of identifying any additional theoretical and analytical approaches that may be relevant based on the issues that surround EMI. To begin, the author conducted a thorough review of cross-cutting literature in the fields of IPE, East Asian studies, and sustainability research with a specific focus on electricity market development, expansion, and integration. Literature reviews took place in separate Microsoft Word Documents identified by theme. Key terms and themes were identified across the literature reviewed and were taken into consideration when analysis of interviews began and throughout the thematic analysis process. In some case new themes were identified in the interview analysis, and then reviewed in relation to already consumed literature (i.e., trust appeared in interviews but was not originally identified as a relevant theme in the literature review). Using information gained from the literature review process, research questions were refined and an initial case study design developed. Preliminary public data was identified and ethical approval and risk assessment were attained prior to conducting initial research trips and background interviews.

#### 3.4.b Documentary Data

Primary and secondary documentary data played a vital role in shaping the research design, data collection methods, and results of this research. Documentary data was collected prior to data collection, during interview collection, and following completion of interviews and during data set analysis. Documentary data includes confidential and public government and policy documents, academic literature, personal notes and "insider" reports collected from interview subjects and relevant online resources. These documents supported interview question development, thematic analysis, and background understanding necessary to approach this cross-disciplinary research. In the case of confidential documents additional research snowballed into identified. relevant areas, expanding the scope of understanding used in approaching eventually data analysis. Secondary data was primarily used for electricity market data collection, in the forms of energy data sets publicly available through research and governmental organisation websites; however, in some cases interview subjects did point the researcher towards additional resources for expansion of electricity market data. In a number of cases data sets had to be combined or updated with new information where existing electricity market data was lacking either some indicators and/or using old data (as was the case with interconnection maps developed and utilised in Chapter 4, Sub-Regional *Market Factors*). In such cases source and organisation information has remained protected and confidential.

Analysis of these different documentary data forms assisted application of IPE themes to the cross-disciplinary data collected via interviews and electricity market mapping. Combining IPE themes with electricity market indicators, EMI pre-requisites as identified by Oseni and Pollitt (2014), and interview subject viewpoints opened up the analysis to include the development of the author's own EMI indicators (identified as "O&PCRB EMI Criteria" in Chapter 5, *National Market Factors*). This was vital to the author's understanding of EMI and cross-border electricity trade potential and expansion in Southeast Asia, and indeed resulted in original contributions to this cross-disciplinary area of study.

#### 3.4.c Interview Data

32<sup>48</sup> elite research interviews were conducted with government, business, and academic leaders in areas of energy policy, electricity market integration, variable renewable energy integration, climate change responses, and East Asian energy and climate policy (Dexter, 1970, p. 19). Subjects included representatives from international organisations, senior level corporate representatives, energy experts from Asia, and senior level retired or active government leaders. The author's experience working in the field of energy policy, with a particular focus in the Asia-Pacific, aided in obtaining these interviews and offered a credible advantage to requests for interviews. The risk assessment discussed in Section 3.3.a of this chapter was submitted to the University of Leeds for ethical review and approval prior to the commencement of interviews (See Appendix B).

The number of participants was estimated by taking into account publication and recommendation goals, prior research experience, and the breadth and importance of this research. The author's own investigation into the number of qualitative interviews necessary for rich, complex, and detailed research has shown that quality is more important than quantity. In addition, in order to formulate a well-informed series of conclusions the author deemed high-level, experienced subjects as the goal interview subject audience. Among social science researchers there is no definitive requirement for number of subjects interviewed. Instead, there is a variety of minimum requirement for sample sizes (See: Gerson & Horowits, 2002; Mason, 2010; Bryman, 2012; Emmel, 2013). Sample saturation was instead deemed as the goal and is discussed in more detail in *Reliability & Validity* subsection. An initial target of interviews was estimated (Emmel, 2013), however, saturation was achieved around 32 interviews.

A portion of interview subjects were already identified and/or confirmed their willingness to participate prior to beginning the interview process. Based on discussion with senior level academics from both the United Kingdom and the United States 68 initial interview subjects were broadly identified based on knowledge and relationship with the author; this was narrowed down to 15 interviews based on responses, availability, expected time constraints, and relevance of interview subject expertise. Utilising snowball sampling additional

<sup>&</sup>lt;sup>48</sup> A total of 68 interview subjects were initially identified, of which 32 interviews were conducted and only 26 provided useful material. Further details can be found in Appendix A.

interviews were included based on interview subject recommendations (Bhattacherjee, 2012), raising the number to 32. Based on these 32 interviews only 26 provided useful feedback, due to quality, information gleaned, and impact of the interview. This figure was finally settled on, as opposed to adding new interview subjects, due to the theoretical saturation of interview material, in which no new data appeared and all primary concepts had been well-developed and further context was targeted for exploration (Emmel, 2013).

In some cases, multiple sources from the same organisation were utilised based on snowball sampling recommendations. This did offer the opportunity to receive multiple viewpoints from within a single organisation, which were in some cases different. However, this tactic must also be used sparingly so as to avoid moving the results in a biased direction, whereby multiple subjects have the same responses based on their internal bias as a result of organisational ties or organisational stances. Fortunately, inter-organisational recommendations only occurred 4 times.

Ethical dilemmas in conducting personal interviews were considered, as a number of interviews were conducted with former colleagues. Considerations include context, conflicts, and openness, and where possible interviews were conducted in person and over multiple meetings (Woliver, 2002). Interviews were also conducted over a yearlong period in a variety of locations, based on schedules, travel, and the occurrence of conferences or workshops that facilitated meeting in person. This method of fieldwork was chosen both for convenience and finance purposes—funding for attending or speaking at conferences and workshops supplemented author funds and time could be maximised for international trips to multiple locations.

In person interviews were conducted in the United Kingdom (London, Leeds), Germany (Berlin), France (Paris), the United States (Washington, D.C.), the Philippines (Manila), Japan (Tokyo), and South Korea (Seoul). Additional interviews via skype were conducted when travel was not available, including interview subjects in Australia (Melbourne), Indonesia (Jakarta), Singapore, China (Beijing, Shanghai), and the United States (New York).<sup>49</sup> In accordance

<sup>&</sup>lt;sup>49</sup> Hine (2005) explores methodology in virtual research, including interviews, in her edited edition, *Virtual Methods*, which was useful in considering the challenges unique to online interview relationships. In particular, Joinson (2005) discusses ways in which virtual (online or phone interviews) can differ from in-person interviews in terms of data collection,

with common data collection recommendations (See: Rudestam & Newton, 2007; Yin, 2011), interviews were recorded with subject permission and transcribed by the author within 24 hours of interview completion. Transcriptions were saved on the University of Leeds document cloud in line with data protection procedures outlined in Section 3.3.e. Transcriptions were analysed by the author following completion, whereby supplementary material, including field notes taken during the interview and resources identified as requiring further research or clarification were addressed and also included in document comments. Transcriptions were formatted following NVIVO requirements for seamless coding, including heading styles and font specifications (Adu, 2019). As transcription occurred in a variety of geographical locations, depending on author location, uploading to NVIVO did not occur until following travel return to Leeds.

As is standard in elite interviewing, open-ended questions were primarily utilised (Leech, 2002). Systemic approaches to common methodological issues associated with elite interviewing were employed, including probe notes, a set of decision rules, subject bridges, and a few, optional, probe questions (Berry, 2002; Rudestam & Newton, 2007). Questions included planned prompts and informal prompts and avoided presuming questions (Leech, 2002). Interview questions also progressed "from nonthreatening to threatening" with "non-judgmental, non-threatening wording" as a means to remain sensitive to interview subjects' professional responsibilities and political sensitivities (Leech, 2002, p. 666).

Cultural standards and norms were also considered when interviewing elites in Asia. Professional experience of the author in the Asia-Pacific, as well as educational and personal experiences living in China, lent to credibility and sensitivity with subjects. In interview particular, mutual, influential professional relationships or ties (called *guanxi 关系* in Chinese) aided in the ability to gain access to and conduct interviews with Chinese or Asia-Pacific other elites (Beijing Foreign Languages Institute. 1979). The concepts (Beijing Foreign Languages Institute, 1979) of favour

trustworthiness, and honesty (pp. 25-234). In this research interview subjects were, for the most part, not complete strangers; this, coupled with the necessary anonymity agreed upon prior to interviews, did not result in any perception in differences between in-person and virtual interviews. Kivits (2005) discusses the building of rapport and strengthening of relationships via email correspondence and virtual interviews (pp. 35-50), while Orgad (2005) discusses strengthening informant trust with both virtual and in-person relationships (51-66), both of which were considered and observed when engaging with interview subjects.

(*renqing* 人情) and prestige face (*mianzi 面子*) were also considered as a means to return professional favours and provide elite subjects the respect their status and position provides them in Asian culture (Wang, 2014). Where interviews were conducted in a setting with which was unfamiliar, consultation with professional contacts took place for advice on proper etiquette and professional norms. While this research does not touch upon any explicitly controversial or taboo topics, respect and consideration was paid to foreign government views on information sharing and conduct with foreign researchers.

In addition, a detailed interview protocol was provided to each interview subject prior to the commencement of interviews both via email and in person. An overview of this protocol can be seen in Figure 3.3 as well as in more detail in Appendix C:

UNIVERSITY OF LEEDS	UNIVERSITY OF LEEDS
Cross Border Electricity Trade in East Asia: Interview Protocol	Background Protocol
Contextual Protocol: Research Objectives The variability of electricity produced from clean energy resources provides an opportunity for sub-regional and international electricity market expansion and the efficient disposal and purchase of power capacity. Based on the resulting potential for cross border electricity trade, this research examines the expansion of these markets in Asia and opportunities for their further development. A number of issues play a role in this research: including national and sub-regional electricity market development, sub- regional governance, interconnectors, and trade governance. A case study examination of (1) a European Union success in electricity trade and (2) a potential growth market in East Asia will be conducted following informative expert interviews. Introductory Protocol You have agreed to take part in a research study entitled 'Cross Border Electricity Trade in the East Asia.' The following information is for your convenience and understanding, but please do not hesitate to ask for clarifications or more details should you need to.	<ol> <li>How long have you been:         <ul> <li>a. Working in your current positions?</li> <li>b. At this institution?</li> <li>c. Retired?</li> </ul> </li> <li>What is your:         <ul> <li>a. Highest degree?</li> <li>b. Area of expertise?</li> <li>c. Probes: Years of relevant experience? Other relevant qualifications?</li> </ul> </li> <li>Briefly describe your role (office, committee, organisation, etc.) as it relates clean energy development / electricity trade / electricity market integration (if appropriate).</li> <li>a. Probes How are you involved in policy making/coordination/responses here? How did you get involved? What motivates your organisation?</li> <li>What are some of the current [research] projects you're conducting?</li> <li>Conversational or Transition Questions: (Pick and choose based on situation)</li> </ol>
This interview will take between 30-60 minutes. During this time, I have several questions that I would like to cover. Follow up via email may be applicable should we run out of time or be interrupted. I may ask you to provide information, recount events, or describe your experiences and understanding about issues of importance to trade and economy policy, electricity markets or clean energy utilisation. With your permission, the interview will be recorded in digital audio and subsequently transcribed. Once the transcript is finished, the voice recording will be disposed of. Participants will remain anonymous unless they explicitly wish to be named in the research. If you prefer anonymity, the data will contain no personal information. With your permission, I may include information on your occupation but this requires your explicit approval and the interview does not depend on it. The data collected during this study may be used in presentation at conferences or in publications. However, all anonymity will be preserved. You have the right to omit or refuse to respond to any question that I may ask. You will be given the right to withdraw at any point up to October 1, 2018. You also have the right to ask that any data you have supplied to me during the interview be withdrawn or destroyed. If you wish to withdraw from this study, let me know by phone or email at any time. My contact information is listed on the release form. Please sign this release form to demonstrate agreement with these stipulations. This document states that: (1) all information will be held confidential, (2) your participation is voluntary and you may stop at any time. Thank you in advance for agreeing to participate.	<ol> <li>In recent years, we've seen numerous positive trends in policy towards the transition to clean energy. Is the speed of change fast enough to meet the climate and efficiency targets?</li> <li>If we look at the East Asia/European Union, what have been the most important legislative initiatives in favour of the energy transition?         <ul> <li>a. Probe: Do you know of specific initiatives geared toward electricity market integration or electricity trade expansion?</li> <li>What is the role of trade policy in climate change adaptation?</li> </ul> </li> <li>What trade policy and/or theoretical approaches can make the best case for optimising clean energy's contribution in addressing climate change and energy security challenges?</li> <li>Key Questions: Electricity Markets:         <ul> <li>What is your understanding of cross-border electricity trade and/or electricity market integration in reaching global emissions reduction targets?</li> <li>What is the role of trade policy in the growth of cross border electricity trade?</li> <li>In what ways can governments incentivize or enable</li></ul></li></ol>

#### Figure 3.3 Interview Protocol Overview

This protocol was created in line with recommendations by Castillo-Montoya (2016) and primarily served to inform the interviewer of subjects to be covered, questions, and processes related to the interview process (anonymity, data protection, deadline for removal from project, etc.). This protocol was drafted and piloted with the first background interview, *Academic A20* and feedback was solicited from *Former Government Official—Informant 1*.

The following subsection will dive deeper into the analysis process of interview data. This includes identifying the coding process used and the resulting key themes that emerged, shaping the progress of this research and the outcomes to research questions.

#### 3.4.d Data Analysis: Interview Coding

NVIVO interview coding was done primarily as a function of content analysis of interview transcripts. Work of Vaismoradi et al. (2013) informed the author's selection of content analysis for data collected via interviews. Emmel (2013) also discusses the benefits and limits of content analysis, or systematic coding, for analysing large amounts of textual information, recommending NVIVO for word or category frequency and cluster analysis. All of these methods require researcher knowledge and methodical analysis of relationships, contexts, and discourses relevant to the material (Emmel, 2013, p. 190). NVIVO is primarily a content analysis tool, where trends and patterns of words are identified using manual (researcher led) and auto (program led) coding.

Interviews were initially coded using NVIVO's auto code function, by which all interviews are analysed for common words and phrases. Initially more than 600 patterns were found, ranging from 2+ mentions to 19 depending on the interview subject, expertise and questions. Subsequent codes were run via NVIVO programming to refine patterns and manual coding based on themes identified in documentary data collection and literature review were manually completed by the author. This led to a slightly refined 100+ patterns that were then manually narrowed down by the author, resulting in an eventual 68 patterns. This narrowing down process included combining related terms or nesting related terms as "child codes" where relevant, deleting similar terms or mentions that didn't pass a certain threshold (5) and adding new codes or "parent codes" where organisation was lacking and terms could be combined under umbrellas of similarity.

The process of content analysis and interview coding was, while visually interesting, very time intensive. It required reviewing the original code patterns

multiple times, as well as combing through the text of each interview individually, in order to make sure no potential data was ignored. As mentioned in the previous subsection, prior to coding interviews had to be individually formatted to NVIVO specifications in order for the coding process to run seamlessly. This required transcribing each interview, formatting each interview, and uploading each interview into NVIVO manually. All of this detail, while time intensive, did allow the author to thoroughly review each interview multiple times. This meant that deeper attention was paid than had a single NVIVO auto-code been run off of a computer-generated transcript.

This process does pose the risk of human error. However, the author found the NVIVO auto coding process created a number of repetitive codes, which required human analysis to decipher. For example, initial coding resulted in 65 mentions of energy in any form. This included mentions like "energy policy" and "energy policy planning", among many others. Deciphering which are of these mentions of energy are repeats of the same conversation, which had not been included in the first place, and which were completely separate entries required the author to review each and every mention independently.

In advance of beginning the coding process the author reviewed the NVIVO 11 and NVIVO 12 manuals, consulting both NVIVO resources and University of Leeds IT department resources. In addition, the author became very familiar with the work of Dr. Philip Adu, who utilises NVIVO for his cross-disciplinary social science-based research. In addition to his book *A Step-by-Step Guide to Qualitative Data Coding* (2019), Dr. Adu has created a treasure trove of online videos for students to learn the software. *Qualitative Data Analysis* by Grbich (2013) also provided an introduction to the software and its usefulness in qualitative, social science research. Learning the software, and how to generate codes automatically as well as manually, was itself a very time-consuming process.

Once interview material was narrowed down into smaller and smaller levels of codes analysis could begin. The process of analysing these codes and identifying common themes will be discussed in the following subsection.

#### 3.4.e Data Analysis: Thematic Analysis

The thematic analysis carried out via NVIVO coding was largely informed by the work of Braun and Clarke (2006), whose work on thematic analysis in a variety of social science fields has informed the author's addition of this mode of analysis. Vaismoradi et al. (2013) posit that while content analysis is more

widely known and often used interchangeably with thematic analysis, thematic analysis offers some yet unseen and unused opportunities for social science research (p. 400). Braun and Clarke (2006) posit that thematic analysis is a flexible method of analysing data across an entire collective data set (in this case interviews, electricity market data, and literature), and involves the discovery of shared commonalities, or themes, across data types (Braun & Clarke, 2006, p. 6). Themes are not merely waiting to be discovered, as with word frequency in content analysis, but links are made via the knowledge of the data that the researcher possesses (p. 7).

Thematic analysis is also not wed to a particular theoretical framework, so flexibility of the research approach is one additional, key benefit (Braun & Clarke, 2006, p. 8). As such, application of thematic analysis to this research, rooted in the epistemological approach of critical realists but informed by the structures and processes of IPE theory, is a natural fit. As Braun and Clarke say, "thematic analysis can be a method which works both to reflect reality, and to unpick or unravel the surface of 'reality'" (p. 9). In addition, researcher judgment is necessary to identify themes, and prevalence can be determined at varying levels and of varying importance. In this case thematic analysis followed a deductive approach, whereby themes line up with theoretical underpinnings of IPE theory, and analysis occurred at a latent level, where themes needed to be interpreted and theorised and relationships are assumed to be present (p. 13).

From the narrowing down of these themes the original mind map, discussed in Section 3.3.b, *Elite & Stakeholder Interviews*, was narrowed down to the key issues identified in thematic analysis of interviews and electricity market data. It can be seen as follows:



#### Figure 3.4 Resulting Mind Map

#### **3.5 Positionality of Research**

The positionality of this research stems from the author's own personal and professional experience and, despite efforts, has no doubt informed the scope of this project. Bourke (2014) identifies positionality of both the researcher and research participants as having an impact on the research product (p. 1). These impacts stem from the identity of those participating, impacting the methods, processes, and eventual product of the research. The author, a Caucasian female from an American family, with extensive educational and professional experience in Asia, has been influenced greatly by her own positionality in the world, with experiences, values, and even identity prejudicing her own perspectives and reflections on the work described in this chapter. This section seeks to briefly explain the positionality and reflexivity of the author's work, offering perspective on the methodology and methods employed in this research.

Prior to beginning the research described in these pages the author worked energy and trade policy research based out of Washington, D.C. but with responsibilities throughout the Asia-Pacific. In this role the author oversaw research, publications, and conferences for a variety of projects focused on global energy issues. In addition to professional experience the author has personal and educational experience in Asia spanning three decades. The global energy issues this author was exposed to professionally are reflected in the story of cross-border electricity trade and this research narrative— as a result of climate change there is a pending imperative for changes to occur in national, regional and global energy mixes, requiring the integration of renewable energies, and cross-border electricity trade provides one solution to the variability of this integration. Consequent to this prior professional experience and influence on the research in these pages, this research falls broadly under the category of applied research, discussed by White (2009): whereby research questions derive from professional settings rather than academic, they principally address immediate issues rather than broader scientific debate, they are often situated in organisational or institutional contexts, and they are concerned with specific sites or populations (p. 30). Even in crafting the research questions themselves the positionality of the author has played a role.

These professional roles and personal experiences have, combined, influenced the author's approach to academic research. The author recognises her personal bias towards China and preference for approaching issues from a perspective of policy solutions. As such policy remains an ever-present consideration as the author conducts research on the pressing energy and electricity challenges described in these pages.

As a result of their bias towards China the author opted, at the outset of this research, to not include China as the only case study or focus, and instead keep a broader approach (East Asia) that would be naturally narrowed down via interviews and research. As was discussed in the *Data Collection & Analysis* section, China emerged as a common consideration via the Belt and Road Initiative (BRI) and its potential funding avenues for electricity market integration and power sector expansion in Southeast Asia. In the chapter 3 section on *Case Study Design*, analysis included a description of how China actually *wasn't* the natural choice for a case study, and as such was omitted in that regard. The later inclusion of China via the theme of trust was fitting for the author professionally, however, may be viewed as being a result of internal bias

based around the author's own interest and experiences in China, in spite of the best efforts to remove this bias from the research.

Due to these experiences the author has had to, at various points throughout the research process and examine her own positionality. This can be referred to as reflexivity (Attia & Edge, 2017, p. 34-35), in which the author has taken a developmental approach to research by engaging in persistent self-analysis, continually assessing author impact on the research (prospective reflexivity) and research impact on the author (retrospective reflexivity). This has guided the development of this project, as questions are adjusted based on feedback and exposure to the issues, theory is adapted based on understanding and positionality, and methods are combined and/or chosen in order to reach a logical point that the data directs to.

During development fieldwork reflexivity was apparent in both its forms, reflexive and retrospective. First, interviews based around conferences and workshops were chosen, in large part, due to access and funding. However, as new opportunities became available the author needed to add or remove certain fieldwork trips. Similarly, when a fieldwork trip did not produce many interviews the author had to reassess the type of event to build fieldwork around, resulting in what Attia and Edge (2017) refer to as "an evolving research practice (retrospective reflexivity)" (p. 38).

The positionality and reflection of the author have shaped the ways in which this research has played out, in particular the methodological approaches chosen (See also Section 3.2, *Methodological Approach*).

# **3.6 Conclusions**

The variability of electricity produced from clean energy resources provides an opportunity for regional and international electricity market expansion and the efficient disposal and purchase of power capacity. Based on the resulting potential for cross border electricity trade, this research examines the expansion of these markets in East Asia and opportunities for their further development and does so in the context of neoliberalism and DS. A number of issues play a role in this research: including national and sub-regional electricity market development, regional governance, interconnectors, and trade governance. This chapter has sought to situate the development of this inter-disciplinary research using methodological triangulation. This chapter introduced the exact methods,

techniques, analysis, and viewpoints utilised to explore cross-border electricity trade in East Asia.

First, the author identified research questions, linked these questions to related research objectives, and identified the relevant methodologies needed to answer these questions. Two separate areas of study were identified as being relevant to this IPE research—East Asian studies and sustainability research. As such, the author chose a mixed method approach to compliment the interdisciplinary nature of this research, using a case study research design with multiple methods of data collection to analyse the realities of cross-border electricity trade in East Asia. The separation of this chapter into two distinct sections methods (the tools / what) and methodology (the justification for use of the tools / why) allowed for a distinction between the two in order to avoid the common academic pitfall of conflating them (Harding, 1987; Clough & Nutbrown, 2012; Schneider, 2014). This also allowed for clearly featuring the epistemological underpinnings of this research, offering a transparent explanation for the methodological approach, theory that informed it, and description of how the research was conducted. Case study choice was narrowed down using documentary analysis and recommendations from initial background interviews. Interview data was explored using content analysis and thematic analysis. In addition, data collection included an examination of electricity market structures in case study countries, providing insight into the needs of countries if they are to engage further in cross-border electricity trade. These two methods shed light on the interrelated nature of cross-border electricity trade and energy market integration (EMI). Subsequent analysis of key themes (chapters 4-6) will introduce EMI more specifically, providing perspective as to its relationship to cross-border electricity trade in the chose case study, BIMP-EAGA.

An exploration of current national policy, sub-regional initiatives, and regional and global energy and climate challenges has provided context to the difficulty and necessity of increasing cross-border electricity trade in East Asia. A qualitative approach of data gathering and analysis was used in the case study design, including an examination of official public and private sector information on electricity market structures in specific case study countries. This information was then contrasted with common recommendations identified in interviews and standard characteristics required in popular research on energy market integration (Oseni & Pollitt, 2014). The development of the methodological framework, methodology, and methods has resulted in the formulation of additional, sub-questions that will be further examined in chapters 4-6. These sub-questions (Andrews, 2003, p. 33-44) aid in exploration of each primary question, adding context to the issues involved. These sub-question are as follows, identified in bullets under the corresponding primary research questions:

#### Table 3.4 Sub-Questions

# **Objective 1: Provide overview of cross-border interconnections and assess market factors in Southeast Asia**

**RQ1:** How do governments understand the IPE of cross-border interconnections in Southeast Asia?

# Sub-Questions:

- What are the energy and trade policy incentives for increased crossborder interconnections?
- How can sub-regional electricity markets diversify their energy mix and reduce emissions? What Incentive to do so exist?

# Resulting Chapter 4 Theme: Sub-Regional Market Factors

Objective 2: Assess Market factors of Southeast Asian case study selection and identify areas of individual and collective reform needed to reach deeper levels of EMI among BIMP-EAGA economies

**RQ2:** How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?

# Sub-Questions:

- How does the political and economic structure of national electricity markets impact the expansion of cross-border interconnections in BIMP-EAGA?
- Is the structure of national electricity markets conducive to subregional and international policy goals in Southeast Asia?

#### Resulting Chapter 5 Theme: National Market Factors

Objective 3: Identify and analyse additional governance challenges to deeper EMI and increased electricity interconnections among ASEAN member states and case study selection

**RQ3:** What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian Development?

#### **Sub-Question:**

• What is the political and economic status of the ASEAN Power Grid?

# Resulting Chapter 6 Theme: Governance Challenges

These sub-questions, while not the primary focus of this research, offer insight into analysis that leads to answering the primary research questions from which they are associated. Each sub-question coincides with the associated objectives, themes, and primary research questions, aiding clarity in corresponding data chapters. The following chapters will explore these questions in more detail, starting with Chapter 4: *Sub-Regional Market Factors*.

# Chapter 4 Sub-Regional Market Factors

# **4.1 Introduction**

Chapter 4 is the first of three data analysis chapters, built around common themes identified in elite interviews. As discussed in Chapter 3 (*Research Design & Plan of Enquiry*), interviews were conducted with elites throughout East Asia, Europe and the United States. These interviews resulted in findings that are reflected in data chapters 4-6. Common themes were identified based on analysis of interviews and interview subject recommendations. Three primary themes, (1) sub-regional market factors, (2) national market factors, and (3) governance challenges are each discussed individually in their corresponding data chapter.

This chapter ties the story of this research together by laying the foundation of the first theme, sub-regional market factors. Recalling Research Objective 1, this chapter will provide an overview of cross-border interconnections and market factors Southeast Asia. Of the three primary research questions identified in Chapter 1, this chapter will address Research Question 1 (RQ1): *How do governments understand the IPE of cross-border interconnections in Southeast Asia?* Two sub-questions identified at the end of Chapter 3 will also be addressed: (RQ1a) *What are the energy and trade policy incentives for increased cross-border interconnections?* (RQ1b) *How can sub-regional electricity markets diversify their energy mix and reduce emissions and what incentives to do so exist?* 

This chapter will be structured as follows. First, based on insight from semistructured interviews, Section 4.2 *(Sub-Regional Interconnections: ASEAN)* will examine the current and projected state of interconnections between ASEAN members. This section will emphasise ASEAN's premier energy connectivity project, the ASEAN Power Grid (APG). This data will paint a picture of the IPE of cross-border electricity trade in the sub-region, making links to specific subsystems that have been identified in the APG project. Section 4.3 *(Subsystem Interconnections: BIMP-EAGA)* will similarly paint a picture of the APG and its progress based on an examination of the BIMP-EAGA subsystem.

Section 4.4 (*Discussion*) will present the contradictions identified in this case study analysis, namely the contradictions between national versus sub-regional policy and necessary sub-regional reforms for EMI to flourish. In addition, the role of neo-developmental statism will be used to explain the IPE of cross-border electricity trade in the sub-region. Section 4.5 (*Conclusions*) will summarise the

importance of ASEAN and its EMI projects, linking the challenges posed between the structure of electricity markets in the region and what is needed to achieve full EMI. This chapter will provide initial reform recommendations that will be built upon in subsequent chapters.

This chapter also sets the stage for subsequent data chapters and corresponding themes—*National Market Factors* (Chapter 5), and *Governance Challenges* (Chapter 6). In this chapter the focus on sub-regional market factors in ASEAN will provide context to the largest sub-regional cross-border electricity trade initiative, situating this initiative within the context of sub-regional development and governance challenges all while utilising IPE analysis to explain the different interests at play.

# 4.2 Sub-Regional Interconnections: ASEAN

In order to determine the status of the APG an analysis of interconnectors is necessary. This is significant to this research for three reasons: (1) in order to assess cross-border electricity trade in Southeast Asia it is necessary to see where interconnections are in place, and which countries require more attention in order to develop sub-regional cross-border electricity trade; (2) mapping current and projected interconnections in Southeast Asia will help answer RQ1 (*How do governments understand the IPE of cross-border interconnections in Southeast Asia?*) and the associated sub-questions, RQ1a (*what are the energy and trade policy incentives for increased cross-border interconnections?*) and RQ1b (*How can sub-regional electricity markets diversify their energy mix and reduce emissions and what incentives to do so exist?*); (3) Answering these questions will go partway to explaining the extent to which climate change has created imperatives for EMI and power sector reform in Southeast Asia.

Chapter 2 (*Cross-Border Electricity Trade: Conceptual Approach & Review of Literature*) introduced the APG and organisational methods for managing the APG, via the HAPUA organisation and their studies via AIMS (ASEAN Interconnection Master Plan Study). An important aspect to understanding the economic, political, and physical dynamics of cross-border electricity trade among ASEAN member economies is the process for increasing interconnections, which have been laid out in the AIMS (ASEAN, 2004). AIMS I identified three primary subsystems within ASEAN that could be realistically targeted for increased interconnections based on geographical location. These identified subsystems have also been adopted in AIMS II (2010) and AIMS III (2018). In the original AIMS I systematic review of these subsystems and the

entire sub-region found that a single, integrated power grid was not economically viable based on an inability of individual ASEAN member states to confirm economic viability of initial interconnection projects (ASEAN, 2004). In 2004 HAPUA was reorganised and the more ambitious AIMS II was subsequently begun.<sup>50</sup> Published in 2010, contrary to AIMS I, AIMS II found that an ASEANwide power grid was economically viable but should be reached via bilateral, subsystem, and then sub-regional interconnections, setting a new strategy for the progression of the APG that better reflected the economic and political ability of individual member states to commit to interconnection projects (ASEAN, 2010). AIMS II also utilised a more complex modelling approach than AIMS I, incorporating the formulation of cost generation capacity and transmission estimates, transmission expansion planning of each system, generation capacity and transmission expansion planning of interconnections and overall determination of total cost savings across the sub-region. AIMS II also identified 12 additional interconnections for additional development (as well as the 5 interconnections in place at the time) expanding the scope of the APG project and building off of developments in the energy markets of individual ASEAN member economies and the national market structures already in place. Following on AIMS I's subsystem identification, AIMS II also adopted the same geographic subsystems, which were subsequently incorporated into HAPUA's EMI efforts—the establishment of an integrated approach to increasing subregional interconnections (APAEC, 2010). This can be visualised as individual markets, progression to multilateral markets, and eventually the joining of multilateral markets into a sub-regional electricity market, where electricity can

<sup>&</sup>lt;sup>50</sup> Despite HAPUA's role in the development of the APG via the AIMS, HAPUA does not have enforcement or financial and political obligations to ASEAN member economies (IEA, 2015, p. 66). Similar to other working groups and committees within ASEAN, HAPUA is based on principles of cooperation with no implementation authority and remit only to manage and consolidate utilities and power authorities (p. 67). HAPUA itself has a total of 8 sub-groups and answers to the leadership of ASEAN Energy Ministers and ASEAN Senior Officials (Srisuping, 2013, pp. 6-7). Other organisations and working groups also play a role in the APG behind the scenes, including: ACE (the ASEAN Centre for Energy), which provides overall administration, coordinating and technical consolidation of all ASEAN energy initiatives (Andrews-Speed, 2013, p. 163); AERN (the ASEAN Energy Regulatory Network), which produces research on policy plans related to regulatory and legal frameworks for international and cross-border trade and investment in the APG (IEA, 2015, 66); and the APGCC (the ASEAN Power Grid Consultative Committee), a sub-group of HAPUA designed to facilitate the APG MoU implementation (p. 66). ASEAN's governance limitations are also the limitations of the variety of ASEAN energy groups and committees, and thus no single sub-group has power to hold member states accountable to ASEAN APG commitments or govern the APG process single-handedly; ASEAN's energy initiatives follow the "ASEAN way" of non-interference that can be found across the institution and its issue areas (Porter & Situmeang, 2005, p. 3).

be seamlessly traded among partners and their synchronised and standardised national electricity markets. An example of this EMI progression can be seen below:





Source: Adapted from AIMS II (ASEAN, 2010, pp. 31-32), integrated approaches to interconnection.

AIMS III (pending release) takes planning a step further than AIMS I and AIMS II. In addition to providing a status overview, AIMS III also sets criteria for assessment of and benchmarks for integrated power system performance (ACE, 2018). The edition of AIMS III will also include a feasibility study (ACE, 2018B) of bilateral and multilateral cross-border electricity trade that incorporates a high rate of renewable energy utilisation (assessing the 23% renewables goal previously mentioned against reliability and identified system needs). According to insight from *Multilateral Organisation—Informant 2* it is also expected that this assessment will include guidelines for policy makers at the highest levels, incorporating sub-regional targets and national capabilities related to an integrated ASEAN Power Market.

Based on data available as of January 2019 projected and planned interconnectors are 30 and 9, respectively (Figure 4.2).<sup>51</sup> Updates are expected

<sup>&</sup>lt;sup>51</sup> All interconnection data in figures 4.2-4.7 and Table 4.1 is estimated based on a variety of sources: APAEC (2004, pp. 5, 11-12, 31-32; 2010, pp. 8-9, 12-14, 16); Srisuping (2013, pp. 5, 17-20); Andrews-Speed & Hezri (2013, p. 17); Ibrahim (2014, p. 9-11); Shi (2014, pp. 18-20, 124; 2016, pp. 675-676); IEA (2015, p. 17-19, p. 64; 2019, p. 17, 19, 24, 37, 39-43); Hermawanto

when new sub-regional assessments become available; however, this data is standardised and compiled by the author, as most data sets don't include interconnectors since AIMS II was first published. Generating capacity is expected to quadruple, and through sub-regional interconnectors ASEAN hopes to reach 100% electrification.





Current status of the APG is fraught, however, and moving slowly. Multilateral trading is virtually non-existent. A number of bilateral interconnectors have been put in place, but multilateral trading is yet to be realised on the scale APG set out to achieve originally (APAEC, 2010; ASEAN, 2017a,b). Forecast interconnections are set out in each subsystem, with the most advanced interconnections in the GMSR subsystem and the least advanced in the BIMP-EAGA subsystem (Pacudan, 2016; APAEC, 2004; APAEC, 2010).

<sup>(2016,</sup> pp. 27-29); Li & Kimura (2016, pp. 23, 30, 32-34, 43, 49, 113-114, 121); Takapong (2016); ASEAN (2017); Halawa, et al. (2018, pp. 7, 9-12); IRENA (2018, pp. 72-73); ACE (2018b, p. 5-7, 12); The author is also grateful to interview subjects for their feedback and suggestions.


Figure 4.3 Forecast ASEAN Interconnections

As of data collection actual ASEAN interconnectors are as follows:

Figure 4.4 Current ASEAN Interconnections



Interviewees within and outside the sub-region are in consensus that many of the barriers to the APG are related to national versus sub-regional perceptions regarding cost, power system needs, and security of supply (Former *Government—Informant 1; Multilateral Organisation—Informant 2; Multilateral* Organisation 18; Academic 21; Academic 26). First, a number of power systems in the region require updating that has not happened at the national level, including modernisation of current physical transmission infrastructure, increased transmission infrastructure, increased capacity, growth in rural-urban connections, and efficiency improvements in old systems. This is due in part to high upfront costs and no immediate, short-term need-as well as the perception that high up-front costs associated with the modernisation of power systems are not immediately necessary for security of supply (Academic 26). However, for EMI to take place the diverse power systems of the region need to be standardised (Navarro & Sambodo, 2013), a view reflected among the same interview subjects and also high-level policy discussions (Former Government— Informant 1; Multilateral Organisation—Informant 2; Multilateral Organisation 18; Academic 21; Academic 26; Government Conference 12; Government Conference 13; Government Workshop 14). This includes a harmonisation of transmission standards and procedures, uniformity among regulators, agreement on sub-regional procedures and rules related to dispute settlement (or agreement on incorporation of outside dispute settlement, i.e., the WTO), and agreement on pricing mechanisms (ASEAN, 2017b). Based on interviews these standardisations also need to exist within national level regulations that are complemented by sub-regional coordination. For example, according to *Multilateral organisation 18:* 

...We look at mostly [sub-]regional, then national, coordination in our monitoring [of interconnection projects], but we have observed there is some lag between the two. This may impact speed of change needed for further electricity trading. So national level is an important part of study, too.

HAPUA, and other working groups within ASEAN, cannot force standardisation or reform; these changes must come from the national level despite 8 HAPUA working groups and their studies on integration and reform (APAEC, 2010). Additional barriers to expansion within ASEAN specifically include other aspects of policy making at the national level—transparency within national markets, political will at the national versus sub-regional level, and national priority of perceived energy security concerns over electricity access motivations (*Former Government Official—Informant 1; Multilateral Organisation 18; Academic 21;*  *Academic 26*). A large portion of the challenges identified in interviews and documentary research also cross over into the 'misinformation' category, whereby understanding at the energy ministerial or other national level is not reflective of policy understanding at the ASEAN governance level. For example, according to *Think Tank 15* there is a fear of energy security being compromised among ministerial level partners:

We [think tanks in this area] focus on increasing energy security and messaging that regional initiatives will increase energy security for all partners... But we do have to fight the misperception that energy security is threatened via the most important regional initiatives.

This subject expanded that while initiatives are moving forward within ASEAN, when national level governments become involved hesitations begin to creep into policy making—including concerns regarding trust of neighbours, how unexpected energy supply pressures might impact interconnection contracts that can't be met, and whether or not energy security will be impacted negatively by cross-border electricity trade (*Think Tank 15*). A variety of sources have said these challenges have been continuous over the last 10+ years, and thus, are more difficult to overcome than may be perceived by outsiders (*Former Government Official—Informant 1; Think Tank 15; Multilateral Organisation 18; Academic 21; Academic 23; Academic 26*). Here it appears that national level incentives for engagement are not as strong as sub-regional incentives—in the form of global climate pressure, sub-regional climate commitments, and AEC interconnection goals versus national interests and perceptions regarding trust and security.

Additional APG barriers lie in affordability and funding challenges—national level concerns over bringing in outside funders (to be explored further in relation to one of the major themes identified in interviews, *Chapter 6: Governance Challenges*), questions regarding immediacy of needs when energy or other concerns 'must' (financially) take precedence (*Academic 26; Business 31*), and the presence and prevalence of fossil fuel subsidies and thus "vested political and economic interests" (*Think Tank 15*) in maintaining the status quo of power systems (despite limitations). The IEA (2017a) predicts that investments of \$2.7 to \$2.9 trillion dollars will be required to meet energy supply and efficiency demands in Southeast Asia by 2040, however, it also identifies fossil fuel subsidies as an obstacle that investors will need to overcome. Similarly, electricity subsidies are also identified by the IEA an obstacle to more efficient energy use, resulting in a potential cumulative cost of \$350 billion from

2017-2040 (IEA, 2017a). Subsidies are perceived by many international organisations in the energy policy space as not only propping up fossil fuels in many countries, but also prolonging the life of older technologies and systems that would otherwise be upgraded (REN21, 2019b). Here, subsidies become not just tied to slow moving renewable energy integration but also efficiencies associated with energy poverty reductions and climate change commitments. It should be noted, however, that many subsidy discussions revolve around standard neoliberal market reforms pushed via the dominant global economic system (subsidies arise again in Chapter 5, *National Market Factors*, and will be discussed in more detail in Section 5.4.c.). A summary of these key barriers is as follows:







One interview subject (*Academic 26*) insists that for progress to be achieved at the level required for deeper diversification of the region's energy mix and increased interconnection, incentive needs to come from the national level, and be encouraged by ASEAN governance at the sub-regional level. Political commitment, this argument goes, is the most important factor in seeing the development of a transparent and operational sub-regional power system. This feedback is also in line with documentary evidence on APG progress and barriers (Navarro & Sambodo, 2013; Andrews-Speed & Hezri, 2013; Wu, 2019) and will be further analysed in Chapter 6, when the discussion moves to the theme of *Governance Challenges*. What is demonstrated via the identification of sub-regional interconnections and barriers to their further development is that barriers exist at the national level, across all ASEAN member states, despite

member state commitment to the APG. These barriers are different depending on the national market, physical infrastructure, policy commitment and considerations, and the presence of economic tools, such as subsidies. The following subsection explores BIMP-EAGA specific factors in cross-border electricity trade at the subsystem level, whereas Chapter 5 will look at national level factors in cross-border electricity trade expansion and sub-regional energy market integration.

# 4.3 Subsystem Interconnections: BIMP-EAGA

The BIMP-EAGA subsystem, also known as the east system, is the least advanced of the three subsystems identified by ASEAN's HAPUA AIMS studies (Pacudan, 2016). Made up of Brunei Darussalam, Indonesia, Malaysia and the Philippines, this subsystem is farther behind GMSR, with only one interconnection to date, established as recently as 2016. However, potential for continued expansion of interconnections, integration of renewable energies, development of national electricity markets, and growth in electricity access is quite high in this subsystem. Central to the opportunity of BIMP-EAGA is the island of Borneo, made up of three different BIMP-EAGA countries: Brunei Darussalam, Indonesia, and Malaysia, and is the largest island in ASEAN (ADB, 2014).

HAPUA envisions expansion of the Borneo system will include inter and intraisland electricity trade, expanding the power systems of neighbouring Southeast Asian islands within BIMP-EAGA (ADB, 2014). This would result in interconnections that would include the peninsular Malaysia, the Philippines and northern Indonesian islands, expanding BIMP-EAGA in line with HAPUA goals (ADB, 2014). Planned interconnections are as follows:



#### Figure 4.6 BIMP-EAGA Planned interconnections

Borneo has quite a large imbalance of electricity supply, with varied infrastructure across the island. Depending on the Borneo country (Brunei Darussalam, Indonesia, or Malaysia), this geographical variation and accompanying infrastructure deficit results in roughly 70-90% electrification ratio, with the exception of Brunei Darussalam, who has a 99% electrification ratio (ADB, 2014). Indonesia and the Philippines have the largest to gain, as they have total populations without electricity reaching 23 million and 11 million respectively (IEA, 2017b,c).

Opportunity within the BIMP-EAGA region is high not just due to the current subregional snapshot, but also as a result of national targets. Each nation has a national target for renewable energy integration, and potential for growth in renewable sectors. However, fuel dependency in each nation is currently dominated by fossil fuels, with the majority of countries dependent on coal and oil (IEA, 2017a,c). An overview of this can be seen as follows:

Country	Actual Intercon.	Planned Intercon.	Mix	Dep- endency	Growth Potential	Power Targets
Brunei	No	Malaysia	Natural Gas & Oil (99%)	Oil	Solar PV	10% by 2035
Indonesia	Malaysia	Singapore	Biofuels, Coal, Natural Gas, Oil	Oil	Geo- thermal, solar potential	26% by 2025
Malaysia	Singapore, Thailand, Indonesia	Brunei, Myanmar	Coal, Natural Gas, Oil	Coal, Natural Gas, Oil	Biomass	20% by 2025
Philippines	No	Malaysia	Coal, Geo- thermal, Hydro, Natural Gas, Oil	Coal (expected to become most coal dependent in region by 2030)	Geo- thermal, Solar	40% by 2020

**Table 4.1** Sub-Regional Snapshot: BIMP-EAGA Renewable Potential

Sources: IEA, 2017a,c.

In addition to a heavy reliance on fossil fuel and fossil fuel imports, the BIMP-EAGA subsystem is made up of a variety of market structures. Brunei Darussalam's power system is managed by both a department and a management company, Malaysia by independent companies, Indonesia's five separate power systems are managed by individual state-owned branch offices, and Malaysia and the Philippines have a combination of state-run corporations and national grid offices (*Multi-lateral Organisation—Informant 2; Academic 21*). Some of the power systems are vertically integrated with liberalisation efforts under initial development, while others have transmission systems and distribution networks that are closed to outside power producers. This variety of structure is not uncommon and is reflective of the tendency towards a combination of state run and private sector management. Further details on these structures will be examined in Chapter 5, *National Market Factors*.

The Philippines offers interesting insight into power sector development in the region, as its structure is quite different from the rest of BIMP-EAGA. It is the only country in the subsystem with market competition and has allowed third party involvement in the transmission of power, as well as an independent regulatory body and independent electricity market operator (Pacudan, 2016). This models global and sub-regional recommendations for development of

power systems (*Academic 21*) and offers a unique example to the remainder of BIMP-EAGA. The Philippines system bodes well for the future coordination of neighbouring BIMP-EAGA systems, as it more feasible to coordinate in a highly diverse international power system than to consolidate (*Academic 21*; IEA, 2014; Pacudan, 2016).

To date a single interconnection exists between Malaysia and Indonesia and is the second smallest current international interconnection in the region, with a capacity of roughly 250MW (Pacudan, 2016; author compilation). This interconnection is seen as a model for future BIMP-EAGA development, with plans for expansion of further interconnections leading out to 2030 (Sarawak Energy, 2017). These are planned for: Malaysia (2), Malaysia-Brunei Darussalam (1), Malaysia-Philippines (1), Malaysia-Indonesia (3), resulting in roughly 5000MW of additional capacity (author compilation). Based on the outlined Borneo interconnection plan, harmonisation of the BIMP-EAGA region will be minimal compared to other subsystems in ASEAN (Pacudan, 2016).



Figure 4.7 BIMP-EAGA Actual Interconnections

Progress for the BIMP-EAGA subsystem hinges on a few key similarities with other APG subsystems. These are: first, and foremost, encouragement of national incentives, which are currently lacking. Where ASEAN has made efforts to present and engage member economies in understanding broad incentives for interconnection (increased energy access, the buying and selling of excess

capacity, modernisation of power systems, financial support, knowledge exchange, and higher shares of renewables with likelihood of reaching renewable integration targets) these incentives are not providing enough movement or engagement among national level policy makers (*Think Tank 5*; Academic 26; Former Government 27). Changes to national power markets and infrastructure must occur before interconnections can be made. These national requirements for sub-regional expansion include infrastructure upgrading, more developed regulatory frameworks, and coordination of operations across subsystem grids at the national level—recommendations that Former Government—Informant 1, Think Tank 5, Academic 6, Academic 21, Development Bank 25, Academic 26, Former Government 27, Former Government 28, and *Workshop 3* all recommended; these recommendations are further supported in the literature on ASEAN EMI (See: Kimura & Shi, 2011; Kimura et al., 2013; Pacudan, 2016; and the breadth of literature from ACE and ASEAN, respectively) and are touched upon previously in Section 4.2 when discussing sub-regional interconnections.

Once national level adjustments have occurred economic exchanges can be developed, third party access can be provided, and multi-buyer, multi-seller market systems can naturally grow (*Think Tank 5; Academic 26*). National incentives for these changes and increased cross-border interconnections are currently tied closely to the sub-regional imperatives for these changes in the first place—greater energy access and reductions in energy poverty, associated climate and energy gains via reductions in emissions, and associate potential economic gains via continued integration into regional and global economic markets via liberalisation (*Academic 6*). Interview subject *Business Leader 19*, who works with energy companies and governments throughout the sub-region, further described these incentives as follows:

Generally speaking, regulatory reforms are clearly needed, but economic liberalisation is easily the most incentivised reform among major economies [in the region]...liberalisation of the power sector has moved very slowly in emerging economies, and unfortunately there are not many strong example of power sector liberalisation in East Asia, with only a few exceptions.

Here the question of liberalisation within the BIMP-EAGA subsystem, and the greater APG, resurfaces in importance. Three prominent EMI experts interviewed (*Multilateral Organisation—Informant 2, Academic 21, Academic 26*) all recommended liberalisation within the regional and sub-regional power systems, but all three pointed to a slow-moving process at the national level. In

fact, the majority of literature on EMI in ASEAN (including ASEAN and HAPUA themselves) all advocate for market liberalisation and increased competition within systems dominated by state run utilities and transmission operators, which most of the region's systems are. *Academic 6* further illustrated the need for liberalisation in the sub-region, stating:

Overall liberalisation will dramatically improve the regional and sub-region[al] energy transition—responding to climate challenges also requires economic mechanisms not possible without access to and engagement in the global market.

In addition to climate incentives, ASEAN appears to understand the IPE of crossborder electricity trade as part-and-parcel to general market liberalisation practices and expectations of emerging economies. BIMP-EAGA economies similarly understand the IPE of cross-border electricity trade, however, integration remains minimal at this time due to market factors not meeting capacity and infrastructure requirements and national level policies not pushing these changes quickly enough. Analysis of current interconnections and market factors did not, unfortunately, yield any major surprises in BIMP-EAGA, further supporting expert insight that BIMP-EAGA is the weakest of ASEAN's EMI subsystems. The underlying issue recurring here, national versus sub-regional incentives and requirements, repeatedly emerges in the interviews as a hinderance to EMI and resurfaces again and again across subsequent chapters and analysis.

### 4.4 Discussion

Data analysis and interviews conducted in this research revealed relationships between elite opinions and political and economic realities in the electricity and energy market space. The aim of this chapter was to cast light on the sub-regional market factors related to cross-border electricity interconnections and subregional EMI goals, uncovering understanding of and incentives for cross-border interconnections. What interviews uncovered is that sub-regional market factors are directly influenced by national market factors, and reforms are necessary at the national level in order for sub-regional EMI to take place. Identified national reforms include: national power system updating, national prioritisation of power system updating and integration, standardisation and harmonisation across national power systems, national political will, and liberalisation. The necessary reforms for deeper EMI to occur are reflective of cross-border electricity needs regionally but emphasise the need for national political and economic priorities to align with sub-regional goals. In addition, analysis of sub-regional interconnections exposed that current interconnections are limited and there is room for diversification of national energy mix given the continued predominance of hydrocarbon resources. This poses the consideration of whether or not sub-regional targets align with national priorities—a consideration that recurs in subsequent chapters. In the following subsection national versus sub-regional phenomenon will be discussed in more detail, stressing the multi-level dynamics at play.

### 4.4.a National vs. Sub-Regional

This chapter reiterated that all of the ASEAN member economies have committed to EMI, ASEAN has merged a number of relevant EMI initiatives into its connectivity agenda, and ASEAN has incorporated EMI into its pledges to reduce emissions and further integrate renewable energy into its power systems, which are reflected in national level targets in BIMP-EAGA. However, in examining the sub-regional market factors it was national issues that kept arising as barriers to sub-regional goals. In addition, what became clear in interviews is that sub-regional policy goals do not necessarily reflect the national level changes required. This finding is also reflected across subsequent data chapters and the themes of national market factors and governance challenges. Without a combination of reforms to the subsystem grouping (BIMP-EAGA) and within the sub-region broadly, increased interconnections and cross-border electricity trade alone will not lower regional CO<sub>2</sub> emissions. Instead, increasing interconnections in current markets, with only modest renewable increases but continued reliance on coal electricity generation as reflected in the sub-regional snapshots, will actually raise regional emissions (Academic 26; confirmed via IEA, 2019a, pp. 146-147). However, economies may gain economically via the buying and selling of excess capacity via cross-border trade (IEA, 2019a).

As far as sub-regional factors are concerned, experts interviewed predict that AIMS III will amend previous goals set in AIMS II based on a relatively low level of progress that has been experienced to date across subsystems; this will also include an updated assessment of interconnectors in the region (*Multilateral Organisation—Informant 2; Academic 21, Academic 26*), including those in the BIMP-EAGA snapshot in Section 4.3. In addition, interviews from outside the region in partner country organisations predict a re-assessment of investment programs and financing tools to meet growth in project funding that is surely to be recognised in the AIMS III assessment (*Think Tank* 12). Further sub-regional and regional funding initiatives will be required if sub-regional renewable energy and climate targets are to be achieved given the limits of funding

initiatives and incentives at the national level to make expensive power sector updates (*Multilateral Organisation 4;* confirmed via IEA, 2019a).

This underscores the importance of increasing growth in renewables generally and using cross-border electricity trade to augment variability, reliability and flexibility concerns in order to do so. However, national level measures must catch up to sub-regional renewable integration and global energy transition goals. Unfortunately, whereas sub-regional understanding of cross-border interconnections are understood as climate, economic and energy-based, national incentives appear to be centred around liberalisation given the emphasis put on neoliberal market reforms, market liberalisation, and finance needs. While ASEAN's interconnectivity targets are laudable, they do not coincide with the realities of renewable integration and generation at the national level. The prevalence of coal in the energy mix of BIMP-EAGA countries specifically (as seen in the subsystem snapshot, Section 4.3), and ASEAN generally, underscores this disconnect (IEA, 2019d). This is further supported in interviews, including this statement by interview subject *Multilateral Organisation 18* regarding cooperation:

cooperation is a tool. But not the most important goal of each member state—national priorities are [the] most important goal, regardless of [ASEAN's] connectivity agenda.

While there is sub-regional commitment to a clean energy transition that commitment is not translated at the national level, where ASEAN commitments are non-binding, ASEAN connectivity goals are not yet clear regarding harmonisation and integration for individual grids, and national priorities are deemed greater than sub-regional targets. For example, energy security concerns are a national priority that, according to interviews discussed in Section 4.3, are often deemed more important than fully embracing ASEAN targets (where the measures to achieve those targets are perceived to decrease energy security).

Overall, national level structures and policies must be conducive to the requirements for increased EMI in order for cross-border electricity trade to take place. Many of these requirements coincide with neoliberal political economy narratives about market structure and the common global economic system that were touched upon in Chapter 2 (*Conceptual Approach & Review of Literature*). What the author observed using a critical realist lens during data analysis is that the underlying market needs are not necessarily representative of both sub-regional and national priorities. Sub-regional goals span multiple

countries, and in the case of BIMP-EAGA, impact economic, political and structural requirements for reaching these goals. Internationally, global climate challenges and targeted responses also impact sub-regional and national policies, increasing or decreasing incentives for economic, political and structural change.

Analysis of the market factors in ASEAN, and how they assist in understanding the IPE of cross-border electricity trade in the sub-region, points predominantly to liberalisation and national level limitations in achieving liberalisation. EMI, experts agreed, requires liberalisation of domestic energy markets, increased competition, and standardisation across markets—traditionally neoliberal solutions for already advanced power sectors that are prescribed as an ideal, without being moderated for an East Asian context. Integration into global markets continues to be a major theme across the literature and data collection, and competition and markets the only well understood route for this integration.

There is also a widespread belief that increased trade among partners and trade liberalisation are a key for increased cross-border electricity trade. In fact, Oseni and Pollitt (2014) find it to be one of the most important criteria for EMI to take place, and the presence of trade agreements prior to cross-border electricity trade and power pooling efforts is also considered helpful for continued development in EMI. Interviews further informed the dominant policy recommendation that increased liberalisation is a requirement of EMI and thus increased cross-border electricity trade. In regions where bilateral trade agreements are already present, barriers to trade will have been reduced and trust between partners will already have been established. One trade expert, Business 22, pointed to established trust as a major benefit of mutually beneficial trade relationships, and indeed a bi-product of international political economics. ASEAN itself has focused primarily on studies pertaining to broader trade liberalisation and the removal of fossil fuel subsidies; according to expert interviews there is space for examination of liberalisation of domestic energy markets, particularly power sectors, where liberalisation is limited to date (Academic 6, Business 22, Academic 26).

Commitments to free trade, amid centrally planned economies that utilise state directed economy policy, further reflects this author's characterisation of neodevelopmental statism and its emerging role in understanding the political economy of cross-border electricity trade. In examining the sub-regional market factors the IPE of cross-border electricity trade emerges as a link between developmental state practices and liberalisation identified by interview subjects and documentary sources for EMI to advance; liberalisation even emerges as a primary requirement noted among interview subjects. However, it appears that states are 'committed' to free markets and liberalisation in so far as free markets and liberalisation benefit state goals; where those goals fall in the hierarchy of needs is debatable, and, based on market needs identified so far, not always in line with the market factors. The political economy of energy policy development in the sub-region or subsystem can influence the development of policies that respond to broader resource concerns—such as commitments to free trade and market competition.

Higher shares of renewable energy in sub-regional and national energy markets compliments sub-regional climate commitments and targets. Increasing renewable energy's share in a national energy mix has the benefit of reducing emissions via a reduction in hydrocarbons and diversification of a country's reliance on hydrocarbon fuels. However, sub-regional commitments are being made as a performative free trade agenda, particularly in sub-regional efforts where countries are seen to be engaging with liberalisation and contributing to market integration, but also slow the progress of projects due to the complexity of sector reform. In this way neoliberal free trade IPE is being performed at the same time as neo-developmental statism is being practiced—regional governance is disabling reform due to its performative nature and the requirements that must occur at the national level but are not highly enough incentivised to do so.

## **4.5 Conclusions**

This chapter has painted a broad picture of the market factors present in subregional and subsystem interconnections among ASEAN members. A defining aspect of this chapter is that analysis was carried out at the sub-regional and subsystem level, but national level analysis emerges to be equally important. Here analysis included a look at projected and actual interconnections, and a snap shot of relevant market factors in the BIMP-EAGA subsystem. It has been illustrated that sub-regional and national governance structures must all respond to the pressure of climate change, but no one is solely responsible where EMI and cross-border electricity is concerned. In ASEAN this has been demonstrated to be true, as national efforts are as important as sub-regional and global ones. This has resulted in the emergence of a key factor in ASEAN crossborder electricity trade—national versus sub-regional policy needs. Here the IPE of interconnections is focused on market liberalisation and increased competition in BIMP-EAGA and EMI in the wider ASEAN sub-region.

In the case of cross-border electricity trade and EMI, power dynamics, national level reform, political economy options including trade liberalisation, and the linking of issues (climate, energy, energy poverty, electricity reliability, etc.) all directly correlate with barriers to EMI. This chapter sought to examine sub-regional interconnections and the market factors impacting them, however, interestingly the majority of EMI barriers identified in Section 4.2 and 4.3 reflect national level needs.

This chapter has addressed one primary research questions and its associated sub-questions—(RQ1) How do governments understand the IPE of cross-border interconnections in Southeast Asia? (RQ1a) What are the energy and trade policy incentives for increased cross-border interconnections? And (RQ1b) How can subregional electricity markets diversify their energy mix and reduce emissions and what incentives to do so exist? The answers are governments are incentivised by climate, economic, and energy imperatives at the sub-regional level to utilise cross-border interconnections and augment VRE challenges while diversifying their energy mix—but progress is intermittent and uneven across ASEAN and in BIMP-EAGA at the national level. In addition, goals are ambitious and national level commitments fall short where actual energy mix and fossil fuel versus renewable energy use is concerned. Instead, it appears that sub-regional institutions of governance (ASEAN) and sub-regional commitments are standing in performatively for concrete action. Where this chapter provided a broad overview of the sub-region, the following chapter will look more specifically at individual national markets in the BIMP-EAGA subsystem, explaining power sectors at a more detailed level.

## Chapter 5 National Market Factors

# **5.1 Introduction**

This chapter continues with the pattern began in Chapter 4 of identifying common themes that emerged from expert interviews and data analysis. Chapter 5 will pick up where Chapter 4 left off by examining national market factors and common national EMI reform recommendations. In addition, this chapter will also highlight how electricity is transacted for each nation via the relations between governments and market actors. This chapter aims to achieve Objective 2, assessing market factors of the case study selection, BIMP-EAGA, identifying areas of individual and collective reform needed for deeper EMI levels.

Chapter 5 will address the following research questions: (RQ2) *How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?* In addition, the two sub-questions associate with RQ2 will be addressed: (RQ2a) *How does the political and economic structure of national electricity markets impact the expansion of cross-border interconnections in BIMP-EAGA?* (RQ2b) *Is the structure of national electricity markets conducive to sub-regional and international policy goals in Southeast Asia?* This chapter will add to prior analysis of sub-regional EMI initiatives by comparing and contrasting commonly identified reform recommendations with the realities of these national markets.

To address these questions Chapter 5 will be structured in the following format: Section 5.2 will examine the development of national electricity markets and current market structures, expanding on the BIMP-EAGA details presented in Chapter 4. In Section 5.3 the EMI criteria developed by Oseni and Pollitt (2014) will be applied to BIMP-EAGA specifically. Application of Oseni and Pollitt's EMI criteria will provide a preliminary conclusion regarding APG and market reform progress, stressing further reform recommendations in both the literature and expert interviews. Next, Section 5.4 will discuss the dominant aspects of reform and assessment in national markets, emphasising strengths and weaknesses based on available data. Section 5.5 will analyse the practice of DS in ASEAN energy markets and further probe its contradictions with common market liberalisation reforms. Finally, in section 5.6, *Conclusions*, national market factors examined in this chapter will be summarised, leading into the subsequent analysis in Chapter 6.

## **5.2 Development of National Electricity Markets**

Development of national electricity markets has been identified widely across the literature<sup>52</sup> as a necessary prerequisite of increased multilateral crossborder interconnection, regional power sector development and energy market integration (EMI). This is because national electricity markets require the physical capacity and regulatory frameworks to manage not just domestic energy but imported energy as well (Li et al., 2016, p.50); where multilateral interconnection develops, system capability, efficiency, flexibility, and reliability become even more important than when bilateral connections occurred. National markets have to be physically developed—have the interconnection and the capability to transport—in order for exchange to occur. Development of national markets includes the implementation of regulatory standards, transmission processes, legal requirements, physical requirements, and even local government support (Li et al., 2016). Owen et al. (2017) argue that while physical bilateral or very basic multilateral interconnection can take place in undeveloped markets, any deep integration and meaningful, long-term trade requires alignment of individual markets and their standards, policies, and practices (p. 150). The majority of experts interviewed argued the progress of developing national electricity markets has been slow and is inhibiting EMI development and multilateral cross-border electricity trade (Former Government—Informant 1, Multilateral Organisation 4, Academic 5, Academic 6, Business 19, Business 22, Former Government 27, Former Government 28, Think Tank 29, Workshop 3). According to Multilateral organisation 4 "...of course the speed of state level change can influence cross-border [electricity] trade." The literature also predominantly supports this premise: cross-border electricity trade takes place predominantly among partners with well-established national markets (Wu et al., 2016); where national markets are strengthened then capacity for cross-border trade grows (Wu, 2019). In addition, while there is some ability of stronger partners to augment the weaknesses of less developed partners by relieving the financial and physical burden of renewable energy development (Chang & Li, 2013), and efficiency gains can be felt by both parties in a bilateral agreement (Matsuo & Tsunoda, 2016), any expansion and growth in demand will require further national development. This is what Phoumin and

<sup>&</sup>lt;sup>52</sup> See for example: Andrews-Speed & Hezri (2014); Oseni & Pollitt (2014); Li & Kimura (2016);
Pacudan (2016); Shi et al. (2019).

Kimura (2014) refer to as increased power sector robustness, a positive side effect of increased EMI in ASEAN.

The following subsection will address the development of BIMP-EAGA markets, providing an overview of national market structures in order to assess renewable energy integration and the IPE of sub-regional cross border electricity trade among these national economies. Once the structure of national markets has been established, they will be contrasted with the market requirements established by Oseni and Pollitt (2014) for EMI and increased cross-border electricity trading to occur.

### 5.2.a Current Market Structures: Overview

Integrated, highly developed national electricity markets have the potential to be a catalyst for growth in EMI among economies wishing to engage in crossborder electricity trade. This is because a modern (reliable and flexible) and integrated (physically connected) power system that can transport electricity from place to place is required for EMI and cross-border interconnections to be developed (IRENA, 2016; Oseni & Pollitt, 2014). If the benefits from EMI and cross-border electricity trade are going to be available to all parts of the market (both rural and urban populations) then modernisation and integration are necessary in order to provide low-cost energy access in an intermittent environment that cross borders (IEA, 2015, p. 20). Analysis of individual power markets in ASEAN shows sector development is varied. As a whole ASEAN energy markets are, like the sub-region itself, highly diverse. Power markets in the sub-region are a mix of public, private, and hybrid forms, with varying levels of state regulation and oversight. The majority of power markets are not unbundled, and indeed development of power markets, including transmission and generation infrastructure, are at varying levels depending on the nation (Wu, 2016, pp. 68-69). Overall ASEAN electricity markets are structured around similar factors: state driven markets with state oversight even in liberalised markets; limited competition generally; a predominance of subsidies; a range of development stages; and varying levels of electrification across the region (See Chapter 2 for more detail on these factors; also Wu, 2019). There is no one-sizefits all recommendation for EMI and electricity market development in ASEAN given the variety across markets, however, there are similar solutions to these limitations across the sub-region. The following section probes these characteristics in more detail.

#### 5.2.b BIMP-EAGA

In Chapter 4 BIMP-EAGA was introduced and the existing and projected interconnections between member economies were discussed. These interconnections were analysed in the context of ASEAN's signature interconnection initiative, the APG. In addition, a summary of the renewable energy integration potential of BIMP-EAGA was provided based on energy mix and identified national targets. Now that overview will be broken down into more detail, providing a comprehensive look at the market structure of BIMP-EAGA economies and give context to the needs of each market.

In BIMP-EAGA the majority of national power markets are on the lower end of development, behind countries like Singapore and Thailand, both of which rank high in terms of grid reliability—with the exception of Brunei, who is just behind them (Huang et al., 2019). Overall the economies of BIMP-EAGA are making physical progress in developing their power sectors; however, market structure is a dominant factor hindering advancement (*Academic 21; Academic 26;* see also: Matsuo & Tsunoda, 2016).

#### Brunei Darussalam

Brunei Darussalam's power market is dominated by fossil fuel interests. The abundance of hydrocarbon resources and heavily subsidised energy and electricity sectors has contributed to a reliance on fossil fuels (Energy and Industry Department of Prime Minister's office, 2016). This overreliance on fossil fuels makes Brunei's economy closely tied to fossil industries, with fossil fuels making up 90% of the country's exports and 44% of its GDP (OECD, 2018b), down from 60% in 2013 (OECD, 2013). Brunei Darussalam's power sector is regulated by a government agency, with separate and unconnected networks for generation, transmission, and distribution (Ahmed & Othman, 2014, p. 156) that operate in a single buyer market that is not open to independent power producers (IPP) (Pacudan, 2016, p. 47). The power sector is still early on in the liberalisation process with restricted private sector participation (*Former Government\_Informant 1, Academic 21*) and a reliance on government fossil fuel subsidies (Navarro & Sambodo, 2013).

A 2019 analysis of ASEAN grid flexibility by Huang et al. found Brunei Darussalam ranking relatively well in most areas and tying for second in overall

grid flexibility,<sup>53</sup> posing a positive outlook for future renewable energy integration and cross-border electricity trade. These flexibility measures are important as they predict the ability of a country's grid to respond to changes in demand, supply, and the uncertainty of renewable sources (p. 711). This sort of flexibility is a requirement for EMI and will have positive spill over effects in the country's greater electricity reliability and access as well as lower costs. Among the factors analysed are individual markets' access, reliability, forecasting system, natural gas generation for electricity, and renewable energy diversity (p. 714). For grid reliability Brunei Darussalam is tied for fourth, closely behind one and two, and its natural gas generation for electricity market access score, ranked based on interconnection capacities "when compared with peak load demands", is quite low, as are the majority of markets in ASEAN (p. 717).

<sup>&</sup>lt;sup>53</sup> In power system research grid flexibility refers to operational flexibility, or "the ability of a power system to respond to changes in electricity demand and generation...that must be balanced to maintain system stability and reliability" as a result of uncertainty in generation due to variable renewable energy sources (NREL, 2015, p. 1). In Huang et al. (2019) overall grid flexibility takes NREL's (2015) definition further to apply a grid flexibility assessment tool to ASEAN member states' individual grids in order to rank flexibility across the region. "Overall grid flexibility" is identified using six indicators: grid reliability, electricity market access, load profile ramp capacity, quality of forecasting tools, proportion of electricity generation from natural gas, and renewable energy diversity (Huang et al., 2019, p. 715). In section 5.2.b (*BIMP-EAGA*) of this research, the author references Huang et al.'s overall grid flexibility ranking to describe the power systems of ASEAN member states within the BIMP-EAGA subsystem.





Source: Author's creation using IEA (2017a, pp. 22-27), Yokota & Kutani (2017, pp. 4-8), and interview feedback.

Brunei's electricity industry is vertically integrated and overseen by the Ministry of Energy and the national power company, Berakas Power Company (BPC), with three national oil and gas companies also participating in select generation and sale of excess electricity. There is limited information available on additional deregulation of the power sector, reflected in both the literature (Yokota & Kutani, 2017) and interviews (*Academic 21; Academic 26*). Brunei's power sector is the most heavily regulated of the BIMP-EAGA economies and data the most limited. However, it is also the BIMP-EAGA economy whose electricity industry is most dominated by fossil fuel industries, reflected in the potential for renewable integration (presented in Chapter 4). Brunei produces a surplus of power domestically, roughly 100-106% of annual consumption being covered by domestic production in 2016 (World Meters, 2020).

### Indonesia

Indonesia is one of the national markets with discrepancies between policy and action, in part due to the intersection of business and policy interests in Indonesia's energy sector (Navarro & Sambodo, 2013). Government control over

Indonesia's energy industry, particularly oil and gas, lack of government clarity over national energy regulations, close-relationships between illegal energy ventures and government leaders, and tight restrictions on foreign investment top the list of conflicting interests and policy messaging (Dutu, 2016). Multiple periods of reform have influenced the continued removal of fossil fuel subsidies; however, the use of subsidies still exists and is closely tied to economic and political influence (Chelminksi, 2018) and the prevalence of fossil fuels in the country's energy mix. Indonesia is one half of the existing cross-border electricity interconnection with Malaysia, with plans for expansion and assessments underway (Pacudan, 2016). Indonesia ranks number 7 on the grid flexibility score analysed by Huang et al. (2019) and discussed in Chapter 2, due in large part to its very low electricity market access score (zero), low forecasting score (zero), low reliability score, and low natural gas for electricity generation score. Overall Indonesia's power sector is dominated by the national utility PLN (Perusahaan Listrik Negara); Indonesia has a state owned, vertically integrated electricity market with little private sector participation and limited independent power producer (IPP) involvement at just over 20% of total generation (PWC, 2016, p. 58).



Figure 5.2 Indonesia Electricity Industry Structure

Source: Author's recreation of IEA (2017a, pp. 27-28, 99-101), Yokota & Kutani (2017, pp. 4-6, 13-18), and interview feedback.

The power sector is regulated by a government agency, the Department of Energy and Mineral Resources. Reforms have been approved, however, delays in implementation have prevented further liberalisation of the national market (Wu, 2019), reflected in the general PLN monopoly and limited IPP involvement. Coal subsidies have encouraged an electricity sector dominated by coal, resulting in high artificial discounts for consumers (Global Subsidies Initiative, 2017), and continued subsidy removal could result in steep government fiscal gains (Burke & Kurniawati, 2018). Indonesia produced 110% of its annual electricity consumption via domestic generation in 2016 (World Meters, 2020).

#### Malaysia

Malaysia makes up the second half of the existing cross-border electricity interconnection, with plans for expansion. Similar to the electricity markets of Brunei and Indonesia Malaysia's market is not fully liberalised, vertically integrated, and allows limited IPP involvement (Yokota & Kutani, 2017). Malaysia is the third most flexible electricity market in the BIMP-EAGA subsystem, with a high grid reliability score but low electricity market access score, very low forecasting score (zero), with some natural gas use for electricity generation and some renewable energy integration (Huang et al., 2019). Electricity market regulation is overseen by an independent regulatory body, unlike Brunei and Indonesia. Malaysia also has vertically integrated transmission and distribution, operated by either Tenaga Nasional Berhard (TNB) or Sarawak Energy Berhard (SEB), depending on the region within the country (Samsudin et al., 2016; A21). Malaysia has significant government funded petroleum subsidies in place (Han & Kimura, 2015). Malaysia also produces more than 100% of its annual consumption needs, with 108% annual consumption generated domestically in 2016 (World Meters, 2020).

Energy Commission – Regulator			
Genting	Port Dickson	TNB	Teknologi Tenaga
Sanyen	Power	Janamanjung	Perlis
NUR Power	<u>Husteq</u> Hydro	Projass	Powertek Portfolio
YTL	Malakoff	TNB GBN	TNB Hydro
Portfolio	Portfolio	Portfolio	Portfolio

Figure 5.3 Malaysia Electricity Industry Structure

Grid	Owner	Grid System Operator		
TNB Transmission & Planning	Single Buyer			
TNB transmission asset development, operation and maintenance of the transmission system	PPA contracts, day ahead scheduling and settlement	Operation planning, real-time rescheduling, dispatching, real-time operation		
	Distri	bution		
TNB Dis	stribution	Other distribution companies		
Customers				

Source: Author recreation of TNB (2020, p. 1), Yokota & Kutani (2017, pp. 4-6, 24-28), and interview feedback.

### Philippines

The Philippines power sector is perhaps the most advanced in the subsystem, as it is farthest along in the liberalisation process. The Philippines market is regulated by an independent body just as Malaysia, but the Philippines has also introduced some competition into the market. Pacudan (2016) describes the

Philippines market as the only subsystem market with "a competitive wholesale electricity market and partial retail market competition" (p. 46), which has resulted in a hybrid public-private model and the only of its kind in the BIMP-EAGA subsystem. However, the Philippines market still only remains partially competitive (Hall & Nguyen, 2017). While the Philippines was one of the first Southeast Asian economies to introduce IPPs into the market, the majority of power is still produced by NAPOCOR, the National Power Corporation, who in 2001 produced more than 50% of the nation's electricity (Wu, 2019, p. 13). Wu (2019) argues that despite a strong national privatisation agenda on record the market is not actually fully competitive, as multiple efforts to privatise have failed or been delayed. Hall & Nguyen (2017) describe these delays as a result of government focus being primarily on security of supply, not competition or equitable access (p. 107). In addition to initial liberalisation practices, the Philippine market also has the most advanced forecasting capabilities among BIMP-EAGA economies (Huang et al., 2019). Domestically the Philippines generated 111% of its total electricity consumption needs in 2018 (DOE, 2019). Among interview subjects the Philippines has been held up as the most advanced of the BIMP-EAGA economies. One subject, Multilateral Organisation—Informant 2, also spoke to the Philippines role in sub-regional power sector development:

[the Philippines] has made efforts behind the scene[s] to support regional power sector trading among ASEAN members, but also set an example for other nations—competition was introduced in the Philippines [power] market, and it is also made policy objective to include trade considerations in its climate and energy policy; this includes tax considerations and additional subsidies for clean energy integration.

Here the electricity industry structure of the Philippines is further illustrated:



### Figure 5.4 Philippines Electricity Industry Structure

Source: Author recreation of Navarro et al. (2016, p. 11), Yokota & Kutani (2017, pp. 31-34), and interview feedback.

This subsection has identified the market structures of ASEAN electricity markets, analysing the electricity markets of ASEAN broadly and BIMP-EAGA specifically. This has included identifying the broad limitations of current market structures based on research into grid flexibility and renewable energy integration, as well as identifying the level of domestic production achieved in each country, point to self-sufficiency of current electricity sectors across the subsystem. In the following subsection subsystem limitations will be contrasted with the dominant research into EMI and combined with the author's own insight based on interviews and additional documentary research, further shedding light on national market factors and the potential impacts on subregional renewable energy growth and development of cross-border interconnections.

# 5.3 Oseni & Pollitt Criteria

The structure of electricity markets in ASEAN explains, in part, the difficulty of developing multilateral cross-border electricity trade, increasing market integration across borders, and establishing a region-wide power grid (APG). It

has been established that cross-border electricity trade is happening in Southeast Asia among ASEAN members on a primarily bilateral scale. However, the goal of increasing cross-border electricity trade beyond bilateral trades requires increased EMI, the process of liberalisation and harmonisation of national electricity markets in order to facilitate electricity cooperation (Grossi et al., 2018; Oseni & Pollitt, 2014). As a result of the harmonisation of structures and processes, and the removal of barriers to trade, countries are able to engage in cross-border electricity trade and, as a result, cooperate on development and modernisation of national and regional electricity markets (Oseni & Pollitt, 2014).

Keys to cross-border electricity trade and power pooling, according to multicountry research by Oseni and Pollitt (2014) for the World Bank, include: (1) the development of national electricity transmission systems; (2) efficient market design (which can include the separation of system operators and transmission ownership, as well as the specific creation of independent institutions responsible for operation); (3) transmission capacity (via physical interconnection of cross-border transmission lines or interconnectors and congestion management); (4) governance support for the building and utilisation of transmission capacity across borders. In addition, (5) "jurisdictional issues" must be thoroughly considered (including energy industry subsidies, transit states); and (6) distributional impacts need to be managed (price impacts or the dispersal of energy intensive industries between two countries based on supply reliability and price increases, for example), and most importantly, (7) "a broader pre-commitment to free trade" (pg. 23).

Oseni and Pollitt's requirements are of relevance to this research due to their alignment with common recommendations for power sector reform and the essentials necessary for increased cross-border electricity trade. Referring back to the interview theme this chapter represents, national market factors, the Oseni and Pollitt EMI criteria reflect factors needed in national markets in order for cross-border electricity trade to expand and greater EMI to take place in the sub-region. The support for these recommendations by interview subjects further solidifies the relevance of Oseni and Pollitt's work and application to EMI and cross-border electricity trade in a variety of regions. The recommendations by Oseni and Pollitt (2016, pp. 23-27) are summed up below with author commentary and specifications included in the column titled 'Author Commentary', including interview subjects that supported these specifications.

# Table 5.1 O&P EMI Criteria

Requirements	Author Commentary
Commitment to free trade	In the form of trade agreements (bilateral or multilateral), removal of barriers to trade, and WTO membership (Former Government—Informant 1; Multilateral Organisation—Informant 2; Multilateral Organisation 4; Academic 6; Think Tank 11; Former Government 16; Former Government 17; Academic 21; Business 22; Development Bank 25; Former Government 27; Former Government 28; Development Bank Conference 4; Policy Conference 7; Policy Conference 9; Academic Conference 11; Government Conference 12);
Efficient market design	Could include separation of system operators and transmission ownership or creation of independent institutions responsible for operation; here the role of "strong, efficient and independent institutions" (p. 25) cannot be overemphasised. See chapter 4 for more discussion on this ( <i>Multilateral Organisation—</i> <i>Informant 2; Think Tank 7; Development Bank 8;</i> <i>Think Tank 12; Think Tank 15; Academic 21; Policy</i> <i>Conference 9; Government Conference 12</i> );
Governance support	For building and utilisation of capacity across borders; for flexibility in regulation in order to respond to global problems (Navarro, 2013, p. 9; Former Government—Informant 1; Development Bank 25; Academic 26; Former Government 27; Former Government 28; Development Bank Conference 4; Policy Conference 9; Academic Conference 11; Government Conference 12; Government Workshop 14);
Jurisdictional considerations / management	Including energy industry subsidies, transit states, etc. (Former Government—Informant 1; Think Tank 12; Academic 26; Former Government 27; Think Tank 29; Policy Conference 9; Government Workshop 14);
Transmission capacity	Via physical cross-border interconnections, transmission lines, congestion management (Multilateral Organisation—Informant 2; Think Tank 5; Think Tank 12; Academic 26; Policy Conference 9; Academic Conference 11; Government Workshop 14);

(Continued from previous)

Requirements	Author Commentary
National electricity transmission systems	Development of or strengthening; harmonisation of rules and standards (Multilateral Organisation— Informant 2; Think Tank 5; Academic 10; Think Tank 12; Multilateral Organisation 18; Former Government—Informant 1; Development Bank 25; Academic 26; Policy Conference 9; Academic Conference 11; Government Conference 13);
Management of distributional effects	Including the dispersal of energy intensive industries between two countries based on reliability of supply and price increases ( <i>Think Tank 5; Academic 6;</i> <i>Development Bank 5; Think Tank 12; Academic 21;</i> <i>Think Tank 22; Development Bank 25; Academic 26;</i> <i>Policy Conference 9; Government Conference 12;</i> <i>Government Workshop 14</i> ).

These recommendations are consistently reiterated across the literature on cross-border electricity trade generally and in East and Southeast Asia specifically (Antweiller et al., 2001; Pritchard, 2003; Wu, 2012; IEA, 2015; Andrews-Speed, 2016; Li & Kimura, 2016; Pollitt, et al., 2017; Grossi et al., 2018). Experts interviewed largely second these recommendations, however, *Academic* 26 reiterated them exactly. This is significant in that this expert is not only a leader in energy policy research in East Asia, but active within two sub-regional institutions responsible for energy policy making. They have written on this topic extensively and referenced the Oseni and Pollitt recommendations specifically. An additional expert, *Former Government Official—Informant 1*, with expertise in East Asian energy policy but no prior work on Southeast Asian EMI (though on South Asian EMI) broadly reiterated the major Oseni and Pollitt criteria, particularly pts 1-5. Similar agreements from other interview subjects abound (Multilateral Organisation—Informant 2, Think Tank 5, Multilateral Organisation 18, Business 19, Academic 21, Development Bank 25, Former *Government Official 27,* and *Think Tank 28*), representing either partial or all of Oseni and Pollitt's criteria. This author also noted that development bank representatives and former government officials interviewed largely reflected all governance-oriented criteria and free trade recommendations—a point that will be further analysed in Section 5.6, Discussion.

Oseni and Pollitt's criteria, while representing specific recommendations with wide-spread support, are only applied in their 2014 research to six specific case

studies in various stages of development—the Sothern African Power Pool, the West African Power Pool, and the Central American Power Pool, the United States, the Single Electricity Market in Ireland, and the South East Europe Market. Application of these criteria to Southeast Asia or the APG does not occur in their research and is original to this Ph.D. dissertation.

In an effort to apply Oseni and Pollitt's criteria to cross-border electricity trade in Southeast Asia the author has created a 'scorecard' of BIMP-EAGA countries applied to O&P's original criteria. This scorecard reflects three rankings—(1) yes, where the country has implemented necessary criteria; (2) partial, where criteria implementation is underway, with political support and/or active initiatives towards criteria; and (3) no, where implementation is limited or nonexistent. These rankings are also visualised in the heat map below (Figure 5.2), whereby green, yellow and red are applied to yes, partial and no answers respectively, drawing attention to areas requiring additional political and economic attention for advancement; the brighter the heat (red) the more help is needed; the cooler the heat (green), the less help. Answers were determined based on the author's review of Oseni and Pollitt's research, documentary research, and interviews. This heat ranking gives a preliminary view of the extent to which cross-border interconnections and related sub-regional energy policy goals are being achieved (RQ2, RQ2a, RQ2b).

Requirements	Brunei	Indonesia	Malaysia	Philippines
Commitment to free trade <sup>54</sup>	yes	yes	yes	yes
Efficient market design	partial	partial	partial	partial
Governance support	partial	partial	partial	partial
Jurisdictional considerations / management	partial	partial	partial	partial
Transmission capacity	no	partial	partial	partial
National electricity transmission systems <sup>55</sup>	partial	partial	no	no
Management of distributional effects <sup>56</sup>	no	no	no	no

### Table 5.2: BIMP-EAGA Scorecard

The scorecard of each BIMP-EAGA member economy tells us a few important things related to market structure and limitations in BIMP-EAGA. First, Brunei is

<sup>&</sup>lt;sup>54</sup> Each BIMP-EAGA economy is fully committed to free trade given their own trade agreements, membership in the WTO, (majority) attempts at liberalisation, and embracing of neoliberal trade policies in the global market (Huang et al., 2019).

<sup>&</sup>lt;sup>55</sup> According to Oseni & Pollitt (2014) nearly all countries have some issue in this area. The development of national transmission systems, transmission capacity, and jurisidictional considerations / management are linked as well due to practical and political problems across markets (physical barriers, governance limitations across jurisdictions, governance capacity, power sector modernisation, etc.). Malaysia and the Philippines are farther away in this regard than their subsystem neighbours, in part because of the presence of multiple unconnected grids (whereas Indonesia has separate island systems that connect multiple power systems within a geographical area [Pacudan, 2016]).

<sup>&</sup>lt;sup>56</sup> All countries scored NO with regards to commitment to the management of distributional effects because of limited associated economic and energy policies, prevalence of energy and fuel poverty, limited access, limits to transport affordability, and lack of climate change policies and instruments at the national level (all of which contribute to management of distributional effects [REN21, 2019a]). Preventing or managing distributional effect in the global transition to a low-carbon economy have proved difficult (McInnes, 2017); Southeast Asia is no different, where affordability of electricity and energy access remain a key challenge across each country in the region but is amplified among the least developed economies in the sub-region.(ACE, 2017).

the least transparent of the BIMP-EAGA countries, and in part its scorecard suffers as a result. While Brunei does have a commitment to free trade, its efficient market design is scored partial as a result of its single buyer model that is dominated by fossil fuel interests and limited private sector involvement (Lawrey & Pillarisetti, 2017); Governance support in Brunei scores a partial as well, due to close government ties for the status quo energy system and lack of transparency (Pacudan, 2016); Jurisdiction considerations are and management similarly partial due to limitations in information (Pacudan, 2016); transmission capacity in the country is low, with no cross-border arrangements or interconnections (Kimura & Shi, 2019); and the national electricity transmission system itself is a work in progress (Energy and Industry Department of Prime Minister's Office, 2016); Similar to other countries on the list Brunei has little to no management of distributional effects beyond surface level climate commitments and requires the development of further sustainable development targets (Kimura & Shi, 2019).

Indonesia scores much better. Indonesia scores partial on efficient market design due to its efforts at liberalisation, reform, and increased competition (which has never fully taken off); Similarly Indonesia has partial commitment to efficient market design (although multiple periods of reform have failed to fully implement); governance support is evident via the nations reform policies; jurisdictional considerations/management, transmission capacity, and national electricity transmission systems are all evident, in part because geographical requirements require a coordinated national market that consists of a number of interconnected systems. Indonesia has what appears to be one of the best scorecards, as it has at least partial in all but one requirement due to government commitment to reforms, but lack of follow through prevents full implementation of many of the requirements. Interestingly, Indonesia also has the highest rate of subsidisation across the BIMP-EAGA subsystem, which limits its commitment to the necessary liberalisation and reform targets. Indonesia does not score higher than partial in many categories as a result of limitations to full implementation.

Malaysia is also partial in the majority of categories, including efficient market design, governance support, jurisidictional considerations and management, and transmission capacity. This partial score is a result of failure to fully implement policies across the market, with some liberalisation but limited IPP involvement (even though a few IPPs are licensed to operate across the country), and the prevalence of monopolies. Malaysia has a very limited national electricity transmission systems and requires significant development of a nationally connected system to progress subsystem interconnection.

The Philippines represents the most liberalised of the BIMP-EAGA economies, with a partial commitment across four requirements: efficient market design, governance support, jurisdictional consideration/management, and transmission capacity (Kimura & Shi, 2019). This partial score is a result of the national liberalisation and reform efforts, including competition and third party access—however, these scores are not green/yes due to the continued monopolies and delayed liberalisation and partial privatisation (Pacudan, 2016).

The O&P BIMP-EAGA score card shows that the majority of criteria are being considered in the majority of countries. However, two areas stand out across all countries—commitment to free trade has a positive score in each BIMP-EAGA economy, and management of distributional effects has a negative score in each BIMP-EAGA economy. This coincides with expert consideration of these criteria (Think Tank 5; Academic 6; Development Bank 5; Think Tank 12; Academic 21; Think Tank 22; Development Bank 25; Academic 26; Academic Conference 1; Academic Workshop 6) and ASEAN review of BIMP-EAGA progress (APAEC, 2010). As evidenced by the table above, the red areas are clustered around physical assets and their impacts which leads this author to posit that physical asset management is one area of weakness within BIMP-EAGA; however, physical costs require large up-front costs which are reflected in a major barrier to cross-border interconnection—investment required as a result of national infrastructure and system improvements that must take place in order for trade to occur (Pacudan, 2016; Owen et al., 2017). According to expert Multilateral Organisation—Informant 2, EMI commercial and regulatory issues are largely centred around investment challenges in Southeast Asia, raising sub-regional questions regarding the establishment of long-term investment guidelines and support from regional and sub-regional financial institutions. This expert posited that the largest immediate barrier to EMI is investment.

Oseni and Pollitt's criteria, while extensive, does not entirely reflect the breadth of recommendations proposed in ASEAN specific literature or interviews. Where interviews and research exposed additional criteria required for EMI this author has added these points into an expansion of Oseni and Pollitt's original criteria. These are represented in italics in the criteria framework now called '*O&PCRB EMI Criteria'*. In addition, the author has added a type distinction to each

commitment, discerning what type of commitment needed—political, physical,<sup>57</sup> relational, or hybrid. This is done in an effort to make additional scoring of these criteria more thorough, as well as allow for expansion of criteria to meet limitations. These type distinctions are indicated by blue shaded cells in the table below, as are any new requirements:

Requirements	Туре	Author Commentary
Commitment to free trade	Policy	In the form of trade agreements (bilateral or multilateral), removal of barriers to trade, and WTO membership;
Market liberalisation efforts	Policy	Including but not limited to free trade commitments; Business buy-in;
Efficient market design	Policy	Could include separation of system operators and transmission ownership or creation of independent institutions responsible for operation; here the role of "strong, efficient and independent institutions" (p. 25) cannot be overemphasised. See chapter 4 for more discussion on this;
Development of multi-buyer and multi-seller market	Policy	According to Pacudan (2016, p. 52) this would occur at a later stage, after a significant number of agreements have been reached (p. 56). This can include the development of an independent, cross- market operator;
Governance support	Hybrid	For building and utilisation of capacity across borders; for flexibility in regulation in order to respond to global problems (Navarro & Sambodo, 2013, p. 9);
Jurisdictional considerations / management	Hybrid	Including energy industry subsidies, transit states, etc.;

## Table 5.3: O&PCRB EMI Criteria

<sup>&</sup>lt;sup>57</sup> The author has noted that physical barriers often require large up-front investments, a view that is reflected among some interviews. However, the author views investment and finance issues as cross-type and therefore they are not distinguished independently in the O&PCRB EMI Criteria.

Requirements	Туре	Author Commentary
Development of transition plans by national power utilities (ASEAN Specific)	Hybrid	Including a road map for development (Navarro & Sambodo, 2013, p. 18) that includes very specific steps and targets more specific than those currently outlined and incorporating business case for development;
Enhance Capacity for Energy Policy and Planning (ASEAN Specific)	Hybrid	Financial and organisational support for ASEAN and country specific goals, including information sharing, capacity building, etc (Andrews-Speed, 2016, p. 4);
Commitment to incremental development	Hybrid	Participation in bilateral agreements, followed by multilateral and eventually sub-regional (Pacudan, 2016, p. 54-56);
Transmission capacity	Physical	Via physical cross-border interconnections, transmission lines, congestion management;
National electricity transmission systems	Physical	Development of or strengthening; harmonisation of rules and standards;
Management of distributional effects	Policy	Including the dispersal of energy intensive industries between two countries based on reliability of supply and price increases;
Presence of trusting or prior relationship	Relational	Including strong bilateral or multilateral relations in the form of agreements, joint initiatives, etc. (Shi et al, 2019, p. 54);

In some cases, the additions to Oseni and Pollitt's criteria are expansions of their ideas but applied directly to ASEAN EMI, providing more detail specific to ASEAN cases. In all cases these recommendations fall under one of four dominant categories (Physical, Policy, relational or hybrid). While it can be argued that each criteria requires some form of governance support, these categories are meant to represent the dominant requirement. There is a roughly even distribution of the four categories, with the exception of relational, of which only trusting relationships is identified (this will be expanded in the following chapter, Chapter 6).

Given the disparate industry structures within BIMP-EAGA (identified in subsection 5.1) what does the application of O&PCRB demonstrate about RQ2a,
how does the political and economic structure of national electricity markets impact the expansion of cross-border interconnections in BIMP-EAGA? And RQ2b, Is the structure of national markets conducive to sub-regional and international policy goals in Southeast Asia? When these expanded criteria are applied to the BIMP-EAGA subsystem, the heat map appears different, and a few outliers begin to emerge. This is depicted in the following expansion of O&PCRB BIMP-EAGA Scorecard, as follows:

Requirements	Туре	Brunei	Indonesia	Malaysia	Philippines
Commitment to free trade	Policy	yes	yes	yes	yes
Market liberalisation efforts	Policy	partial	partial	partial	partial
Efficient market design	Policy	partial	partial	partial	partial
Development of multi-buyer and multi-seller market (Pacudan, 2016) <sup>58</sup>	Policy	n/a	n/a	n/a	partial
Governance support	Hybrid	partial	partial	partial	partial
Jurisdictional considerations / management	Hybrid	partial	partial	partial	partial
Development of transition plans by national power utilities (ASEAN Specific) <sup>59</sup>	Hybrid	unknown	unknown	unknown	partial

#### Table 5.4 O&PCRB BIMP-EAGA Scorecard

<sup>&</sup>lt;sup>58</sup> While development of a multi-buyer, multi-seller market has not occurred anywhere except the Philippines, progression to this market model across the other BIMP-EAGA markets is an expected next step in the integration process across BIMP-EAGA (Pacudan, 2016, pp. 52-53), although not applicable given the current market structures in Brunei, Indonesia and Malaysia.

<sup>&</sup>lt;sup>59</sup> Only the Philippines has record of a transition plan by its utilities; the author has found no record of additional transition plans across the subsystem.

Requirements	Туре	Brunei	Indonesia	Malaysia	Philippines
Enhance Capacity for Energy Policy & Planning (ASEAN Specific)(Andrews- Speed, 2016) <sup>60</sup>	Hybrid	partial	partial	partial	partial
Commitment to incremental development <sup>61</sup>	Hybrid	no	partial	partial	yes
Transmission capacity	Physical	no	partial	partial	partial
National electricity transmission systems	Physical	partial	partial	no	no
Management of distributional effects	Policy	no	no	no	no
Presence of trusting or prior relationship <sup>62</sup>	Relational	partial	partial	partial	partial

<sup>&</sup>lt;sup>60</sup> According to Andrews-Speed (2016) enhancing capacity via cooperation and planning is necessary for sub-regional and regional interconnectivity, including "coherent and effective national and collaborative energy policy and planning" with "the support of a region-wide cadre of energy professionals in government, research institutes, think-tanks and universities" (p. 4). This recommendation is also supported via interviews (*Former Government— Informant 1; Multilateral Organisation—Informant 2*), where recommendations regarding information sharing and continued study centered on expert coordination in both policy and academic spheres.

<sup>&</sup>lt;sup>61</sup> Commitment to incremental development is present across the sub-region with the exception of Brunei, where national policies and transmission development do not reflect an immediate commitment to bilateral development; there is hope via the upgrading of Brunei's transmission capacity, however, this is in progress and currently limited, requiring further harmonisation of Brunei's power systems and national policy commitment to structural changes. The Philippines presents the most committed to incremental development, offering leadership for sub-regional development projects and the most developed subsystem power sector (Pacudan, 2016).

<sup>&</sup>lt;sup>62</sup> Membership in ASEAN and subsystem cooperation via ASEAN initiatives (see Chapter 2 discussion on ASEAN membership and sub-regional cooperation) presents evidence of existing trusting relationships among ASEAN members; however, challenges of trusting energy resources

While the majority of BIMP-EAGA economies are in the partial category for progress in implementing criteria, the Philippines is farthest along (albeit not by much). Brunei is in the red more than any other economy, indicating the least progress in implementing the O&PCRB EMI Criteria. Based on the score card criteria that fall under the policy type are slightly more dominant, but for the most part there is an even spread across type. As reflected in the table above, the weakest areas remain with physical assets and their impacts.

Overall the BIMP-EAGA subsystem is making progress in reaching criteria, but is nowhere near complete, and as a result EMI progress is limited. The only criteria that all four economies have firmly established is a commitment to free trade one of the priorities of ASEAN's own economic goals (i.e. the AEC), and thus achieved with ASEAN support and via ASEAN economic initiatives. While the level of commitment to free trade varies, focus here is on broad criteria performance. In this case neo-developmental statism offers a lens from which to view this commitment—some commitment is all that matters, and therefore EMI criteria is being met via participation in global and region markets, engagement in free trade via WTO membership, trade agreements, and ASEAN free trade initiatives, all while prioritising state directed economic policy making domestically. Therefore, while EMI criteria is technically being met, free and open markets and commitment to neoliberal economic ideals may not be fully reflected in the national policy making apparatus; as a result, full EMI might be difficult to achieve based on the primacy of free trade and market liberalisation efforts in the EMI criteria above.

Of consistent importance among interview subjects and research are four particular points: commitment to free trade / market liberalisation; harmonisation of rules and procedures; transmission capacity, and policy or governance support. Additional points are often recommended as well, however, these four are consistent across both Oseni and Pollitt's work, the broader literature, and expert interviews conducted for this research. These four criteria evenly fall into the categories of physical or policy requirements.

In analysing the criteria established by Oseni and Pollitt and expanded into O&PCRB EMI Criteria, the author did discover that an actual score card with a number range would be beneficial for additional analysis. With a minimally

with one another prevents a higher score across BIMP-EAGA economies. Trust is discussed in more detail in Chapter 6.

sufficient quantity of requirements under each type, each country and the subregion's performance across the four core proposed requirement types could be more accurately measured and thus enable easier identification of weak areas. With a heat map, gradients of success or failure can be more easily identified with a more impactful narrative. Due to data limitations the author chose not to expand this analysis, an issue discussed in more detail later in Chapter 7, *Conclusions*.

The criteria established by Oseni and Pollitt and expanded in the O&PCRB EMI Criteria also coincide with reform recommendations first introduced in Chapter 4. These are expanded upon in the following section, *Reform & Assessment*, with a focus on application to BIMP-EAGA specifically and the reform needs of national markets based on the O&PCRB EMI Criteria covered previously.

## 5.4 Reform & Assessment

This chapter seeks to address Research Question 2 and two associated subquestions: (RQ2) How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets? (RQ2a) How does the political and economic structure of national electricity markets impact the expansion of cross-border interconnections in BIMP-EAGA? (RQ2b) Is the structure of national electricity markets conducive to sub-regional and international policy goals in Southeast Asia? Based on sub-regional market factors discussed in Chapter 4 it has been demonstrated that there is potential for renewable expansion based on a) energy mix and b) renewable targets. However, Chapter 4 also exposed national level changes are necessary for subregional level EMI. Based on the O&PCRB EMI Criteria discussed in Section 5.3, there are EMI limitations based on current national market factors; of relevance to renewable integration is efficient market design and governance support. In assessing the sub-region's opportunities for renewable energy to participate in national markets it should also be considered that EMI itself provides an incentive for deeper renewable integration-excess production has the potential to be bought and sold among EMI partners, contributing to economic gains, energy poverty reductions, and sub-regional and national climate targets. Since sub-regional and national market factors point to limits in national government commitment, this leads to a consideration of the necessary reforms and assessment of national needs for greater EMI and renewable energy integration across the BIMP-EAGA subsystem. Where the market structures are not currently conducive to BIMP-EAGA interconnection expansion (RQ2a) and sub-regional and international policy goals (RQ2b), exploring potential reforms and assessments of these issues will help highlight how these market structures can be improved in order to realise ASEAN power sector connectivity goals and renewable integration targets (if at all).

Reform and assessment of current market structures is a priority among the research already assessing ASEAN EMI and the progress of the APG (Navarro & Sambodo, 2013; Andrews-speed, 2016; Pacudan, 2016; among many others). Interviews also highlighted the need for structural and policy changes in national electricity markets, pointing to the need for continued reform and also assessments of these reforms as EMI becomes deeper and cross-border electricity trade grows (Multilateral Organisation 4, Academic 10, and Academic 26). Multilateral Organisation 4, an expert interviewed in the Philippines during an international policy conference, works on energy market integration in South and Southeast Asia, assessing programs in one sub-region and their applicability in another. This expert spoke passionately about the requirement of continued appraisal of progress in order to deepen physical integration but also align political goals as the environment for EMI and renewable energy integration becomes more welcoming. This is an important overarching point regarding EMI and cross-border electricity trade—reform is not static, but continuous, just as energy policy adapts to technical, physical, and political changes, so, too, should ASEAN's EMI efforts.

Reforms are quite often associated with costs—political, business, and financial. However, there is agreement that the costs of electricity market reform are met by gains in productivity, efficiency, trade and "far outweigh the costs" (Porter & Situmeang, 2005, p.2). ASEAN itself has acknowledged these gains, and as such EMI and the required reforms needed remain a signature of connectivity and economic development plans introduced in Chapter 4 (See APAEC, 1999, 2004, and 2010, for example). Expert interview *Think Tank 25* acknowledges that costs associated with sector reforms are generally accepted as necessary by governments, but often resisted among businesses interests and suppliers. *Multilateral Organisation—Informant 2* echoes this point in ASEAN specifically:

Energy policy reforms are resisted at some points, such as reluctance from the utilities or based on the nature of [the] national system... The awareness of stakeholders and [their] participation level can also be a point of resistance for us [organisation], especially in promoting renewable energy and the use of more efficient technologies in national power systems. include those highlight

- 172 -

Common reforms proposed include those highlighted in the prior section within both O&P's criteria as well as the author's expansion of this criteria in the O&PCRB EMI Criteria framework. These can be grouped under economic reforms (market design, liberalisation, management of distribution impacts, etc.), physical reforms (transmission expansion across borders, adjustment to rules and standards across trade partners for harmonisation) and political reforms (capacity building among leadership, policy commitments, development and implementation of policy plans for transitions among utilities); Similarly, barriers to these reforms are also economic (sunk costs in current inefficient or ineffective systems; high upfront costs, and also lack of investor confidence), physical barriers (capacity deficits, infrastructure limits, and transmission capability), and political barriers (capacity deficits, entrenched interests in the status quo, power dynamics). In some cases, individual nations are ripe and ready for increased EMI and cross-border electricity trade (Singapore), some economies are already making progress (the Philippines and Thailand), and others are lagging behind (Myanmar and Cambodia). In the case of BIMP-EAGA all of the O&PCRB EMI Criteria need to be further addressed, with the exception of commitments to free trade, which are largely underway or already in place but require continued political support. Free trade commitments are also tied up with market liberalisation reforms and reflected in some of the language used surrounding this issue (see ADB, 2014 for example).

Regardless of the type of reform needed in BIMP-EAGA or ASEAN each O&PCRB EMI Criteria will benefit national and sub-regional electricity markets. In particular, reform to meet these criteria will expand national electricity access, increase modernisation (operational efficiency, reliability, and flexibility of power systems) and technical updates to existing infrastructure (generation and transmission capacity, development of national grid infrastructure including domestic interconnected grid networks), strengthen national commitments to diversification of energy resources and electricity production from non-fossil fuel resources, encourage technology sharing and increase investment from regional, sub-regional and international partners (Shi, et al., 2019, p. 452). Interview subjects questioned about limitations of EMI and cross-border electricity trade in ASEAN unanimously cited financial concerns and upfront costs, perhaps the largest perceived cross-sector limitation to sub-region EMI reforms. Interview subject Academic 26, among the top 5 experts immersed in this topic globally, had this to say about financial concerns: "Investment problems, those never leave. They are always there. Instead, making progress and finding solutions should be [the] focus."

Additional relevant EMI reforms can be grouped into three key areas: (1) market liberalisation, (2) coordination, harmonisation and standardisation of markets and infrastructure, and (3) removal of subsidies. These three reforms issues each have political-economy linkages and have controversial reception among experts interviewed. In the following subsections these three reform issues will be explored in more detail, highlighting their relevance to ASEAN and BIMP-EAGA, as well as incorporating previous analysis of O&PCRB EMI Criteria for increased EMI. Analysing the legitimacy of these reforms in individual markets, and their likelihood in being implemented fully, is beyond the scope of this study; however, a broad overview will provide context to the needs of ASEAN for greater cross-border electricity trade and increased EMI. This will further aid in addressing RQ2a and RQ2b—how does the political and economic structure of national electricity markets impact the expansion of cross-border interconnections in BIMP-EAGA and is the structure of national electricity markets conducive to sub-regional and international policy goals in Southeast Asia?

#### 5.4.a Market Liberalisation

In Chapter 2 (*Cross-Border Electricity Trade: Conceptual Approach & Review of Literature*) international trade was introduced, explaining the common, perceived benefits gained from liberalisation and increased trade among neighbours. Under this line of thinking, increased competition results in market efficiency, which benefits the economy, lowering costs and thus prices for consumers (Jamasb & Pollitt, 2005). Electricity markets are no different, as the wider global experience and literature (Han & Kimura, 2015; IEA, 2015) reflects recommendations for liberalisation of electricity markets in order to foster what Jamasb and Pollitt (2005) refer to as a "well-functioning, market-oriented industry" (p. 2). Similar to recommendations for reform, in order to achieve EMI the necessary liberalisation reforms for an open market are interrelated and interconnected, including one or more of the following:

...sector restructuring, introduction of competition in wholesale generation and retail supply, incentive regulation of transmission and distribution networks, establishing an independent regulator, and privatisation (Jamasb, 2002; Joskow, 1998; Newbery, 2002; cited in Jamasb & Pollitt, 2005, p. 2).

These reforms, of course, reflect a neoliberal way of thinking about trade, where policies focus on economic efficiency with few market barriers and the "...greatest creation of wealth at the lowest possible cost" (Strange, 1988, p. 179).

In the case of electricity markets this means well-functioning regulation and oversight, strong infrastructure and transmission networks, and market competition in order to provide the lowest costs to both power sector businesses and consumers of electricity. This also reflects the implementation of neoliberalism in theory versus in practice. Chapter 2 discussed ways in which liberalisation in electricity markets is not always reflected as it is theorised—losers emerge and, in some cases, prices actually increase and reform is halted in favour of more state intervention and state support of private investment (Hall & Nguyen, 2017, p. 114).

Regardless of critical power sector liberalisation research (Anderson, 2009; Hall & Nguyen, 2017; Aris et al., 2020) the assertion by Oseni and Pollitt (2014) that national commitments to trade liberalisation are required for cross-border electricity trading is also consistent with the literature on power sector reform in ASEAN (Porter & Situmeang, 2005; Wu, 2012; Sheng & Shi, 2013; Shi, 2014; IEA, 2015; Pacudan, 2016; among many more) and expert interviews and feedback (*Former Government—Informant 1; Multilateral Organisation—Informant 2; Multilateral Organisation 4; Academic 6; Academic 10; Think Tank 11; Former Government 16; Former Government 17; Academic 21; Business 22; Development Bank 25; Academic 26; Former Government 27; Former Government 28; Academic Workshop 6; Policy Conference 9; Government Workshop 14) and among international institutions (World Bank, 2008; IEA, 2014, 2015; IRENA, 2018a). According to Oseni and Pollitt:* 

greater trade openness leads to more cross-border trade in electricity, even keeping the potential gains from trade constant (p. 23).

The presence of regional trade agreements prior to cross-border electricity trade and power pooling efforts is also viewed as helpful for continued development in EMI. In regions where bilateral trade agreements are already present, barriers to trade will have been reduced and trust between partners will already have been established. Experts interviewed on this topic unanimously called for increased market liberalisation, further regional trade integration, and alignment of national markets. Interview subject *Academic 6* even went so far as to assert that free trade and engagement in trade agreements is good not just for EMI but for renewable integration specifically:

trade liberalisation will help scale up clean technology use in the region, and renewable utilisation...through demand-side effects.

Following the logic of neoliberal reform recommendations, ASEAN's present commitments to free trade, via the AEC and other initiatives, is one step in

further EMI and cross-border electricity trade, boding well for the development of further market integration and continued trust among parties (trust will be further explored in Chapter 6).

In the case of electricity market liberalisation, reform suggestions are not only referring to removing barriers to trade, although that is the dominant refrain (see Academic 6, above). There is also the matter of increasing competition in national electricity markets by opening markets up to third parties or IPPs (Hall & Nguyen, 2017). The majority of BIMP-EAGA nations have bundled transmission and distribution, as opposed to TSOs and independent distribution (see Section 5.2 for more on individual markets). This can also include the separation of distribution and transmission costs, resulting in increased investor and customer confidence, encourage competition, and safeguard "nondiscriminatory network access" (Jamasb & Pollitt, 2005, p. 3). Similarly, formal rules and regulations can maintain investor confidence and encouraging uniformity across ASEAN's national power sectors. (Porter & Situmeang, 2005, p. 2). Think Tank 15 discussed at great length the dominance of bundled distribution and transmission in the sub-region, and the prevalence of stateowned monopolies as a major obstacle to "any significant integration and modernisation" in the sub-region. This particular trait of national markets does not appear conducive to expansion of cross-border interconnections or subregional and international energy policy and climate goals (RQ2a, RQ2b).

While the majority of interview subjects (Former Government—Informant 1; *Multilateral Organisation—Informant 2; Multilateral Organisation 4; Academic 6;* Academic 10; Think Tank 11; Former Government 16; Former Government 17; Academic 21; Business 22; Development Bank 25; Academic 26; Former Government 27; Former Government 28; Development Bank Conference 4; Policy Conference 7; Policy Conference 9; Academic Conference 11; Government *Conference* 12) and literature on the issue reflects liberalisation recommendations. Porter and Situmeang (2005)preface their recommendations with a qualifier that reflects the common 'ASEAN way' of noninterference: the structure of power sectors "is a matter for national policy in each sovereign [ASEAN] country" (pg. 3). Multilateral Organisation—Informant 2 spoke of anti-competition policy at the national level as a major barrier to ASEAN governance measures in the EMI space:

Anti-competition is the largest obstacle at national level. ASEAN recognizes that it is part of the national policy of individual countries to not allow foreign competition or maintain a monopoly structure in its [national] electricity supply industry.

The complex nature of electricity markets and interconnections means that each country must make the necessary decisions on their own, based on what is best for national industries and citizens.

Following the development of a modernised national electricity market multiple experts (Bacon & Beasant-Jones, 2001; Porter & Situmeang, 2005; Kimura & Shi, 2011; Wu, 2019) argue that reform almost always includes the introduction of competition and liberalisation of traditionally state-controlled power sectors (*Think Tank 12; Think Tank 15; Academic 26*). These reforms often lead to improvements in supply security, overall electricity access, and price impacts. *Think Tank 15* repeatedly referenced the positive price impacts liberalisation would have on sub-regional electricity markets, as well as the broader East Asian region should liberalisation of power sectors be uniformly embraced. The majority of experts interviewed echoed these recommendations, and indeed cited some of these authors (For example: Pacudan, 2016; Shi, 2014) as support. There are some discrepancies among ASEAN-based experts as to the speed of liberalisation reforms considering various national economic and political structures.

Research shows that while neoliberal power sector reforms are meant to benefit consumers via price reductions and efficiency gains, reforms actually create a constellation of winners and losers (Anderson, 2009). Hall & Nguyen (2017) demonstrate how reforms are traditionally sought for economic and efficiency gains, but in some cases reforms later require more reforms, with a tendency towards developing economies reverting back to state involvement in power sectors (pp. 112-114). Where reforms may create dis-benefits in the short term, such as through price increases, there are also long-term benefits via investment or access for those who can pay (Anderson, 2009, p. 70-72). Reform is both positive and negative, with winners and losers depending on the market, government, and level of restructuring.

These reform and deregulation recommendations are in line with status quo recommendations for overall market liberalisation and are indeed supported by the majority of international development organisations as well. This does raise the question of how unbiased liberalisation recommendations are, and whether or not they are actually conducive to structures and systems at the national level, or whether these recommendations are reflecting an international system that favours predominantly western, neoliberal policy and economic reforms. This dichotomy will be explored in more detail in section 5.6, *Discussion*.

#### 5.4.b Coordination, Harmonisation & Standardisation

Coordination, harmonisation and standardisation of power sectors are tightly intertwined policy recommendations in the ASEAN region. Here these recommendations will be explained, including interview support for the inclusion of these reform and assessment procedures in order to optimise subregional and regional power sector connectivity.

Previously in this chapter two different methods of energy market integration were introduced—consolidation and coordination. In the case of ASEAN, where power sector development is happening at varying rates within different system structures, coordination is recommended (Academic 21) followed by a hybrid combination of the two as markets develop (Multilateral Organisation-Informant 2). Consolidation, whereby a single body controls power systems in a particular area, merges systems; coordination requires organisation to optimise and harmonise electricity trade among neighbours (Pacudan, 2016, p. 50). Coordination allows markets that are not already integrated to trade and expand their market access, although it still requires close cooperation (Government Conference 12; Government Conference 13; Government Workshop 14). Coordinating system operators and distributors requires the standardisation of practices utilised and a harmonisation of regulation, planning, costs, networks, and monitoring associated with electricity production and distribution (Pacudan, 2016, p. 41). Here the role of ASEAN is key, as working under guidelines for specific coordinating rules would benefit the reform and development of integrated markets, as well as potentially expand national markets in a manner that would benefit customers and electricity access, a key goal of ASEAN as discussed in Chapter 4. Multilateral Organisation—Informant 2 stressed the importance of coordination in "facilitating the integration of energy strategies", the development of national markets and systems, and the broader "harmonisation and standardisation required for APG" and sub-regional EMI. Coordination is important across the reform process.

Pacudan (2016) identified 5 coordinating arrangements relevant to ASEAN broadly and BIMP-EAGA specifically (pp. 51-52). A combination of coordinating arrangements could be employed given different market structures, including unidirectional trade, bidirectional trade, power purchase from an IPP (already common in the GMSR subsystem, whereby neighbouring IPPs can sell to a national level utility), third party access (where already existing infrastructure could be utilised to link two different markets) and a multi-buyer, multi seller market (allowing trading between utilities under a variety of power sector

structures). These coordination arrangements take advantage of current market structures, utilise liberalisation and modernisation efforts of individual countries, and have the potential to be applicable across ASEAN. In addition, as coordination arrangements develop, integration would increase, and the wider APG grow, moving from coordination to a hybrid coordination-consolidation model, and eventual consolidation arrangement (p. 53). This type of progression is a useful by-product of market reforms, potentially facilitating ASEAN subsystem development as markets develop (*Multilateral Organisation—Informant 2*). "More regional and bilateral agreements in power trades" will be possible as coordination arrangements are coupled with "harmonised rules" among ASEAN members (*Multilateral Organisation—Informant 2*).

This is also reflected in the model of bilateral interconnection, limited network interconnection, and multilateral, full system interconnection discussed in Chapter 4. Similarly, *Academic 21* advocate for transitional development whereby a hybrid collection of arrangements can be used to the benefit of ASEAN economies with different institutional and structural arrangements. The type of development, regulations, or arrangements pursued can be tailored to the countries wishing to engage with each other and organised around BIMP-EAGA interconnection goals. This was echoed in interviews with those close to ASEAN EMI policymaking, including *Academic 26*, who suggested this sort of coordination would suit the variety of market factors found in ASEAN. This is backed up in the literature, whereby incremental development of infrastructure, coordination arrangements, technical standards, and trade arrangements are discussed as suiting a variety of development needs (Porter & Situmeang, 2005; IEA, 2019a,d; Wu, 2019).

Harmonisation refers to the synchronisation of protocols and principles related to: consumer protection, contracts, legal structures, licensing, safety requirements, tariff-setting, tax issues, third party access, and trading systems all related to the transmission of electricity across borders and between markets (Andrews-Speed, 2016; Pacudan, 2016; Huang et al., 2019; Wu, 2019). Developing infrastructure based around transmission needs along borders, coordinating resources for domestic and cross-border power sectors, and utilising existing infrastructure are all a part of coordinating and harmonising not only market structures but also power sector development plans at the national level that would benefit sub-regional power sector goals (*Multilateral Organisation—Informant 2; Think Tank 5*). Information sharing and synchronisation of investments and planning would also benefit sub-regional

# and national goals (Government Conference 12; Government Conference 13; Government Workshop 14).

Technical harmonisation in BIMP-EAGA and ASEAN are perceived as less complicated than in fully liberalised and competitive markets elsewhere in the world, due in part to simplicity of procedures, and lack of established processes and standards (Academic 16). However, this does not mean that harmonisation of regulations and standards is not important; quite the contrary, many experts argue that harmonisation is a requirement if eventual consolidation is to occur (Porter & Situmeang, 2005; Andrews-Speed, 2016; Pacudan, 2016; Huang et al., 2019; Wu, 2019). By harmonising best practices across the region ASEAN members will assist one-another in power system and energy policy development across the region (Multilateral Organisation 26), and eventually "catch up" with global best practices (Wu, 2019, p. 17)—perhaps one day "providing aspirational EMI targets" outside the region (Former Government— Informant 1). The need for harmonisation or regulations and standards was the topic of an East Asian-based conference attended by the author, where they spoke on the challenges of grid-code harmonisation among ASEAN members (Government Conference 12). In this setting multiple experts who agreed to be quoted advocated for the prioritisation of harmonisation—however, the author did find discussion to be quite high-level, with few micro-level, tangible recommendations beyond cooperation and communication across power systems.

According to a 2014 report by the ADB on EMI in ASEAN, significant harmonisation problems do not exist in the BIMP-EAGA subsystem; voltage and stability are less a problem than elsewhere in ASEAN. Like other island areas long-distance transmission lines do pose technical problems (*Multilateral Organisation—Informant 2; Academic 16; Academic 26*), however, these can be overcome given some of the proposed coordination arrangements discussed above, including multi-buyer, multi-seller markets. Selective harmonisation and coordination can be employed, making incremental progress as markets develop, and incorporating solutions to harmonisation problems as they arise. This approach is reflective of ASEAN's broader energy and environmental policy making strategies (*Former Government 27*).

Coordination, harmonisation and standardisation of practices are not to be approached as independent issues, but as one expert put it "interdependent recommendations" (*Think Tank 5*). In their widely cited 2005 report Porter and Situmeang argue that "transmission, interconnection, regulation and pricing

rules need to be considered as a package" (p. 5). Given the focus of this research the role of ASEAN is not to be overlooked, as coordination and consolidation of energy and electricity development plans is a key role the institution can play. This is reflected in the many committees and initiatives ASEAN has that were outlined in Chapter 2, as well as subsequent author recommendations for strengthening or coordinating mechanisms (Section 5.2 and Section 5.3). While these changes are primarily national level there is also room for sub-regional collaboration and oversight as well as uniform ASEAN policies pertaining to EMI and APG coordination, harmonisation, and standardisation (*Think Tank 15*). To be clear, coordination, harmonisation, and standardisation are not the major barriers to sub-regional EMI and cross-border interconnections, but their absence does hinder the development of national electricity markets. In this regard coordination, harmonisation, and standardisation needs to be embraced across ASEAN member economies simultaneously and approached in tandem with other power sector reforms.

#### 5.4.c Subsidies

The challenge of subsidies, and need for subsidy reform, was not one the author expected to be as prevalent across interviews as it was (*Former Government—Informant 1; Think Tank 5; Development Bank 8; Academic 10; Think Tank 15; Former Government 17; Development Bank 25; Academic 26; Government Conference 12; Government Conference 13*).<sup>63</sup> However, subsidy reform was an additional area that came highly discussed—and controversially so—among respondents.

The role of subsidies in energy markets is a common area of debate across regions. Often subsidies are considered in relation to fossil fuels, drawing attention to the unbalanced competition between cheaper fossil fuels or abundant hydrocarbon resources and more expensive renewables (Ren21, 2019a). The persistence of fossil fuel subsidies is recognised by neoliberal, Western-led multilateral organisations as one barrier to renewable energy integration in Southeast Asia; and, as a result, an additional barrier to increased shares of renewable energy in a country's energy mix (ADB, 2016a). When it comes to cross-border electricity trade the role of renewables has been

<sup>&</sup>lt;sup>63</sup> Here it is interesting to note that each of these respondents works either (1) primarily in development banks or (2) has prior work experience in development banks. For example, *Think Tank 15* is currently affiliated with a think tank but gained early career experience working in a development bank setting.

discussed at great length throughout this research—but it is not a requirement for cross-border electricity trade. However, in order to realise efficiency gains, cost reductions and CO<sub>2</sub> emissions reductions gained from increased renewable energy integration and the management of variability, flexibility and uncertainty associated with cross-border electricity trade, the role of renewables is necessary for diversified national energy markets.

In the matter of electricity markets in Southeast Asian subsidies play two roles. 1) supply subsidies: in this instance subsidies are applied to energy resources to keep costs below production and thus cheaper for producers and consumers. These can be subsidies on fossil fuels (the overwhelming majority),<sup>64</sup> or subsidies on renewable resources (although the latter is debated as one potential solution to limited growth in renewable energy integration, according to interviews Former Government 17 and Former Government 27 and literature such as Whitley 2013a; Whitley 2013b; Lockwood, 2015a; Phoumin & Kimura, 2015; among other). The other relevant role subsidies play in Southeast Asia is 2) electricity subsidies: whereby the cost of electricity is reduced via government funding, IPP involvement, and with higher rates of utility debt.<sup>65</sup> These electricity subsidies are often aimed at the poor or those with the lowest access to energy (ADB, 2016a), although in many countries the most populous regions actually gain the most (For example, in the case of Indonesia 70% of the population benefits from subsidies and in the national sub-region with the highest population density. See: Global Subsidies Initiative, 2017). Whether the subsidies benefit supply or electricity the two are intertwined as coal, gas or oil subsidies often reduce electricity prices for consumers, propping up a reliance on fossil fuels for energy

 $<sup>^{64}</sup>$  In the case of fossil fuel subsidies, the economic and environmental costs (in the form of CO<sub>2</sub> emissions) are widely acknowledge and studied. See Lockwood (2015) for an overview of fossil fuel subsidies and impacts on developing economies; Svogaard and Van Asselt (2018) for analysis of the global fight against fossil fuel subsidies; Whitley (2013a, 2013b) for research on the intersection of fossil fuel subsidies and climate change.

<sup>&</sup>lt;sup>65</sup> Subsidies and debt are highly intertwined within the power sectors of ASEAN economies. Across the sub-region deregulation has been motivated by a desire to raise capital investment in order to keep up with electricity demand growth (IEA, 2015, p. 33). As the sub-region experienced an economic boom in the 1980s and 1990s, ASEAN economies took on extensive debt to drive public spending programmes (IEA, 2015, p. 33). Governments were similarly unable to keep up with power sector developments, and as such allowed IPP involvement in national markets; however, as discussed in Chapters 4 & 5, much of these effort were scrapped between 1993 and 2012 as a result of poor regulatory oversight, and the Philippines remains the only (nearly) independent power market in the BIMP-EAGA subsystem (IEA, 2015, p. 33-36). As a result of increasing debt utilities have had to either raise prices or renegotiate power production agreements (PPAs), which has upset consumers and reduced investor confidence, respectively (IEA, 2015, p. 36).

consumption, encouraging inefficient electricity use, continuing higher emissions, and perpetuating systems that are not compatible with clean climate development (Burke & Kurniawati, 2018; Whitley, 2013a). 2020 subsidies among BIMP-EAGA economies are not insignificant.

Country	Total* Subsidies in 2020		
Brunei	\$186.2 million USD		
Indonesia	\$19, 217.7 million USD		
Malaysia	\$1,811.6 million USD		
Philippines**	n/a		

\*Including Coal, Electricity, Gas, and Oil subsidies;

\*\*Philippines data not included by IEA as a result of its very low subsidy ranking internationally and progress in subsidy reform.

Source: IEA, 2020, no pagination.

In part as a result of high costs and financial limitations, subsidy reform in ASEAN remains an important and contentious issue (Phoumin & Kimura, 2015), with implications for growth in renewables and the sub-region's energy transition. Previous perceptions regarding energy subsidies favouring the poor are actually being rebuked by many studies (ADB, 2016a; International Institute for Sustainable Development [IISD], 2014; Phoumin & Kimura, 2015), with an array of winners and losers being created as a result of energy subsidies throughout the region (APEC, 2012). In 2012 estimates of total Southeast Asian fossil fuel subsidies equaled roughly \$51 billion USD—a cost that is much higher if environmental and health cost are included (Phoumin & Kimura, 2015, p. viii). In 2011, the share of after- tax fossil fuel subsidies to GDP were roughly 8.41% in Brunei Darussalam, 7.21 % in Malaysia, 5.36 % in Indonesia (IEA, 2012b, p. 3) and only 0.7% in the Philippines (IISD, 2014, p. 62; also 2011 data). Over half of the subsidies in each country are attributed to low-priced petroleum products, with total subsidization strongly believed to be higher than estimates, reflecting calls for transparency among global institutions (ADB, 2016a, p. 2).

For the majority of ASEAN members removing or even just reducing energy subsidies would have positive impacts on growth, poverty alleviation, equity of access, and environmental impacts as a result of continued fossil fuel (IEA, 2012b). Of the many unilateral political commitments to lower subsidies across

the world, a 2016 ADB study found that of those in Asia, Indonesia, Malaysia and the Philippines had significant commitments on paper (ADB, 2016a, p. 2). Subsidy policies differ from nation to nation in southeast Asia, much like the other reforms identified. The majority of ASEAN economies and BIMP-EAGA countries have supply subsidies in place, increasing reliance on hydrocarbons (IMF, 2013). Three of the 4 BIMP-EAGA countries have considerable subsidy reforms to undertake—Brunei Darussalam, Indonesia and Malaysia. The Philippines is the only country to have actually undergone subsidy reform (IISD, 2014, pp. 60-62), with primarily oil subsidies remaining. Data from Brunei Darussalam remains limited, but recommendations regarding subsidy reform largely hinge on reducing distortions that might limit energy mix diversification, with concerns regarding transparency from Indonesia and Malaysia as well (ADB, 2016a). Previously unsuccessful reforms in Indonesia and Malaysia (as a result of high prices for consumers and government that resulted in protests, riots, and in the case of Malaysia, threats of military intervention) look posed for second and third attempts (APEC, 2012).

In the case of the majority of Southeast Asia, with the exception of Singapore and Malaysia in BIMP-EAGA, the removal of supply subsidies would increase costs in the short term; in Malaysia's case removing government petroleum subsidies would positively impact GDP and increase economic efficiency in the short term (Phoumin & Kimura, 2015, pp. viii-ix). In the case of sub-region wideimplementation, while there may be short-term costs, alterations of fossil fuel subsidies are believed to have the potential to reduce ASEAN member economies' budget deficits and improve economic efficiency in the medium and long-term (ADB, 2016a, p. 17). Augmenting short term costs in order to achieve long term gains associated with the removal of supply subsidies would require targeting policies for those who are poor or already lack energy access (Development Bank 25), government transparency (Multilateral Organisation— Informant 2; Academic 26), consistency in programming and messaging (Academic 10), and policy support at both the national and subnational levels, as well as ASEAN support and incentivisation measures (Think Tank 5, Academic 10; Former Government 17; Academic 26). The gradual phase out of subsidies is expected to remain a priority if renewable commitments are to be achieved (REN21, 2020).

The role of fossil fuel subsidies in preventing renewable energy growth globally, within the East Asian region, and in the Southeast Asian sub-region is agreed upon by experts and international organisations (See: IMF, 2013; ADB, 2016a;

IRENA, 2016; Global Subsidies Initiative, 2017; REN21, 2019a, 2019b), as well as interview subjects (*Former Government—Informant 1; Think Tank 5; Development Bank 8; Academic 10; Think Tank 15; Former Government 17; Development Bank 25; Academic 26; Conference Event 6; Workshop 1*). In addition, the difficulty in removing subsidies is equally represented in interviews and cited as one area where considerable political will is needed in order to make any changes. The uniformity of these recommendations and observations will be discussed in more detail in the discussion section of this chapter, however, it is worth noting many of the organisations and individuals represented here come from a similar neoliberal viewpoint.

One particular interview subject who works on power sector reform in developing economies in the global south, *Development Bank 25*, spoke at great length about the positive progression of power sector subsidies when developing electricity markets in poor communities:

I also see a role for power sector reform in fiscally deficit countries to move from first, providing subsidies for access, to then more sustainable prices for power generation...this must be managed, but it could provide relief where [energy poverty] is a concern...the high upfront costs of transmission lines and interconnections could also be augmented by subsidies for poor consumers.

Among reform targets proposed by a variety of authors (Navarro & Sambodo, 2013; Han & Kimura, 2015; IRENA, 2016), the removal of supply subsidies is described as a necessary prerequisite for an integrated ASEAN electricity market (*Academic 10; Academic 26*). In this regard ASEAN has a role to play in improving the messaging regarding the efficacy of supply subsidies, their ineffectiveness in reducing energy poverty, and their long-term contribution to sub-regional, regional, and global CO<sub>2</sub> emissions. Reform of subsidies is also closely linked to other reform recommendations, particularly liberalisation recommendations. According to *Academic 10*:

ASEAN EMI relies in part on market-based pricing and regional market liberalisation practices so as to fully integrate. But the subsidy policy of ASEAN economies contradicts these mechanisms. False pricing, high consumption, monopolies can be a result of these subsidy practices. SO, EMI goals must also phase-out subsidies.

Work by the IMF (2013), ADB (2016a), IEA (2016), IRENA (2018), Skovgaard and Van Asselt (2018), and REN21 (2019) point to the difficulty removing subsidies in societies where fossil fuels, subsidisation, and (in some cases)

energy poverty are prevalent. Without national and sub-/regional incentives subsidy reform will be very difficult to incorporate at the level necessary for a fully integrated ASEAN energy market and the APG (*Academic 10*). If ASEAN is indeed serious about the APG this issue will have to be tackled more seriously, and with more serious repercussions than it has been to date (*Academic 26*). According to *Academic 10*, the challenge here is the 'ASEAN way' and the practice of non-interference that is prevalent across ASEAN initiatives (a challenge which was also present in other reform recommendations). At this point in time subsidy removal and ASEAN enforcement are in direct competition with each other, which does not bode well for ASEAN EMI in the long term.

Expert interviews shared three dominant reform recommendations in order for EMI to take place: (1) market liberalisation; (2) coordination, harmonisation and standardisation of markets and practices; and (3) the removal of subsidies. These reform recommendations do not replace O&PCRB EMI Criteria but support the largely neoliberal criteria in aiding the physical, policy, and relational market requirements in order for EMI to flourish in BIMP-EAGA and Southeast Asia. The following section, *Discussion*, will examine the common national market factors and how the previously discussed EMI criteria and reform proposals fit within the market dynamics of BIMP-EAGA, providing context to EMI and electricity market reform and expansion within the sub-region.

## 5.5 Discussion

The aim of this chapter is to address RQ2 and its relevant sub-questions: (RQ2) How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets? (RQ2a) how does the political and economic structure of national electricity markets impact the expansion of cross-border interconnections in BIMP-EAGA? (RQ2b) is the structure of national electricity markets conducive to sub-regional and international policy goals in Southeast Asia? So far this chapter has revealed that the variety of economic and political structures present in BIMP-EAGA are not currently conducive to increased cross-border electricity trade based on policy, physical, and relational requirements for EMI to flourish (RQ2a). In addition, current structures of national electricity markets are not conducive to subregional and international policy goals related to energy and climate targets given a continued reliance on subsidies, dominance of fossil fuels in sub-regional and subsystem energy mixes, and the market structure of national electricity systems, whereby there is a reliance on bundled systems, lack of competition, and varied capacity (RQ2b). This section will further expand on these answers, and address Research Question 2 by discussing two specific points related to the theme and title of this chapter, *National Market Factors*. First, the primacy of national priorities, and second, the usefulness of neo-developmental statism to explain these priorities and the associated challenges in EMI and cross-border electricity trade growth in the sub-region and BIMP-EAGA subsystem.

# 5.5.a National Priorities

In Chapter 5 (*National Market Factors*) a number of necessary requirements were introduced using the Oseni and Pollitt (2014) criteria coupled with results from author interviews and a review of the literature on electricity market integration. This combined list of requirements was subsequently named O&PCRB EMI Criteria and contrasted with market factors of individual BIMP-EAGA nations, highlighting the requirements that are missing in order for increased EMI to take place within this ASEAN subsystem. Examining the O&PCRB BIMP-EAGA Scorecard, as it was named in Section 5.3, against national versus sub-regional level requirements, highlights the importance of national requirements. The majority of proposed criteria are, in fact, national level needs or a hybrid of both national and sub-regional—further supporting the findings in Chapter 4 that sub-regional targets cannot be achieved without national level initiatives and prioritisation of national level reform. In the table that follows national and sub-regional criteria have been designated, highlighting this point further.

Requirements	Туре	Level
Commitment to free trade	Policy	Hybrid
Market liberalisation efforts	Policy	National
Efficient market design	Policy	National
Development of multi-buyer and multi-seller market (Pacudan, 2016)	Policy	National
Governance support	Hybrid	Hybrid
Jurisdictional considerations / management	Hybrid	Hybrid
Development of transition plans by national power utilities (ASEAN Specific)	Hybrid	Hybrid
Enhance Capacity for Energy Policy and Planning (ASEAN Specific) (Andrews-Speed, 2016)	Hybrid	Hybrid

Table 5.6	<b>O&amp;PCRB</b>	<b>BIMP-EAGA</b>	Scorecard	with	Reform	Level
I UDIC DIO	our und	DINII DIIGII	Deorecura	VVICII	I CIOI III	LC V CI

Commitment to incremental development	Hybrid	Hybrid
Transmission capacity	Physical	National
National electricity transmission systems	Physical	National
Management of distributional effects	Policy	Hybrid
Presence of trusting or prior relationship	Relational	Hybrid

Based on these designations, this table demonstrates that national level power sector requirements make up the majority of EMI criteria, followed by incremental sub-regional integration, leading to eventual EMI in designated ASEAN subsystems—subsystems that could eventually look different from those originally drawn by ASEAN and HAPUA. The conclusion the author has drawn here, based on feedback and prior analysis, is that some nations may be left behind if their national power sector policies do not move with the speed of their neighbours and make the necessary requirements for deeper EMI and eventual cross-border interconnections.

Power grids are, currently, confined by national level boundaries, therefore reinforcement or growth of power grids needs to first happen at the national level. However, political will and governance strength are required for this buttressing to occur. Where state owned utilities dominate the power sector (Indonesia and Brunei, for example, as discussed in Section 5.2) there may be difficulty in expanding renewable energy options within the national energy mix; for example, the prevalence of fossil fuel subsidies and the nationalised companies that benefit from their continued use is one barrier in such a system (Shi, 2016, p. 674). In addition, power dynamics, in the form of dominant national economic actors, influence political will. *Former Government 27* highlighted this in describing these power dynamics among ASEAN members:

Their [ASEAN members] interests are not the same across the board, because corporate powers have a vested interest in some of these systems—political, economic or energy. It's hard to say how that will play out in the connectivity agenda, just like it's hard to say how it will play out in the South China Sea. Add outside actors to the mix and its' even harder to predict how power will make policy move.

Here *Former Government 27* is making the link between national level concerns and sub-regional power considerations, using the influence of China over the

territorial South China Sea dispute<sup>66</sup> with ASEAN members as an effective illustration. This link is an interesting exemplification because China is similarly involved in power plays throughout Southeast Asia through its BRI initiatives and connectivity agenda in ASEAN, all of which have contributed to issues of trust between ASEAN actors and Chinese. These Chinese dynamics will be explored in Chapter 6 (Governance Challenges) and contribute to national hesitancies in taking Chinese investment for power sectors. However, a number of countries in Southeast Asia (Cambodia, Lao PDR, Myanmar, Vietnam) have already allowed Chinese corporations to not just invest in but also build national power grids (Shi et al., 2019).<sup>67</sup> The author has observed that, given the development competencies, geographical location (China's closest neighbours), and electrification needs that these four economies have independently weighed the relative gains from Chinese involvement in their domestic markets as greater than the losses; among these considerations are naturally going to be energy security and supply concerns. Seemingly, if nations can individually decide to accept Chinese involvement in increasing electrification and power grid development, the region as a whole can similarly agree (Although likely not without some ASEAN leadership among those countries that have already made the decision to include China in their domestic power sector development).

National governments are quite tied up; there are multiple vested interests informing policy makers and multiple factors pushing policy many different directions. It is no wonder that connectivity is not fully clarified and imagined yet in ASEAN. Two interviews (*Think Tank 5* and *Academic A26*) pointed the author towards an example of ASEAN failure that has, they believe, been largely due to similar national energy security priorities: the abandoning of the Malaysia-Brunei interconnection that was meant to deliver hydropower from Malaysia to Brunei (Shi, 2016). This line, formally named the Sabah-Brunei line, was removed from the APG plan over Brunei's concerns of security of supply

<sup>&</sup>lt;sup>66</sup> The South China Sea dispute refers to the years-long territorial dispute over competing and overlapping territorial claims in the South China Sea between China, Taiwan, the Philippines, Brunei, Malaysia and Vietnam. The disputed area has estimated oil and natural gas reserves and falls within an important area for fishing and transport, making it valuable economically and politically. Within ASEAN the South China Sea dispute has been a topic of contention, both in terms of overlapping claims among members and with regards to ASEAN's policy of non-intervention (Angela, 2020).

<sup>&</sup>lt;sup>67</sup> It should be noted that these four economies, commonly referred to as the CLMV countries, are among the least economically advanced in Southeast Asia but also considered to be quickly growing (Ambashi, 2017).

from Malaysia (p.676). Details regarding Brunei's concerns are limited, and to the author's knowledge no academic literature has expanded on Brunei's motivations behind the removal (ADB, 2014; Shi, 2014, 2016; IEA, 2015).<sup>68</sup> *Think Tank 5* believes this demonstrates the prioritisation of energy security concerns over sub-regional projects and targets, a sentiment echoed by *Academic A26.* Ultimately this doesn't bode well for renewable integration—if energy security concerns take precedence, and national governments perceive energy security to be most strongly guaranteed through domestic hydrocarbon supplies, integration of renewables may continue to take a back seat.

Energy security concerns emerged as a primary national level worry—national ministries, regardless of subject focus, have perceived domestic energy security as independent from sub-regional energy security cooperation. In fact, it appears that national ministries perceive energy security to be zero-sum—if your neighbour has it you don't. In addition, some ministries have different incentives when it comes to reform of market structure generally. This was echoed in *Government Conference 12* where the following remark was made by a representative of a global multilateral institution:

We have policy barriers when it comes to [energy] transition issues and different ministries; there are silos of communication among ministries within the same government, with different issues being prioritised with no regard for long-term regional and global energy goals...we have issues of decoupling and liberalisation, for example, where two different ministries may be playing a role in the power sector but refusing to coordinate necessary reforms.

In the same meeting (*Government Conference 12*) another expert from an East Asian based think tank echoed the challenge of national level coordination and action where a ministry perceives the reform as negative yet sub-regional institutions, such as ASEAN, perceive reform as positive:

The challenge is to combine national with regional and local level initiatives; we need to bring them together for cooperation, otherwise we run the risk of sectors directing the game...national

<sup>&</sup>lt;sup>68</sup> This is reflective of a repeated challenge with interconnection research—access to information on motives is limited, and, in the case of Brunei, transparency of data and information is lacking (ADB, 2014, p. xiii). Sabah has experienced supply shortages as a result of rising demand and aging power systems (IEA, 2015, p. 18), however, these are not explicitly cited as cause of security of supply concerns in Brunei.

### and [sub-]regional competencies could be better utilised, combining practical experiences with [sub-]regional initiatives.

How these vested interests and multiple levels of power at play (sub-regional and national) interact within the sub-regional political economy of cross-border electricity trade has impacts on the ability of renewable energy to participate in national markets—where sub-regional and national level targets point towards room for growth (and commitment to growth) in renewables, national measures and national market factors point to continued reliance on the status-quo. Where sub-regional and national level targets point to growth in cross-border interconnections, national level measures point to hesitancy in making the necessary political and economic changes in order for renewable-dominant EMI to flourish. The data on national market factors also points to a startling undercurrent—excess energy, in the form of hydrocarbons, can also be traded across subsystem and sub-regional interconnections; if national measures are not committed to renewable integration, but instead have the ability to continue to operate as-is via energy consumption and also gain from trading excess capacity, perhaps the echoes of climate change commitments are not the real impetus behind increased interconnections. Instead, economic gain via trade in excess capacity to neighbours may be the actual motivating factor.

Neo-developmental statism offers an additional explanation from which to understand these dynamics—and uncertain motivations and potential outcomes. The following section will bring this explanatory tool into the discussion of national market factors, adding additional insight to this multifaceted political and economic phenomenon in Southeast Asia.

### 5.5.b Neo-Developmental Statism & National Market Factors

The development story of East Asia is one that is quite familiar to political economists due to the expanse of its use, resulting in extreme & sustained economic growth, and global repercussions of the development practices utilised in the region. This developmental model can be summed up as a competition between the dynamics of state-led and market oriented economic policies; this includes employing developmental state growth practices such as state-led economic development programs targeted at specific industries, encouraging public-private partnerships, and utilising export-led industrialisation practices (World Bank, 1993). This balanced state-led market approach has been utilised to varying degrees of success and with varying levels of state or market dominance in East Asia (White and Wade, 1988), challenging western economic and political beliefs that economic development comes at a cost—picking more distinctly between one and the other.

For the most part the traditional developmental state model is reflected in the electricity markets of Southeast Asia today. State run, the presence of strong transnational corporations, undergoing various stages of restructuring around IPPs or increased (although limited or in very early stages) competition. Close relationships between business and government are not unique to developmental states—however, they are one of the key characteristics of East Asian developmental states, including those found in the sub-region of Southeast Asia. State-business relationships in Southeast Asia are not as clear cut as those found among the Northeast Asian tigers—instead, state-business relationships in Southeast Asia are strongly "influenced by inter-ethnic distributive pressures and the resulting need to manage challenging political and economic issues at the same time" (Kasahara, 2013, p. 7). Criticism that southeast Asian states are not 'developmental' because of corruption or rent-seeking is a point of contention in developmental state literature (see, for example, Woo-Cummings, 1999). Nonetheless, this thesis concurs with Hayashi (2010) that Southeast Asian nations possess the characteristics of plan-rational economies (p. 52) and exhibit a variety of nuanced, developmental state characteristics that are in and of themselves unique in degree and application to developmental statism in Southeast Asia. Where DS is reflected throughout individual ASEAN economies, it is also reflected in the economic and structural institutions of the sub-region as well.

While the reform recommendations laid out in Section 5.4 make sense in the context of electricity access and grid integration of variable renewable energy resources, they don't necessarily make sense in terms of the broader, underlying issues of economic development, continued industrialisation, and energy intensity that have moved the region to a place that calls for increased renewables in the first place. If global climate goals, specifically emissions reductions, are to be met, capitalistic, neoliberal systems that encourage development-as-is do not appear conducive to long term changes to the global political economy. "Neoliberal market restructuring", whereby neoliberal norms are placed not just on a market, in this case electricity in nature, but trickle down into society and all aspects of management and socialisation (Wiegratz, 2016), is clearly visible in the reform practices proposed.

One particular voice of opposition to standard reform proposals presented among interview subjects comes in the work of Andrews-Speed (2016) on lessons from the Nord Pool countries (discussed in Chapter 1, section 1.2). Andrews-Speed's critique can be examined with BIMP-EAGA in mind, providing political and economic ways forward that don't require liberalisation or removal of subsidies. Andrews-Speed uses the example of SAPP (previously discussed in Chapter 1), which trades electricity between 12 southern African economies with electricity systems that are vertically integrated, state dominated, and have limited liberalisation (p. 3). Andrews-Speed examines how the Southern African Power Pool has integrated Nord Pool lessons, and how these might shed light on a way forward for ASEAN-without the removal of subsidies and complete liberalisation. The SAPP is an interesting juxtaposition to standard neoliberal power sector recommendations-here, the 12 member nations defy the common economic model and trade via national power companies that act as single buyers and sellers, with continued subsidies to consumers and only some IPP involvement (IEA, 2019d, p. 61). SAPP economies also only trade excess power, after national needs have been met (IEA, 2019d, p. 62), thus limiting potential concerns over security of domestic supply. Both differences in SAPP power trading offer hope for ASEAN. To the author's knowledge comparison of these two different markets (SAPP and ASEAN) has not been expanded on further than the work of Andrews-Speed (2016) and a brief comparison by the IEA (2019d); additional comparison of SAPP in future ASEAN EMI studies could add further nuance to the neo-developmental statism explanation of competing market factors in the BIMP-EAGA subsystem.

An additional point to make regarding an alternative approach to reform is the issue of subsidy removal. This author would like to note that the removal of fossil fuel subsidies in Section 5.4.c is largely advocated by neoliberal economic and governance institutions (ADB, IMF, etc.) and many of the interview subjects are associated with similar organisational and political views on neoliberal economic policy making. In this sense, the uniform recommendations are no surprise and reflective of a global economic and energy policy trend to advocate for the complete removal of subsidies along with other liberalisation efforts. This standard, blanket approach to subsidy removal is often approached with little concern for hybrid models of subsidisation, the role of subsidies in developing economy policy making, within corrupt political systems, or in communities where trust in transition policies is lacking (See Lockwood, 2013, 2015a). The power dynamics at play within the Southeast Asian communities where subsidies are most dominant, notably throughout BIMP-EAGA at varying levels, is indicative of the balance needed between policy making, climate compatible

development, and economic interests in hybrid, centrally planned, market economies.

One option for reconciling reform proposals and state-led economic practices, based on this author's understanding, is a hybrid developmental state model. In a hybrid developmental state model a combination of state-run and minimal liberalisation efforts are employed in order to integrate markets and maintain order under current, slowly adjusting, national political and economic systems. Work on a hybrid developmental state model in various forms has been done by Bishop et al. (2018c), Chikozho and Mapedza (2017), Chowdhury (1999), Gainsborough (2009), Radice (2008) among many others, but this research is not applied to the power sectors specifically and does not include BIMP-EAGA in their application. Much of this prior research posits that instead of focusing on the developmental state as an advocate of industrialisation and state supported industries, hybrid, 'new' models of DS instead focus on states picking the freemarket mechanisms that work for them (See: Bishop et al., 2018, for Northeast Asian examples of growth in this area generally). Based on this research and the authors understanding this can also be applied to electricity markets, EMI, and the structural needs for cross-border electricity trade to flourish in Southeast Asia, with some specifications tailored to the ASEAN experience and market structures. A revised, hybrid model of DS, such as neo-developmental statism, illustrates the challenges between free market and developmental state policies. Neo-developmental statism, for example, explains that choices between free market and developmental state policies are made based on the political and economic realities of the particular nation or sub-region, and can result in a hybrid model, such as those BIMP-EAGA economies described earlier; they are committed to free trade and liberalisation as far as market participation is concerned, but they also favour state dominated electricity systems with limited competition. This suggests there is no one-size fits all model, and also that dominant pressures within a state system will often have influence on the outcome of subsequent governance models.

It is the author's premise here that hybrid developmental state practices could be not just tailored to individual states, but to national electricity markets, their goals, and the overall goals of ASEAN. States and sub-regional institutions could, under this line of thinking, prioritize semi-liberalisation, reduce subsidies, increase actual competition, and focus on sustainable development practices that complement sub-regional EMI goals and national electrification priorities. This would allow for the continued drive of policy and economic function from states but open up a higher prioritisation of liberalisation if it aims to meet power sector needs. Under this hybrid model, however, some liberalisation needs may need to be more realistically adjusted—and, in fact, this could be better for sustainable development goals overall.

Take for example, the case of Brunei and Malaysia, where there is limited private sector involvement. the earliest signs of liberalisation, and heavily subsidised energy and electricity sectors (Pacudan, 2016, pp. 43-44, p. 46; See section 5.2.a for more details on individual BIMP-EAGA economies). In Brunei, a primary focus on subsidy removal, balanced with improving transparent regulation and transmission capacity might be the best combination of policy priorities, rather than say, regional and global calls for more market liberalisation and increased competition in the Brunei power sector. In Malaysia, a focus on subsidy removal and jurisdictional management may be more realistic than development of a national electricity transmission system. In the case of Brunei this author believes it is not realistic to prioritise market liberalisation and increased competition given the deep integration of hydrocarbon companies into the electricity and energy sectors. Similarly, in the case of Malaysia, where geographical constraints limit the development of national electricity transmission systems, this may not be a realistic priority. These two examples offer illustrations of hybrid models of developmental statism, in this case identified as neo-developmental statism, where the reforms most realistic to the individual market are prioritised.

Further, neo-developmental statism and hybrid state practices also offer explanations for the imperatives pushing cross-border interconnection in the sub-region and renewable energy's ability to participate in national markets: while there is incentive for renewable energy integration as a result of international and sub-regional climate initiatives and energy poverty targets, states could choose instead to focus on the economic gains of selling current excess capacity, thus the continued dominance of hydrocarbons and hydrocarbon power structures in national markets. States are, again, picking and choosing the developmental state practices that best serve them at this point in time. For some that means actual growth in renewables, for others that means immediate economic gain and slow or gradual growth in renewables as it is seen to benefit or when it makes economic and political sense to prioritise.

How hybrid-models shape out will be dependent on national and sub-regional governance institutions and the relationships between interconnection partners. Analysis of this will be expanded upon in Chapter 7, incorporating the

forthcoming arguments in Chapter 6, where trust will be brought into this discussion.

#### 5.6 Conclusions

While the initial goal of this research was to provide context to the use of trade as a tool for growth in renewable energy use, and in the case of East Asia, growth in renewable energy use through electricity market integration, the proposed reforms needed to make this growth possible pose a number of theoretical questions relating to the political economy of electricity markets in the broader East Asian region, the sub-region of Southeast Asia, and the microcosm of BIMP-EAGA. This chapter has sought to answer Research Question 2 (RQ2) and its relevant sub-questions: (RQ2) How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets? (RQ2a) How does the political and economic structure of national electricity markets impact the expansion of cross-border interconnections in BIMP-EAGA? And (RQ2b) Is the structure of national electricity markets conducive to sub-regional and international policy goals in Southeast Asia? This chapter has disclosed that the variety of economic and political structures present in BIMP-EAGA are not currently conducive to increased cross-border electricity trade based on requirements for EMI to flourish. In addition, the present structures of national electricity markets are not conducive to subregional and international policy goals related to energy and climate targets given a continued reliance on subsidies, dominance of fossil fuels in sub-regional and subsystem energy mixes, and the market structure of national electricity systems. In addition, the IPE of these national market factors and sub-regional interconnections have exposed that while there is room for growth in renewables, countries do not have an electricity deficit, and therefore the impetus for increasing renewable integration as a means to trade surplus electricity is not actually present. As a result, based on RQ2, renewable energy has potential for participation in national markets, however, current market factors point to limits in its ability given the present national power systems.

Strengthened EMI and increased cross-border electricity trade are two sides of the same coin in power sector reform and renewable energy integration. This chapter has discussed the national market factors of BIMP-EAGA nations, how these two tools (EMI and cross-border electricity trade) can be achieved in BIMP-EAGA, and the reforms necessary for their growth in Southeast Asia. Major reforms have been grouped in three categories: (1) market liberalisation, (2) coordination, harmonisation, and standardisation; and (3) subsidies. Analysis of the market factors in the BIMP-EAGA subsystem, reforms needed, and national level challenges to implementation of reforms, has probed whether or not renewable energy and climate target incentives are really pushing cross-border interconnections and sub-regional EMI forward. Instead, the author posits, economic gains via trade in surplus energy (renewable or not), may be behind the regional and sub-regional interest as well as national measures.

This chapter has contrasted identified necessary and common reform proposals against the market structures of the region as a whole. This has resulted in the use of a hybrid form of DS—neo-developmental statism—as a means to understand the power sector needs of the region within the context of the East Asian developmental story.

# Chapter 6 Governance Challenges

## **6.1 Introduction**

This chapter continues where the previous two chapters (Chapter 4, *Sub-Regional Market Factors* and Chapter 5, *National Market Factors*) left off: picking up on the final theme that emerged from expert interviews and data analysis, governance challenges. Chapter 6 will address Research Question 3 and its associated sub-question: (RQ3) *What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian Development?* and (RQ3a) *What is the political and economic status of the ASEAN Power Grid?* This chapter will build on the narrative of what political and economic policy options exist to further cross-border electricity trade by addressing Objective 3, identifying and analysing additional governance challenges to deeper EMI and increased electricity interconnections among ASEAN member states.

Stakeholder interviews conducted with current and past government, business, and academic leaders throughout East Asia inform the governance challenges outlined in the following sections. These governance challenges have been grouped by the author into three inter-related categories that all fall under the broad umbrella of governance: (1) sub-regional versus national interests, (2) trust among ASEAN members; (3) trust of outsiders. According to elite interviews, there is a serious perception challenge as it relates to the need for examination of domestic energy market and sub-regional governance shortcomings and the interests that drive these decisions. The challenge of national perceptions repeatedly arose in interviews—ASEAN economies perceive there to be security of supply risks at the national level if they are dependent on imported electricity from their neighbours. However, this perception is, in fact, not based on experience—there are no known instances of electricity being withheld among existing ASEAN trading partners as a result of political (or other) conflict. In fact, ASEAN as a whole is reliant on fossil fuel imports and numerous energy trading relationships already exist across the subregion (IEA, 2019a). Regardless, the perception that imported electricity will pose a supply risk is prevalent. It is the author's interpretation based on interviews that this perception is reflective of the challenge of trust among ASEAN member states, and not actually preventative of future electricity trading relationships.

In addition, the challenge of political will, coupled with questions of trust, have created an environment where progress is limited in scope and impact but ASEAN targets remain high. This chapter also represents the most surprising and also potentially interesting interdisciplinary and topical cross-over of this research, China's Belt and Road Initiative (BRI) and its impacts on sub-regional trust and power sector development via investment and financing.

Chapter 6 will be structured as follows: first, Section 6.2 will explore the different sub-regional versus national level interests embedded within ASEAN's EMI and cross-border electricity trade initiatives. This section will highlight the conflicting sub-regional and national expectations and EMI considerations. This will include sub-regional and national perceptions of energy poverty versus energy security, drawing attention to a mismatch in sub-regional and national goals and policy maker perceptions on how to manage priorities that may conflict depending on the predominance of renewable energy integration in national energy policies (i.e. renewable energy integration vs. hydrocarbons for electricity access). In Section 6.3 trust within ASEAN will be described, analysing how trust relates to cross-border electricity trade and EMI in Southeast Asia and among ASEAN members. Mistrust among actors will be examined, relating trust and mistrust in regional relations and the use of bilateral and multilateral electricity trading in trusting relationships. Section 6.4 will cover the issue of trust of outsiders, including an overview of China's BRI, what it means for energy sector investment in Southeast Asia, and general concerns with the Chinese initiative and Chinese driven financing. Section 6.5 will survey the status of APG to date, making links between ASEAN EMI projects, national interests, and trust. Section 6.6 will bring these points together in the discussion, making links between governance challenges and the economic and political realities that have emerged from analysis of the data and the use of neo-developmental statism as an explanatory tool. Finally, section 6.7 will reiterate the main points of each section, tying these key issues together in the framework of understanding, evaluating, and progressing cross-border electricity trade and EMI in East Asia and answering research questions 3 and 3a.

## 6.2 Sub-Regional versus National Interests

Among interview subjects with an expertise in regional governance the role of ASEAN was discussed quite extensively. Nearly all respondents noted the governance limitations of ASEAN in ensuring cooperation with non-binding measures, managing expectations and interests of multiple parties with similar

but different interests, and the challenges of working within the ASEAN bureaucracy. However, all also noted the power of ASEAN's sub-regional grouping, and the impact it has had on regional cooperation and policy cohesion. In short, the general opinion is as a governance institution ASEAN has flaws (such as the governance limits of non-binding measures, difficulty managing expectations, multiple competing interests, and institutional bureaucracy), but the overall strengths of cooperation, coordination, and cohesion outweigh the weaknesses and the sub-region is better off for its existence (*Former Government—Informant 1; Academic 26; Former Government 27; Workshop 1; see also section 2.3.c for further discussion on ASEAN governance limitations)*. This viewpoint is also reflected in the academic literature on ASEAN (Kivimaki, 2009; Kurlantzick, 2012).

In Chapter 2 key governance limitations of ASEAN were highlighted, mainly related to the potential governance of sub-regional EMI within ASEAN, including the practice of non-interventionism, non-binding agreements, polite non-interference, voluntary compliance, and a lack of legality within the organisation itself. Here, the discussion moves to a different governance challenge within ASEAN, sub-regional versus national interests. Chapter 4, *Sub-Regional Market Factors*, disclosed that sub-regional targets require national level changes; Chapter 5, *National Market Factors*, expanded on this finding by focusing on national-level electricity systems and the needs at the BIMP-EAGA subsystem level. Here analysis will more deeply examine the balance between national and regional interests and the corresponding differing expectations, as well as how these challenges are being incorporated into the IPE of cross-border interconnections in the sub-region.

### 6.2.a Sub-Regional Interests

An interesting point made by interview subjects was the issue of trust and communication breakdown at the national and sub-regional levels. On the one hand, ASEAN has made sub-regional interconnection a key component of its regional connectivity agenda (see prior reference to the AEC in Chapter 4; APAEC 1999, 2004, 2010) and energy policy initiatives. Regional targets, commitments, and forward-looking programs are all peppered with mention of the importance of EMI in ASEAN's growth (ASEAN, 1997; ASEAN, 2016). However, as previously demonstrated (Chapters 4-5) national level adjustments are required in order to realise sub-regional EMI and cross-border electricity trade. These adjustments include standardisation of practices and rules, modernisation of infrastructure including increased grid capacity, expansion of national grid interconnections,

institutional strengthening, capacity building within relevant institutions (government or otherwise), and in some cases a review of current energy subsidies. Unfortunately, no manner of ASEAN pledges can push national governments to make changes if reforms are perceived to be against national interests.

On the surface, increased access to electricity is a priority of governments around the world, and indeed of significance to ASEAN and its member economies. According to three primary interview subjects (Former Government—Informant 1; Multilateral Organisation—Informant 2; Academic 26) while electricity access is a priority, other concerns, financial or policy, make take precedence at the national level, contradicting ASEAN targets. National governments have expectations that pledges to ASEAN initiatives will result in improvements and financial support that would be otherwise difficult to realise without ASEAN assistance; all while maintaining primacy as an independent actor with national needs that "must be met and take precedence over regional policy making" (Academic 26). In addition, regional recommendations for trade and economic liberalisation are not always met with support at the national level, and indeed become entrenched with other economic interests and incentivisation limits at the national level. For example, the introduction of competition into national electricity markets has been quite challenging among BIMP-EAGA economies (with the exception of the Philippines) and East Asian examples, such as Japan and China, show hybrid public-private power system models remain limited.

Infant industry arguments are among the most popular protectionist policies used among ASEAN members (Chandra, 2016) and countering liberalisation recommendations (*Academic 6*). Infant industry arguments are employed among ASEAN members, as well as strategic industry arguments geared towards protecting particular, strategic industries (See also Chapter 2 for brief discussion on this phenomenon in the context of the developmental state model of economic development). In the case of EMI and power sector integration, resistance at the national level is sometimes attributed to protection of strategic industries, notably those related to energy (*Academic 6; Academic 10*). 2018 Electricity tariffs in Indonesia, for example, are the most competitive for consumers (just behind Malaysia) among BIMP-EAGA economies and the single most competitive for industries, with Malaysia closely behind (Digital Energy Asia, 2018). 2018 prices in the Philippines are the highest for both consumers and industries (Yokota & Kutani, 2018) among BIMP-EAGA economies. In

countries like Indonesia and Brunei, where the links between industry and government are very strong and opaque, such arguments are more widely visible in current power sector structures. In addition, protection of national subsidies and resistance to fully liberalise the power sector can also be naturally linked to industry protections and economic nationalism.

Protectionism and economic nationalism have limited the success of the AEC and other sub-regional community building initiatives (See: Chandra, 2016; Hutt, 2017; Jones, 2016). This has been explored by experts in a variety of topical areas, but is quite obvious in sub-regional energy initiatives, including the APG and TAGP *(Academic 6).* A number of experts from the Murdoch School<sup>69</sup> of critical political economy note how ASEAN reforms can redistribute and potentially disrupt established power structures and are thus met with opposition domestically (Rodan et al., 2006; Chandra, 2016; Jones, 2016; among many others)—this author has viewed these same dynamics via interviews and market factor analysis introduced in chapters 4-5. As a result, it appears that shielding of established power structures can compromise ASEAN initiatives, leading to protectionist domestic policies. According to *Academic 6:* 

this opposition nationally has spill over effects that impact both regional and national liberalisation, market reforms, EMI, and the progress on the APG.

Proposed reforms to national power sectors and economic structures pose similar potential challenges to established power structures within domestic economies (*Academic 21; Academic 26*), thus impacting sub-regional policy making priorities.

The preponderance of economic nationalism among ASEAN member economies, and close relations between business and government, aligns with the common East Asian developmental state practices outlined in Chapter 2 and Chapter 5. The East Asian developmental practices of state-led development programmes and export-led industrialisation are complimented by public-private partnerships, many of which are still around in Southeast Asia today (Indonesia, Malaysia, Philippines, and Singapore are among the most well-known Southeast

<sup>&</sup>lt;sup>69</sup> The Murdoch School is named for the critical political economy tradition that developed from the Asia Research Centre at Murdoch University in Australia. This school of thought argues that institutions are highly influenced by power and resources that are present within both state and civil society and that as a result certain interests are preserved within institutions, closely aligning with political and social ideologies and agendas of those in power (Hewison et al., 1993; Jones, 2016). This school of thought has been widely used to critique development and progress in Southeast Asia.

Asian examples due to the suspected role of corporate-political relationships and oversight in their experience and response to the 1997 East Asian Financial Crisis).<sup>70</sup> Chapter 2 discussed how developmental state practices created political environments agreeable to non-state actors and therefore solidified strong state-business relationships Jones (2016); these strong relations have in turn nurtured lasting interdependent relationships that are widely present today in national energy markets in BIMP-EAGA and similarly reflected in the BIMP-EAGA EMI Scorecard in section 5.3. This author posits that these competing interests are in part to be blamed for APG progress, a sentiment echoed by Shi (2015, 2016), and reflected in Brunei Darussalam and Indonesia, where interdependent business-political relationships in the energy sector encourage business-as-usual. Just as ASEAN members have slowly progressed the AEC, so too have they slowly progressed the APG (Jones, 2016) and subsystem EMI development. This is not because there are no incentives, as has been demonstrated, but because competing national interests and protectionism have presented road blocks to national reform.

The challenge of national interests does not just impact protectionist economic policies but also perceptions about energy security and thus prioritisation of interests. This issue crosses over into the follow subsection and will also be discussed in the discussion section of this chapter, however it is worth noting here ASEAN's ability to respond to this challenge. There may be room for ASEAN to play a role in education at the national level—according to three interviews (*Former Government—Informant 1, Former Government 17, Academic 26*) there is pushback at the ministerial level to enact reforms that are perceived to go against national energy security interests, and ASEAN has the unique opportunity to provide increased education for ministers and government offices where delays may be taking place at the national level.

As a result of the national level EMI barriers analysed throughout this research, sub-regional considerations of national interests and motivations must, therefore, include regard for the various forms of power at play within individual ASEAN member states. The author discussed with *Former Government— Informant 1,* at great lengths whether the UNFCCC concept of CBDR (first introduced in Chapter 1) might be a relevant solution to regional and sub-

<sup>&</sup>lt;sup>70</sup> For further discussion on the intricacies of policy responses to the 1997 East Asian Financial Crisis see Drysdale (2000), which covers the relationship between the state and corporate sectors in East Asia with a number of relevant Southeast Asian examples.
regional market considerations. CBDR differentiates climate responsibility based on the acceptance that different countries have different capabilities in their responses to pressures on the global climate system (Pauw, et al., 2014). *Former Government—Informant 1,* posited that a similar method could be proposed for power sector reform across ASEAN EMI projects. Following this line of thinking, subsystems identified by ASEAN as priorities areas, such as BIMP-EAGA where conflicting national interests are at play, will require more development and capacity support in order for the required reforms to occur. This sort of CBDR-EMI approach, as this author has dubbed it, could aid in subsystem development via incremental progression of sub-regional interconnections and complement practices already in place.

One option for addressing the challenge of ASEAN EMI governance considering these national versus sub-regional dynamics is to dispense with the ASEAN practice of consensus-based decision making for majority voting (*Former Government—Informant 1;* supported by the work of Albert, 2017). This is different from a two-speed or multi-speed approach among ASEAN member states, whereby a sub-grouping of ASEAN economies proceeds with reforms, while other sub-groupings don't. In this example, the subsystem approach to electricity market integration, whereby interconnections take place among a smaller group of countries, progressing to larger groupings gradually until the entire system is included (refer to Figures 2.1 and 4.1), could also be utilised in the reform process. This could include progression to other reforms as well and would allow subsystem to move at their own speeds.

Additionally, strengthening the ASEAN secretariat and leadership within it would further organisational progress and encourage adaptation to political and diplomatic changes regionally and sub-regionally (*Multilateral Organisation— Informant 2*). Grouping ASEAN issues in security vs territoriality decision making would simplify some of these issues (*Former Government—Informant 1*). This specification would come with the stipulation that consensus is required for type 1, while majority vote is used for type 2. This would in effect remove the challenge of domestic autonomy to mutually shared problems such as climate change but maintain a reverence for independence and non-intervention when it comes to sovereignty issues. According to *Former Government—Informant 1* and *Multilateral Organisation—Informant 2* this practice would be achievable based on their own experiences with ASEAN governance, and indeed could be complemented by current ASEAN issue clustering.

Additional expert interviews echoed similar reform suggestions, particularly in reference to consensus building and lack of movement on energy and environmental issues within ASEAN (*Think Tank 15; Former Government 17; Academic 27*). In particular, *Former Government 17* advocated for a restructuring of ASEAN participation, in which countries who are keen engage and act on certain issues, while non-interested or unable member economies participate when they are ready. This recommendation is also reflected in the literature, and historically was practiced behind the scenes among ASEAN members (Albert, 2017; Nair, 2016). In the case of climate and energy policy this recommendation reflects previously mentioned CBDR practices and is therefore complimentary to current global policy expectations for developing economies.

These ASEAN-wide governance challenges are commonly debated across the field of East Asian studies and across the literature (ADB, 2008; Etsy and Ivanova, 2002; Guan, 2004; Heilmann, 2015;), however, their prior application to real-world observations about power sector development and sub-regional and regional EMI are unique to the interviews carried out in this research. Additional governance challenges, particularly at the national level of governance making, are discussed in the following sub-section, contrasting regional versus national level interests in the context of sub-regional EMI and increased interconnections.

## **6.2.b National Interests**

As national political environments become more complicated, so too do the decision-making processes by ASEAN member states. According to Bilahari Kausikan (2017), former permanent secretary of Singapore's Ministry of Foreign Affairs, sub-regional and national concerns are constantly evaluated in decision making processes among ASEAN members, and "calculations of interests" are constantly being weighed (pg. 5). This supports expert interviews and the observation that energy security receives much different consideration than environmental security (*Former Government—Informant* 1, *Former Government* 17, *Former Government* 27, *Former Government* 28). It should be noted here that each expert interview who echoed that sentiment is a former government representative.

In addition to national perceptions regarding energy security considerations, another example is the financial crush felt by many developing economies and the need to allocate resources carefully (*Academic 21*). In the case of Indonesia, for example, *Development Bank 25* questioned whether or not cross-border interconnection measures should be taken when simple household electrification and off-grid measures could be employed to bring electricity to

the remaining 23 million people without access (supported by IEA, 2017a). This expert also pointed out that reliable access should be of higher focus of national governments, thus preventing intermittency and strengthening reliability of those who do have access. Further, if a national government perceives the strengthening or expansion of access as an 'either or choice', national concerns inevitably win out (*Development Bank 25*). Long term, net costs savings of subregional power sector integration are estimated at 20.9 billion USD (3.0%) for 20% of demand met by trade and 29.0 billion USD (3.9%), if 50% of demand was met by trade (Wu et al., 2012, p. 4). However, as is similar in other investment scenarios, if upfront costs are perceived to be prohibitive, investment will not occur or take much longer than cheaper options (*Business 31*).

Additional investment concerns colour priorities at the national level. One interview subject *Business 31*, with experience on large- and small-scale investment projects (energy and non-energy), spoke to contradictions in the national business environment as an illustration of the push and pull that is present in many infrastructure projects in emerging and developed economies alike. This expert argued that the national business environment must be enhanced in order for some emerging economies to be an attractive investment environment. Competing powers in these markets, for example state-business relations in Indonesia, can make additional foreign investments in the energy sector difficult where financial risk is perceived to be too high given uncertainty over agreements that may not serve prevailing political or national business interests. Unfortunately, results vary depending on global markets and geopolitics, among other, national issues, and predictions of outcomes for improving investment environments tend to vary in accuracy (*Business 31*).

There is some contradiction between interview subjects and the literature on economic factors' influence on national level governance of EMI. According to Wu (2016), in addition to economic factors, political factors remain the largest driver of EU power sector integration, and a limit that ASEAN must come to terms with. Here Wu argues that political will, coupled with energy security concerns, can lead to policy solutions that coincide with and include climate change commitments, hence pushing EMI. However, this is slightly contradictory with a very specific piece of feedback from interview subjects (*Academic 26, Think Tank 12*). If the national level governments perceive energy security concerns to exist separate from power sector integration, energy security will win out in terms of policy focus. As a result, national measures required for international power sector integration will take a back seat to national measures

that are perceived to be more relevant to increasing energy security. For example, if an argument was made by national level public and private actors that oil and gas subsidies are necessary in order to provide secure, independent energy resources, those subsidies may not be removed in favour of market liberalisation and subsidy support for clean energy integration. Another example would be the importance placed on energy poverty versus energy security. *Academic 26* argues that if a national government perceives that coal, a cheaper resource cost upfront, is able to provide affordable access to electricity now, while renewables require a higher cost upfront, and further development in order to integrate into a national grid, then coal may be given priority.<sup>71</sup>

This is not to say that Wu's point isn't correct, clearly, as a leading scholar and researcher in this area Wu's analysis of EU lessons for ASEAN holds ground (and are explored in more detail in Chapter 2, Cross-Border Electricity Trade: Conceptual Approach & Review of Literature) when global examples of EMI are introduced. In this case the need and circumstances within ASEAN at the national level has altered the environment just enough that notions of strengthening political will are not enough. Jones (2016) notes that political will is often a buzzword used to represent slow moving ASEAN processes like the AEC and, in the case of this research, relevant to the APG as well. In fact, political will is an issue because regional agreements are driven by political not economic incentives, therefore members engage or join because they don't want to miss out on strengthened alliances, increased sub-regional or regional power, or perceptions of being left behind (Jones, 2016). Academic 26 echoed these concerns about political will at the national level, noting that progress is not slow within ASEAN itself, but within national electricity markets and the necessary regulatory changes that must be enacted in power sectors.

In Chapter 5 analysis of market factors confirmed that coordination, harmonisation and standardisation are important reforms needed in order to achieve the O&PCRB EMI Criteria. According to *Multilateral Organisation— Informant 2* and feedback received at *Government Conference 12*, these standardised practices can be better coordination between the national governments and ASEAN, streamlining national reforms so that they coincide or support ASEAN APG goals. In addition to this coordination, it is the author's

<sup>&</sup>lt;sup>71</sup> This example is supported in IEA literature and research, whereby coal is projected to remain the dominant fuel source in Southeast Asia despite cost reductions to renewables (IEA, 2017a, 2019). In a 2019 report produced by REN21 similar conclusions were reached for select Southeast Asian Countries.

position that better messaging and education at the national level must also occur, otherwise these reforms may never occur in the first place. Building political will through education, messaging, and financing is preferable to empty agreements.

An additional unique theme that emerged from interview subjects that work directly with ASEAN was the intersection of energy security and trust in crossborder electricity trade groupings (This was briefly mentioned in the prior section in relation to ASEAN support for ministerial level education. It should also be noted that trust will be more specifically explored in the following section, however, here the focus is on national level concerns regarding energy security). Former Government 27 spoke of a general perception by outsiders that within organisations of voluntary membership issues of trust would be limited. However, this interview subject has found that instead, behind the scenes there are questions of trust among and between ASEAN members as a result of national energy security concerns; Former Government—Informant 1 echoed these remarks about behind the scenes challenges, as did *Former Government* 17. Based on interviews the author has concluded that a common EMI concern among ASEAN members is whether or not they can trust their electricity access to a neighbour, and whether or not they are seceding control over their energy access to another actor who may later hold that energy for ransom. This poses the challenge of whether or not voluntary compliance—a staple of ASEAN membership—outweighs national governance concerns, and if not, what alternative do ASEAN members have. There is no evidence that these perceptions are valid in ASEAN, as there are no examples of electricity services being withheld from transit neighbours in the sub-region (Shi, 2014).

According to *Multilateral Organisation 18* and *Academic 13* power imbalances in the electricity sector are a concern for participating EMI economies. There are varying levels of grid preparedness (ability of current grids to integrate and adaptation of current infrastructure for further renewable energy integration into the existing grid) and flexibility in current power systems within the region. This means some economies, such as the Philippines and Brunei Darussalam, will have a stronger ability to integrate their grids, but may be reticent to integrate with neighbours who have very low grid reliability, such as Indonesia. Additional imbalances include access to resources, presence of national policies that enforce investment and logistical rules, adherence to climate policy, and availability of natural resources (*Academic 13*). As discussed in Chapter 5, where bilateral cooperation already exists EMI agreements will be easier to develop,

and so integration with strong partners may be favoured. Ultimately, countries like Myanmar and Cambodia, who have the most undeveloped power sectors and lowest electrification rates in the sub-region (*Multilateral Organisation 18*), may struggle to integrate at the same level as a result of some of these imbalances— impacting the progress of the APG and subsystem development.

In this section sub-regional vs national level dynamics been discussed, drawing attention to EMI challenges that fall within either national or sub-regional governance interests and perceptions. What the prior two subsections have exposed is that trust further complicates national versus sub-regional interests in ASEAN EMI efforts. These complications include expectations, conflicts between sub-regional and national governance concerns, finance challenges, and perceptions of energy security. The following section will pick up on the challenge of trust, examine this issue among ASEAN members, expanding on where institutional membership is supposed to help reduce trust and uncertainty, and where sub-regional and national dynamics may impact the outcome for the APG and wider East Asian power sector development.

## **6.3 Trust Among ASEAN Members**

In analysis of interview data it became clear that perceptions of trust are a major factor in cross-border electricity trade and EMI, not just in East Asia or the subregion of Southeast Asia, but also globally. Even under the guise of a sub-regional, voluntary organisation perceptions of trust colour agreements and alliances. Past trust concerns can impact future agreements, and issues of trust can persist for years. In some cases, trust falls along lines of shared histories, territorial or colonial occupation, language, or development status; in other cases, trust stems from global, regional and national perceptions of trustworthiness based on multiple factors. In interviews with one former US government official, Former Government—Informant 1, who played a role in ASEAN energy policy development over a 30+ year career, trust was explored as a challenge between national perceptions and ASEAN goals. A different organisational leader turned academic, Academic 10, whose views are widely represented in the literature, echoed these concerns, but was troubled more specifically with energy security perceptions and dispute settlements in cross-border interconnections. Once identified these views jumped out among other interviewers as common barriers to EMI progress in ASEAN (Think Tank 5, Think Tank 7, Think Tank 11, Think Tank 12, Multilateral organisation 23).

A second related issue that emerged is how ASEAN has addressed perceptions of trust in EMI relationships, notably by engaging in cross-border electricity trading relationships on a bilateral versus multilateral scale. The topic of bilateral relationships was first introduced in Chapter 4, illustrating the development of APG interconnections and their gradual progression. In this section that same topic will be explored in relation to its role as a tool to overcome trust barriers and engage in agreements on smaller, more realistically perceived scales.

First, a very brief background on trust that is relevant to the subsequent data: the study of trust in international relations is widely conducted but very conflicted. A wide breadth of information is available on how trust impacts relations among nation states, international institutions, and multinational enterprises, however, a common definition is difficult to pin down.<sup>72</sup> Finding a definition of trust often means narrowing down what trust is not (Hoffman, 2002; Considine, 2015). Hoffman (2002, p. 379) sums up a variety of different definitions of trust across fields related to international relations as being a combination of trust expectations, perceptions, obligations, and risk taking. Considine (2015) takes issue with commonly used definitions of trust, pointing to the fact that often when trust is used in politics, the phrase includes some sort of interpretation of policy and political expectations; trust is never simply about confidence or risk, it is about political agendas, and thus loaded beyond a simple definition. In Considine's work, they also point to how terms commonly used in association with trust frame the questions and answers asked about international political events, and therefore our answers as researchers become biased. In this case of this research it is the author's opinion that their perception of trust in Southeast Asia has been coloured by their time in China and their understanding of Chinese economic influence in the region. Therefore, this author, in line with Considine's recommendations, recognises that there is some bias in their interpretation of trust, its impact on EMI, and the relevance of China's role (for further discussion on research bias please refer to Chapter 3: Research Design and Plan).

In terms of governance and international relations, trust intersects with power and the balancing of interests. Hoffman (2002, p. 394) makes links to the

<sup>&</sup>lt;sup>72</sup> See also: Baier (1986); Barber (1983); Brenkert (1998); Coleman (1990); Dasgupta (1988); Deuttsch (1958); Gambetta (1998); Holzner (1973); Keohane (1984); Sztompka (1999); for a variety of viewpoints and disciplines that utilise trust in theoretical and analytical research.

complicated relationship between trust and power, noting how power and opportunism can influence trust, or lack thereof:

Very simply, opportunism destroys the expectation of trustworthiness, i.e. the belief that trustees will advance and/protect the interests of trustors in spite of their capacity to do harm. Absent basic confidence that other will do what is right, trusting relationships cannot be developed.

Prantl (2014) echoes the sentiment that power dynamics deserve more attention in international relationships and are vital to understanding politics in a changing international order. Power, or perceptions of power, have the ability to indirectly impact regional and sub-regional relations and markets. Joint responses to shared concerns, such as those used by Northeast Asian economies, encourage regional integration and counterbalance power dynamics (Atanassova-Cornelius, 2017, p. 7), shifting focus to cooperation and integration as opposed to focusing on regional differences in responding to such concerns. Reinke de Buitrago (2009) similarly agrees, positing that where threats are perceived, East Asian states have overcome them by engaging in multilateral institutions where trust is built through education and "confidence-building measures...[that] reduce uncertainty and allow actors to trust one another more" (p. 739). When economies engage in multilateral governance organisations, such as ASEAN, trust is fostered; but we must also consider issues of trust within the organisation itself and how these may impact joint-governance initiatives.

Friendship is commonly used to study and analyse bilateral relationships in the international system (Oelsner & Vion, 2011) and is relevant here when discussing the challenges of trust in ASEAN. ASEAN itself has provided all of the components of friendship<sup>73</sup> to its members, creating an institutional framework for which friendship is rewarded and used as a political and economic tool. In international relations friendship represents the antithesis of enmity. Based on these interpretations of trust and friendship it is logical to deduce that an absence of friendship and predominance of enmity would lead to distrust, and a reticence to engage in cooperative agreements or relations. It is reasonable to posit that trust, however, exists on a spectrum like many social and cultural

<sup>&</sup>lt;sup>73</sup> Oelsner and Vion (2011) combine multiple disciplinary approaches to friendship in order to frame its role in the international system. This includes the institutionalisation of friendship, which requires the production of a common good or shared sense, the formalisation of international relations and institutions to guarantee sovereignty, structure mutually beneficial relationships, promote cooperation, foster social justice, and result in the creation of order in a chaotic world system (p. 5-6).

phenomenon, and therefore an absence of distrust among partners does not mean 100% trust exists between partners, and vice versa.

While ASEAN operates with relative success in reaching agreements and producing data on shared issues, how to address them, and progress made, as well as incremental assessments of projects, there is concern about trust among members. This has been reflected by interview subjects and in literature on the limitations of cross-border electricity trade and EMI in the sub-region (Pacudan, 2016; Shi, 2015; Former Government—Informant 1; Academic 26; Former Government 27). How, a member economy might question, will disputes be settled should a member hold electricity hostage in the event of a conflict? How can members ensure a potential conflict won't lead to electricity shortages? What if a member economy decides to stop electricity trading arbitrarily and with no warning? These are common questions in cross-border electricity trade literature and reflected in expert opinions on the challenges associated with cross-border electricity trade expansion and regional power sector reform, not just in ASEAN but elsewhere in the world as well (See Chapter 1 for further global EMI examples). These member economy concerns are reflected across a variety of experts interviewed inside and outside of the region.

According to *Think Tank 11*, trust within ASEAN is fraught. Member countries, while joined together in numerous economic agreements, do have histories of conflict that have overshadowed policy development in the past. In addition, placing dependence (whether full or in part) of electricity access on a neighbouring country does raise security of supply concerns at the national level *(Development Bank 5)*. In the case of an economic or political conflict among members, *Academic 26* said a common concern among members—expressed at the national level—is could cross-border electricity trade cease or access be restricted in order to be used as a bargaining chip? An additional challenge to this point according to the same expert, is that "often ministerial level [political] will puts priority on security concerns over energy poverty concerns" (*Academic 26*). Following this argument, if a ministry had the perception that trade of excess capacity was hindering national security, trade may be cut-off in order to assuage security of supply concerns.

In addition, bilateral agreements that are expanded incrementally to include more and more neighbours, or joined with other, local bilateral agreements, are more realistic than larger, multilateral agreements given the variety of concerns that must be addressed among partner nations, including coordination, deregulation, harmonisation and standardisation among others previously

discussed (*Multilateral Organisation—Informant 2*; supported by IEA, 2019a, pp. 153-159). Also, the physical needs for increased trade and EMI are high—physical infrastructure connection is required first (grid to grid), and then institutional coordination, management and regulations can be established. Without actual grid connection electricity cannot be traded. Referring back to Chapter 5 analysis of current interconnections in the BIMP-EAGA subsystem, physical connections in the sub-region are limited at the moment, and updates are needed at a local level before sub-regional connection can occur. Finally, liberalisation of national electricity markets, through increased competition and standardised rules of practice and operation, are often identified as equally necessary for physical connection to be successful (see section 5.3-5.4). ASEAN, however, can help with this through funding, policy support, and information exchange—if trust among partners can be established. Here the role of bilateral agreements is further relevant, this time in augmenting against national level concerns and progressing slowly given previously proposed CBDR-EMI approaches.

In ASEAN aspects of trust may also be related to the concept of saving face, introduced in Chapter 3 when discussing working in East Asian contexts. *Former Government—Informant 1* provided anecdotes about their experience working within ASEAN, and questions of trust among members—in meetings members tend to agree with each other but after meetings and in one-on-one settings will express dissatisfaction or concerns:

They [ASEAN] like to avoid conflict and have harmony and consensus. Issues where they can't get amenity, they avoid and just don't discuss. For instance, with India and Pakistan if there is conflict they discuss it. With ASEAN they just avoid it. It's a difficult environment to get tough issues addressed. (Former Government—Informant 1)

In addition, saving face, instead of making conflict, is a commonly observed trait of politics within ASEAN, and is reflected in the voicing of concerns regarding trust behind the scenes (Guan, 2004). Where conflict could result in embarrassment, partners avoid voicing conflict in meetings. Take for example this anecdote by *Former Government 17*, when describing the challenges of energy policy consulting in ASEAN power sectors:

...The job is so challenging. There are times that it feels like we are constantly being told two or three different things—no one wants to lose face! Everyone wants to avoid conflict and as a result my team experiences a lot of conflict behind the scenes trying to figure

# out what our partners want. It's very hard. (Former Government 17)

In this sense, saving face is the prevention of embarrassment via conflict or disagreement, and therefore requires the feigning of harmony when in a group, and reversal of that agreement behind the scenes. This false harmony, therefore, becomes untrustworthy when no disagreement or conflict is ever truly hashed out, and complete agreement is never reached. In the ASEAN sense this means, for example, agreeing to climate targets or subsidy removal in a group, but refusing to prioritise these issues or even work seriously towards the agreed upon goals at the national or local level. This can also, according to Former *Government*—A27, mean that targets among ASEAN members are only as strong as individual members national priorities, and therefore agreements "can't always be trusted at face value". Here the intersection of national and subregional priorities is an interesting addition to the challenge of trust in policy making. If this is true, that sub-regional agreements will only ever be as strong as national priorities, then if the two don't fully align structural economic and political change may not happen at the ambitious level regional groupings like ASEAN aim towards. This does not bode well for the APG or subsystem-tosubsystem development, where trusting relationships may be lacking and EMI capabilities mismatched based on differing levels of power sector development.

### 6.3.a Mistrust Moving Actors to Trust

Where trust is viewed as a foundational requirement for personal relationships, government and governance, and human success (Reinke de Buitrago, 2010), mistrust is also a factor in moving relationships forward. Mistrust is defined as the absence of trust,<sup>74</sup> which Carey (2017) bases on the notion that "familiarity is insufficient ground for trust" and therefore expectations about behaviour cannot be based on familiarity in relationships (p. 7). In his ethnography on the absence of trust Carey posits that mistrust moves politics and institutional relationships forward, and should be equally considered when studying trust, as opposed to its general absence in the majority of trust literature (p. 5). The concept of mistrust, while not widely applied outside of Carey's work, piqued the author's interest based on feedback during interview data analysis. As a result of

<sup>&</sup>lt;sup>74</sup> Distrust, however, would be a lack of trust as a result of conflict, disagreement, or experience. As referenced earlier in this chapter, if trust existed on a spectrum, one end would be absolute trust and the opposite end would be distrust. Mistrust is harder to place on this hypothetical spectrum, however, as it is a lack of either but familiarity with the subject to which trust could be applied.

expert suggestions, the author began exploring how this concept might be applied to ASEAN and the trust challenges highlighted among interview subjects.

In Carey's conceptualisation of mistrust he examines politico-organisation structures as a response to mistrust—they provide a solution to uncertainty and fear, a place for structure to be built out of uncertainty (p. 12). In the case of ASEAN this characterisation is quite realistic; the creation of ASEAN in 1987 was in part a response to lack of structure for post-colonial states in Southeast Asia (Zhao, 2016), part response to fear of the spread of communism and rising tensions in the region (Albert, 2017). Where bilateral relationships between the founding ASEAN members did exist prior, mistrust was present between all countries—there was no prior experience of cooperative organisational development based around these specific, five founding members and the formation of ASEAN provided a means for structure in a tumultuous sub-regional environment. In the context of ASEAN, trust was built based on mistrust and the need for a balance to the anarchy of no relationships or politico-organisations in the sub-region. Regional institutional governance was built upon the existence of mistrust, and the formation of a sub-regional institution to address shared governance challenges was born; in this sense, mistrust is a useful addition to governance, and in fact aids in the building and maintaining of institutional governance relationships as long as distrust does not also exist between all participating parties. Using this conceptualisation of mistrust, we can go even further with ASEAN, applying mistrust to cross-border electricity trade; this can be further applied to BIMP-EAGA and governance challenges highlighted among interview subjects.

There are few (if any) recent experiences of ASEAN countries using energy access against one another in disagreements. There are few (if any) examples of cross-border interconnections being hijacked by partner countries. There are few (if any) examples of ASEAN members using energy as a political tool in general. Therefore, there is no distrust in the power sector, but there is mistrust (no subject interviewed was aware of a clear example where distrust had been fostered via electricity trade conflict). Mistrust in power sector reform, climate change mitigation, and EMI in ASEAN can, based on an expansion of Carey's original conceptualisation, move these mistrusting countries together in cooperation as a response to the lack of structure in these areas. In fact, that is already happening as illustrated by the various EMI agreements built into ASEAN's APG initiative and the gradual progression of bilateral to multilateral cooperation. Mistrust is, in a sense, the invisible hand that is being used to tie

BIMP-EAGA nations together in climate change, energy policy, and EMI cooperation at the sub-regional level.

- 215 -

Applying this logic to wider ASEAN relations, outside of subsystem cooperation, there is opportunity for deeper EMI between national markets. While subregional and subsystem EMI is overseen by ASEAN and HAPUA, there has been little movement in coordinating national markets up to this point. As bilateral and multilateral EMI cooperation is expanded, the opportunities for success based on absence of distrust and presence of mistrust are positive. This can also further incorporate [national] market-to-market coordination, harmonisation and standardisation, criteria and reform needs established as necessary for EMI and cross-border electricity trade. While there are challenges of trust, distrust does not yet exist and mistrust may actually bode well for the long-term development of these subsystems, ASEAN's governance role, and the eventual creation of the APG.

## 6.3.b Bilateral vs. Multilateral Cooperation

Bilateral agreements emerged as another common talking point in interviews (*Former Government—Informant 1, Multilateral Organisation—Informant 2, Academic 10, Think Tank 7, Think Tank 11, Former Government 17, Former Government 27*), reflecting the breadth of ASEAN's own policy and research on EMI throughout the region (APAEC 2009, 2014). In the case of the APG, some progress is better than no progress—if trust, capacity, and national priorities are all barriers, moving forward where able is appreciated as "better than nothing" (*Academic 10*).<sup>75</sup> This is where bilateral agreements can play an important role in APG progress.

Among experts interviewed it is clear that ASEAN is moving forward with the APG and an integrated, sub-regional electricity market is the ultimate aim of this connectivity (every interview subject with expertise in ASEAN agreed upon this point). A significant limit to APG progress is, unfortunately, the pace of development (*Former Government—Informant 2; Multilateral Organisation—Informant 2; Think Tank 5; Think Tank 11; Think Tank 12; Academic 26; Former Government 27; Conference Event 2; Workshop 1).* According to these

<sup>&</sup>lt;sup>75</sup> There is quite a lot of research on bilateral and multilateral cooperation in trust literature and East Asian studies. However, as this research is focused on the IPE of cross-border electricity trade, the author has opted not to critique this approach, simply provide context to the utilisation of these two forms of cooperation in relation to regional power sector development. For more on bilateral and multilateral cooperation see: Low (2003); Akaha (2004); Qi (2009); Oelsner & Vion (2011); Choo (2014); Atanassova-Cornelius (2017); among others.

interviews, the initiative is moving quite slow as a result of common limitations identified in this research. While progress has been incremental, for example the grid-to-grid connection between Lao PDR-Thailand-Malaysia-Singapore (LTMS), the 100 MW power agreement is not enough to reasonably meet region wide or ASEAN specific clean energy goals (*Think Tank 5; Think Tank 11*). This raises the question, how is any trade possible given the barriers countries encounter? The example of LTMS power trading provides interesting insight into the dynamics of multilateral versus bilateral trade, and the difficulty in achieving either one. The following subsection examines three examples of existing trade in the region, highlight the fraught LTMS multilateral trading scheme and two bilateral trading relationships in the BIMP-EAGA subsystem. This analysis adds context to the difficulty ASEAN member states have experienced in achieving any cross-border trade relationship, while highlighting the mutually beneficial terms such relationships can provide.

#### Existing Interconnections

LTMS represents the very first multilateral project in the sub-region, originally initiated in 2013 (Owen et al., 2017) and representing ASEAN's attempt at progressing APG development incrementally (ACE, 2018b). Pilot projects for LTMS interconnection were further developed in 2014, transmitting electricity from Lao PDR to Singapore via existing infrastructure in the transit countries of Thailand and Malaysia (Andrews-Speed, 2016). Later successful iterations of the project developed without the inclusion of Singapore—The largest transmission to occur is the Lao PDR-Thailand-Malaysia interconnection, which was signed in 2017 to transfer hydropower from Lao PDR to Malaysia via transmission lines in Thailand (Atmo & Otsuki, 2018, p. 1). According to Owen et al. (2017) it is unknown why Singapore decided not to join this particular interconnection, although the authors point out Singapore's energy needs and landscape are quite different from the other three partners (p. 146), as is Singapore's power sector structure, which exclusively allows private generation to participate in generation and trade—meaning Singapore has no counterpart to join the LTMS inter-utility agreement (p. 154). It is not known if this is why Singapore did not join later iterations of the LTMS interconnection (Owen et al, 2017, p. 145-146).

Motivators behind the multilateral LTMS interconnection are closely aligned with broader ASEAN interconnection motivations—sustainability and energy mix goals, rural electrification, and financial gain (Owen et al., 2017; ACE, 2018b; Atmo & Otsuki, 2018). The needs of individual LTMS economies can be identified as follows (Owen et al, 2017): (1) all the partner economies have growing

electricity demand that needs to be met, motivating the search for alternative power sources to add to system reliability (p. 146, 169, 177, ). (2) In the case of Lao PDR that demand growth will put stress on the undersupplied domestic market and stress the need for expansion of reliable transmission and distribution systems for rural electrification and industry growth (p. 169). In addition. domestic government focus on export-oriented bilateral interconnection has established policy incentives for engaging with neighbours in interconnection projects (p. 170). (3) In the case of Malaysia, energy mix motivations have encouraged diversification of power sources, as domestic renewables growth is expected to remain small for the next 10-15 years as a result of prevalent fossil fuel subsidies and cost perceptions (p. 177); In addition, in order to increase rural electrification of its island communities the Malaysian government has placed importance on the growth of bilateral interconnections between Malaysia and its neighbours (p. 180). (4) With regards to Thailand, electricity imports allow the country to augment its reliance on natural gas (p. 188) and diversify its own energy mix with more renewables; In addition, acting as a transit country reaps economic benefits on its own (p. ). (5) Singapore is motivated by its energy security and energy mix concerns, seeking out diversification via imports (p. 194). In all cases the domestic motivations also benefit the sub-region as a whole—development of a regional power trading system has benefits for both the importers and exporters (p. 148-149).

Two particular examples of successful bilateral interconnection are the Thailand-Lao PDR interconnection and the Malaysia-Indonesia interconnection. The Thailand-Lao PDR interconnection is one of the oldest interconnections, having been established in 1972 (IEA, 2015, p. 13). Via this interconnection both countries gain from each other—First, Lao PDR is able to export electricity to Thailand that is fueled by hydropower, offering Thailand the opportunity to diversify its energy mix; Second, in exchange, Lao PDR is able to not only gain financially but also technically, as it uses Thailand's developed transmission network to provide electricity to its own remote communities (IEA, 2015, p. 13). In addition, this relationship has allowed Thailand to act as a developer of power projects in Lao PDR, further growing the developing economies power system and improving infrastructure while deepening the two countries' continued electricity exchange relationship. Similar models of exchange are being employed by Lao PDR in the development of future exports to Vietnam.

The only successful BIMP-EAGA interconnection to date is the bilateral interconnection between Sarawak, Malaysia and Kalimantan, Indonesia. This

interconnection began formally exchanging electricity from Malaysia to Indonesia in 2016 (IEA, 2015). Kalimantan, Indonesia, like the rest of the country, is heavily reliant on oil fueled power generation. In 2012 the two countries signed a power exchange agreement for the length of 20 years (ADB, 2013). This project was made possibly largely by an ADB loan that funded the development of the Indonesian side's transmission infrastructure; Malaysia funded the expansion across the border and into Sarawak (ADB, 2017). The project, while benefiting Indonesia's energy mix with hydropower and gas produced electricity from Malaysia (Pacudan, 2015, p. 48), has also sought to solve the problem of power scarcity in this particular region of Indonesia—as a result of the Malaysia interconnection Indonesia has had the opportunity to increase its national electricity access with more reliable excess Malaysian power (ADB, 2017, p. 5). Malaysia, on the other hand, has gained economically from the sale of excess electricity. In addition, expansion of the Malaysia-Indonesia interconnections will allow continued improvements in efficiency and cost reductions—interconnections allow utilities to invest in larger and more efficient systems (Pacudan, 2016, p.49). Furthermore, this project represents an important political and economic expansion of the BIMP-EAGA subsystem's role in the APG and potential future expansion of this bilateral interconnection into a multilateral network that includes Brunei and the Philippines (ADB, 2013).

While the progress of the multilateral Lao PDR-Thailand-Malaysia 100MW transmission is an important development in ASEAN's long-term multilateral power trading goals, progress was slow and the removal of Singapore in later iterations of the project signal the difficulty in engaging in cross-border electricity trade on a multilateral level. Owen et al (2017) find that financing, reform of market structure, coordination, harmonisation, and standardisation of systems, and trust to be barriers to quicker LTMS development. This supports the author's findings regarding BIMP-EAGA and the relative ease of bilateral versus multilateral transmission, as well as the breadth of literature on the topic (See also: Footnote 54 on page 199; also the work of: Low (2003); Akaha (2004); ADB & ADBI (2009); Qi (2009); Oelsner & Vion (2011); Choo (2014); Atanassova-Cornelius (2017); and ACE (2018b), among others).

Experts interviewed also remain optimistic that ASEAN power connectivity will be accelerated as more bilateral agreements come on-line, and eventually these bilateral agreements will increase trust and form the basis for an ASEAN electricity market (*Multilateral Organisation—Informant 1; Academic 10; Multilateral Organisation 18; Academic 26*). This logic of progression is supported by Choo's (2014) characterisation of the unique, East Asian style of institutionalism whereby institution building leads to trust as opposed to the common western model which is opposite; in the case of the APG bilateral agreements lead to trust, which leads to further interconnections and additional partners (*Former Government—Informant 1; Multilateral Organisation—Informant 2; Workshop 1*). This complements ASEAN's subsystem development, and the ASEAN model for expansion based on multiple subsystems' gradual linking (See Figure 4.1 in Chapter 4), as well as the ADB's findings regarding the primacy of bilateral agreements in the progression towards multilateral, sub-region-wide interconnection (ADB & ADBI, 2009, p. 35).

In this section trust has been framed in the ASEAN context—this includes examining trust in ASEAN among member economies and exploring how this relates to cross-border electricity trade, EMI, and the development of the APG. In addition, this section has covered the challenge of bilateral versus multilateral EMI, demonstrating the contradiction between bilateral engagement and ASEAN wide, multilateral EMI goals. This has also included expert interview testimony and literature to support the findings that bilateral EMI cooperation is a steppingstone to eventual multilateral integration. In the following section trust of outsiders will be introduced in the context of ASEAN engagement with China and Chinese EMI investment. Included in this will be an analysis of APG needs and interlinked Chinese connectivity goals.

## 6.4 Trust of Outsiders

Trust of outsiders is an obstacle in many relationships, between nations, corporations, communities or other. In East Asia trust of outsiders is largely attributed to historical memory. This is principally associated with Chinese-Japanese relations (Drysdale & Zhang, 2000; Togo, 2008; Prantl, 2014; among many others); however, Southeast Asia's historical memory of China is that of an often benign but sometimes aggressive empire that influenced regional politics and sought leadership over the larger East Asian region—a perspective that is reflected in Chinese foreign policy today (Weatherbee, 2009).

In Northeast Asia (China, Japan, the Republic of Korea, the People's Republic of Korea, and sometimes Mongolia) trust has played a large role in regional relations. Atanassova-Cornelius (2017) argues that the duelling factors of competition and cooperation have created a unique practice of trust building in Northeast Asia; mutual distrust has been exacerbated by geopolitical tensions, but ultimately economic interdependence and shared security concerns have

increased trust and fostered a shared understanding of the value of trust for regional stability. Non-traditional security (NTS) concerns in Northeast Asia have driven trilateral (China-Japan-Republic of Korea [ROK]) cooperation, necessitating a joint response to shared challenges. In the case of energy and power sector reform climate change is the clear challenge, requiring joint and individual action.

Following Choo's characterisation of institutional development first, followed by trust, it is logical to posit that trust-building between ASEAN +China will follow after increased institutional cooperation. NTS issues, particularly related to power sector reform and climate change mitigation, are perhaps the most important concerns in the wider East Asian region, representing the amalgamation of both environmental and energy security, both of which are truly transnational regional issues. In the case of EMI and power sector reform suspicions abound due to power sector imbalances and perceptions about energy security, as discussed previously in this chapter. Suspicion of outside influence and control over a country's resources or electricity is similarly evident, particularly in the ASEAN-China relationship. This has been made evident by China's global Belt and Road Initiative (BRI). The following subsections will bring BRI investment into the discussion about trust as a barrier to ASEAN's APG. This will include an introduction to BRI, an examination of its potential contribution to the APG, and also an analysis of its shortcomings and concerns with trust. Finally, this section will examine ASEAN-Chinese trust and how it may be a barrier to BRI assistance with development of EMI in Southeast Asia.

## 6.4.a Chinese Investment through the Belt & Road Initiative (BRI)

A dominant theme that emerged in conversations about trust was expert interest in BRI as one potential solution to the shortcomings and difficulty of realising power sector integration in ASEAN through the ASEAN Power Grid. This is where similarities between BRI Infrastructure investment and ASEAN Power Sector Integration began to emerge. In initial interviews the most common theme was BRI & Trust. 6 out of 10 respondents mentioned both of these issues in a variety of forms across multiple discussions.

BRI refers to China's infrastructure investment project, announced by Chinese President Xi Jinping in September 2013. BRI is made up of two avenues of investment, China's [overland] Silk Road Economic Belt and the 21<sup>st</sup> Century Maritime Silk Road, collectively making up the largest infrastructure investment project in history (Chatzky & McBride, 2019). If realised, the two routes of the BRI would connect China with Europe via Central Asia, Southeast Asia, South Asia and Africa. Infrastructure ventures that include railways, ports, roads, accommodations, power projects, etc. would boost China's impact in the regions developed and encourage economic development among emerging economies along the BRI routes.

The BRI is an ambitious project when looked at from above—over 10,000 projects spread across three continents (South China Morning Post [SCMP], 2019), ranging from \$1-8 trillion USD (Hillman, 2018). In 2014 President Xi announced the creation of the Silk Road Fund, a "medium to long-term fund" which would contribute \$40 billion USD to projects that promote connectivity along the BRI routes (Silk Road Fund, no date). Additional BRI funding comes from a combination of Chinese, state-owned commercial banks, development banks, sovereign wealth funds, and foreign exchange reserves (Deloitte, 2018).

A challenge with examination of BRI is in part the distinction of what BRI actually is-state-sponsored financial investment with an expectation of return or government aid (Wharton Business Daily, 2019, p. 2). Hillman (2018) points out the difficulty in making this distinction based the nature of BRI financing—BRI financing is Chinese government supported aid with the expectation of return via the winning of contracts by Chinese companies (p 2-3). BRI contracts in the Reconnecting Asia Database (2018), which tracks BRI funds and projects, found that 89% of all contracts were held by Chinese companies (Hillman, 2018, p. 4). The general complication of identifying BRI projects and whether they are investment or aid is in part due to the process BRI investment goes through: State sponsorship of BRI investment leads to capital provided by Chinese financial institutions, these Chinese financial institutions give loans to partner countries, but Chinese companies win the majority of the contracts required, which means partner countries in turn hold the debt-if the investments returned don't pay off this debt then China repossesses the project in what becomes "foreign appropriation of energy sector assets" (Lowder et al., 2020, p.6). Chinese state sponsorship also leads to support for difficult to complete projects—lending credence to projects that otherwise might not find outside investment. This is not to say that outside investment of the APG does not exist it does—but in this context BRI is relevant as an oft cited example by interviews of a potential investment source for APG projects; and it is a big potential source.

Along the BRI routes the World Bank (2019) has identified 70 countries, however, the BRI is open for any country to participate in. For the 70 countries used in World Bank analysis the investment is estimated to be closer to \$575

billion USD; of this, 46% of BRI investment is in energy and electricity, the largest focus of BRI investment. 34% of all BRI investment is focused in East Asia and the Asia-Pacific. While qualifying projects that were underway prior to the creation of the BRI are eligible for funding and are in some cases lumped in with BRI projects that are being scoped,<sup>76</sup> the World Bank has found that the majority of investment is in projects that are in the construction or planning stages (p. 45).

While the investment levels of BRI range depending on the source reporting and the types of projects examined, BRI projects fit within a five priority areas that encourage cooperation among parties: promote policy coordination, facilitate connectivity, unimpeded-trade, embolden financial integration, and foster people-to-people bonds (NDRC, 2015). Similar to ASEAN, the BRI is meant to provide mutual benefit for all parties and shared security (NRDC, 2015). Each BRI focus area is reflected in the key sectors funded, namely transport, communications and energy (Renwick et al., 2018).

Many of the BRI projects fall under the broad umbrella of infrastructure, but that does not mean it is just an infrastructure investment map; it is also the catalyst for what could be the largest improvement to trade in modern history (Mckinsey&Co, 2016). The BRI includes roughly 65 percent of the global population and nearly one-third of global GDP, but it also includes just under one quarter of goods and services moved around the globe (Mckinsey&Co, 2016). In addition to providing investment the BRI could be the world's largest platform for regional trade and economic collaboration (*Business 22*), pushing regional economic development where lack of access to funds previously prevented it from going—providing an "unrivalled form of investment for the APG" (*Business 22*).

While energy and electricity may be the focus of the majority of BRI funds, not all of this is investment in clean energy sources or renewable energy integration. In fact, a large focus of state-sponsored BRI energy investment is in hydrocarbon projects, including oil, gas, and coal, accounting for roughly 90% of the Silk Road Fund's investments (Nakano, 2019, p. 1). Many of the countries along the BRI routes are also rich in hydrocarbon resources, however, a common perception is this will revive overseas investment from Chinese oil companies (*Workshop 3*). In addition, questions regarding China's true commitment to 'greening the BRI'

<sup>&</sup>lt;sup>76</sup> The World Bank found that only \$66billion of total funds went to projects that were completed by 2016.

have arisen, as "Chinese financial institutions have invested \$15 billion in coal projects abroad" (Nakano, 2019, p. 1). Suspicions regarding China's true intentions, or eagerness to put its own gain above that of its partners, abound. Oil and gas investments at a time of global movement away from fossil fuels and ASEAN sustainability and emission reductions pledges, seem at odds with the reality of BRI investments.

One interview subject particularly concerned with this dynamic, *Think Tank 11*, an energy analyst with a 25-year career working in East Asia and research director for an international research institute questioned whether, in fact, BRI is bad for ASEAN's energy goals. Perhaps, this subject posited:

BRI is useful for investment, but really the goals do not really align with ASEAN [sub-]regional emissions reductions priorities...they don't seem to even align with China's emissions targets.<sup>77</sup>

This is similarly reflected in the struggle between ASEAN's region-wide goals and the policy making at the national level—where targets are agreed upon with the sub-regional governance structure they are not reflected in national level energy policies, particularly when looking at energy mix and renewable targets (see, for example, chapters 4-5).

Another common criticism of BRI (in addition to its state-sponsored investment model) is its breadth—BRI covers a wide range of projects, but also a broad timeline. Projects started years before the initiative was even announced are being included in its progress and monitoring (Hillman, 2018). Additional criticisms concern China's ability to manage BRI investment and training, with questions regarding the country's ability to engage and also let go of projects once they have been established or terms of agreements have been made (*Former Government 28*). According to *Former Government 28*, a Chinese expert in energy policy development, there are lessons learned from BRI that are not yet being fully integrated in the power sector projects:

Including the relevant local and regional parties needs to also be made a larger priority in the power sector...we learned this in development projects years ago. China's [BRI] projects need to reflect this lesson as well. We get this criticism all the time—it's low hanging fruit! We know it already.

<sup>&</sup>lt;sup>77</sup> Or, the author wonders, are ASEAN priorities actually in-line with the goals of member economies? As discussed earlier in this chapter these appear to be at odds, as well.

An interview with a leading BRI scholar, creator and manager of one of the largest BRI databases, put into words some of the challenges associated with just understanding BRI, let alone tracking its progress in specific sectors (*Think Tank 12*). As stated by this scholar, because the BRI is broad and virtually all inclusive in scope and objective, identifying, assessing, and tracking projects linked to the initiative is quite difficult:

The variety of institutions involved in BRI projects further complicates how BRI is viewed from the outside—It's no wonder ASEAN is sceptical (Think Tank 12).

Given the broad nature of BRI in terms of process, scope, and objective, assessing infrastructure projects linked to the initiative as a whole proves difficult. This is further complicated by the vast number of private, public and international institutions involved in the initiative.

According to *Former Government 28* a large proportion of BRI projects remain in the planning stages and a number of leading, high profile projects were cancelled following their announcement; this statement was later confirmed by the author's own research (Economist Intelligence Unit [EIU], 2019). For example, high-speed rail projects that fall under the BRI remit have seen very little development in the years since their first proposal. China's proposed rail network that link Southeast Asia with its northern neighbour is one such project (EIU, 2019). As stated by The Belt and Road Portal,<sup>78</sup> there are 22 Power Plant projects and three energy transmission projects seeking funding in Southeast Asia as of June 2019 that fall under BRI. These include projects in Cambodia, Indonesia, Lao PDR, Malaysia, Myanmar, the Philippines, and Singapore at a variety of phases of development—from planning stages to follow-on investment to construction. Projects range from traditional power plant projects to waste-to-energy projects. The likelihood of whether all or some of these projects will be completed is unknown based on the data available.

The number of projects in the tendering stage does not mean China isn't taking this seriously, however, or that power sector connectivity isn't a priority. Approximately 44% of BRI construction projects are energy related (Nakano, 2019), with an estimated \$7 trillion USD going towards power grid development specifically (Office of the Leading Group for Promoting the Belt & Road Initiative,

<sup>&</sup>lt;sup>78</sup> The Portal was developed by the Hong Kong Trade Development Council to provide a platform for global businesses to take advantage of BRI opportunities via Hong Kong. More information can be found at <u>https://beltandroad.hktdc.com/</u>.

2019). While concern over Chinese commitment to clean energy is based on current investment, discussed previously in this subsection, there are positive investment projects and interests of China that ASEAN could utilize. In particular, China's Global Energy Interconnection (GEI) initiative may be a tool for deployment of BRI funds while fulfilling ASEAN's APG and power sector connectivity interests.

The GEI is a strategic plan to develop and connect the power grids and systems of the countries of greater Eurasia (Cornell, 2019). While initial interviews pointed to BRI, subsequent interviews pointed to GEI as a strategic tool that could assist APG progress where it has stalled. Conversations at a conference in June 2019 provided valuable feedback on the exploration of GEI and recommendations for inclusion in this research (*Academic Conference 11*). Later workshop conversations (*Government Workshop 14*) with GEI officials pointed the author towards additional resources and insight into the power behind these Chinese proposals.

GEI was announced in 2015 by President Xi as an initiative with the goal of eradicating energy poverty, enhancing global power sector integration, all with the utilisation of clean energy resources, for the same reasons covered in this research. In addition, in 2016 the State Grid Corporation of China (SGCC or State Grid) created the Global Energy Interconnection Development and Cooperation Organization (GEIDCO) to oversee GEI (Cornell, 2019). Interestingly, GEI largely promotes neoliberal value of free trade, economic interdependence, and the removal of barriers to trade and financial exchange. Even infrastructure development coming from China is embracing the neoliberal values advocated as a requirement for power sector development in Southeast Asia.

The positive environmental and infrastructure impacts Chinese BRI investment broadly, and GEI specifically, could have on Southeast Asia and the APG are overwhelming. However, the national interests of energy security and sovereignty mentioned earlier in this chapter are at play here, as well. Cornell (2019) notes that even in regional electricity or power sharing agreements sovereignty concerns are present, and so at an international level, and bringing in China, exasperates these concerns (p. 8). Cornell further suggests that one solution is piecemeal interconnection, or, as this author posits, on a bilateral scale, subsequently growing interconnections with additional partners as trust is built. This is where ASEAN and China may be able to find some common ground, if mistrust doesn't get in the way.

## 6.4.b ASEAN-China Bilateral Relations

Historically China and ASEAN have engaged both positively and negatively along their development paths. Chinese soft power in the region is quite strong and felt in a variety of sectors and socio-cultural environments. This includes Chinese diaspora, pop culture, food, history, education and even economic interests via initiatives and agreements such as GEI, infrastructure development, or trade agreements (Roasa, 2012; Cho & Jeong, 2008). Chinese and ASEAN history are intertwined, as are their people and economies.

ASEAN-China relations, while strong today and largely integrated into each other's economies as well as the larger regional and global economies, is not based on entirely confident relations. The creation of ASEAN (discussed in more detail in Chapter 2) was positioned around a fear of communism, and suspicion of China's intentions in Southeast Asia have persisted (Astarita, 2008). Mutually beneficial economic relations have largely driven the development of this relationship, however, ASEAN's western partners have maintained a hesitancy when cooperating with China, as evidenced by the tumultuous relationship of China and the United States, one of ASEAN's largest strategic partners (with a similarly fraught relationship in the region). Astarita (2008) points to a familiar dynamic among Chinese relations with other Asian economies, ASEAN members included—China engages in multilateral institutions with a pattern of seeking dominance, while other Asian economies engage out of a fear of missing out (p. 85). This is a similar motive identified earlier when discussing trust; often trusting relationships are build out of a fear of missing out, as are trade agreements (See Chapter 5 for more discussion on trade engagement and EMI cooperation).

An interesting point in the ASEAN – BRI power sector convergence is the issue of bilateral vs multilateral cooperation. As discussed previously, within ASEAN power sector integration has happened at a primarily bilateral level. ASEAN itself has advocated for bilateral power sector integration before multilateral, and ASEAN's own APAEC (ASEAN Plan of Action for Energy Cooperation) identifies bilateral cooperation in power sector integration as a means to eventually achieve its goal of an integrated ASEAN power sector via the APG.

Similarly, ASEAN and other Southeast Asian experts interviewed advocated for bilateral cooperation and investment with BRI. Two former government experts interviewed went so far as to say China itself will have an easier time hurdling trust issues on a bilateral as opposed to multilateral basis (*Former Government 27; Former Government 28*). Other experts (*Academic Workshop 2*) pointed to this

decision being a strategic one, whereby China has chosen to engage on a bilateral basis as their strength will be greater one-on-one as opposed to with ASEAN as a collective. In July 2018 preliminary interview results that reflected this sentiment were presented at an academic workshop *(Academic Workshop 6).* The session these views were presented at included dominant European academic experts with research interest in China's BRI. A number of experts gave positive feedback and advice for further research in this area, for which the author is grateful. Two particular experts suggested a more critical approach to China's motivations for BRI investment in Southeast Asia, pointing to historical relations between China and ASEAN. This led the author to include an additional interview *(Academic 32)* with a historian and Chinese studies expert who explored China-ASEAN relations in more detail, providing context to some of the scepticism expressed in April 2018.

This new expert (*Academic 32*) pointed to predominantly positive historical relations between China and ASEAN member economies, with a few exceptions (notably disputes over the South and East China Seas and the sometimes-tumultuous Vietnam-China relationship). While Chinese influence in the region is a predominantly western focused topic, and there is some mention of Southeast Asian concern in the news (See: Gong, 2019 for one example among many), there is little reflection of this concern among ASEAN members publicly. However, the ASEAN pattern of non-interference and saving face, discussed in detail with interview Academic 10 (See Chapter 2 for references) is likely in partly a cause. In addition, ASEAN has taken a public stance of accepting Chinese infrastructure and economic investment pledges, while individual states are able to make their own decisions regarding accepting proposed packages; Malaysia, for example, put three projects on pause in 2018, two of which were resumed under fairer terms in April 2019 (Reuters, 2019). In short, ASEAN-Chinese relations are stable above the surface, but below there are ripples.

The viewpoint that China is invested in bilateral relations for strategic purposes, expressed in interviews and conference conversations, is contradictory to some of the literature that argues China is an advocate of multilateralism in East Asia and has actually advanced multilateral institutional development in the region (Kuik, 2005; Prantl, 2014). Unfortunately, this leads the author to the conclusion that Chinese motives are likely sensationalised, but also in part unknown. This viewpoint is also reflected in the work of interview subject *Think Tank* 29, a leading author and analyst in Asia-Pacific affairs. This interview subject cautions that a negative interpretation of ASEAN response to China's BRI is exceedingly

narrow and a misread of sub-regional motivations. Instead, the current state of BRI developments in Southeast Asia points to:

...actors weighing the pros and the cons of engagement with China's BRI, and in the end the majority find more positives than negatives.

For the time being, caution in policy making at the national level of ASEAN member states will likely be the norm as further agreements and investments with China are developed. Until then, strong ASEAN relations are likely to continue, but perhaps not "reflect the reality of mixed responses" at the sub-regional and national level (*Think Tank 29*).

There is also the issue of trusting BRI investment management and training; however, this seems to just be a perception of critics, and is not widely accepted by experts interviewed as an actual shortcoming of BRI investment in Southeast Asia (*Former Government 28; Government Workshop 14*). Instead the reality is quite the opposite: China has successfully managed power sector integration and the reduction of energy poverty on its own and in other countries. Engineers and electrification experts are often trained in managerial roles. The capacity of Chinese companies outweighs that of many other global actors and is therefore a moot point. Therefore, ASEAN concerns over Chinese management and training capacity should not be an issue, but instead be addressed via ASEAN and Chinese leadership as a non-existent barrier. Here the narrative and education issue discussed previously in relation to national level considerations is similarly present, and similarly surmountable.

Interviews with ASEAN experts and members demonstrated that there is ASEAN political will, there is infrastructure need, but there is a lack of money for the envisioned APG to be completed. ASEAN itself recognizes the limits of APG progress to date and is eager to incorporate outside funding sources in order to move the initiative forward (*Multilateral Organisation—Informant 2*). The baseline ADB estimate for infrastructure investment needs in developing Asia and the Pacific's power sector is roughly \$12.6 tn. USD; including climate mitigation and proofing costs this number goes up to \$15 tn USD for the wider region (ADB, 2017). A combination of BRI funds can provide \$2-3 tn USD a year that is well over estimated financing needs, roughly \$20 billion USD for the entire APG (IEA, 2015).

Without Chinese financial support the APG will undoubtably reach some investment roadblocks, whether slowing down further or halting full expansion. The IEA (2015) estimates that power sector investment, excluding

interconnections, will reach \$308 billion USD (p. 14) in order to support growing demand projected out to 2040. This coupled with the estimated \$20 billion USD expense of the APG will be a heavy price for the developed and developing ASEAN member economies to fund. Chinese aid and investment appears to come with some strings attached, and motives are unknown; however, as in other relationships of trust, priorities and risks will have to be weighed. It is the author's position that Chinese funding of the APG is inevitable, however, that does not mean ASEAN has no bargaining power over terms. In fact, China's eagerness to engage in Southeast Asia is to the sub-region's advantage, whether done on an incremental, bilateral scale or multilateral. In the subsequent subsection progress of APG will be examined, as well as limits to progress based around previously discussed governance challenges associated with trust and Chinese financial involvement.

## 6.5 ASEAN Power Grid (APG): Status

Power Sector integration in Southeast Asia is an important component of EMI initiatives focused on power connectivity. ASEAN itself is a prime place for interconnectivity to be encouraged given the availability of clean energy resources, multilateral institutions, and governance structures that operate there, plus the economic and physical geography of countries within the region. The proposed APG is one potential solution to meet sub-regional clean energy goals and offers the benefits of energy poverty reduction, deeper energy and economic integration and the amplification of ASEAN's regional impact—all common ASEAN justifications for engaging in the APG. It has been argued across the EMI literature previously presented that an integrated electricity market can improve efficiency in electricity supply, reduce costs of production and hence electricity prices, and raise standards of services due to increased competition. As global concerns for climate change increase, regional power integration is an effective way to reduce carbon emissions. In addition, connectivity and infrastructure development are an important priority for China's BRI, and some regional experts have interpreted BRI goals to align with ASEAN's energy and power sector integration goals, not the least of which is because BRI's infrastructure commitments are a significant positive force for the eradication of energy poverty and increases in development throughout the region.

Unfortunately, APG progress to date is quite limited. This is in part attributed to, among financing challenges, difficulty implementing the necessary national power sector reforms in a timely manner. As discussed in Chapter 4, progress to date is largely bilateral in nature, with no region-wide interconnections. While ASEAN has attributed this to be the first step to wider connection (*Academic 21*; supported by APAEC, 2004). This is attributed by *Former Government— Informant 1, Multilateral Organisation—Informant 2, and Academic 24* to be a way for ASEAN member economies to mitigate trust concerns, engaging first with those they have trusting, cooperative relationships with, then slowly incorporate additional members when agreements can be reached as a larger negotiating block. Complicated power relations make bilateral agreements more realistic as well, as governments must work within complicated systems of interests, including business, consumer, and public interest groups. APG literature supports this as well (Jones, 2016). How the APG progresses is unknown, but current interconnection goals are not being realised as originally set out in AIMS I-II.

Many of the trust issues identified in ASEAN states, in relation to both bilateral cooperation among ASEAN member states and between ASEAN member states and BRI investments, can in theory be solved. *Academic 26* suggested that a consideration of the political forces and historical relationships that colour understandings of present-day dynamics on both sides has the potential to push BRI investments in ASEAN power sectors further. Experts in *Workshop 3* also suggested that ASEAN itself needs to more seriously push transparency in the power sector if it wants transparency in power sector funding. This reflects recommendations by Wu (2016) as well as the IEA (2015) and the APAEC Blueprint 2025 (2010) itself.

The addition of trust among ASEAN member states has brought new considerations to light as the author considers the policy and societal implications of findings thus far. The addition of trust has also opened up analysis to an inclusion of the fluid dynamics of cooperation on a country-by-country basis as opposed to a multilateral effort. According to the majority of interviews, there is a serious need for the continued examination of domestic energy market liberalisation and sub-regional governance shortcomings, particularly related to trust and bilateral versus multilateral cooperation. The following section will continue this examination by focusing on the market and political realities that this research has exposed, discussing their relevance to the broader aims of EMI and cross-border interconnections in Southeast Asia.

#### 6.6 Discussion

So far chapter 6 has examined additional governance challenges to EMI and cross-border electricity trade that have arose throughout interviews, documentary research, and data analysis. These additional governance challenges fall within 3 different broad categories: (1) Sub-regional versus National interests; (2) trust among ASEAN members; (3) trust of outsiders. Examination and analysis of these three challenges to EMI also highlighted for the author the unique economic, political, and social dynamics that play a role in understanding the overall EMI and cross-border electricity trade picture in Southeast Asia. This discussion section will introduce the additional dynamics that emerged for the author via analysis of data and interviews, concluding with a neo-developmental statism-based explanation of the IPE of EMI governance challenges.

In addition to a misalignment between ASEAN and national priorities there appears to be a misalignment between market and political realities. In other words, the sub-regional and national electricity markets involved in ASEAN EMI are able to take on the reforms proposed as necessary for deeper integration; however, the political reality is many governments are not yet able or willing to make these changes. In Chapter 5 (National Market Factors) the market needs were discussed in detail, drawing on recommendations from Oseni and Pollitt (2014) and recommendations gleaned from interviews and literature. These were discussed above in relation to the O&PCRB BIMP-EAGA Scorecard, whereby the economic, political and societal reforms were ranked against the realities of the individual BIMP-EAGA markets. None of these reforms are insurmountable—in fact, at the heart of the very design of many of these reforms is their approachability and feasibility for developing economies. However, there are clearly obstacles as many of these reforms are not new or novel in their imagining. Chapter 6 (Governance Challenges) explored some of the additional factors preventing such reforms and deeper sub-regional and subsystem integration. These include ASEAN governance shortcomings, particularly perception at the national versus sub-regional level, political will, trust among ASEAN members, and also trust of outsiders. Here these barriers are analysed in relation to their overlap, bringing these three overarching challenges together to map the difficulty of EMI within the neo-developmental state structure of ASEAN and its market subsystems.

Arguments for and against globalisation, which is largely driven by neoliberal world economic order, as outlined in Chapter 2 (*Cross-Border Electricity Trade: Conceptual Approach & Review of Literature*), predominantly hinge on issues of power shifts and equity—who benefits and how? In the case of cross-border electricity trade these same arguments are relevant. Electricity is directed from major suppliers and producers into the grid; those connected to the grid already benefit. Those connected but with unreliable access might have some benefit, while those not currently connected may be connected in the future, if funding is secured and projects for infrastructure expansion occur; but it is more likely that those rural, impoverished communities will be the last to benefit from increased cross-border electricity trade. Without grid connections they will never benefit from those new, more affordable or reliable, electricity imports. Grid expansion can absolutely happen. However, it will have to happen in tandem with other technical and physical measures, and have the political will behind it, to be felt in rural, island communities of Southeast Asia.

This highlights yet another challenge of balancing EMI and cross-border electricity given the multiple levels of governance-political will is largely lacking at the national level. This is supported via the conflicting realities of subregional level targets and national level shortcomings discussed in Chapter 4 (Sub-Regional Market Factors), via data analysis discussed in Chapter 5 (National *Market Factors*) that highlight coal is still king and renewable energy targets are far from being reached, as well as via interviews in Chapter 6 (Governance *Challenges*) that highlight interview feedback on political barriers. As one expert said (Academic A26) "Climate Change is not [a] national priority but is treated publicly as though it is." Evidence across sources demonstrates that the subregional energy mix is not defined by ASEAN goals, targets, or even sub-regional connectivity vision, and as a result national level action is not defined by these factors, either. This is difficult to come to terms with in part because there is a logical cross-over between economic development policies, increased energy access, and renewable energy integration (Shi, 2016); however, governments are clearly prioritising economic factors over climate, which is thus reflected in coal dominated regional energy outlooks and heavily subsidised national energy sectors discussed in Chapter 5 (Market Factors).

Unfortunately, the challenge of garnering political will at both the national and sub-regional level hinges on an aspect of the neo-developmental state model that was discussed previously in this chapter—ASEAN'S political and economic

systems need to include considerations of equity and people. Who benefits, who gains, and how to make those two issues intersect with regional energy and climate transition goals. This is similarly reflected in the emerging rural and urban divide that the author has caught glimpses of in this research.

## 6.6.b Rural vs Urban

There is a distinction to be made between poorer populations and electricity access, one with important implications for this research and broader liberalisation and economic development goals. It has been mentioned repeatedly in prior chapters that access to reliable electricity is a fundamental barrier to improved quality of life and increased economic development among the world's disenfranchised populations. Despite growth in Southeast Asian electricity access there is still a significant portion of the population (65 million people according to IRENA, 2018b) without reliable electricity access. This is magnified by the fact that those without access are in large part in poor communities and/or rural locations—in urban communities it is the poorest that remain without access, and in rural locations the cost of building new infrastructure and connecting to the grid can be prohibitive. Thus, paying for better access, and funding new projects, is largely driven by neoliberal economic expectations about payment for goods and services. Physical and technical barriers, like generation capacity, poor infrastructure, remote service costs, lack of financing, and affordability all come down to economic factors, which in economically disenfranchised populations are near to impossible to overcome without new political commitments and financing options. However, the supremacy of neoliberal economic policy is based in part on the idea that bigger markets mean more access. Here, another contradiction emerges, one that has followed neoliberalism throughout its varied history-does more market access translate to improvements for the world's poorest? Where ASEAN's connectivity agenda includes these poorer populations, perhaps, but it is apparent that they are not the primary focus, regardless of messaging, considering the realities of power grid growth among island economies.

ASEAN's connectivity agenda itself is a replication of the growth pushed by the dominant global capitalist system. The Master Plan on ASEAN Connectivity (MPAC) was introduced in Chapter 2 (*Cross-Border Electricity Trade: Conceptual Approach & Review of Literature*) as the basis for ASEAN's power sector connectivity goals, but it is also related to a variety of connectivity measures, including transport and trade (ASEAN, 2017). This agenda, in fact, reflects the common, neoliberal push for infrastructure development that enforces global

trade progress and development. This connectivity agenda also reflects many of the neoliberal policies underscoring the current economic system as discussed in Chapter 2—reflecting ASEAN's own sub-regional commitment to the global capitalist market system insofar as it benefits ASEAN members and their role in the global system. ASEAN's connectivity agenda, however, largely focuses on urban populations and urban growth and support, leaving the rural ASEAN communities out of the connectivity agenda.<sup>79</sup> This point was made by the author during discussions at three high-level policy events, *Government Conference 12* & 13, and *Government Workshop* 14, and was widely well-received by participants from within the ASEAN community. One interview subject, Multilateral Organisation 4, echoed these sentiments, saying "The rural population isn't forgotten in broader electrification goals but the reality is they aren't a priority for connectivity." This might seem contradictory to ASEAN messaging echoed in this research, which highlights the ability of the APG and other cross-border electricity trade or EMI projects for increasing electrification, but the distinction is where and how electrification occurs. In these projects gridconnected electrification is the priority, and while that can include rural populations, the truth is off-grid connectivity is more likely for these groups (ACE, 2013a).

This poses an interesting additional point—are sub-regional and national connectivity agendas further fragmenting the rural versus urban divide that already exists in ASEAN (See Table 1.1 in Chapter 1 for a reference to the urban versus rural electricity divide)? Within ASEAN 55% of the population lives in rural communities (ACE, 2013a) and roughly 10% of the total population in ASEAN remains without electricity (IEA, 2019). Much of ASEAN's growth has included the movement of large populations from rural to urban settings, and dramatic growth in urban centres as ASEAN's regional production network has also grown and become further integrated into the global and regional economic

<sup>&</sup>lt;sup>79</sup> Fünfgeld (2019) conducted a study of ASEAN connectivity imagery, finding that the focus of this major ASEAN initiative is almost exclusively on the urban ASEAN population. The Economic Research Institute for ASEAN and East Asia's (ERIA) own report The ASEAN Economic Community into 2025 and Beyond (Maria, et al., 2017) highlights a vast array of business cases for connectivity, but only refers to rural communities in relation to movement of workers; similarly the updated Master Plan on ASEAN Connectivity (ASEAN, 2017a) largely refers to urban communities in relation to sustainable urbanisation only, with little mention (if any) of rural communities and increased connectivity among rural populations. Instead the urban narrative tangentially references agricultural communities, but only in relation to job movement and growth, and never explicitly quality of life improvements. While there is an ASEAN Guideline on Off-grid Rural Electrification Approaches (ACE, 2013b) this is not a part of the connectivity agenda in so far as it is discussed at the policy level, identified in the literature, or defined in interviews, including with ASEAN officials.

system (see Singapore and Thailand, for example; Drysdale, 2000). Urbanisation growth is expected to continue for the foreseeable future, with 90 million people moving to ASEAN's urban centres by 2030, with McKinsey's Global Institute (2014) projecting that as a result \$520-930 billion could be added to ASEAN's annual GDP by 2030. It's no surprise then, that there is much focus on urban populations within ASEAN. However, this does conflict with the ASEAN policy messaging, and perhaps explains some of the national level restraint on echoing this connectivity agenda through domestic policies that contribute to it. Failing to consider and include rural concerns and needs in both national and subregional/regional policy making would be a great failure, particularly considering the sustainability challenges that already exist. This could not only exasperate environmental and energy challenges, but rural urban divides globally as well as regionally.

Uneven development in rural and urban communities is no new concept to ASEAN policy makers and is already an important part of the AEC (ASEAN, 2015). However, failing to similarly prioritise rural electrification, or at least make it an equal priority to urban connectivity, will not only weaken the ability of ASEAN to achieve its connectivity agenda and sub-regional EMI goals, but it could weaken trust in national level policy making and further expose existing development divides among rural and urban populations, as well as weaken trust in ASEAN governance, sub-regional level targets and the reasoning behind them. Prior criticisms of neoliberal economic policies (Chapter 2, *Conceptual Approach & Review of Literature)* and equitable economic development are echoed here as well. Focusing on urban centres, traditional cross-border linkages between urban centres, and connectivity that supports international trade among partners does not ring true to ASEAN's message of connectivity for all.

A neo-developmental state model that includes considerations of power and power balancing factors (such as subsidy removal or transition from hydrocarbon to renewable sources) along with alignment of sub-regional and national goals would better fit the combined policy approach of neodevelopmental statism, instead of being driven from a strictly neo-liberal reform processes applied to regional developmental state systems with no considerations for market or social realities of individual economies within the wider system.

### 6.6.c A Comprehensive, Neo-Developmental Statism Explanation

One of the unique factors of this research is the application of the author's adapted theoretical framework of neo-developmental statism to sub-regional EMI and cross-border electricity trade efforts. Chapter 2 (Conceptual Approach & Review of Literature) discussed the ways in which neoliberal, free market ideology is translated in common proposed modifications of national and subregional electricity markets in order for deeper integration to occur. A commitment to free trade, increased competition, and liberalisation efforts are all recommendations referenced in the O&PCRB BIMP-EAGA Scorecard; these recommendations are similar across a variety of literature on EMI, across expert interviews conducted for this research, and in Southeast Asia and elsewhere globally. Developmental statism (DS), the common, state directed economic policy that drove many East Asian economies farther along in the development process from the 1970s onward, is also present throughout the sub-region of Southeast Asia. Chapter 2 considered, in part, how the two seemingly contradictory economic processes could operate in tandem in Southeast Asian energy markets—neoliberalism, driven by free markets, and DS, driven by government policy planning.

In response the author has proposed neo-developmental statism as a useful explanatory tool for the examination of the IPE of sub-regional electricity markets. Neo-developmental statism is a process of state directed and market liberalisation policies, where free market processes are embraced alongside government economic planning, including targeted industries and directed growth; power dynamics play a role in compelling directions and outcomes of policy planning and neoliberal market pressures similarly coerce economies in their policy choices. In the following section the concept of neo-developmental statism will be discussed in more detail in relation to proposed electricity market reforms, bringing into this analysis a discussion of where sustainability and energy transition discussions can fit within this state driven, neoliberal market structure. This will include examples of where neo-developmental statism is currently in use within power sector reform, and how reform and policy making can evolve given current market realities and neo-developmental statism practices.

Regional and national shifts towards a combined policy approach, one that utilises both neoliberal economic policies and traditional developmental state ideology, has become a common policy strategy within the broader East Asian region. ASEAN has further progressed this combined policy approach in its subregional development strategies, visible in such examples as the MPAC and the ASEAN Economic Community (AEC) and EMI agendas. Analysis of electricity markets within each of the BIMP-EAGA countries shows that further liberalisation is needed; liberalisation does appear likely to occur given the pressure from ASEAN to do so and the economic gains perceived to be realised from such reforms. At the same time electricity markets remain highly controlled by the state, and power dynamics remain structured between energy and political interests, similarly represented elsewhere in global energy markets. Regional shifts towards a neo-developmental statism are already being realised and are visible across the span of sub-regional energy and electricity markets via a combination of state directed and neoliberal economic policy making. The sub-regional shift towards a hybrid, neoliberal-oriented developmental strategy has resulted in benefits across the variety of developmental paths within ASEAN and corresponded with region-wide assimilation into the global capitalist market system.

In short, neo-developmental statism has already been realised in ASEAN through a combination of state-led planning and control of energy and electricity markets, coupled with implementation of neoliberal trade practices, a focus on the removal of trade barriers, and commitments to increasing trade across the region. These reform combinations are already in place in many Southeast Asian economies or are planned for further inclusion as a result of ASEAN EMI policy planning. Interview subject *Academic A6*, an economist with expertise in Asian clean energy development, directly discussed the challenge state intervention poses to traditional notions of free markets:

The movement to decarbonise electricity markets is really about reconciling liberalism with interventionism, which is also the main theoretical challenge when considering trade's role in climate action. More broadly, I believe the arguments in support [of deeper renewable energy integration] require states to intervene with smart policy and governance approaches to support the development of clean energy techs and power sector integration. However, this depends on which development stage these states are at, and also goes against standard neoliberal expectations of free markets.

This author has observed in a reading of the literature and analysis of interviews that there does seem to be an expectation among global institutions that neoliberal ideology will be broadly adapted by the economic systems employing these practices and thus deeply integrated in the economic system in questionrecommendations by the World Bank, International Renewable Energy Agency, International Energy Agency, ADB and others all include a focus on liberalisation as key to further energy transitions and necessary for continually higher stages of development. What is instead happening in ASEAN is a combined policy approach with collectivist Asian values remaining at the forefront of politics domestically but complemented by the picking and choosing of neoliberal economic policies where and when they may be beneficial to the government employing them; essentially, another example of traditional DS, whereby the government directs policy based on domestic political and economic choices and/or needs.

An area where there is evidence of room for neo-developmental statism is in investment and financing of power sector development and electricity access in developing economies. It was discussed in Chapter 4 (*Regional Market Factors*) that reforms to electricity markets are varied in their outcomes; in particular, where neoliberal market reforms result in higher prices for consumers, such reforms are detrimental to welfare indicators while being positive for overall economic growth. The policy disconnects between privatisation goals and consumer access, where price becomes a factor, is another example of the delicate policy balancing act that is employed regionally. Governments can offset this price, as can regulators, but this may have implications for financing or project development in the first place. One multi-country study (Sen et al., 2016) recommends that reforms include transferring surplus gained from increased competition and liberalisation to underprivileged consumers, thus increasing wellbeing (p. 39).

These concerns were echoed in an interview with a development bank economist whose work focuses specifically on developing country electricity market reform. When asked about the biggest obstacle to power sector reform among emerging economies, interview subject *Development Bank 25* brought together a few of the finance and investment obstacles identified across Southeast Asia—some of which are not actually consistent with truly liberal market structures, but instead a combined, neo-developmental statist approach:

[The largest barrier is] Absolutely cost. Due to the expenses involved in setting up power infrastructure, the costs are very high for consumers. As a result, governments provide subsidies. This in turn puts pressure on the government's finances... [Governments] need to attract more investment. In order to attract investment, power sector reform is crucial. While transmission and distribution continue to be monopolized [sic] by
government, power generation has been open for investment. Crucial reforms like tariffs continue to impede potential growth in the energy sector. It's a cycle! But also, I also see a role for power sector reform in fiscally deficit countries to move from providing subsidies to more sustainable prices for power generation.

These comments highlight the need for a well-rounded approach in reform that supports the neo-developmental state model: holistic reform approaches need to be taken by each ASEAN member, considering communities, capabilities, finances, investors, and market biases in order to structure national electricity market transformations that will have a lasting impact on not just energy mix and access, but communities that benefit as well. While parts of this approach are being used already, the challenges in local communities (see prior subsection) and the performative nature of governance institution targets point to limits in the current approach. Based on the realities of national market factors outlined in Chapter 5, a holistic approach to sub-regional EMI would be difficult considering that the ideal reform package would include reductions in coal generation or removal of subsidies. Here the dynamics of power in electricity markets are obvious-private and public interests make up both the current shortcomings and the potential solutions necessary in developing economies where physical, economic and policy structure of power sectors require some balancing.

## **6.7 Conclusions**

By examining the governance challenges that emerged in expert interviews this chapter has addressed Research Question 3 and its associated sub-question: (RQ3) *What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian Development?* And (RQ3a) *What is the political and economic status of the APG?* First, national markets factors and challenges between sub-regional and national interests complicate the ability of EMI to address renewable energy integration challenges as a result of limits to market integration; challenges of trust and relationship power dynamics also limit the ability of the APG to develop in line with ASEAN's EMI plans. In addition, progress to the APG is slow and complicated by political and economic uncertainty associated with power sector development.

In order to reach these conclusions this chapter has examined trust in ASEAN, the potential impact of China's Belt and Road Initiative (BRI) on power sector integration efforts among ASEAN member economies, and developments within

- 240 -

support for power sector infrastructure development in Southeast Asia, and the development of the ASEAN Power Grid (APG) through bilateral, sub-regional and regional initiatives.

The significance of addressing the potential impact that BRI may have on regional power sector integration lies in the pressing need for coordinated, global responses to climate change. The role of BRI as a supporting mechanism for regional growth in clean energy utilisation and integration is poorly understood in relation to national and sub-regional power sectors.

While the theme of trust emerged over time, it is a thread linking many governance challenges associated with cross-border electricity trade and EMI in ASEAN. This includes trust within ASEAN and trust with outsiders. The challenge of managing sub-regional and national incentives, priorities and perceptions is also a further complication to EMI governance and barrier to APG development and EMI growth. Trust adds an additional challenge to the management of interests in the development process of the APG, as ultimately trust is required for strong relations. As discussed in Section 6.1, trust does not necessarily have to precede organisational membership or multilateral cooperation—in some cases trust can develop as relationships are developed within these groupings. For ASEAN and China this may be one avenue towards stronger EMI cooperation.

Based on themes identified in this chapter the author has developed 2 additional findings: (1) an all-of-the-above approach is necessary in order to fully realize the potential of trade as a means to respond to intermittency concerns in regional and international electricity markets. This means in addition to electricity trade on a bilateral scale, progression towards multilateral should also be included. In addition, utilising a broad array of energy efficient policies, practices, and partners is necessary in order to reach regional climate goals; and (2) BRI financial and technical assistance offers one necessary solution to a multitude of shared problems. In some cases this is progressing, but ASEAN advocacy of BRI funds and movement of BRI funded power sector projects is needed.

## Chapter 7 Conclusions

## 7.1 Introduction

This concluding chapter will summarise the main findings of the research as they relate to the research questions and objectives set out at the beginning of this research. This will begin with Section 7.2, Summary of the Thesis, where the research questions, objectives, and aims of this research will be presented. This section will also summarise each prior chapter and connect the main themes discussed in the corresponding data chapters (chapters 4-6) with the key findings and research questions and answers. Next, Section 7.3, Original Contributions, will discuss the original contributions of this research to the intersection of IPE, East Asian studies and sustainability research, including the author's theoretical framework, neo-developmental statism, data collection, and elite interviews. Section 7.4, Research Limitations, will discuss limits that the author identified and aspects of this research that challenged the author. Section 7.5, *Suggestions for Further Study*, will discuss a few key ideas that the author is interested in exploring in future research as an expansion of the key concepts identified here and related to the IPE of cross-border electricity trade. This section will also highlight the ways in which the key themes of sub-regional market factors, national market factors, and governance challenges create options for further political and economic exploration of the phenomenon and drivers behind them. The penultimate section in this chapter, Section 7.6, *Policy Implications*, will discuss the repercussions for policy that have emerged through the analysis of the key themes. These policy implications highlight the overarching challenge of climate change imperatives for cross-border electricity trade—the energy transition towards less reliance on hydrocarbon resources and more utilisation of renewable energy sources could be improved. Ultimately, power sector needs and the reforms necessary to achieve them are reflective of the priority of responses to climate challenges, and the way in which economic and market factors take precedence within national policy decisions. The final section of this chapter, *Concluding Remarks*, will briefly summarise the main points and arguments made.

## 7.2 Summary of the Thesis

This thesis was led by a focus on the three primary research questions set out at the beginning of this research. The three themes identified in data chapters 4-6

were drawn together and examined alongside the three primary research questions and the three research objectives that were originally conceived. As research progressed, and the research plan took shape, relevant sub-questions were added, providing context to the key findings that were identified. These objectives, questions, and themes were paired together as follows:

# Table 7.1 Consolidation of Research Aims

# **Objective 1: Provide an overview of cross-border interconnections and assess market factors in Southeast Asia;**

**RQ1:** How do governments understand the IPE of cross-border interconnections in Southeast Asia?

# **Chapter 4: Sub-Regional Market Factors**

Sub-questions:

- What are the energy and trade policy incentives for increased crossborder interconnections?
- How can sub-regional electricity markets diversify their energy mix and reduce emissions? What incentives exist to do so?

Objective 2: Assess market factors in sub-region of Southeast Asia and case study selection and identify areas of individual and collective reform needed to reach deeper levels of EMI among BIMP-EAGA member states;

**RQ2:** How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets?

# **Chapter 5: National Market Factors**

Sub-Questions:

- How does the political and economic structure of national electricity markets impact the expansion of cross-border interconnections in BIMP-EAGA?
- Is the structure of national electricity markets conducive to subregional and international policy goals in Southeast Asia?

Objective 3: Identify and analyse additional governance challenges to deeper EMI and increased electricity interconnections among ASEAN member states and case study selection;

**RQ3:** What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian Development?

## **Chapter 6: Governance Barriers**

Sub-Questions:

• What is the political and economic status of the ASEAN Power Grid?

When research questions were originally conceived they were intended to demonstrate policy considerations for cross-border electricity trade and increased interconnections. The research findings demonstrate that there are tensions between different levels and types of interests—national versus subregional interests, economic versus sustainable interests, and liberal versus state-led markets. This dichotomy draws on the themes that emerged in data collection and discussed in the corresponding data chapters—sub-regional market factors (Chapter 4), national market factors (Chapter 5), and governance challenges (Chapter 6); within each of these themes there are additional factors, such as national priorities, interests, political will, and the rural versus urban divide, all of which have been explored in chapters 4-6. Increased interconnections are motivated by multiple elements that are weighted differently depending on the national and sub-regional interpretation of global goals and the economic and political structure within which the interpretation is rooted. There is, the author believes, a balance between economics, climate challenges, and trade (here, cross-border electricity trade, specifically) that is demonstrated via the three themes identified within this research. Whether or not governments can equally balance these three considerations is yet to be seen, as the analysis will suggest.

In order to make this case each data chapter was framed around a single, corresponding theme and the associated research question that informed the theme. Chapter 4, Sub-regional Market Factors, examined the state of electricity interconnections in the ASEAN sub-region and the BIMP-EAGA subsystem. This included an overview of the current and projected cross-border interconnections in the sub-region and BIMP-EAGA subsystem. Analysis of the state of interconnections also revealed the incentives and reform needs for further interconnection and diversification of the sub-region's energy mix. This section also first introduced the contradictions between national versus subregional practices and the priorities employed at both levels. Chapter 4 addressed Research Question 1 and its associated sub-questions—(RQ1) How do governments understand the IPE of cross-border interconnections in Southeast Asia? (RQ1a) What are the energy and trade policy incentives for increased crossborder interconnections? And (RQ1b) How can sub-regional electricity markets diversify their energy mix and reduce emissions and what incentives to do so exist? In response to these questions this chapter revealed: at the ASEAN level governments are incentivised to participate in cross-border electricity trade by economic imperatives to utilise cross-border climate, energy and interconnections and augment VRE challenges while also diversifying their energy mix; however, economic incentives drive EMI at the national level, exposing the need for better coordination across multiple governance levels. EMI and interconnection progress is intermittent and uneven across ASEAN and in the BIMP-EAGA subsystem due to a variety of different market factors related to climate targets, electricity supply, energy mix, and sustainability. In addition, goals are ambitious and national level commitments fall short where diversification of the energy mix is concerned. Instead, it appears that subregional institutions of governance (ASEAN) and sub-regional commitments are standing in performatively for concrete action.

Chapter 5 sought to answer Research Question 2 and its associated subquestions: (RQ2) How does the IPE of sub-regional cross-border electricity trade affect the opportunities for renewable energy to participate in national markets? (RQ2a) How does the political and economic structure of national electricity markets impact the expansion of cross-border interconnections in BIMP-EAGA? And (RQ2b) Is the structure of national electricity markets conducive to subregional and international policy goals in Southeast Asia? Chapter 5 explored the theme of *National Market Factors*, explaining the realities of individual markets within ASEAN and the BIMP-EAGA subsystem, while describing the weaknesses and strengths within these markets using the criteria established by Oseni and Pollitt (2014). The author also created their own market criteria for increased EMI (O&PCRB EMI Criteria) by expanding upon work done by Oseni and Pollitt and details elucidated in data collection via documentary and data analysis and elite interviews. Chapter 5 also highlighted the relationship between liberalisation and DS in East Asia, looking at the contradiction between policies in both camps and their materialisation in the broader ASEAN system and within the BIMP-EAGA subsystem. This chapter built upon prior discussions in Chapter 4 about national versus sub-regional priorities, expanding analysis to emphasise the primacy of national concerns and need for three types of reform: (1) market liberalisation; (2) coordination, harmonisation & standardisation; (3) subsidies. Discussion in this chapter included a revised, liberal development strategy, neodevelopmental statism.

In response to RQ2 and associated sub-questions, Chapter 5 revealed the findings that the variety of economic and political structures present in BIMP-EAGA are not currently conducive to increased cross-border electricity trade based on requirements for EMI to flourish. The present structures of national electricity markets are also not conducive to sub-regional and international policy goals related to energy and climate targets given a continued reliance on

subsidies, dominance of fossil fuels in sub-regional and subsystem energy mixes, and the market structure of national electricity systems. In addition, the IPE of these national market factors and sub-regional interconnections have revealed that while there is room for growth in renewables, countries do not have an electricity deficit, and therefore the impetus for increasing renewable integration as a means to trade surplus electricity is not actually present. As a result, while renewable energy has potential for participation in national markets, current market factors point to limits in its ability given the present national power systems and the corresponding economic and political structures.

Chapter 6, *Governance Challenges*, addressed Research Question 3 and its associated sub-question: (RQ3) *What can the case of cross-border electricity trade and renewable energy integration tell us about the IPE of East Asian Development?* And (RQ3a) *What is the political and economic status of the APG?* This chapter discussed governance barriers to EMI and cross-border electricity trade in ASEAN as they relate to issues of interest. These included national versus sub-regional interests, trust within ASEAN, and trust of outsiders, including China and its role in the sub-region via power sector investment. Discussion in this chapter probed the market and political realities in ASEAN, political will and prioritisation of responses to global climate challenges, rural versus urban divides, and the impacts these issues have on a fully integrated ASEAN energy market and increased EMI. This chapter further used neo-developmental statism to explain the existence of these dynamics in a hybrid system that requires holistic responses to pressing climate and energy challenges that greatly impacted by economic factors.

In response to RQ3 and RQ3a, Chapter 6 revealed: First, national markets factors and challenges between sub-regional and national perceptions and interests complicate the ability of EMI to address renewable energy integration challenges as a result of limits to market integration and commitment to EMI goals; and second, challenges of trust and relationship power dynamics also limit the ability of the APG to develop in line with ASEAN's EMI plans. In addition, APG progress is slow and complicated by political and economic uncertainty associated with power sector development and commitment to climate targets amid economic pressures. Overall cross-border electricity trade and renewable energy integration in Southeast Asia are limited by performative sub-regional commitments that enable economies to appear to be engaging with liberalisation and contributing to market integration, but with slow moving progress as a result of complex systems. ASEAN economies perform neo-liberalisation while practicing neo-developmental statism; the ASEAN governance of sub-regional EMI is disabling reform due to its performative nature.

Here it is the author's intent to draw a thread through these three independent data chapters: sub-regional targets (discussed in Chapter 4, Sub-Regional Market Factors) and the realities of those interconnection goals are not possible without national reforms and system changes (discussed in Chapter 5, National Market Factors), which in this research are framed around the work of Oseni and Pollitt (2014) as well as interview results; these reforms are further complicated by additional barriers, including the role of competing interests, political will, and dynamics of trust (discussed in Chapter 6, Governance Challenges). ASEAN and ASEAN member economies are limiting the development of sub-regional EMI via the performative goal setting employed by member economies that are driven by neoliberal market incentives. Neo-developmental statism offers a lens through which to understand this research thread, whereby the traditional development paradigm of East Asian economies has evolved into a more complex and nuanced method for operating in neoliberal markets and gaining from liberalisation while maintaining traditional state-driven economic policy making practices.

# 7.3 Original Contributions

The thesis makes a number of original contributions to the conceptualisation of cross-border electricity trade in Southeast Asia, understanding of the economic development practices employed throughout East Asia, and how these practices intersect with global, regional, and sub-regional EMI and cross-border interconnections.

First, this research exposed the need for new modes of IPE analysis in interconnection research in East Asia. While there are some academic studies of the BIMP-EAGA sub-region and its potential for EMI, this thesis is the first to analyse the IPE of individual electricity markets in order to explain incentives behind cross-border electricity trade and the policies necessary for it to be carried out. In addition, this research expands on Oseni and Pollitt's (2014) work on EMI criteria to include additional criteria relevant to developing Asia and the BIMP-EAGA subsystem, contrasting these criteria against the market factors and IPE of cross-border electricity trade and EMI in Southeast Asia specifically.

Secondly, the application of neo-developmental statism to electricity markets is novel in its interpretation and application. While various iterations of DS have been applied to IPE topics in East and Southeast Asia, the application of the author's conceptualisation of neo-developmental statism offers insight into the very specific challenges of energy policy development amid state-driven economic policy making. In particular, neo-developmental statism offers an explanation for the difficulty of EMI in Southeast Asia and within BIMP-EAGA not articulated previously—the challenge of applying neo-liberal policy reforms in state-dominated markets has resulted in gaps between the identification of needs, interpretation of those needs and policy action in EMI development. Neodevelopmental statism also explains the gaps between national and sub-regional EMI developments, and the delays that have been observed between these two levels in this research. In addition, neo-developmental statism explains the evolved economic development practices present among ASEAN member economies, whereby a balance is struck between engagement in neo-liberalism and commitment to centrally planned economic policy making.

Finally, the elite interviews conducted in this research represent novel contributions to the IPE of cross-border electricity trade. Among respondents were established experts in their fields, high-level energy analysts from across a swath of energy market issues sub-regionally and regionally, and practitioners on the ground dealing with energy and electricity issues daily in their respective sub-regions. In addition, contributions from representatives of development banks, research institutions, and national governments offered high-level policy opinions that have not been combined in this format previously because of access challenges. These interviews, combined with the data collection of individual BIMP-EAGA markets, offer an original contribution to the study of EMI in Southeast Asia and the fields of IPE and sustainability research as they are applied to EMI issues.

This Ph.D. research also makes an original contribution to the approach to studies of EMI by combining IPE, East Asian studies, and sustainability research to explain the drivers behind and reality of cross-border electricity trade in East Asia. Ultimately this cross-disciplinary research has utilised conceptual and methodological approaches from each individual field, combining them to achieve new results and deeper insight into a very specific area of study. The empirical, conceptual and methodological choices carried out and described in previous chapters have resulted in the collection of unique data and analysis. These methods have aided in addressing the primary research questions and, as

a result, made multiple contributions to our understanding of the IPE of crossborder electricity trade. While these results are not necessarily prescriptive, and in some ways leave only aspirational policy recommendations, they do offer insight into the difficulty of EMI and cross-border electricity trade expansion in an environment where policy is conflicting and not always complementary with broader sustainability goals.

# 7.4 Research Limitations

Research limitations in this Ph.D. dissertation are largely based around access, data and methods. While the author chose to utilise a mixed method approach, access to electricity market data for analysis did prove difficult, resulting in a tendency to utilise interview data and elite opinions where market data was not available. In addition, the difficulty in finding current, reliable electricity market data across all BIMP-EAGA countries means that historical data was used in some cases. Tracking interconnections in Southeast Asia was largely done via ASEAN data and contacts. Contradictions between sources did mean that in some cases the author had to rely on older data or make judgement calls based on multiple data sources. Broadly, however, the author's mapping of interconnections complements related work done by previous scholars.

If additional data access and time were available an expansion of analysis of the case study could benefit further research. For example, in Chapter 5, a heat map was used to visualise relative progress across EMI requirements, type of requirement and country. The author noted that further insights could be acquired by adding a number range to the score card. However, this would require expanding the number of requirements for each type to create a more quantifiable measurement of success and failure, i.e. 1-10. A colour gradient would further visualise success or failure by type. This expansion would require more data and further study, which was beyond the time and data available. This data could include a review of actual government policies, statements, and assessment of individual projects or initiatives. Financial analysis could also be included to gauge commitment. The addition of a SWOT analysis (strengths, weaknesses, opportunities and threats) of each individual BIMP-EAGA country might also yield beneficial insights into progress and development of EMI reform, further expanding the data collected. However, time limitations and data access prevented both of these additions.

Time limitations also meant that the author was not able to do as many interviews as they had imagined at the earliest stages of this research development; initially 68 potential interview subjects were identified. This was eventually narrowed down to 32. Follow up with interview subjects and analysis of interview materials based on new data from data collection or documentary research meant that additional time was needed for further expansion beyond 32 interviews. As research interviews began revealing similar information and saturation was approached the author made the difficult decision to cease interviews despite access to additional subjects. Expansion of interviews could be one avenue for future research in this area as well, such as the inclusion of civil society representatives and interest groups among interview subjects. There were a number of additional issues that were not covered in this research in serious depth, but their inclusion would likely have contributed salience to this project were there no time limitations. These additional issues will be covered in the following section.

# 7.5 Suggestions for Further Study

This research represents a thorough examination of the IPE of electricity market integration in Southeast Asia and within the context of East Asian economic development practices. However, this research is only a small microcosm of the entire picture of policy responses to climate change and the global energy transition, both in terms of issues included, elite opinions gathered, and time period covered. As research progressed beyond the original research design and data collection began to take shape into analysis, a number of additional questions arose. However, due to the parameters of research and time limitations these issues were touched upon briefly, where relevant, but ultimately saved for deeper analysis in future study.<sup>80</sup> Four primary issues revealed themselves: (1) the intersection of water policy management and crossborder electricity trade, which overlaps with (2) the role of China and the Belt and Road Initiative (BRI); (3) Moral International Political Economy (Moral IPE) considerations; and (4) grid connected distributed generation. These four areas are explored further in the following subsections.

## 7.5.a Water Policy in the GMSR

At the outset of this research it was clear time limitations would hinder the number of case studies that could be employed. The author chose to use the least

<sup>&</sup>lt;sup>80</sup> The availability of additional data and further interviews with civil society groups would also be avenues for further research; however, here the focus will be on additional topical areas as opposed to expansion of the current project.

successful ASEAN subsystem as the primary case study, BIMP-EAGA. Considerations for further study include expansion of case study to include other ASEAN subsystems identified in the ASEAN Interconnection Master Plans (AIMS), such as the GMSR. Were this analysis to include a deeper examination of GMSR, the focus would likely switch towards a success story (GMSR) and a developing cross-border electricity trade subsystem (BIMP-EAGA) with comparisons across the two subsystems. Unfortunately, time and parameters set out early on did not allow for this deeper comparison. One avenue to exploring the lessons learned between these two subsystems would be a greater analysis of water policy and its implications for cross-border electricity trade regionally. Via interviews and data collection the author learned that GMSR water policy has some interesting implications for cross-border electricity trade within the subregion. A greater analysis of regional and subsystem water policy would be beneficial to explorations of cross-border electricity trade as water policy considerations are alive and well among ASEAN countries that border China, as damn management is a major challenge in both upper Mekong countries (China, Lao PDR, Myanmar, Thailand) and lower Mekong (Cambodia and Vietnam) countries, all of which fall within the GMSR. In addition, including a deeper analysis and exploration of water policy in GMSR would also add to sustainability and climate change considerations and arguments within cross-border electricity trade and EMI topics.

#### 7.5.b China & BRI

Interestingly, considerations of greater water policy in GMSR also highlight the dominant role of China and its BRI initiative in sub-regional and regional power sector development and reform. Chapter 6 (*Governance Challenges*) explored China's role in sub-regional cross-border electricity trade as it relates to issues of trust among ASEAN members. However, research and interviews revealed China's BRI also intersects with water policy management in the GMSR and across ASEAN broadly. An expansion of this research that considers China's BRI and water policy impacts, as well as the cross-over with power sector development, could include a variety of country case studies and bridge a previously limited area of research, China's water policy development projects and ASEAN connectivity goals.

Additional avenues of exploration that incorporate China's BRI projects could focus on a variety of subsystem case studies, building on discussions this author had with experts across a variety of fields, both in interviews and at conferences and workshops. In particular, at *Government Workshop 14*, discussions focused

on China's GEI initiative (which was discussed in Chapter 6). In the context of this workshop the value GEI and BRI could add to power sector modernisation and coordination in ASEAN was a dominant focus. Primary recommendations and considerations included utilisation of GEI and BRI to aid in the necessary standardisation of grid codes and physical needs necessary for EMI to take place in ASEAN's primary APG interconnections. While these issues are already being addressed among ASEAN members, workshop discussion focused on the limited priority standardisation and coordination seem to have at the national level, and the role knowledge sharing via GEI and BRI could play. China, it was pointed out in discussions, is already an important global example of electricity grid development and integration, with many successful and pending projects on national grid interconnection and synchronisation with internet connectivity (Chen, 2018). Exploration of the physical and structural needs of electricity grids at the national level, and Chinese investment and knowledge-sharing, would be beneficial for the broader knowledge base in this area. The author has, in fact, received two offers for deeper knowledge exchange with GEI representatives, aiding in potential development of future projects in this area given access opportunities. Additional BRI considerations could include analysis of China's 'greening the economy' initiatives, which would add useful economic analysis.

Finally, given the repeated emergence of power in this research, an additional approach for further research is to focus more specifically on the power dynamics entrenched in China's BRI investment in ASEAN's power sector. By taking a more specific look at structural and relational power in ASEAN-China power sector cooperation, theoretical contributions could potentially span not just IPE but international relations or politics & government as well, broadening the application of this research across fields. In addition, an examination of power dynamics between specific ASEAN member economies and China could prove quite political and timely given sub-regional and regional affairs (for example, Myanmar and China in light of border conflict and hydropower concerns).

## 7.5.c Moral IPE Considerations

An additional consideration arose via conversations about the IPE of crossborder electricity trade with political economists in a variety of fields. While attending *Academic Workshop 1*, the author was introduced to the concept of Moral Political Economy (Moral IPE). This concept was expanded upon further at *Academic Workshop 2*, where the author had the opportunity to explore ways in which moral IPE could be applied in energy sectors. The following description expands on some of the ideas introduced at these two workshops, framing the author's thoughts for additional IPE research across energy policy fields.

Mainstream neoliberal economic policy recommendations may make sense in the context of electricity access and grid integration of variable renewable energy resources, but they don't necessarily account for the broader, underlying issues of economic development, continued industrialisation, and energy intensity that have led East Asia to a place that calls for increased renewables in the first place. If global climate goals, specifically emissions reductions, are to be met, capitalistic, neoliberal systems that encourage development-as-is do not appear conducive to long term changes to the global political economy. "Neoliberal market restructuring", whereby neoliberal norms are placed not just on a market, in this case electricity in nature, but trickle down into society and all aspects of management and socialisation (Wiegratz, 2016), is clearly visible proposed. How can liberalisation in the reform practices these recommendations be further critiqued in the context of East Asian development practices, current political-economy structures of regional electricity markets, and broader climate-related goals? Moral political economy provides a lens through which the economic order and relations of an economy are viewed with moral dimensions, preconditions and implications that are reflected in economic relations and practices of governments (Wiegratz, 2016).

Based on the analysis conducted in this research it has become more apparent to the author that the reform proposals and neoliberal market restructuring aimed at ASEAN electricity markets does not reflect the moral political economy of the region or sub-regional organisation represented in this context, particularly given the focus on climate change responses. Moral political economy creates space for the environment in political and economic conversations; it prioritises the moral considerations societies (and economies) base their actions on, instead of the traditional, neoliberal economic ideology that prioritises growth and consumption as a means to improve quality of life but on a seemingly unending scale (Sayer, 2016).

This, of course, would be a hard pill for any national government or regional institution to swallow—values can be arbitrary, differ from person-to-person or institution-to-institution. A neoliberal may place importance on the values of free and fair trade, whereas a Marxist may place more value on the wellbeing of workers and workers' rights amid neoliberal structures of labour. However, refining of values and goals is at the very root of ASEAN, so, then, perhaps a refinement of economic, political, and energy goals would better reflect not only

the energy transition imperative but also the state-driven structures that already exist in the region. Thus, a neo-developmental state model that places clean energy transitions at the top of the priority list may be just such a structure where moral political economy decisions are incorporated into policy choices at the national and also sub-regional and regional level. This would allow a better alignment of ASEAN goals with national goals but would require political will and incentives to occur. This aspirational, moral political economy-policy making aligns with market factors, interests and perceptions discussed across chapters 4-6, most notably the market and political realities within ASEAN and BIMP-EAGA and their shortcomings. Further research that examines aspirational policy making, moral IPE, and collectivist Asian values would be a unique addition to the canon of political economy-based energy research and offer additional insight into policy making within the ASEAN context. However, time limits and research parameters established at the outset make the exploration of moral IPE outside the scope of this research and instead an avenue for future IPE research in this area.

#### 7.5.d Grid-Connected Distributed Generation

The electricity focus of this research was centralised grids. However, gridconnected distributed generation is being deployed at increasing rates globally and holds relevance to ASEAN's own geographical challenges. As a sub-region made up of a variety of island nations and rural communities with limited grid access, distributed generation holds a lot of promise for increased electricity access and energy poverty targets in the sub-region. Distributed generation refers to small-scale renewable energy technology "options that connect to the electrical distribution network" including bioenergy, small-scale wind, photovoltaics (PV), and others (Passey, et al., 2011). Research into the challenges associated with grid-connected distributed generation exists, however, expansion into its relevance and role in the APG and ASEAN power connectivity targets is limited. Exploration into the minutia of grid-connected distributed generation in ASEAN would be a relevant and timely area of research with potential disruptive impacts on sub-regional and regional targets, as well as opportunities for innovative financing mechanisms among rural communities. Questions regarding distributed generation arose when the author was presenting research findings (Development Bank Conference 4) and would be a logical direction for this research to go if focus remains on ASEAN and the challenges associated with national versus sub-regional targets and interests.

## 7.6 Policy Implications

A number of policy implications have emerged through the key themes subregional market factors, national market factors, and governance challenges. Some of these are standard energy policy recommendations, such as: better coordination of policy goals and initiatives with sub-regional and national competencies, a consolidation of needs across sectors, and prioritisation of deregulation in order to encourage other power sector needs, such as liberalisation and increased FDI. These recommendations are reflected in the literature on electricity market integration, they are reflected among the opinions of elites interviewed, and they are reflected in the work of organisations and multi-national institutions on the ground in the region pushing for EMI. What is clear from the consistency in responses is that these needs are not being adequately addressed or at a pace in line with sub-regional and national energy goals. The energy transition towards less reliance on hydrocarbon resources and higher utilisation of renewable energy sources, with higher levels of efficiency, could be improved.

The primary policy implications identified by this author, based on data and interview analysis described in the previous pages, are as follows:

- 1. National: Physical infrastructure development, market liberalisation, and refinement of market design are all necessary at national levels. Without these three overarching adjustments EMI progress will not be realised among ASEAN members and BIMP-EAGA countries. Decarbonisation of national power sectors should also be a prioritised. Current national interconnection efforts appear to be driven by economics, not decarbonisation goals and climate targets.
- 2. Sub-Regional Alignment: National priorities and sub-regional targets are not aligned in policy responses at either level. Better coordination and development of join initiatives / support policies (political and economic / financial) could help in this area, but, ultimately, national needs may not align with sub-regional targets for many years. Regionally engagement in the global, neoliberal market system remains a priority.
- 3. Technical Concerns: More technical targets, including enhanced capacity, sub-regional coordination, including grid codes, sub-regional transmission systems, and other sub-regional jurisdiction considerations are secondary to national power sector reform. These needs are years away among BIMP-EAGA countries if EMI is to be realised.

- 4. Power Dynamics & Political Will: Strong political and economic relationships need to be present in tandem with national power sector reform. These trusting relationships are a necessary by-product for national and sub-regional reform measures to be realised. Political and economic power dynamics will be challenged if EMI measures are realised; this will be difficult in the short term but have benefits for the sub-regional and global energy transition. The importance of political will cannot be over emphasised.
- 5. Global Climate Targets: sub-regional, regional, and global climate targets are not fully integrated into national policy making and local level concerns. Engagement in global and sub-regional climate commitments is performative.
- 6. IPE Explanation: neo-developmental statism provides a long-term, development paradigm by combining power and equity to provide context to sub-regional and regional energy and climate priorities. It is a mistake to expect or predict power sector reforms on the trajectory towards a fully liberalised power sector instead of a hybrid model, which is in fact the legitimate form of East Asian developmentalism. In this regard neo-developmental statism has moved on one step forward from the traditional developmental state model and offers a better explanation of economic and political decision making in Southeast Asia.

These policy implications can be summarised around the above key issues as follows:



## Figure 7.1 Policy Implications

These power sector needs and the reforms necessary to achieve them are one piece of the energy transition and global responses to climate change. But they are reflective of the priority of responses to climate challenges, and the way in which economic and market factors take precedence within national policy decisions in this area.

# 7.7 Concluding Remarks

There is, from this author's reading of the materials and data collected, a further, but more complicated implication of the sub-regional EMI development in Southeast Asia: the identified needs and reforms necessary for deeper energy market integration are, ultimately, liberal in nature, and reflect the needs of a neoliberal market system; decarbonisation is itself reliant on neoliberal market reform. Unfortunately, national governments seem to be prioritising the continued use of hydrocarbon energy resources and selling of excess electricity across borders for economic reasons, not energy transitions reasons. The neoliberal, market-driven push for EMI appears to be a contradictory dichotomy given the responsibility for climate and environmental challenges that falls upon the neoliberal market system. Industrialisation, carbonisation of world economies, and global trade development are a part of the cycle that has created climate imperatives. Where increased share of renewables can play a role in responding to these imperatives, neoliberal economic policies are also a necessary side effect—and, in fact, take priority over the renewable energy transition.

This intersection of power system development and economic development poses the question of whether or not economic development can actually continue as is, with market reforms and technological advancements that also lessen environmental impact while simultaneously contributing to improvements in access. Perhaps the solution is not, in fact, deeper integration and reform alone, but instead a restructuring of priorities, placing environment, climate and equity above even electricity access where it may be detrimental to sustainable choices.

The data analysis of the BIMP-EAGA case study resulted in the findings that market realities do not necessarily reflect a current positive outcome for increased EMI and cross-border electricity trade in Southeast Asia. This is for two reasons in particular: (1) decarbonisation of the power sectors, via increased shares of renewables for electricity generation, is not evident via the current sub-regional and subsystem energy mix; coal is still the dominant resource in individual markets and projected to continue as the dominant fuel source. (2) Because, ultimately, there is very little (if any) gap between electricity needs and domestic supply—according to data collected and interviews (*Former Government Informant A1, Academic A23, Academic A26, Development Bank A25*), each country is able to supply its own electricity needs based on 2016 rates, however, this could change depending on future added electricity demand (Zheng, et al., 2017).<sup>81</sup> As *Development Bank A25* said: "The need is not [there] for demand—most countries can provide for their [own] needs. There are not many exceptions, and those are the emerging countries." Combine this reality with point number 1, and there is no incentive for larger structural change if ASEAN and BIMP-EAGA economies are ultimately providing for electricity needs via already embedded hydrocarbon sources. ASEAN, instead, is providing a performative role in its sub-regional targets, while national economies are aimed at economic gains for partial engagement in these initiatives.

These two points, coal dominance and electricity needs, ultimately signal that ASEAN's BIMP-EAGA subsystem is not, system-wide, committed to immediately decarbonising electricity sectors and energy mixes. Instead of short-term necessities these goals have been relegated as medium to long-term ambitions with market realities that echo these findings. Fear of short-term energy security, economic impact, and distrust are driving national policy, instead of sub-regional commitments to engage in relevant energy policy changes as a response to man-made climate change and CO<sub>2</sub> emissions. This research, while not focused solely on energy diversity, but the usefulness of cross-border electricity trade for responding to the variability, uncertainty and flexibility challenges of increased renewable energy integration, exposes what this author now considers an uncomfortable truth: cross-border electricity for renewable energy integration is not the same priority as cross-border electricity trade for economic gain. The selling of excess electricity and continued use of hydrocarbon resources in said electricity generation is a very real possibility for EMI in Southeast Asia. Combining the moral, climate-driven choices for subregional, regional and global energy transformation is a necessary addition to sub-regional and regional EMI initiatives.

<sup>&</sup>lt;sup>81</sup> According to the IEA (2019b) renewables will only meet roughly 1/3 of the region's total electricity demand by 2030, with coal remaining the dominant resource unless regional and domestic policy push for renewables changes dramatically.

This research has combined the barriers to EMI discussed in chapters 4-6 to illustrate the competing interests at both sub-regional and national levels of EMI. A variety of different reforms have been proposed to address these barriers. These political and economic reforms were observed via interviews and data collection to be realistic but contradictory in how they are carried out at different levels. In addition to the contradiction between sub-regional vs national level policy implementation and priority, there is the possibility of certain populations being left behind in the benefits to be gained from increased renewable energy integration and electricity market expansion and integration. As such, neodevelopmental statism can combine power and equity in explaining energy and climate priorities; perhaps, ultimately, this is an unrealistic call for action considering the dominant sub-regional, regional and global neoliberal economic system and the climate change mitigation linkages discussed.

## **Bibliography**

Aaron, C. 2011. *Renewable Energy Subsidies and the WTO: The Wrong Law and the Wrong Venue*. Geneva: Global Subsidies Initiative.

Abbott, K.W. & Snidal, D. 2000. Hard and Soft Law in International Governance. *International Organization*. 54(3), pp. 421-456.

Abbott, K.W., Keohane, R.O., Moravcsik, A., Slaughter, A.M., & Snidal, D. 2000. The Concept of Legalization. *International Organization*. 54(3), pp. 401–419.

Aberbach, J.D. & Rockman, B.A. 2002. Conducting and Coding Elite Interviews. *Political Science and Politics.* 35(4), pp. 673-676.

ACE. 2013a. ASEAN Energy Market Integration (AEMI): From Coordination to Integration. AEMI Group. Bangkok: ASEAN Studies Center.

ACE. 2013b. ASEAN Guideline on Off-grid Rural Electrification Approaches. Jakarta: ACE.

ACE. 2017. The 5<sup>th</sup> ASEAN Energy Outlook (AEO) 2015-2040. Jakarta: ACE.

ACE. 2018a. *ASEAN Power Grid*. [Online]. [Accessed 04 August 2018]. Available from: <u>http://www.aseanenergy.org/programme-area/apg/</u>

ACE. 2018b. Provision of the ASEAN Interconnection Masterplan Study (AIMS) III. Jakarta: ACE.

ACE. 2019. *About ACE.* [Online]. [Accessed 25 May 2019]. Available from: <u>http://www.aseanenergy.org/</u>

Acharya, A. 2009. *Whose Ideas Matter? Agency and Power in Asian Regionalism.* Ithaca: Cornell University Press.

Acharya, A. 2018. The Evolution and Limitations of ASEAN Identity. *Building ASEAN Community: Political–Security and Socio-cultural Reflections.* 4, pp. 24-38.

ADB & ADBI. 2009. Infrastructure for a Seamless Asia. Tokyo: ADBI.

ADB. 2008a. Emerging Asian Regionalism. Manila: ADB.

ADB. 2008b. Key Indicators for Asia and the Pacific 2008. Manila: ADB.

ADB. 2012. ADB GMS Power Trade and Interconnection: Two Decades of Cooperation. Manila: ADB.

ADB. 2013. ADB's First BIMP-EAGA Project to Bring Clean Energy to West Kalimantan. News Release. Manila, Philippines: ADB.

ADB. 2013. Asian Development Outlook 2013: Asia's Energy Challenge. Manila: ADB.

ADB. 2014. An Evaluation of the Prospects for Interconnections Among the Borneo and Mindanao Power Systems. Manila: ADB.

ADB. 2015. Fossil Fuel Subsidies in Asia. Manila: ADB.

ADB. 2016a. Fossil Fuel Subsidies in Asia: Trends, Impacts, and Reforms. Manila: ADB.

ADB. 2016b. Greater Mekong Subregion: Energy Sector Assessment, Strategy, and Road Map. Manila: ADB.

ADB. 2017. Grid Unlocked: The Mechanics of Cross-Border Electricity Sharing. *Development Asia*. Manila, Philippines: ADB.

ADBI. 2020. Energy Insecurity and Renewable Energy Policy: Comparison Between the People's Republic of China and Japan. ADBI Working Paper 1085, pp. 1-17.

Adu, P. 2019. A Step-by-Step Guide to Qualitative Data Coding. London: Routledge.

Agyeman, J., Bullard, R.D., & Evans, B. 2003. *Just Sustainabilities: Development in an Unequal World.* Boston: MIT Press.

Ahmed, A. & Othman, H.M. 2014. Electricity Consumption in Brunei Darussalam: Challenges in Energy Conservation. *International Energy Journal.* 14, pp. 155-166.

Ahmed, T., Mekhilef, S., Shah, R., Mithulananthan, N., Seyedmahmoudian, M., & Horan, B. 2017. ASEAN Power Grid: A Secure Transmission Infrastructure for Clean and Sustainable Energy for South-East Asia. *Renewable and Sustainable Energy Reviews*. No. 67, pp. 1420-1435.

AIMS Study Group. 2003. ASEAN Interconnection Master Plan Study. ASEAN: Jakarta, Indonesia.

Akaha, Tsuneo. 2004. Non-traditional Security Cooperation in Northeast Asia. In: Thakur, R. and Newman, E. eds. *Broadening Asia's Security Discourse and Agenda*. New York: United Nations University Press.

Albert, E. 2017. ASEAN: The Association of Southeast Asian Nations. *CFR Backgrounder.* Washington, D.C.: Council on Foreign Relations.

Aldy, J.E. & Stavins, R.N. 2009. *Post-Kyoto International Climate Policy: Summary for Policymakers*. Cambridge: Cambridge University Press.

Ambashi, M. 2017. ASEAN as an FDI Attractor: How do Multinationals Look at ASEAN? *ERIA Policy Brief.* 04, pp. 1-8.

Amsden, A. 1989. *Asia's Next Giant: South Korea and Late Industrialization.* New York: Oxford University Press.

Amsden, A. 1991. "Diffusion of Development: The Late-Industrializing Model and Greater East Asia." *American Economic Review* 81(2), pp. 282–6.

Amsden, A. 1994. "Why Isn't the Whole World Experimenting with the East Asian Model to Develop? Review of The East Asian Miracle." *World Development* 22(4), pp. 627–33.

Amsden, A. 2001. *The Rise of "The Rest": Challenges to the West from Late-Industrializing Economies*. New York: Oxford University Press.

Amsden, A., & Chu, W.W. 2003. *Beyond Late Development: Taiwan's Upgrading Policies*. Cambridge, MA: MIT Press.

Amundsen, E. & Bergman, L.O. 2006. Why has the Nordic electricity market worked so well? *Utilities Policy.* 14(3), pp. 148-157.

Amundsen, E. & Bergman, L.O. 2007. Provision of Operating Reserve Capacity: Principles and Practices on the Nordic Electricity Market. *Competition and Regulation in Network Industries.* 2(1), pp. 73-98.

Anderson, J.A. 2009. Electricity Restructuring: A Review of Efforts around the World and the Consumer Response. *The Electricity Journal.* 22(3), pp.70-86.

Andrew Hurrell. 2013. Effective Multilateralism and Global Order. In: Prantl, J. ed. *Effective Multilateralism: Through the Looking Glass of East Asia*. Basingstoke: Palgrave Macmillan, pp. 21-42.

Andrews-Speed, P. & Hezri, A. 2013. Institutional and Governance Dimension of ASEAN Energy Market Integration. In: ASEAN Studies Centre. *ASEAN Energy Market Integration (AEMI): From Coordination to Political Integration*. Bangkok: ASEAN Studies Centre, pp. 149-174.

Andrews-Speed, P. & Len, C. 2013. The Political Economy of ASEAN Energy Market Integration. In: ASEAN Studies Centre. *ASEAN Energy Market Integration (AIMI): From Coordination to Political Integration*. Bangkok: ASEAN Studies Centre, pp. 197-213.

Andrews-Speed, P. 2016. Connecting ASEAN Through the Power Grid: Next Steps. *Energy Studies Institute Policy Brief*. 11, pp. 1-4.

Andrews, R. 2003. Research Questions. London: Continuum.

Ang, G. 2015. Overcoming Barriers to International Investment in Clean Energy. *OECD Investment Insights.* Paris: OECD.

Angela, M. 2020. China's South China Sea Moves Raise Concerns. *The ASEAN Post*. [Online]. 1 May. [Accessed 1 May 2020]. Available from: <u>https://theaseanpost.com/</u>

Antweiler, W. 2016. Cross-Border Trade in Electricity. *Journal of Economics*. 101, pp. 42-51.

Antweiler, W. Copeland, B.R., & Taylor, M.S. 2001. Is Free Trade Good for the Environment? *The American Economic Review*. Pp. 877-908.

APAEC. 1999. ASEAN Plan of Actions for Energy Cooperation 1999-2004. Jakarta: ASEAN Secretariat.

APAEC. 2004. ASEAN Plan of Actions for Energy Cooperation 2004-2009. Jakarta: ASEAN Secretariat.

APAEC. 2010. ASEAN Plan of Actions for Energy Cooperation 2010-2015. Jakarta: ASEAN Secretariat.

APEC. 2012. *Reforming Fossil-Fuel Subsidies to Reduce Waste and Limit CO*<sub>2</sub> *Emissions while Protecting the Poor.* APEC Energy Working Group: Singapore.

Apergis, N. & Payne, J.E. 2011. A Dynamic Panel Study of Economic Development and the Electricity Consumption-Growth Nexus. *Energy Economics.* 33, pp. 770-781.

Arat, Z.F. 1998. Democracy and Economic Development: Modernisation Theory Revisited. *Comparative Politics*. 12(2), pp. 21-36.

Araya, M. 2016. Can the Environmental Goods Agreement Help Advance the Paris Agreement and Sustainable Development Goals? *International Centre for Trade and Sustainable Development*. [Online]. [Accessed 05 December 2017]. Available from: <u>https://www.ictsd.org/</u>

Archer, M., Bhaskar, R., Collier, A., Lawson, T., & Norrie, A. eds. 1998. *Critical Realism: Essential Readings.* Oxon: Routledge.

Archer, M., Sharp, R., Stones, R., & Woodiwiss, T. 1999. Critical Realism and Research Methodology. *Alethia*. 2(1), pp. 12-16.

Arias-Maldonado, M. 2013. Rethinking Sustainability in the Anthropocene. *Environmental Politics.* 22, pp. 428–446.

Arndt, H.W. & Hill, H. eds. 1999. *Southeast Asia's Economic Crisis.* Singapore: Institute of Southeast Asian Studies.

ASEAN. 1997. *ASEAN Vision 2020.* [Online]. [Accessed 13 October 2018]. Available from: <u>https://asean.org/</u>

ASEAN. 2004. MPAC. (AIMS I). Jakarta: ASEAN Secretariat.

ASEAN. 2007. 2007 Memorandum of Understanding on the ASEAN Power Grid. Jakarta, Indonesia: ASEAN.

ASEAN. 2010. MPAC (AIMS II). Jakarta: ASEAN Secretariat.

ASEAN. 2011. MPAC 2025. Jakarta: ASEAN.

ASEAN. 2012. BIMP-EAGA Implementation Blueprint 2012-2016. Jakarta: ASEAN.

ASEAN. 2012b. Economic Achievement. [Online]. [Accessed 13 October 2019]. Available from: <u>https://asean.org/</u>

ASEAN. 2015. ASEAN Economic Community Blueprint 2025. Jakarta: ASEAN.

ASEAN. 2016. MPAC 2025. Jakarta: ASEAN.

ASEAN. 2017a. MPAC 2025. Jakarta: ASEAN.

ASEAN. 2017b. The ASEAN Electricity Exchange (AEE): An International Perspective. Jakarta: HAPUA-UNESCAP.

ASEAN. 2020. Economic Achievement. [Online]. [Accessed 14 October 2019]. Available from: <u>https://asean.org/</u>

Asia Pacific Energy Research Centre. 2004. Electric Power Grid Interconnections in the APEC Region. Asia Pacific Energy Research Centre: Tokyo, Japan.

Atanassova-Cornelis, E. 2017. Northeast Asia's Evolving Security Order: Power, Politics, Trust Building, and the Role of the EU. *IAI Working Papers.* 17, pp. 1-12.

Atkinson, A. 1991. *Principles of Political Ecology.* London: Belhaven Press.

Atmo, G. U. & Otsuki, T. 2018. ASEAN Power Grid Cooperation: Cross-Border Electricity Trade and its Influence on Carbon Emissions. Tokyo: International Association of Energy Economics Japan.

Attia, M. & Edge, J. 2017. Be(com)ing a Reflexive Researcher: A Developmental Approach to Research Methodology. *Open Review of Education Research*. 4(1), pp. 33-45.

Attia, M. & Edge, J. 2017. Be(Com)Ing A Reflexive Researcher: A Developmental Approach to Research Methodology. *Open Review of Educational Research*. 4(1), pp. 33-45.

Axelrod R.M. 1984. The Evolution of Cooperation. New York: BASIC Books.

Ayers, R.U., van den Bergh, J.C.J.M. & Gowdy, J.M. 2001. Weak Versus Strong Sustainability. *Environmental Ethics.* 23(2), pp. 155-168.

Bacon, R.W. & Besant-Jones, J. 2001. Global Electric Power Reform, Privatization, and Liberalization of the Electric Power Industry in Developing Countries. Washington, D.C.: The International Bank for Reconstruction and Development and The World Bank.

Bahar, J. & Sauvage, J. 2013. Cross-border Trade in Electricity and the Development of Renewables-based Electric Power: Lessons from Europe. [Online]. *OECD Trade and Environment Paper*. 2013(02). [Accessed 25 November 2014]. Available from: <u>http://www.oecd.org/</u>

Baier, A. 1986. Trust and Antitrust. *Ethics*. 96, pp. 231-260.

Bailey, I., Gouldson, A. and Newell, P. 2011. Ecological Modernisation and the Governance of Carbon: A Critical Analysis. *Antipode.* 43(2), pp. 682-703.

Ban, C. 2012. A New Developmental State? April 24, 2012. European Economics.[Online].[Accessed 21 August 2019]. Available from:https://europeaneconomics.wordpress.com/

Ban, C. 2013. Brazil's Liberal Neo-Developmentalism: New Paradigm or Edited Orthodoxy? *Review of International Political Economy*. 20(2), pp. 298-331.

Barber, B. 1983. *The Logic and Limits of Trust.* New Brunswick: Rutgers University Press.

Bardhan, P. 1997. Corruption and Development: A Review of Issues. *Journal of Economic Literature*. 35(3), pp. 1320-1346.

Barker, J.Jr., Tenenbaum, B. & Woolf, F. 1997. Governance and Regulation of Power Pools and System Operators. *World Bank Technical Paper*. 382, pp. 1-82.

Barlow, C., ed. 1999. *Institutions and Economic Change in Southeast Asia.* Cheltenham: Edward Elgar Publishing House.

Bartle, I. 2007. The Political Economy of Power Sector Reform: The Experience of Five Major Developing Countries. *Capital & Class.* 98, pp. 165-168.

Bauer, S., Brandi, C., Pauw, P., Richerzhagen, C., & Schmole, H. 2014. *Different Perspectives on Differentiated Responsibilities.* [Online]. Bonn: German Development Institute. [Accessed 25 November 2014]. Available from: <u>http://www.die-gdi.de/</u>

Beeson, M. & Pham, H.H. 2012. Developmentalism with Vietnamese Characteristics: The Persistence of State-Led Development in East Asia. *Journal of Contemporary Asia.* 42(4), pp. 539-559.

Beeson, M. 2003. Sovereignty Under Siege: Globalisation and the State in Southeast Asia. *Third World Quarterly.* 24(2), pp. 357-374.

Beeson, M. 2009. Contemporary Southeast Asia. Basingstoke: Palgrave.

Beeson, M. 2014. *Regionalism and Globalization in East Asia: Politics, Security, and Economic Development.* Basingstoke: Palgrave Macmillan.

Beeson, M.K. 2006. Politics and Markets in East Asia: Is the Developmental State Compatible with Globalisation? In: Stubbs, R. & Underhill, G.R.D. eds. *Political Economy and the Changing Global Order.* Ontario: Oxford University Press, pp. 443-453.

Beijing Foreign Languages Institute. 1979. *The Pinyin Chinese-English Dictionary.* Beijing: The Commercial Press, LTD.

Bernstein, H. 1971. Modernization Theory and the Sociological Study of Development. *The Journal of Development Studies*. 7(2), pp. 141-160.

Berry, J.M. 2002. Validity and Reliability Issues in Elite Interviewing. *Political Science and Politics.* 35(4), pp. 679-682.

Bhagwati, J. 1995. US Trade Policy: The Infatuation with FTAs. Discussion Paper Series No. 726. [Online]. American Economic Association meetings, January 1995, Washington, D.C. [Accessed 01 July 2017]. Available from: <u>http://www.columbia.edu/</u>

Bhaskar, R. 1979. *The Possibility of Naturalism: A Philosophical Critique of the Contemporary Human Sciences*. Atlantic Highlands: Humanities Press.

Bhaskar, R. 1997. *A Realist Theory of Science*. 2<sup>nd</sup> ed. London: Verso.

Bhattacharya, A. & Kojima, S. 2008. Impact of Cross-Border Energy Infrastructure Investment on Regional Environment, Society and Climate Change. *Infrastructure for Asia.* Tokyo: Asian Development Bank Institute.

Bhattacharya, A. & Kojima, S. 2010. Economic Impact Analysis of East Asia Energy Market Integration. In: Shi, X. & Kimura, F. eds. *Energy Market Integration in the East Asia Summit Region: Review of Initiatives and Estimation of Benefits.* Jakarta: ERIA.

Bhattacherjee A. 2012. Social Science Research: Principles, Methods, and Practices. CreateSpace Independent Publishing Platform.

Biernacki P, Waldorf D. 1981. Snowball Sampling—Problems and Techniques of Chain Referral Sampling. *Social Methods Research*. 10(2), pp. 141–63.

Bishop, M.L. & Payne, A. 2018a. Introduction: Why Revisit the Developmental State? In: Bishop, M.L., Payne, A., Sen, K., Breslin, S., Öniş, Z., Muzaka, V., Booth, D., Lindsay, C., and Yeung, H.W-C., eds. 2018. Revisiting the Developmental State. *Sheffield Political Economy Research Institute.* Sheffield: The University of Sheffield.

Bishop, M.L. & Payne, A. 2018b. Conclusion: Where Does the Developmental State Stand Now? In: Bishop, M.L., Payne, A., Sen, K., Breslin, S., Öniş, Z., Muzaka, V., Booth, D., Lindsay, C., & Yeung, H.W-C., eds. 2018. Revisiting the Developmental State. *Sheffield Political Economy Research Institute.* Sheffield: The University of Sheffield.

Bishop, M.L. Payne, A., Sen, K. Breslin, S. Onis, Z., Muzaka, V., Booth, D. Lindsay, C., & Yeung, H.W.C. 2018c. Revisiting the Developmental State. Sheffield Political Economy Research Institute. SPERI Paper 43.

Blakers, A.W., Luther, J., & Nadolny, A. 2012. Asia Pacific Super Grid—Solar Electricity Generation, Storage and Distribution. *Green*. 2, pp. 189–202.

Boffey, D. 2020. Brexit: EU to Start Talks with UK over Future Relationship on 3 March. *The Guardian.* [Online]. [Accessed 01 February 2020]. Available from: <u>https://www.theguardian.com/</u>

Booth, D. 2011. Introduction: Working with the Grain? The Africa Power and Politics Programme. *IDS Bulletin.* 42(2), pp. 1–10.

Bourke, B. 2014. Positionality: Reflecting on the Research Process. *The Qualitative Report*. 19(33), pp. 1-9.

Bowen, A. & Fankhauser, S. 2011. The Green Growth Narrative: Paradigm Shift or Just Spin? [Online]. *Global Environmental Change*. [Accessed 03 May 2016]. 21(4), pp. 1157-1159. Available from: <u>http://ac.els-cdn.com/</u>

Brander, J. & Spencer, B. 1985. Export Subsidies and International Market Share Rivalry. *Journal of International Economics*. 18(1-2), pp. 83-100.

Brander, J. 1986. Rationales for Strategic Trade and Industrial Policy. In Krugman, P.R. ed. 1986. *Strategic Trade Policy and the New International Economics*. Massachusetts: The MIT Press, pp. 23-46.

Braun, V. & Clarke V. 2006. Using Thematic Analysis in Psychology. *Qual. Res. Psych.* 3, pp. 77–101.

Brekke, K.A.. 1997. *Economic Growth and the Environment. On the Measurement of Income and Welfare.* Cheltenham: Edward Elgar Publishing.

Brenkert, G. G. 1998. Trust, Morality, and International Business. In: Lane, C. & Bachmann, R. eds. *Trust Within and Between Organizations: Conceptual Issues and Empirical Applications*. New York: Oxford University Press, pp. 273–97.

Breslin, S. 2018. Fitting China in to the Debate. In: Bishop, M.L., Payne, A., Sen, K., Breslin, S., Öniş, Z., Muzaka, V., Booth, D., Lindsay, C., & Yeung, H.W-C. 2018. Revisiting the Developmental State. *Sheffield Political Economy Research Institute.* Sheffield: The University of Sheffield.

Bresser-Pereira, L.C. & Theuer, D. 2012. Latin America: After the Neoliberal Years, Is the Developmental State Back In? Paper presented in the panel models of development in Latin America: continuity and change under the left turn. LASA-2012 Congress, 25 May, San Francisco, California.

Bresser-Pereira, L.C. 2006. The New Developmentalism and Conventional Orthodoxy. Économie Appliquée. 59(3), pp. 95–126.

Bresser-Pereira, L.C. 2007a. Macroeconomics of Stagnation and New Developmentalism in Latin America. In: Forstater, M. & Wray, R. L. eds. *The General Theory after 70 Years*. Houndsmills: Palgrave Publishing, pp. 145-174.

Bresser-Pereira, L.C. 2007b. The New Developmentalism and Conventional Orthodoxy. *Iberoamericana: Nordic Journal of Latin American and Caribbean Studies*. XXXVII(1), pp. 31–58.

Bresser-Pereira, L.C. 2008. Globalization, Nation-State and Catching Up. *Brazilian Journal of Political Economy*. 28(4), pp. 557–576.

Bresser-Pereira, L.C. 2009. From Old to New Developmentalism in Latin America. In: Ocampo, J.A. & Ros, J. eds. *The Oxford Handbook of Latin America Economics.* Oxford: Oxford University Press, pp. 108–30.

Bresser-Pereira, L.C. 2010a. The 2008 Financial Crisis and Neoclassical Economics. *Revista de Economia Pol'ítica*. 30(1), pp. 3–26.

Bresser-Pereira, L.C. 2010b. The Global Financial Crisis, Neoclassical Economics and the Neoliberal Years of Capitalism. *Revue de la r'egulation Capitalisme, institutions, pouvoirs* (7).

Broadbent, J. 1998. *Environmental Politics in Japan*. United Kingdom: Cambridge University Press.

Brohman, J. 1996. Postwar Development in the Asian NICS: Does the Neoliberal Model Fit Reality? *Economic Geography.* 72(2), pp. 107-130.

Brown, A. 2014. Critical Realism in Social Research: Approach with Caution. *Work, Employment and Society.* 28(1), pp. 112-123.

Brown, E., Cloke, J., Gent, D., Johnson, P.H. & Hill, C. 2014. Green Growth of Ecological Commodification: Debating the Green Economy in the Global South. *Geografiska Anneler: Series B, Human Geography*. 96(3), pp. 245-259.

Brown, W. 2005 *Edgework*: *Critical Essays on Knowledge and Politics*. Princeton: Princeton University Press.

Bryman, A. 2012. Social Research Methods. Oxford: Oxford University Press.

Buchanan, J.M. 1980. Rent-Seeking and Profit-Seeking, in J. Buchanan, R.D. Tollison and G. Tullock (eds.), Towards a Theory of the Rent-Seeking Society. College Station: Texas A&M University Press.

Bultheel, C., Morel, R., & Alberola, E. 2016. Climate Governance & the Paris Agreement: The Bold Gains of Transnational Cooperation. *Institute for Climate Economics.* 40, pp. 1-12.

Burke, P.J. & Kurniawati, S. 2018. Electricity Subsidy Reform in Indonesia: Demand-Side Effects on Electricity Use. Australian National University. Working Paper 2018(01), pp. 1-35.

Butler, N. 2020. We Need a World Institutions for Climate and Energy. *Financial Times.* [Online]. [Accessed 24 February 2020]. Available from: <u>http://www.ft.com/</u>

Cai, K.G. 2008. The Political Economy of East Asia. New York: Palgrave Macmillan.

Cammack, P. 1990. Statism, New Institutionalism, and Marxism. In: Miliband, R. & Panitch, L. eds. *Socialist Register 1990: The Retreat of the Intellectuals.* 26, pp. 148-172.

Campbell, K.M. & Doshi, R. 2020. The Coronavirus Could Reshape Global Order. *Foreign Affairs.* [Online]. [Accessed 20 March 2020]. Available from: <u>https://www.foreignaffairs.com/</u>

Campinha-Bacote, J. 1994. Cultural competence in psychiatric mental health nursing: A conceptual model. *Journal of Nursing Clinics of North America*. 29, pp. 1-8.

Campinha-Bacote, J. 2002. The process of cultural competence in the delivery of healthcare services: A model of care. *Journal of Transcultural Nursing*. 13, pp. 181-184.

Caprotti, F. 2009. China's Cleantech Landscape: The Renewable Energy Technology Paradox. [Online]. *Sustainable Development Law & Policy*. 9(3), pp. 6-10. [Accessed 14 September 2014]. Available from: http://digitalcommons.wcl.american.edu/ Carey, M. 2017. *Mistrust: An Ethnographic Theory*. Chicago: Hau Books.

Carliner, G. 1986. Industrial Policy for Emerging Industries. In Krugman, P.R., ed. 1986. *Strategic Trade Policy and the New International Economics*. Boston: The MIT Press, pp. 147-168.

Carroll, T. 2018. Context and Conflict: The Political Economy of Southeast Asia's Development from Independence to Hyperglobalisation. *Southeast Asia Research Centre Working Paper Series*. 195, pp. 3-45.

Castillo-Montoya, M. (2016). Preparing for Interview Research: The Interview Protocol Refinement Framework. *The Qualitative Report*. 21(5), pp. 811-831.

Cavanagh S. 1997. Content Analysis: Concepts, Methods and Applications. *Nursing Research*. 4, pp. 5–16.

Cerny, P. 2005. Capturing Benefits, Avoiding Losses: The United States, Japan, and the Politics of Constraint. In: Soederberg, S. ed. *Internalizing Globalization: The Rise of Neoliberalism and the Decline of National Varieties of Capitalism.* London: Palgrave, pp. 1-32.

Chan, C.M.V. 2006. A New Pathway for Regional Governance in East Asia: From Economic-driven Regionalization to Crisis-driven Regionalism. *Ritsumeikan Journal of Asia Pacific Studies*. 1, pp.149-161.

Chan, S., Clark, C. & Lam, D. 1998. Looking Beyond the Developmental State. In: Chan, S., Lutz, H., Lam, D., & Clark, C. eds. *Beyond the Developmental State: East Asia's Political Economies Reconsidered.* London: Palgrave Macmillan, pp. 1-8.

Chandra, A. 2016. Regional Economic Community Building Amidst Rising Protectionism and Economic Nationalism in ASEAN. *Journal of ASEAN Studies.* 4(1), pp. 1-19.

Chang, C.P., & A.N. Berdiev. 2011. The Political Economy of Energy Regulation in OECD Countries. *Energy Economics.* 33(5), pp. 816–825.

Chang, H.J. 2010. Industrial Policy: Can we go Beyond an Unproductive Confrontation? *Turkish Economic Association Discussion Paper*. 2010/01. Ankara: Turkish Economic Association.

Chang, K-S. 2012. Predicaments of Neoliberalism in a Post-Developmental State. Chang, K-S., Fine, B., & Weiss, L. 2012. *Developmental Politics in Transition*. Hampshire: Palgrave Macmillan, pp. 274-298.

Chang, Y. & Li, Y. 2012. Power Generation and Cross-Border Grid Planning for the Integrated ASEAN Electricity Market: A Dynamic Linear Programming Model. In: Wu, Y., Shi, X., & Kimura, F. eds. *Energy Market Integration in East Asia: Theories, Electricity Sector and Subsidies.* Jakarta: ERIA.

Chang, Y. & Li, Y. 2013. Power Generation and Cross-Border Grid Planning for the Integrated ASEAN Electricity Market: A Dynamic Linear Programming Model. *Energy Strategy Reviews*. 2, pp. 153-160.

Chang, Y. & Li, Y. 2015. Renewable Energy and Policy Options in an Integrated ASEAN Electricity Market: Quantitative Assessments and Policy Implications. *Energy Policy*. 85, pp. 39-49.

Chang, Y., Lee, J. Ang, W.X., & Chua, J.Y. 2019. Energy Market Integration in ASEAN: Locational Marginal Pricing and Welfare Implications. Journal of Asian Economic Integration. 1(1), pp. 48-72.

Charlton, B. & Andras, P. 2003. *The Modernization Imperative.* London: Imprint Academic.

Chatterjee, S. 2005. Poverty Reduction Strategies—Lessons from the Asia and Pacific Region on Inclusive Development. *Asian Development Review.* [Online]. 22(1), pp. 12-44. [Accessed 14 June 2016]. Available from: <u>https://think-asia.org/</u>

Chattopadhyay, D. & Jha, S. 2014. The Impact of Energy Subsidies on the Power Sector in Southeast Asia. *The Electricity Journal.* 27(4), pp. 70-83.

Chatzky, A. & McBride, J. 2019. China's Massive Belt and Road Initiative. [Online]. New York: Council on Foreign Relations. [Accessed 01 June 2019]. Available from: <u>https://www.cfr.org/</u>

Chelminski, K. 2018. Fossil Fuel Subsidy Reform in Indonesia: The Struggle for Successful Reform. In: Skovgaard, J. & Van Asselt, H. eds. The Politics of Fossil Fuel Subsidies and Their Reform. Cambridge: Cambridge University Press.

Chen, X. 2018. Globalisation Redux: Can China's Inside-Out Strategy Catalyse Economic Development and Integration Across its Asian Borderlands and Beyond? *Cambridge Journal of Regions, Economy, and Society.* 11, pp. 35-58.

Chikozho, C. & Mapedza, E. 2017. Free Market Economics and Developmental Statism as Political Paradigms: Implications for Water Governance Theory and Practice in Developing Countries. In: Karar, E. ed. 2017. *Freshwater Governance for the 21stCentury*. Global Issues in Water Policy. 6, pp. 51-79.

Cho, Y.N. & Jeong, J.H. 2008. China's Soft Power: Discussions, Resources, and Prospects. *Asian Survey*. 48(3), pp. 453-472.

Choo, J. 2014. Non-Traditional Security Cooperation and Northeast Asian Regional Order. In: Atanassova-Cornelis, E. eds. *Changing Security Dynamics in East Asia.* New York: Palgrave Macmillan, pp.171-190.

Chowdhury, S.R. 1999. Neo-Statism in Third World studies: A Critique. *Third World Quarterly*. 20(6), pp. 1089-1107.

Christoff, P. 1996. Ecological Modernisation, Ecological Modernities. *Environmental Politics.* 5(3), pp. 476-500.

Chung, S.H. 2003. Wriggling into Regionalism? Political Economy of Free Trade Agreement in Northeast Asia. *Korean Journal of International Relations.* 43(5), pp. 89-115.

Churchman, C.W. 1967. Wicked Problems. *Management Science*. 14, pp. B-141-B-142.

Claudia Astarita, C. 2008. China's Role in the Evolution of Southeast Asian Regional Organizations. *China Perspectives.* 3, pp. 78-86.

Clough, P. & Nutbrown, C. 2012. A Student's Guide to Methodology. London: Sage.

Cohn, T.H. 2000. *Global Political Economy: Theory and Practice*. New York: Longman.

Coleman, J. S. 1990. *Foundations of Social Theory.* Cambridge: Cambridge University Press.

Colgan, J. 2020. Governments Haven't Managed to Reduce Greenhouse Gases.Here's Who's Taking Charge in the Next Phase. The Washington Post. [Online].[Accessed17February2020].Availablefrom:https://www.washingtonpost.com/

Commander, S. 2012. A Guide to the Political Economy of Reforming Energy Subsidies. IZA Policy Paper 52. Bonn: Institute for the Study of Labour.

Conklin, J. 2005. *Wicked Problems and Social Complexity.* [online]. Wiley. [Accessed 05 July 2019]. Available from: <u>https://cognexus.org/</u>

Conlon, P. 2009. The Integration of Electricity Markets in Ireland under the ISO Model. ESB International.

Considine, L. 2015. 'Back to the Rough Ground!' A Grammatical Approach to Trust and International Relations. *Journal of International Studies.* 44(1), pp. 109-127.

Cook, J.W. ed. 2019. *Sustainability, Human Well-Being, & The Future of Education*. Cham: Palgrave Macmillan.

COP21 Paris. 2015. *About*. [Online]. [Accessed 13 May 2016]. Available from: <u>http://www.cop21paris.org/</u>

Corfee-Morlot, J, Kamal-Chaoui, L., Donovan, M.G., Cochran, I., Robert, A., & Teasdale, P.J. 2009. Cities, Climate Change and Multilevel Governance. *OECD Environment Working Papers 14.* Paris: OECD Publishing.

Cornell, P. 2019. Energy Governance and China's Bid for Grid Integration. Washington, D.C.: *The Atlantic Council.* 

Creswell, J.W. 2005. Educational Research. Planning, Conducting, and Evaluating Quantitative and Qualitative Research. Upper Sadle River: Pearson Education.

Crotty, M. 2003. The Foundations of Social Research. London: Sage Publications.

Crouch, C. 2005. Models of Capitalism. *New Political Economy.* 10(4), pp. 439-456.

Crousillat, E. 1998. Developing International Power Markets in East Asia. *The World Bank Group: Public Policy for the Private Sector.* [Online]. 143, pp. 1-7. [Accessed: 01 November 2017]. Available from: <u>http://siteresources.worldbank.org/</u>

Cumings, B. 1984. The Origin and Development of the Northeast Asian political Economy: Industrial Sectors, Product Cycle, and Political Consequences. International Organization. 38(1), pp. 1-40.

D'Souza, C. 2003. An inference of gift-giving within Asian business culture. *Asia Pacific Journal of Marketing and Logistics.* 15(½), pp. 27-38.

Danermark, B., Ekström, M., Jakobsen, L. & Karlsson, J. 2002. *Explaining Society: Critical Realism in Social Sciences*. London: Routledge.

Dang, K.N., Nguyen V.B., & Nguyen H.T. 2010. Water Use and Competition in the Mekong Delta. Vietnam. In: Be, T.T., Sinh, B.T., & Miller, F. eds. *Challenges to sustainable development in the Mekong Delta: Regional and national policy issues and research needs: Literature analysis.* Bangkok, Thailand: The Sustainable Mekong Research Network.

Dasgupta, P. 1988. Trust as a Commodity. In: Gambetta, D. ed. *Trust: Making and Breaking Cooperative Relations.* New York: Basil Blackwell, pp. 49–76.

de Jong, H.W. & Shepherd, W.G. eds. 2007. *Pioneers of Industrial Organization: How the Economics of Competition and Monopoly Took Shape*. London: Edward Elger.

Decuypere, M., Hoet, H. & Vandenabeele, J. 2019. Learning to Navigate (in) the Anthropocene. *Sustainability*. 11(547), pp. 1-16.

Deloitte. 2018. Embracing the BRI Ecosystem in 2018. Beijing, China: Deloitte Insights.

Dent, C.M. 2008. East Asian Regionalism. London: Routledge.

Dent, C.M. 2014. *Renewable Energy in East Asia: Towards a New Developmentalism*. London: Routledge.

Dent, C.M. 2016. Clean Energy Trade Governance: Reconciling Trade Liberalism and Climate Interventionism. *New Political Economy*. 23(6), pp. 728-747.

Dent, C.M., Huang, D.W.F. 2013. Northeast Asian Regionalism: Lessons from the European Experience. London: Routledge.

Dent, C.M., Richardson-Barlow, C.G. 2016. Trade in Clean Energy: Policy Issues & Governance Solutions. [Online]. The National Bureau of Asian Research. [Accessed 07 June 2016]. Available from: <u>http://nbr.org/</u>

Denzin, N.K. & Lincoln, Y.S. 2011. The Discipline and Practice of Qualitative Research. In: Denzin, N.K. & Lincoln, Y.S. eds. *The Sage Handbook of Qualitative Research*. 4<sup>th</sup> ed. London: Sage.

Department for International Development. 2009. *Political Economy Analysis*. [Online]. [Accessed 07 November 2014]. Available from: <u>http://www.gsdrc.org/</u>

Deutsch, K.W. 1968. The Analysis of International Relations. Englewood Cliffs: Prentice-Hall.

Dexter, L. 1970. Elite and Specialized Interviewing. London: ECPR Press.

Diala-McBride, C. 2018. OBOR and the Future of Energy Trade. [Online]. [Accessed 08 June 2018]. Eniday. Available from: <u>http://www.eniday.com/</u>

Digital Energy Asia. 2018. Competitive Indonesia Electricity Rates in the ASEAN Region. [Online]. [Accessed 19, October, 2020]. Available from: https://digitalenergyasia.com/

DOE. 2019. Power Statistics. [Online]. [Accessed 21 January 2020]. Available from: <u>https://www.doe.gov.ph/</u>

Doner, R.F. 1991. Approaches to the Politics of Economic Growth in Southeast Asia. *The Journal of Asian Studies*. 50(4), pp. 818-849.

Drysdale, P. & Zhang, D. 2000. Uneasy bedfellows in East Asia. In: Drysdale, P. and Zhang, D.D. eds. *Japan and China: Rivalry or Cooperation in East Asia?* Canberra: Asia Pacific Press, pp. 1-12.

Drysdale, P. ed. 2000. *Reform and Recovery in East Asia*. London: Routledge.

Dryzek, J.S. 2013. *The Politics of the Earth: Environmental Discourses.* Third Edition. Oxford: Oxford University Press.

Dutton, J. & Lockwood, M. 2017. Ideas, Institutions and Interests in the Politics of Cross-Border Electricity Interconnection: Greenlink, Britain and Ireland. *Energy Policy.* 105(2017), pp. 375-385.

Dutu, R. 2016. Challenges and Policies in Indonesia's Energy Sector. *Energy Policy.* 98, pp. 513-519.

Dyer, H.C. 2014. Climate Anarchy: Creative Disorder in World Politics. *International Political Sociology*. 8, pp. 182-200.

Easton, G. 2010. Critical Realism in Case Study Research. *Industrial Marketing Management.* 39, pp. 118-128.

Eckersley, R. 2004. The Big Chill: The WTO and Multilateral Environmental Agreements. *Global Environmental Politics.* 4(2), pp. 24–50.

EIA. 2019. Electricity Explained. [Online]. [Accessed 23 May 2019]. Available from <u>https://www.eia.gov/</u>

Eilperin, J. & Mufson, S. 2009. Renewable Energy's Environmental Paradox. *The Washington Post*. [Online]. April 16. [Accessed 14 September 2014]. Available from: <u>http://www.washingtonpost.com/</u>

EIU. 2019. China's Reboot of the Belt and Road Initiative. [Online]. EIU. [Accessed 01 September 2019]. Available from: <u>http://country.eiu.com/</u>

Emmel, N. 2013. *Sampling and Choosing Cases in Qualitative Research: A Realist Approach.* London: Sage.

Energy and Industry Department of Prime Minister's Office. 2016. Brunei Darussalam Country Report. In: Kimura, S. & Han, P. eds. Energy Outlook and Energy Saving Potential in East Asia 2016. ERIA Research Project Report 2015(5). Jakarta: Economic Research Institute for ASEAN & East Asia, pp.79-90.

Epps, T. & Green, A. 2010. *Reconciling Trade and Climate: How the WTO Can Help Address Climate Change*. London: Edward Elgar.

Erdogdu, E. 2014. The Political Economy of Electricity Market Liberalization: A Cross-Country Approach. *The Energy Journal*. 3(35), pp. 91-128.

<u>ESMAP & IFC. 2017.</u> Energy Storage Trends and Opportunities in Emerging Markets. Boulder, CO: Navigant Consulting, Inc.

Esty, D.C. & Ivanova, M.H. 2002. Global Environmental Governance: Options & Opportunities. *Forestry & Environmental Studies Publications Series*. 8, pp. 1-249.
EU. 2017. ASEAN–EU Joint Statement on the Implementation of the Paris Agreement. Brussels: EU.

European Commission. 2013. *Report from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of Regions*. Brussels: European Commission.

European Commission. 2014. *EU Greenhouse gas Emissions and Targets*. [Online]. [Accessed 25 September 2014] Available from: <u>http://ec.europa.eu/</u>

Evans, J.P. 2012. Environmental Governance. London: Routledge.

Evans, P. B. 1995. *Embedded Autonomy: States and Industrial Transformation*. Princeton, NJ: Princeton University Press.

Finnemore, A. & Toope, S.J. 2001. Alternatives to "Legalization": Richer Views of Law and Politics. *International Organization*. 55(3), pp. 743-758.

Fischer, C. & Preonas, L. 2010. Combining Policies for Renewable Energy: Is the Whole Less than the Sum of its Parts? *Discussion Paper Series* 10(19). Washington, D.C.: Resources for the Future.

Fletcher, A.J. 2017. Applying Critical Realism in Qualitative Research: Methodology Meets Method. *International Journal of Social Research Methodology*. 20(2), pp. 181-194.

Flynn, N. 1999. *Miracle to Meltdown in Asia*. Oxford, U.K.: Oxford University Press.

Foot, R. 2011. The Role of East Asian Regional Organizations in Regional Governance: Constraints and Contributions. [Online]. [Accessed 09 March 2019]. Available from: <u>http://carnegieendowment.org/</u>

Forsyth, T. 1999. *International Investment and Climate Change*. London: The Royal Institute of International Affairs.

Forsyth, T. 2003. *Critical Political Ecology: The Politics of Environmental Science.* London: Routledge.

Foucault, M. 1991. Discipline and Punish: The Birth of a Prison. London: Penguin.

Foucault, M. 2003. Technologies of the Self. In: Rabinow , P. & Rose N. eds. *The Essential Foucault: Selections from Essential Works of Foucault, 1954–1984*. New York: New Press, pp. 146-169.

Froggart, A., Wright, G., & Lockwood, M. 2017. Staying Connected: Key Elements for UK–EU27 Energy Cooperation After Brexit. *Chatham House: Royal Institute of International Affairs*. London: Chatham House. Fünfgeld, A. The Dream of ASEAN Connectivity: Imagining Infrastructure in Southeast Asia. *Pacific Affairs.* 92(2), pp. 287-311.

Gainsborough, M. 2009. The (Neglected) Statist Bias and the Developmental State: The Case of Singapore and Vietnam. *Third World Quarterly.* 30(7), pp. 1317-1328.

Gambetta, D. ed. 1988. Trust, Making and Breaking Cooperative Relations. Oxford: Blackwell.

Gao, B. 1996. *Economic Ideology and Japanese Industrial Policy.* Cambridge.: Cambridge University Press.

Gârdu, D. 2015. Making Sense of the East Asian Development Model: An Overview of Neoliberal and Statist Accounts. *International Journal of Business and Social Science.* 6(11), pp. 25-34.

Gereffi, G. 1989. Rethinking Development Theory: Insights from East Asia and Latin America. *Sociological Forum.* 4(4), pp. 505-533.

Gereffi, G. 1998. More than the Market, More than the State: Global Commodity Chains and Industrial Upgrading in East Asia. In: Chan, S., Lutz, H., Lam, D., & Clark, C. eds. *Beyond the Developmental State: East Asia's Political Economies Reconsidered.* London: Palgrave Macmillan, pp. 26-38.

Gerson, K., & Horowitz, R. 2002. Observation and Interviewing: Options and Choices. In: May, T. ed. *Qualitative Research in Action*. London: Sage.

Gezmis, H. 2018. From Neoliberalism to Neo-Developmentalist? The Political Economy of Post-Crisis Argentina (2002-2015). *New Political Economy.* 23(1), pp. 66-87.

Ghiollarnath, C. 2011. Renewable Energy Tax Incentives and WTO Law: Irreconcilably Incompatible? An Examination of the WTO-Consistency of Direct Corporate Tax Incentives for the Development of Renewable Energy. Netherlands: Maastricht University.

Giddens, A. 1990. The Consequences of Modernity. London: Polity Press.

Giddens, A. 2009. The Politics of Climate Change. Cambridge: Polity Press.

Glaser, B., & Strauss, A. 1967. *The Discovery of Grounded Theory: Strategies for Qualitative Research*. Mill Valley: Sociology Press.

Global Subsidies Initiative. 2017. How Electricity Subsidy Reform Can Benefit Indonesia. International Institute for Sustainable Development. [Online]. [Accessed 23 May 2010]. Available from: <u>https://www.iisd.org/</u> Gong, X. 2019. Will China Undermine its own Influence in Southeast Asia Through the Belt and Road? *The Diplomat.* [Online]. 13 April. [Accessed 21 June 2019]. Available from: <u>https://thediplomat.com/</u>

Gorski, P.S. 2013. What is Critical Realism? And Why Should You Care? *Contemporary Sociology*. 42(5), pp. 658-670.

Gouldson, A., Colenbrander, S., Sudmant, A., Papargyropoulou, E., Kerr, N., McAnulla, F., & Hall, S. 2016. Cities and Climate Change Mitigation: Economic Opportunities and Governance Challenges in Asia. *Cities.* [Online]. 54, pp. 11-19. [Accessed 20 October 2017]. Available from: <u>http://www.sciencedirect.com/</u>

Gratwick, K.N. & Eberhard, A. 2008. Demise of the standard model for power sector reform and the emergence of hybrid power markets. *Energy Policy*. 36(2008), pp. 3948–3960.

Grbich, C. 2013. *Qualitative Data Analysis: An Introduction*. London: Sage.

Grossi, L., Heim, S., HÜschelrath, K. & Watreson, M. 2018. Electricity Market Integration and the Impact of Unilateral Policy Reforms. *Oxford Economic Paper*. 70(3), pp. 799-820.

Grossman, G.M. & Richardson, J.D. 1985. Strategic Trade Policy: A Survey of Issues and Early Analysis. Special Papers in International Economics. [Online]. Princeton, NJ: Princeton University. [Accessed 26 April 2014]. Available from: <u>https://www.princeton.edu/</u>

Grossman, G.M. 1986. Strategic Export Promotion: A Critique. In Krugman, P.R. ed. 1986. *Strategic Trade Policy and the New International Economics*. Massachusetts: The MIT Press, pp. 47-68.

Grubb, M. 2003. The Economics of the Kyoto Protocol. *World Economics*. 4(3), pp. 143-190.

Guan, B.T.C. 2004. ASEAN's Regional Integration Challenge: The ASEAN Process. *The Copenhagen Journal of Asian Studies*. 20, pp. 70-94.

Gulagi, A., Bogdaanov, D., Fasihi, M., & Breyer, C. 2017. Can Australia Power the Energy-Hungry Asia with Renewable Energy? *Sustainability.* 9(233), p. 1-26.

Habito, C.F., Aldaba, F.T., & Templo, O.M. 2004. An Assessment Study on the Progress of ASEAN Regional Integration: The Ha Noi Plan of Action toward ASEAN Vision 2020. *Ateneo Center for Economic Research and Development.* 03(006b), pp. 1-136.

Haggard, S. 1990. *Pathways from the Periphery: The Politics of Growth in the Newly Industrializing Countries*. Ithaca: Cornell University Press.

Haggard, S. 2015. *The Developmental State is Dead: Long Live the Developmental State!* In: Mahoney, J. & Thelen, K. *Advances in Comparative-Historical Analysis.* Cambridge: Cambridge University Press, pp. 39-66.

Halawa, E., James, G., Shi, X., Sari, N.H., & Nepal, R. 2018. The Prospect for an Australian-Asian Power Grid: Critical Appraisal. *Energies*. 11(20), pp. 2-23.

Hall, S. & Roelich, K. 2016. Business Model Innovation in Electricity Supply Markets: The Role of Complex Value in the United Kingdom. *Energy Policy*. 92(2016), pp. 286-298.

Hamilton, A. 1791. *Report on Manufacturers.* [Online]. Washington, D.C.: US Government Printing Office. [Accessed 27 April 2016]. Available from: <u>https://books.google.co.uk/</u>

Harada, I. 2019. US Overtaken by Southeast Asia as China's No. 2 Trade Partner. *Nikkei Asia Review*. [Online]. July 13. [Accessed 15 July 2019]. Available from: <u>https://asia.nikkei.com/</u>

Harding, S. 1987. The Method Question. *Hypatia*. 2(3), pp. 19-35.

Harris, P.G. & Lang, G. eds. 2015. *Routledge Handbook of Environment and Asia*. Oxon, U.K.: Routledge.

Harris, P.G. 2011. Reconceptualizing Global Governance. In: Dryzek, J.S., Norgaard, R.B., & Schlosberg, D. *The Oxford Handbook of Climate Change and Society*. Oxford: Oxford University Press, pp. 639-652.

Harvey, D. 2005. *A Brief History of Neoliberalism*. Oxford: Oxford University Press.

Hayashi, S. 2010. The Developmental State in the Era of Globalization: Beyond the Northeast Asian Model of Political Economy. *The Pacific Review*. 23(1), pp. 45-69.

Head, B. 2008. Wicked Problems in Public Policy. *Public Policy. 3*, pp. 101–118.

Heijmans, M. 2007. The Future of the East Asian Political Economy: China, Japan and Regional Integration. IAAS Newsletter. [Online]. 44, pp. 22-24 [Accessed February 20, 2017]. Available from: <u>https://iias.asia/</u>

Heinberg, R. 2007. Five Axioms of Sustainability. [online]. [Accessed September 9, 2019]. Available from: <u>http://richardheinberg.com/</u>

Heldt, E.C. & Schmidtke, H. 2019. Explaining Coherence in International Regime Complexes: How the World Bank Shapes the Field of Multilateral Development Finance. *Review of International Political Economy*. 26(6), pp. 1160-1186.

Hermawanto, B. 2016. *ASEAN Power Grid.* 4<sup>th</sup> North East Asia Energy Security Forum, December 15, 2016, Seoul, Republic of Korea.

Hiebert, M. 2018. Southeast Asia Financial Integration and Infrastructure Investment. *Center for Strategic & International Studies Briefs.* Washington, D.C.: Center for Strategic & International Studies. Pp. 1-10.

Hiebert, M., Pumphrey, D.L., Poling, G.B., & Walton, M.A. 2012. Sustainable Energy Futures in Southeast Asia. Washington, D.C.: Center for Strategic & International Studies.

Hill, H. 1997. Towards a Political Economy Explanation of Rapid Growth in ASEAN: A Survey and Analysis. *ASEAN Economic Bulletin*. 14(2), pp. 131-149.

Hill, R.C. 2007. *Neoliberalism & Developmentalism: The East Asian Experience.* Taipei Discussion Forum, December 20, 2007, Taipei.

Hillman, J. 2018. China's Belt and Road Initiative: Five Years Later. January 25, Dirksen Senate Office Building, Washington, D.C.

Hines, C. ed. 2005. Virtual Methods: Issues in Social Research on the Internet. Oxford: Berg.

Hoekman, B. & Kostecki, M. 1995. *The Political Economy of the World Trading System.* Oxford: Oxford University Press.

Hoffman, A.M. 2002. A Conceptualization of Trust in International Relations. *European Journal of International Relations.* SAGE Publications and ECPR. 8(3), pp. 375-401.

Holzner, B. 1973. Sociological Reflections on Trust. *Humanitas*. 9(3), pp. 333–345.

Hope, E. 2008. The Political Economy of Power Sector Reform: The Experiences of Five Major Developing Countries by David G. Victor and Thomas C. Heller. *Energy Journal.* 29(2), pp. 186-188.

Hoselitz, B.F. & Moore, W.E. eds. 1963. *Industrialization and Society*. Paris: UNESCO.

Houston, D. 2013. Crisis is where we live: Environmental justice for the Anthropocene. *Globalizations.* 10, Pp. 439–450.

Huang, Y.W., Kittner, N. and Kammen, D.M. 2019. ASEAN grid Flexibility: Preparedness for Grid Integration of Renewable Energy. *Energy Policy.* 128, pp. 711-726.

Huber, J. 2000. Towards Industrial Ecology: Sustainable Development as a Concept of Ecological Modernization. *Journal of Environmental Policy & Planning.* [Online]. 2, pp 269–285. [Accessed 15 April 2016]. Available from: <u>http://onlinelibrary.wiley.com/</u>

Huber, M., Roger, A. & Hamacher, T. 2015. Optimizing Long-Term Investments for a Sustainable Development of the ASEAN Power System. *Energy*. 88, p. 180-193.

Huff, W.G. 1995. The Developmental State, Government, and Singapore's Economic Development since 1960. *World Development.* 23(8), pp. 1421-1438.

Hughes, S. 2002. *Global Governance: Critical Perspectives*. London: Routledge.

Hume, D. 1752. *Essays, Moral, Political and Literary.* [Online]. [Accessed 20 February 2016]. Indianapolis: Liberty Fund, Inc. Available from: <u>http://www.econlib.org/</u>

Hunt, S. D. 2010. *Marketing Theory*. New York: M. E. Sharpe.

Huotari M., Rüland J. & Schlehe J. eds. 2014. *Methodology and Research Practice in Southeast Asian Studies*. London: Palgrave Macmillan.

Hutt, D. 2017. How Nationalism Undermines ASEAN Integration. *The Diplomat*. [Online]. June 07. [Accessed 18 June 2019]. Available from: <u>http://www.thediplomat.com/</u>

Ibrahim, S. 2014. *Barriers and Opportunities for Electricity Interconnection the Southeast Asian Experience*. APEC Conference, March 27, 2014, Tokyo, Japan.

ICTSD. 2016. Enabling the Energy Transition and Scale-up of Clean Energy Technologies: Options for the Global Trading System. [Online]. [Accessed 20 January 2016]. Available from: <u>http://www.ictsd.org/</u>

IEA. 2012a. *Global Carbon-Dioxide Emissions Increase by 1.0 Gt in 2011 to Record High.* [Online]. Paris: IEA. [Accessed 25 September 2015]. Available from: <u>http://www.iea.org/</u>

IEA. 2012b. World Energy Outlook. Paris: IEA.

IEA. 2014. Seamless Power Markets: Regional Integration of Electricity Markets in IEA Member Countries. Paris: IEA.

IEA. 2015. Development Prospects of the ASEAN Power Sector. Paris: IEA.

IEA. 2017b. *Country Profile: Indonesia.* [Online]. [Accessed 01 May 2020]. Available from: <u>https://www-pub.iaea.org/</u>

IEA. 2017c. Electricity Database 2017. [Online]. [Accessed 02 April 2019]. Available from: <u>https://webstore.iea.org/</u>

IEA. 2019a. Southeast Asia Energy Outlook. Paris: IEA.

IEA. 2019b. Renewables 2019. Paris: IEA.

IEA. 2019c. World Energy Outlook 2018. [Online]. [Accessed 24 May 2019]. Available from: <u>https://www.iea.org/</u>

IEA. 2019d. Establishing Multilateral Power Trade in ASEAN. Paris: IEA.

IEA. 2020. IEA Fossil Fuels Database. Paris: IEA.

IISD. 2013. Fossil-Fuel Subsidy Reform: Challenges and Opportunities. *A Forum for South East Asian Policy-Makers.* November 20–21, 2012, Bali, Indonesia. Indonesia: IISD, pp. 1-29.

IMF. 2013. Energy Subsidy Reform: Lessons and Implications. Washington, D.C.: IMF.

IMF. 2015. Regional Economic Outlook: Asia and Pacific. Washington, D.C.: IMF.

IMF. 2016. *World Economic Outlook Database*. [Online]. [Accessed 05 May 2016]. Available from: <u>http://www.imf.org/</u>

Impram, S., Nese, S.V., & Oral, B. 2020. Challenges of Renewable Energy Penetration on Power System Flexibility: A Survey. *Energy Strategy Reviews.* 31(100539), pp. 1-12.

Incropera, F.P. 2015. *Climate Change: A Wicked Problem.* New York: Cambridge University Press.

Integrated Research and Action for Development. 2016. Learning from Nord Pool Region: Power Market Development. Washington, D.C.: SARI/EI programme.

IPCC. 2019. *Global Warming of 1.5°C.* Geneva: IPCC Secretariat.

IRENA & ACE. 2016. Renewable Energy Outlook for ASEAN. Abu Dhabi: IRENA & Jakarta: ACE.

IRENA, OECD/IEA and REN21. 2018. Renewable Energy Policies in a Time of Transition. Abu Dhabi: IRENA and Paris: OECD, IEA, & REN21.

IRENA. 2013. Renewable Energy Innovation Policy: Success Criteria and Strategies. Abu Dhabi: IRENA.

IRENA. 2018a. Power System Flexibility for the Energy Transition. Abu Dhabi: IRENA.

IRENA. 2018b. Renewable Energy Market Analysis: Southeast Asia. Abu Dhabi: IRENA.

IRENA. 2019a. A New World Order: The Geopolitics of the Energy Transformation. Abu Dhabi: IRENA.

IRENA. 2019b. Renewable Capacity Statistics 2019. Abu Dhabi: IRENA.

Jackson, T. 1991. Renewable Energy: Great Hope or False Promise? *Energy Policy.* 19(1), pp. 2-7.

Jacobs, M. 2013. Green Growth. In Falkner, R. ed. *Handbook of Global Climate and Environmental Policy*. Oxford: Wiley Blackwell.

Jamasb, T. & Pollitt, M. 2005. Electricity Market Reform in the European Union: Review of Progress Toward Liberalization & Integration. *Center for Energy and Environmental Policy Research*. Working Paper 05(003), pp. 1-31.

Jamasb, T. 2002. Reform and Regulation of the Electricity Sectors in Developing Countries. *DAE Working Paper* 0226. Cambridge: Department of Applied Economics, University of Cambridge.

Jayasuriya, K. 2005. Beyond Institutional Fetishism: From the Developmental to the Regulatory State. *New Political Economy.* 10(3), pp. 381-387.

Johansson, T.B., Patwardhan, A., Nakicenovic, N., & Gomez-Echeverri, L. eds. 2012. Global Energy Assessment: Toward a Sustainable Future. New York: Cambridge University Press.

Johnson, C. 1982. *MITI and the Japanese Miracle*. California: Stanford University Press.

Johnson, C. 1987. Political Institutions and Economic Performance: The Government Business Relationship in Japan, Korea, and Taiwan. In: Deyo, F., ed. *The Political Economy of the New Asian Industrialism.* Ithaca, NY: Cornell University Press, pp. 136–64.

Johnson, C. 1999. The Developmental State: Odyssey of a Concept. In: Woo-Cumings, M. ed. *The Developmental State.* Cornell: Cornell University Press, pp. 32-60. Johnson, E. & Morehouse, H. eds. 2014. After the Anthropocene: Politics and Geographic Inquiry for a New Epoch. *Progress in Human Geography.* 38, pp. 439–456.

Johnston, L.A. 2019. The Belt and Road Initiative: What is in it for China? *Asia and the Pacific Policy Studies*. 6, pp. 40-58.

Joinson, A.N. 2005. Internet Behaviour and the Design of Virtual Methods. In: Hines, C. ed. 2005. Virtual Methods: Issues in Social Research on the Internet. Oxford: Berg.

Jones, L. 2014. *Renewable Energy Integration: Practical Management of Variability, Uncertainty and Flexibility in Power Grids*. London: Academic Press.

Jones, L. 2016. Explaining the Failure of the ASEAN Economic Community: The Primacy of Domestic Political Economy. *The Pacific Review*. 29(5), pp. 647-670.

Joskow, P.L. 1998. Electricity Sectors in Transition. Energy Journal. 19, pp. 25–62

Joy-Perez, C. & Scissors, D. 2018. Be Wary of Spending on the Belt and Road. *American Enterprise Institute.* Pp. 1-6.

Junlakarn, S. & Wangjiraniran, W. 2016. Assessing Potential of Power Generation Investment in ASEAN Countries. *Energy Procedia*. 100, pp. 475-479.

Kahler, M. 2000. Legalization as Strategy: The Asia-Pacific Case. *International Organization.* 54(3), pp. 549-571.

Karplus, V.J. & Von Hirschhausen, C. 2019. *Economics of Energy & Environmental Policy*. 8(1), pp. 1-18.

Kasahara, S. 2013. The Asian Developmental State and the Flying Geese Paradigm. *UNCTAD Discussion Papers.* 213, pp. 1-30.

Kasahara, S. 2013. The Asian Developmental State and the Flying Geese Paradigm. Geneva: UNCTAD.

Katz, B. & Lee, J. 2011. Alexander Hamilton's Manufacturing Message. [Online]. The Brookings Institution. [Accessed 27 April 2016]. Available from: <u>http://www.brookings.edu/</u>

Kausikan, B. 2017. Growing Complexities Threaten ASEAN Consensus Politics. *Nikkei Asia Review.* [Online]. 11 October. [Accessed 18 June 2019]. Available from: <u>https://www.asia.nikkei.com/</u>

Kelsall, T. & Booth, T. 2010. Developmental Patrimonialism? London: Oversees Development Institute.

Kelsall, T. 2011. Rethinking the Relationship between Neo-Patrimonialism and Economic Development in Africa. *IDS Bulletin.* (42)2, pp. 76–87.

Keohane, R. O. & Nye, J. S. 1989. *Power and Interdependence.* Second edition. New York: Harper Collins Publishers.

Keohane, R. O. 1984. *After Hegemony: Cooperation and Discord in the World Political Economy*. Princeton: Princeton University Press.

Keohane, R.O. & Victor, D.G. 2011. The Regime Complex for Climate Change. *Perspectives on Politics.* [Online]. 9(1), pp. 7-23 [Accessed 21 September 2017]. Available from: <u>http://www.jstor.org/</u>

Keynes, J.M. 1936. *The General Theory of Employment, Interest and Money*. London: Palgrave Macmillan.

Kim, Y. Horta, H. & Jung, J. 2017. Higher Education Research in Hong Kong, Japan, China and Malaysia: Exploring Research Community Cohesion and the Integration of Thematic Approaches. *Studies in Higher Education.* 42(1), pp.149-168.

Kim, Y.T. 1999. Neoliberalism and the Decline of the Developmental State. *Journal of Contemporary Asia.* 29(4), pp. 441-461.

Kimura, F. & Obashi, A. 2011. Production Networks in East Asia: What we Know so Far. ADBI. 320, pp. 1-26.

Kimura, F. & Phoumin, H. 2014. Energy Market Integration in East Asia: Energy Trade, Cross Border Electricity, and Price Mechanism. ERIA Research Project Report 29. Jakarta: ERIA.

Kimura, F. & Phoumin, H. 2019. Energy Outlook & Energy Savings Potential in East Asia 2019. Jakarta: ERIA.

Kimura, F. & Shi, X. 2011. *Deepen Understanding and Move Forward: Energy Market Integration in East Asia.* Jakarta: ERIA.

Kimura, F. 2015. International Production Networks and Economic Diplomacy in Japan. In: Mulgan, A.G. & Honma, M. eds. *The Political Economy of Japanese Trade Policy*. London: Palgrave Macmillan, pp. 175-195.

Kimura, S., Phoumin, H. & Jacobs, B. eds. 2013. *Energy Market Integration in East Asia: Renewable Energy and its Deployment into the Power System*. Jakarta: ERIA.

Kincheloe, J.L., McLaren, P.L., & Steinberg, S. R. 2011. Critical Pedagogy and Qualitative Research. In: Denizin, N.K. & Lincoln, Y.S. eds. *The Sage Handbook of Qualitative Research.* Thousand Oaks: Sage, pp. 163-177.

Kirchherr J. & Charles, K. 2018. Enhancing the Sample Diversity of Snowball Samples: Recommendations from a Research Project on Anti- Dam Movements in Southeast Asia. *PLoS ONE.* 13(8), pp. 1-17.

Kivimaki, T. 2014. Soft Power and Global Governance with Chinese Characteristics. *The Chinese Journal of International Politics.* [Online]. 7(4), pp. 421-447. [Accessed 07 July 2017]. Available from: <u>https://academic.oup.com/</u>

Kivits, J. 2005. Online Interviewing and the Research Relationship. In: Hines, C. ed. 2005. *Virtual Methods: Issues in Social Research on the Internet*. Oxford: Berg.

Knight, J. & Harrison, S. 2014. Limitations of Uniformitarianism in the Anthropocene. *Anthropocene*. 5, pp. 71–75.

Knight, J. 2015. Anthropocene Futures: People, Resources and Sustainability. *The Anthropocene Review.* 2(2), pp. 152-148.

Koeberle, S., Bedoya, H., Silarsky, P. & Verheyen, G. eds. 2005. *Conditionality Revisited: Concepts, Experiences, and Lessons.* Washington, D.C.: The World Bank Group.

Kojima, K. 2000. The "Flying Geese" Model of Asian Economic Development: Origin, Theoretical Extensions, and Regional Policy Implications. *Journal of Asian Economics.* 11, pp. 375-401.

Kojima, M. 2009. Changes in End-User Petroleum Product Prices: A Comparison of 48 Countries. Washington, D.C.: World Bank.

Kojima, S. & Bhattacharya, A. 2011. Pricing Reform and Enhanced Investment in the Energy Sector: A Way Towards East Asian Economic Development. In: Kimura, F. and Shi, X. eds. *Deepen Understanding and Move Forward: Energy Market Integration in East Asia.* Jakarta: ERIA.

Komori, Y. 2009. Regional Governance in East Asia and the Asia-Pacific. *East Asia.* 26, pp. 321–341.

Komori, Y. 2012. In Search of Regional Governance in East Asia: Processes and Outcomes. *Asian Journal of Political Science*. 20(3), pp. 221-243.

Konings, M. 2012. Neoliberalism and the State. *Uniting Struggles: Critical Social Research in Critical Times.* 23, pp. 85-98.

Krueger, A. 1995. Free Trade Agreements Versus Customs Unions. Working Paper No. 5084. [Online]. National Bureau of Economic Research. [Accessed 01 July 2017]. Available from: http://www.nber.org/ Krueger, A. 2012. *Struggling with Success: Challenges Facing the International Economy.* New Jersey: World Scientific Publishing, Co.

Krueger. A. 1999. Trade Creation and Trade Diversion Under NAFTA. Cambridge: National Bureau of Economic Research.

Krugman, P.R. & Obstfeld, M. 2009. *International Economics: Theory and Practice.* 8<sup>th</sup> edition. Boston: Pearson.

Krugman, P.R. 1983. Targeted Industrial Policies: Theory and Evidence. In *Structural Change and Public Policy.* Kansas City: Federal Reserve Bank of Kansas City.

Krugman, P.R. 1986. *Strategic Trade Policy and the New International Economics*. The MIT Press.

Krugman, P.R. 1991. Increasing Returns and Economic Geography. *The Journal* of Political Economy. 99(3), pp. 483-499.

Krugman, P.R. 1993. Regionalism Versus Multilateralism: Analytical Notes. In *New Dimensions in Regional Integration,* de Melo, J. & Panagariya, A. eds. New York: Cambridge University Press, pp. 58-79.

Kuik, C.C. 2005. *Contemporary Southeast Asia*. 27(1), pp. 102-122.

Kulovesi, K. 2014. International Trade Disputes on Renewable Energy: Testing Ground for the Mutual Supportiveness of WTO Law and Climate Change Law. *Review of European Community & International Law*. [Online]. 23(3), pp. 342-353. [Accessed 14 August 2017]. Available from: <u>http://onlinelibrary.wiley.com/</u>

Kunstýř, J. & Mano, S. 2013. "Energy Security and Cross-Border Electricity Trade: Can the Asian Super Grid Project Pose Security Risks for Japan? Can the Electricity Imports be Used as an Extortion Weapon?" [Online]. Renewable Energy Institute. [Accessed November 24, 2017]. Available from: https://www.renewable-ei.org/

Kurlantzick, J. 2012. ASEAN's Future and Asian Integration. *Council on Foreign Relations (CFR) Working Paper.* New York: CFR.

Kutani, I. & Li, F. eds. 2014. Investing in Power Grid Interconnection in East Asia. *ERIA Research Project Report 2013*. 26, pp. 1-79.

Lajčiak, M. 2017. East Asian Economies and Their Philosophy Behind Success: Manifestation of Social Constructs in Economic Policies. *Journal of International Studies*. 10(1), pp. 180-192. Langhelle, O. 2000. Why Ecological Modernization and Sustainable Development should not be conflated. *Journal of Environmental Policy & Planning.* [Online]. 2, pp. 303-322. [Accessed 02 April 2016]. Available from: <u>http://onlinelibrary.wiley.com/</u>

Lawrey, R. & Pillarisetti, J.R. 2017. Energy Subsidies versus Economic Efficiency: Practical and Theoretical Issues in the Case of Brunei Darussalam. *Asian Social Science.* 7(3), pp. 108-116.

Lawson, T. 1997. Economics & Reality. London: Routledge.

Layder, D. 1990. *The Realist Image in Social Science*. London: Macmillan Press.

Lazarus, R.J. 2009. Super Wicked Problems and Climate Change: Restraining the Present to Liberate the Future. *Cornell Law Review.* 94, pp. 1153–1234.

Leech, B. 2002. Asking Questions: Techniques for Semi-structured Interviews. *Political Science and Politics.* 35(4), pp. 665-668.

Leech, B. L. 2002. Interviewing Methods in Political Science. *Political Science and Politics.* 35(4), pp. 663-664.

Leftwich, A. 1995. Bringing Politics Back in: Towards a Model of the Development State. *The Journal of Development Studies*. 31(3), pp. 400-427.

Leonard, T. 2018. *Energy Transition Outlook 2017*. ASEAN Centre for Energy, Webinar.

Leu, G. 2011. ASEAN's Preferential Trade Agreements (PTA) Strategy. *Journal of Current Southeast Asian Affairs.* 30(2), pp. 31-64.

Levin, K., Bernstein, S., Cashore, B. & Auld, G. 2007. Playing it Forward: Path Dependency, Progressive Incrementalism, and the 'Super Wicked' Problem of Global Climate Change. In Proceedings of the International Studies Association Convention, Chicago, IL, USA. Pp. 1–26.

Levin, K., Bernstein, S., Cashore, B., & Auld, G. 2012. Overcoming the Tragedy of Super Wicked Problems: Constraining our Future Selves to Ameliorate Global Climate Change. *Policy Sciences.* 37(2), pp. 123-152.

Lewis, J. 2012. Emerging Conflicts in Renewable Energy Policy and International Trade Law. *Global Environmental Politics*. 14(4), pp. 10-35.

Lewis, J. 2014. The Rise of Renewable Energy Protectionism: Emerging Trade Conflicts and Implications for Low Carbon Development. *Global Environmental Politics.* 14(4), pp. 10-35.

Li, Y. & Chang, Y. 2014. Infrastructure Investments for Power Trade and Transmission in ASEAN+2: Costs, Benefits, Long- Term Contracts, and Prioritised Development. *ERIA Discussion Paper Series*. Tokyo: ERIA.

Li, Y. & Kimura, S. eds. 2016. *Achieving an Integrated Electricity Market in Southeast Asia: Addressing the Economic, Technical, Institutional, and Geopolitical Barriers.* ERIA Research Project Report FY2015(16). Jakarta: Economic Research Institute for ASEAN & East Asia, pp. 41-58.

Liftin, K.T. 1994. Ozone Discourses. New York: Columbia University press.

Lincoln, Y.S. & Guba, E.C. 1985. *Naturalistic Inquiry.* Newbury Park: Blackwell Science.

Lindbald, T. 1995. Current Trends in the Economic History of Southeast Asia. *Journal of Southeast Asian Studies.* 26(1), pp. 159-168.

Liow, E.D. 2011. The Neoliberal-Developmental State: Singapore as Case Study. *Critical Sociology.* Pp. 1-24.

List, F. 1909. *The National System of Political Economy.* [Online]. London: Longmans, Green and Co. [Accessed 25 April 2016]. Available from: <u>http://oll.libertyfund.org/</u>

Liu, W.D. & Dunford, M. 2016. Inclusive Globalization: Unpacking China's Belt and Road Initiative. *Area Development and Policy*. 1(1), pp. 323–340.

Liu, W.D., Dunford, M. & Gao, B.Y. 2018. A Discursive Construction of the Belt and Road Initiative: From Neo-Liberal to Inclusive Globalization. *Journal of Geographical Sciences.* 28(9), pp. 1199-1214.

Liu, Y., Sheng, Z., and Azhgaliyeva, D. 2019. Toward Energy Security in ASEAN: Impacts of Regional Integration, Renewables and Energy Efficiency. *ADBI Working Paper 1040.* Tokyo: Asian Development Bank Institute.

Lockwood, M. 2013. The Political Economy of Low Carbon Development. In: Urban, F. & Nordensvärd, J. eds. *Low Carbon Development: Key Issues.* London: Routledge, pp. 25-37.

Lockwood, M. 2015a. Fossil Fuel Subsidy Reform, Rent Management and Political Fragmentation in Developing Countries. *New Political Economy*. 20(4), pp. 475-494.

Lockwood, M. 2015b. Stern Review 2.0? The Report of the Global Commission on the Economy and Climate. *The Political Quarterly.* 86(1), pp. 146-151.

Lockwood, M. 2015c. The Political Dynamics of Green Transformations: Feedback Effects and Institutional Context. In: Scoones, I., Leach, M. & Newell, P. eds. *The Politics of Green Transformations.* Oxford: Routledge, pp. 86-101.

Low, L. & Johnston, D.M. 2001. *Singapore Inc.* Singapore: Times Publishing Group.

Low, L. 2003. Multilateralism, Regionalism, Bilateral and Cross Regional Free Trade Arrangements: All Paved with Good Intentions for ASEAN? *Asian Economic Journal.* 17(1), pp. 65-86.

Lu, Z., Suehiro, S., Matsuo, Y., Yamashita, Y. & Ito, K. 2017. Asia/World Energy Outlook: Focusing on ASEAN Economies Toward Market Integration. Tokyo: The Institute of Energy Economics.

Lukes, S. 2005. Power: A Radical View. London: Palgrave Macmillan.

MacGregor, A. 2008. Southeast Asian Development. London: Routledge.

MacNeil, R. & Paterson, M. 2012. Neoliberal Climate Policy: From Market Fetishism to the Developmental State. *Environmental Politics*. 21(2), pp. 230-247.

Maria, R.S., Urata, S. & Intal, P.S. eds. 2017. The ASEAN Economic Community into 2025 and Beyond. ASEAN @50(5). Jakarta: Economic Research Institute for ASEAN & East Asia, pp., 1-252.

Martinot, E. 2016. Grid Integration of Renewable Energy: Flexibility, Innovation, and Experience. *Annual Review of Environment and Resources.* 41, pp. 223-251.

Mason, M. 2010. Sample Size and Saturation in PhD Studies Using Qualitative Interviews. *Qualitative Social Research.* 11(3), pp. 1-19.

Massachusetts Institute of Technology (MIT). 2017. Pathways to Paris: ASEAN. Massachusetts: MIT Press.

Matsuo, Y. & Tsunoda, M. 2016. Integrated, Trans-Boundary Energy and Electricity Markets in the BIMP Region—A Quantitative Assessment. In: Li, Y. & Kimura, S. eds. *Achieving an Integrated Electricity Market in Southeast Asia: Addressing the Economic, Technical, Institutional, and Geo-political Barriers.* Jakarta: Economic Research Institute for ASEAN & East Asia, pp. 41-58.

McGregor, A. 2008. Southeast Asian Development. London: Routledge.

McInnes, G. 2017. Understanding the Distributional and Household Effects of the Low-Carbon Transition in the G20 Countries. Paris: OECD.

*McKinsey Global Institute*. 2014. Southeast Asia at the Crossroads: Three Paths to Prosperity. New York: McKinsey&Co.

McKinsey&Co. 2016. China's One Belt, One Road: Will it Reshape Global Trade? *McKinsey Podcast.* [Podcast]. [Accessed 08 September 2019]. Available from: <u>https://www.mckinsey.com/</u>

Ministry of Commerce of the People's Republic of China. 2012. Announcement No. 26 of 2012 of the Ministry of Commerce of the People's Republic of China on the Preliminary Investigation Conclusion on the US Policy Support and Subsidies for Its Renewable Energy Sector. [Online]. [Accessed 17 June 2013]. Available from: <u>http://english.mofcom.gov.cn/</u>

Mkhwanazi, X. 2003. Power Sector Development in Africa. In: *The Workshop for African Energy Experts on Operationalizing the NGPAD Energy Initiative, 2/3 June 2003, Senegal.* Senegal & Paris: Republic of Senegal & United Nations, pp. 1-15.

Mol, A.P.J. & Spargaren, G. 2000. Ecological Modernisation Theory in Debate: A Review. *Environmental Politics.* [Online]. 9(1), pp. 17-49. [Accessed 01 May 2016]. Available from: <u>https://www.researchgate.net/</u>

Moon, C. & Prasad, R. 1998. Networks, Politics, and Institutions. In: Chan, S., Lutz, H., Lam, D., & Clark, C. eds. *Beyond the Developmental State: East Asia's Political Economies Reconsidered.* London: Palgrave Macmillan, pp. 9-24.

Moon, C. 1999. Political Economy of East Asian Development and Pacific Economic Cooperation. *The Pacific Review.* 12(2), pp. 199-224.

Morse, J. M. 1998. Validity by Committee. *Qualitative Health Research*. 8, pp. 443-445.

Morse, J.M. 1995. The Significance of Saturation. *Qualitative Health Research*. 5(2), pp. 147-149.

Morse, J.M., Barrett, M., Mayan, M., Olson, K., & Spiers, J. 2002. Verification Strategies for Establishing Reliability and Validity in Qualitative Research. *International Journal of Qualitative Methods.* 1(2), pp. 13-22.

Moudud, J.K. & Botchway, K. 2008. The Search for a New Developmental State. *International Journal of Political Economy*. 37(3), pp. 5-26.

Mundaca, L. & Richter, J.L. 2015. Assessing 'Green Energy Economy' Stimulus Packages: Evidence from the US Programs Targeting Renewable Energy. *Renewable and Sustainable Energy Reviews.* 43, pp 1174-1186.

Mundaca, L. T., C. Dalhammar, C. & Harnesk, D. 2013. The Integrated NORDIC Power Market and the Deployment of Renewable Energy Technologies: Key Lessons and Potential Implications for the Future ASEAN Integrated Power Market. In Kimura, S., Phoumin, H. and Jacobs, B. eds. 2013. *Energy Market*  *Integration in East Asia: Renewable Energy and its Deployment into the Power System.* Jakarta: Economic Research Institute for ASEAN & East Asia, pp.25-97.

Mungiu-Pippidi, A. & Hartmann, T. 2019. Corruption and Development: A Reappraisal. *Oxford Research Encyclopedia Economics & Finance.* USA: Oxford University Press.

Murphy, J., Gouldson, A. 2000. Environmental Policy and Industrial Innovation: Integrating Environment and Economy through Ecological Modernisation. *Geoforum.* 31(1), pp. 33-44.

Naimoli, S. & Nakano, J. 2018. Fuel Competition in the Expanding Power Market. *Center for Strategic & International Studies-Pertamina Southeast Asia Energy Initiative.* 

Nair, D. 2016. A Strong Secretariat, A Strong ASEAN? A Re-evaluation. ISEAS 2016(8), pp. 1-12.

Nakano, J. 2019. Greening or Greenwashing the Belt and Road Initiative? Washington, D.C.: Center for Strategic & International Studies. Pp. 1-2.

Navarro, M. A. & Sambodo, M.T. 2013. The Pathway to ASEAN Energy Market Integration. Philippines Institute for Development Studies Discussion Paper Series. 40, pp. 1-21.

Navarro, M.A., Detros, K.C., & dela Cruz, K.J. 2016. Post-EPIRA Impacts of Electric Power Industry Competition Policies. Philippines Institute for Development Studies Discussion Paper Series. 15, pp. 1-45.

Newbery, D. 2002. Issues and Options for Restructuring Electricity Supply Industries. *Cambridge Working Papers in Economics* 0210. Cambridge: Department of Applied Economics, University of Cambridge.

Nguitragool P. 2014. Learning from Locals: Doing Interviews in Southeast Asia. In: Huotari M., Rüland J., Schlehe J. eds. *Methodology and Research Practice in Southeast Asian Studies.* London: Palgrave Macmillan.

Noy C. 2009. Sampling Knowledge: The Hermeneutics of Snowball Sampling in Qualitative Research. *International Journal of Social Science Research*. 11(4), pp. 327–44.

NRDC, People's Republic of China. 2015. Vision and Actions on Jointly Building Silk Road Economic Belt and 21st-Century Maritime Silk Road. Beijing: NRDC.

NREL. 2014. Flexibility in 21<sup>st</sup> Century Power Systems. *21<sup>st</sup> Century Power Partnership*. Colorado: NREL.

NREL. 2015. Source of Operational Flexibility. *21st Century Power Partnership.* Colorado: NREL.

O'Neill, K. 2009. *The Environment and International Relations*. Themes in international relations, New York: Cambridge University Press.

Odell, J.S. 2001. Case Study Methods in International Political Economy. *International Studies Perspectives.* 2, pp. 161-176.

OECD. 1990. Strategic Industries in a Global Economy. OECD: Paris.

OECD. 2010. *The Political Economy of Poor Growth.* [Online]. [Accessed 07 November 2015]. United Kingdom: OECD. Available from: <u>http://www.oecd.org/</u>

OECD. 2011. Maximum Sustainability Yield, Fishing (MSY). [Online]. [Accessed 03 June 2016]. Available from: <u>https://stats.oecd.org/</u>

OECD. 2012. Mapping Global Value Chains. [Online]. [Accessed 12 September 2017]. Available from: <u>https://www.oecd.org/</u>

OECD. 2013. Economic Outlook for Southeast Asia, China and India 2014: Beyond the Middle-Income Trap. Paris: OECD.

OECD. 2017. Towards Green Growth in Southeast Asia. Paris: OECD.

OECD. 2018a. China's Progress Towards Green Growth. *OECD Green Growth Papers.* 2018(05), pp. 1-40.

OECD. 2018b. Economic Outlook for Southeast Asia, China and India 2019. Paris: OECD.

OECD. 2018c. Economic Outlook for Southeast Asia, China and India 2019: Towards Smart Urban Transportation. Paris: OECD.

Oelsner, A. & Vion, A. 2011. Special Issue: Friendship in International Relations. *International Politics.* 48(1), pp. 1-9.

Office of the Leading Group for Promoting the Belt and Road Initiative. 2019. The Belt and Road Initiative: Progress, Contributions, and Prospects. Beijing: Foreign Language Press.

Olivier, J.G.J., Janssens-Maenhout, G., Muntean, M. & Peters. J.A.H.A. 2013. Trends in Global CO<sub>2</sub> Emissions: 2013 Report. PBL Netherlands Environmental Assessment Agency. Brussels: The Hague.

Öniş, Z. 2018. The 'Beijing Consensus' and Prospects for Democratic Development in China and Beyond. In: Bishop, M.L., Payne, A., Sen, K., Breslin, S., Öniş, Z., Muzaka, V., Booth, D., Lindsay, C., & Yeung, H.W-C. 2018. Revisiting the

Developmental State. *Sheffield Political Economy Research Institute.* Sheffield: The University of Sheffield.

Orgad, S. 2005. From Online to Offline and Back: Moving from Online to Offline Relationships with Research Informants. In: Hines, C. ed. 2005. *Virtual Methods: Issues in Social Research on the Internet*. Oxford: Berg.

Oseni, M. O. & Pollitt, M. G. 2014. *Institutional Arrangements for the Promotion of Regional Integration of Electricity Markets: International Experience*. Policy Research Working Paper: No. WPS 6947. Washington, DC: World Bank Group.

Owen, A.D., Finenko, A. & Tao, J. 2017. Power Interconnection in ASEAN Region: Lessons Learned from International Experiences. Singapore: Energy Studies Institute.

Owusu, P.A. & Asumadu-Sarkodie, S. 2016. A Review of Renewable Energy Sources, Sustainability Issues, and Climate Change Mitigation. *Cogent Engineering.* 3, pp. 2-14.

Pacudan, R. 2016. Road Map for Power Market Integration in the Brunei-Indonesia-Malaysia-Philippines (BIMP) Region. In: Li, Y. & Kimura, S. eds. *Achieving an Integrated Electricity Market in Southeast Asia: Addressing the Economic, Technical, Institutional, and Geo-political Barriers.* ERIA Research Project Report FY2015(16), pp. 41-58. Jakarta: ERIA.

Panezi, M. 2016. The WTO and The Spaghetti Bowl of Free Trade Agreements: Four Proposals for Moving Forward. *CIGI.* Policy Brief, 87.

Pang, E-S. 2000. The Financial Crisis of 1997-98 and the End of the Asian Developmental State. *Contemporary South East Asia.* 22(3), pp. 570-593.

Passey, R., Spooner, T., MacGill, I., Watt, M. & Syngellakis, K. 2011. The Potential Impacts of Grid-Connected Distributed Generation and How to Address Them: A Review of Technical and Non-Technical Factors. *Energy Policy*. 39, pp. 6280-6290.

Pauw, P., Bauer, S., Richerzhagen, C., Brandi, C. & Schmole, H. 2014. Different Perspectives on Differentiated Responsibilities. German Development Institute Discussion Paper, 6. Bonn: German Development Institute.

Pavarini, C. & Mattion, F. 2019. Tracking the Decoupling of Electricity Demand and Associated CO<sub>2</sub> Emissions. Paris: IEA.

Peng, D. & Poudineh, R. 2017. Electricity Market Design for a Decarbonised Future: An Integrated Approach. Oxford Institute for Energy Studies Paper Series, EL 26. Oxford: OIES. Pereira, M.G., Sena, J.A., Freitas, M.A.V., & da Silva, N.F. 2011. Evaluation of the Impact of Access to Electricity: A Comparative Analysis of South Africa, China, India, and Brazil. *Renewable and Sustainable Energy Reviews.* 15, pp. 1427-1441.

Perreault, T., Bridge, G., & McCarthy, J. Ed. 2015. *The Routledge Handbook of Political Ecology.* London: Routledge.

Peters, L.D., Pressey, A.D., Vanharanta, M., and Johnston, W.J. 2013. Constructivism and Critical Realism as Alternative Approaches to the Study of Business Networks: Convergences and Divergences in Theory and in Research Practice. *Industrial Marketing Management*. 42, pp. 336–346.

Phoumin, H. & Kimura, S. 2015. Institutional Policy and Economic Impacts of Energy Subsidies Removal in East Asia. *ERIA Project Report.* 15(23), pp. 1-101.

Pillai, A.V. & Prasai, S. 2019. The Political Economy of Electricity Trade in and Hydropower Development in Eastern South Asia. *International Journal of Water Resource Development.* Pp. 1-13.

Pirie, I. 2005. The New Korean State. *New Political Economy*. 10(1), pp. 25-42.

Pirie, I. 2008. *The Korean Developmental State: From Dirigisme to Neoliberalism*. London: Routledge.

Poch, K. & S. Tuy. 2012. Cambodia's Electricity Sector in the Context of Regional Electricity Market Integration. In Wu, Y., Shi, X., & Kimura, F. eds. *Energy Market Integration in East Asia: Theories, Electricity Sector and Subsidies.* ERIA Research Project Report 2011(17). Jakarta: Economic Research Institute for ASEAN & East Asia, pp.141-172.

Pollitt, M.G. 2004. Electricity Reform in Chile: Lessons for Developing Countries. Cambridge Working Papers in Economics 0448. Cambridge: Electricity Policy Research Group.

Pollitt, M.G. 2008. The Future of Electricity (and Gas) Regulation in a Low-carbon Policy World. *The Energy Journal*. Pp. 63-94.

Pollitt, M.G. 2012. The role of Policy in Energy Transitions: Lessons from the Energy Liberalisation Era. *Energy Policy.* 50, pp. 128–37.

Pollitt, M.G., Yang, C.H. & Chen, H. 2017. Reforming the Chinese Electricity Supply Sector: Lessons from the International Experience. *Energy Policy Research Group*, University of Cambridge. 1704, pp. 1-69. Porter, M. & Situmeang, H. 2005. Preparing for Electricity Trading in ASEAN. REPSF Project 03(022). Australia: Australian Agency for International Development.

Porter, M. 1990. The Competitive Advantage of Nations. New York: Free Press.

Poudineh, R. & Rubino, A. 2016. Business Model for Cross-Border Interconnections in the Mediterranean Basin. Oxford Institute for Energy Studies Paper Series, EL 19. Oxford: OIES.

Prantl, J. 2014. Multilateralism in East Asia: The Good, The Bad, and the Ugly. *The Asan Forum.* Pp. 1-16.

Pritchard, R. 2003. East Asian Electricity Trade—A Compelling Case for Interconnection. *Journal of Energy & Natural Resources Law.* 21(4), pp. 470-487.

Puka, L. & Szulecki, K. 2014. The politics and economics of cross-border electricity infrastructure: A framework for analysis. *Energy Research & Social Science*. 4(2014), pp. 124–134.

Punch, K.F. 2014. *Introduction to Social Research: Quantitative & Qualitative Approaches.* London: Sage.

PWC. 2016. Power in Indonesia. Jakarta: PWC.

Qi, H. 2009. Contradictions and Compatibility between China's Multilateral Cooperation System and the US-Led Bilateral Alliance Structure in East Asia. *China International Studies*. 14, pp. 118-141.

Quiggin, D. & Froggatt, A. 2018. Flexibility: Shifting the Power Balance. *Chatham House.* [Online]. [Accessed 20 August 2018]. Available from: <u>https://hoffmanncentre.chathamhouse.org/</u>

Radice, H. 2008. The Developmental State under Global Neoliberalism. *Third World Quarterly.* 29(6), pp. 1153-1174.

Rahman, S.H., Wijayatunga, P.D.C., Gunatilake, H. & Fernando, P.N. 2013. Energy Trade in South Asia. Manila: ADB.

Rajah, A. 1999. Southeast Asia: Comparatist Errors and the Construction of a Region. *Southeast Asian Journal of Social Science*. 27(2), pp. 41-53.

Reinert, E.S. & Reinert, S.A. 2011. Mercantilism and Economic Development: Schumpeterian Dynamics, Institution Building, and International Benchmarking. *OIKOS*. 10(1), pp. 8-37.

Reinke de Buitrago, S. 2009. What Role Can Trust Play? Conflicts and Trust-Building in East Asia. *Asian Politics & Policy*. 1(4), pp. 739-761.

REN21. 2017. Renewables 2017 Global Status Report. Paris. REN21 Secretariat.

REN21. 2019a. *Asia and the Pacific Renewable Status Report 2019*. Paris: REN21 Secretariat.

REN21. 2019b. Renewables 2019 Global Status Report. Paris. REN21 Secretariat.

Renwick, N., Gu, J. & Gong, S. 2018. The Impact of BRI Investment in Infrastructure on Achieving the Sustainable Development Goals. *K4D Emerging Issues Report*. Brighton: Institute of Development Studies.

Reuters. 2019. RPT-Malaysia Says it is Keen for More 'Fair' Deals with China. Reuters. [Online]. 24 April. [Accessed 13 June 2019]. Available from: <u>https://uk.reuters.com/</u>

Ricardo, D. 1815. *Essay on the Influence of a Low Price of Corn on the Profits of Stock*. London: Oxford University.

Roasa, D. 2012. China's Soft Power Surge. Foreign Policy. Pp. 1-6.

Robison, R. 1996. The Politics of 'Asian Values'. *The Pacific Review*. 9(3), pp. 309-327.

Robitaille, D.F. & Garden, R.A. eds. 1996. *Research Questions & Study Design.* Vancouver: Pacific Educational Press.

Rodan, G., Hewison, K. & Robison, R. 2006. *The Political Economy of South East Asia: Markets, Power and Contestation*. Oxford: Oxford University Press.

Rodrik, D. & Rosenzweig, M.R. 2010. Development Economics. 5. Oxford: Elsevier.

Rodrik, D. 2003. *In Search of Prosperity: Analytical Narratives on Economic Growth.* Princeton: Princeton University Press.

Rodrik, D. 2018. What Do Trade Agreements Really Do? *Journal of Economic Perspectives.* 32(2), pp. 73-90.

Roncancio, I.V., Temper, L., Sterlin, J., Smolyar, N.L., Sellers, S., Moore, M., Melgar-Melgar, R. Larson, J., Homer, C., Erickson, J.D., Egler, M., Brown, P.G., Boulot, E., Beigi, T. & Babcock, M. 2019. From the Anthropocene to Mutual Thriving: An Agenda for Higher Education in the Ecozoic. *Sustainability*. 11(1312), pp. 1-19.

Rosa, L.P., dos Santos, M.A., Matvienko, B., dos Santos, E.O., & Sikar, E. 2004. Greenhouse Gas Emissions from Hydroelectric Reservoirs in Tropical Regions. *Climate Change*. 66, pp. 9-21. Rowlands, I.H. 2009. Renewable Electricity Politics across Borders. *Changing Climates in North American Politics Institutions, Policymaking, and Multilevel Governance*. Boston: MIT Press, pp. 181–198.

Rubini, L. 2011. *The Subsidization of Renewable Energy in the WTO: Issues and Perspectives.* Alabama: University of Birmingham, School of Law.

Rudestam, K.E. & Newton, R.R. 2007. *Surviving Your Dissertation – A Comprehensive Guide to Content and Process* (3rd ed.). Thousand Oaks: Sage.

Russett, B.M. 1967. The Ecology of Future International Politics. *International Studies Quarterly.* [Online]. 11(1), pp. 12-31. [Accessed 09 April 2015]. Available from: <u>http://www.jstor.org/</u>

*Russian International Affairs Council.* 2013. Challenges of Interviewing Elites. [Online]. [Accessed 29 April 2016]. Available from: <u>http://russiancouncil.ru/</u>

Salvatore, D., 2013. International Economics: Trade and Finance. 11th Edition. Hoboken: John Wiley & Sons.

Samsudin, M.S.N., Rahman, M.M. & Wahid, M.A. 2016. Power Generation Sources in Malaysia: Status and Prospects for Sustainable Development. *Journal of Advance Review on Scientific Research.* 25(1), pp. 11-28.

Samuelson, P.A. 1962. The Gains from International Trade once Again. *The Economic Journal*. 72(288), pp. 820-829.

Sandu, S., Yang, M., Mahlia, T.M.I., Wonsapai, W., Ong, H.C., Putra, N., and Rahman, S.M.A. 2019. Energy-Related CO2 Emissions Growth in ASEAN Countries: Trends, Drivers and Policy Implications. *Energies.* 12(4650), pp. 1-15.

Sarawak Energy. 2017. Realising the Borneo Grid. [Press Release]. [Accessed 12 August 2018]. Available from <u>https://www.sarawakenergy.com/</u>

Sawin, J.L. 2004. Policy Lessons for the Advancement & Diffusion of Renewable Energy Technologies Around the World. In: International Conference for Renewable Energies, Bonn, Germany. Bonn: Secretariat of the International Conference for Renewable Energies, Bonn 2014, pp. 1-52.

Sayer, A. 1992. *Method in Social Science: A Realist Approach*. London: Routledge.

Sayer, A. 2016. Moral Economy: A Different Way of Thinking about the Future. [Online]. [Accessed 12 March 2019]. Available from: https://www.opendemocracy.net/

Schmookler, J. 1966. *Invention and Economic Growth*. Boston: Harvard University Press.

Schneider, F. 2014. *What's in a Methodology.* [Online]. [Accessed 01 February 2020]. Available from: <u>http://www.politicseastasia.com/</u>

Schumpeter, J. 1934. *The Theory of Economic Development*. Boston: Harvard University Press.

Schumpeter, J. 1935. The Analysis of Economic Change. *The Review of Economics and Statistics.* 17(4), pp. 1-10.

Schumpeter, J. 1942. *Capitalism, Socialism, and Democracy*. New York: Harper Collins.

SCMP. 2019. Explained: Belt and Road Initiative. [Online]. *South China Morning Post.* [Accessed 13 March 2019]. Available from: <u>https://www.scmp.com/</u>

Seidman, I. 2013. *Interviewing as Qualitative Research*. 4<sup>th</sup> ed. New York: Teachers College Press.

Selivanova, Y. 2015. Clean Energy and Access to Infrastructure: Implications for the Global Trade System. Geneva: ICTSD.

Sen, A. 2014. Divergent Paths to a Common Goal: An Overview of Challenges to Electricity Sector Reform in Developing versus Developed Countries. *Oxford Institute for Energy Studies*. EL Working paper 10, pp. 221-241.

Sen, A. Nepal, R. & Jamasb, T. 2016. Reforming Electricity Reforms? Empirical Evidence from Asian Economies. Oxford Institute for Energy Studies. OIES PAPER, EL 18, pp. 1-50.

Sen, A., Nepal, R. & Jamasb, T. 2017. Rethinking Electricity Reform in Asia: Balancing Economic and Environmental Objectives. In Bhattacharya, S. ed. *Handbook of Energy in Asia*. London: Routledge, pp. 221-240.

Sen, K. 2018. Why Have we seen so few Developmental States? In: Bishop, M.L., Payne, A., Sen, K., Breslin, S., Öniş, Z., Muzaka, V., Booth, D., Lindsay, C. & Yeung, H.W-C. Revisiting the Developmental State. *Sheffield Political Economy Research Institute.* Sheffield: The University of Sheffield.

Serra, N. & Stiglitz, J.E. 2008. *The Washington Consensus Reconsidered*. London: Oxford University Press.

Shafaeddin, M. 1998. How Did Developed Countries Industrialize? The History of Trade and Industrial Policy: The Cases of Great Britain and the USA. [Online]. Geneva: UNCTAD. [Accessed 24 April 2016]. Available from: <u>http://unctad.org/</u>

Shafaeddin, M. 2000. What Did List Actually Say? Some Clarifications on the Infant Industry Argument. [Online]. Geneva: UNCTAD. [Accessed 24 April 2016]. Available from: <u>http://unctad.org/</u>

Shanghai WTO Affairs Consultation Centre. 2011. China's Trade Performance Report: World Economy Restructuring and China's Trade. Shanghai: SCPG Publishing Corporation.

Shani, N. 2017. How Power Grids Could Embrace the Era of Renewable Integration. ASEAN Centre for Energy. [Online]. [Accessed 01 February 2018]. Available from: <u>http://www.aseanenergy.org/</u>

Sheng, Y. & Shi, X. 2012. Economic Development, Energy Market Integration and Energy Demand: Implication for East Asia. In: Wu, Y., Shi, X., and Kimura, F. eds. *Energy Market Integration in East Asia: Theories, Electricity Sector and Subsidies*. Jakarta: ERIA.

Sheng, Y. & Shi, X. 2013. Energy Market Integration and Equitable Growth Across Countries. *Applied Energy*. 104, pp. 319-325.

Shi, X. & Widodo, T. 2014. Benefits of ASEAN Energy Market Integration: A Survey of the Literature. ASEAN Energy Market Integration Working Paper Series. Bangkok: ASEAN Studies Centre.

Shi, X. 2014. ASEAN Power Grid, Trans-ASEAN Gas Pipelines and ASEAN Economic Community: Vision, Plan and the Reality. *Global Review.* Pp. 116-131.

Shi, X. 2015. Development of ASEAN Power Grid: Implications for Northeast Asian Cooperation. *The 3<sup>rd</sup> Northeast Asia Energy Security Forum, 17 December 2015, Seoul, Republic of Korea.* Singapore: Energy Studies Institute, The National University of Singapore.

Shi, X. 2016. The Future of ASEAN Energy Mix: A SWOT Analysis. *Renewable and Sustainable Energy Reviews.* 53, pp. 672-680.

Shi, X., Yao, L. & Jiang, H. 2019. Regional Power Connectivity in Southeast Asia: The Role of Regional Cooperation. Global Energy Interconnection Development and Cooperation Organization. 2(5), pp. 444-456.

Silk Road Fund. [no date]. About Us. [Online]. [Accessed 13 May 2018]. Available from: <u>http://www.silkroadfund.com.cn/</u>

Simpson, A. 2007. The Environment – Energy Security Nexus: Critical Analysis of an Energy 'Love Triangle' in Southeast Asia. *Third World Quarterly*. 28(3), pp. 539-554.

Singh, J.N. & Ovadia, J.S. 2018a. The Theory and Practice of Building Developmental States in the Global South. *Third World Quarterly.* 39(6), pp. 1033-1055.

Singh, J.N. & Ovadia, J.S. eds. 2018b. Developmental States Beyond East Asia. *Third World Quarterly*. 39(6), pp. 1033-1055.

Skaburskis, A. 2008, The Origin of "Wicked Problems". Planning Theory & Practice. (9)2, pp. 277-280.

Skovgaard, J. & Van Asselt, H. 2018. *The Politics of Fossil Fuel Subsidies and Their Reform.* Cambridge: Cambridge University Press.

Smith, A. 1776. *An Inquiry into the Nature and Causes of the Wealth of Nations*. London: Strahan W. and Cadell T.

Snyder, J. 2019. The Broken Bargain: How Nationalism Came Back. *Foreign Affairs.* [online]. [Accessed 03 February 2020]. Available from: <u>https://www.foreignaffairs.com/</u>

Sovacool, B.K & Brown, M.A. 2009. Addressing Climate Change: Global vs. Local Scales of Jurisdiction? In: Sioshansi, F.P. ed. *Generating Electricity in a Carbon-Constrained World*. Amsterdam: Elsevier, pp. 109-124.

Sovacool, B.K., Axsen, J. & Sorrell, S. 2018. Promoting Novelty, Rigor, and Style in Energy Social Science: Towards Codes of Practice for Appropriate Methods and Research Design. *Energy Research & Social Science.* 45, pp. 12-42.

Spencer B. & Brander J.A. 2008. Strategic Trade Policy. In: Durlauf S.N. & Blume L.E. eds. *The New Palgrave Dictionary of Economics*. Hampshire: Palgrave Macmillan.

Spencer, B. 1986. What Should Trade Policy Target? In Krugman, P.R. ed. 1986. *Strategic Trade Policy and the New International Economics*. Massachusetts: The MIT Press, pp. 69-90.

Springett, D. 2013. Editorial: Critical Perspectives on Sustainable Development. *Sustainable Development.* 21, pp. 73–82.

Srisuping, K. 2013. *ASEAN Power Grid*. Expert Group Meeting on Conceptualizing the Asian Energy Highway, September 3-5, 2013, Urumqi, China.

Statistics Netherlands. 2013. Sustainable Development and Green Growth: Comparisons of the Measurement Frameworks at Statistics Netherlands. The Netherlands: Statistics Netherlands.

Stegemann, K. 1985. Strategic Trade Policy. In: Greenway, D. eds. *Current Issues in International Trade*. London: Macmillan Press, pp. 82-99.

Stern, N. 2014. Growth, Climate and Collaboration: Towards Agreement in Paris 2015. [Online]. Centre for Climate Change Economics and Policy and the

Grantham Research Institute on Climate Change and the Environment. [Accessed 20 April 2016]. Available from: <u>http://www.lse.ac.uk/</u>

Stern, N.,S. Peters, V. Bakhshi, A. Bowen, C. Cameron, S. Catovsky, D. Crane, S. Cruickshank, S. Dietz, N. Edmonson, S.-L. Garbett, L. Hamid, G. Hoffman, D. Ingram, B. Jones, N. Patmore, H. Radcliffe, R. Sathiyarajah, M. Stock, C. Taylor, T. Vernon, H. Wanjie, & D. Zenghelis. 2007. Stern Review: The Economics of Climate Change. London: HM Treasury.

Strange, S. 1988. *States and Markets*. London: Pinter Publishers.

Sudo, S. 2003. Regional Governance and East and Southeast Asia: Towards the Regulatory State? *Japanese Journal of Political Science*. 4(2), pp. 331-347.

Suh, E.E., Kagan, S. Strumpf, N. 2009. Cultural Competence in Qualitative Interview Methods with Asian Immigrants. *Journal of Transcultural Nursing.* 20(2), pp. 194-201.

Sun, J. & Yang, K. 2016. The Wicked Problem of Climate Change: A New Approach Based on Social Mess and Fragmentation. *Sustainability.* 8(1312), pp. 1-14.

Suryadi, B. 2020. Energy Priorities in High-Performing ASEAN. [Online]. [Accessed September 20, 2020]. Available from: https://www.energyforgrowth.org/

Sztompka, P. 1999. *Trust: A Sociological Theory.* Cambridge: Cambridge University Press.

Taggart, B.S., James, G., Dong, Z., and Russell, C. 2012. The Future of Renewables Linked by a Transnational Asian Grid. *Proceedings of the IEEE*. 100(2), pp. 348–359.

Taggart, S. 2013. *A Pan-Asian Energy Infrastructure by 2050*. Grenatec: Sydney, Australia.

Taguchi, M. & Murofushi, H. 2014. International Production Networks in ASEAN Economies. *Munich Personal RePEc Archive.* MPRA paper. 64409, pp. 1-21.

Takapong, S. 2016. *Regional Connectivity and Grid Interconnection*. Sustainable Energy & Technology Asia, March 23, 2016, Bangkok, Thailand.

Tan, G. 1999. *The End of the Asian Miracle?* Singapore: Times Academic Press.

Tan, G. 2000. *The Asian Currency Crisis.* Singapore: Times Academic Press.

Tawney, L. 2012 *Taking Renewable Energy to Scale in Asia*. Washington, D.C.: The National Bureau of Asian Research.

The Economist. 2013. Towards the End of Poverty. [Online]. [Accessed 15 February 2016]. Available from: <u>http://www.economist.com/</u>

The Oxford Dictionary. [Online]. 2016. Asia-Pacific. [Accessed 15 September 2014]. Available from: <u>http://www.oxforddictionaries.com/</u>

Thurbon, E. 2012. From Developmentalism to Neoliberalism and Back Again? In: Chang, K-S., Fine, B., and Weiss, L. *Developmental Politics in Transition*. Hampshire: Palgrave Macmillan, pp. 274-298.

Thurbon, E. 2014. The Resurgence of the Developmental State: A Conceptual Defence. *Critique Internationale.* 63, pp. 59-75.

TNB (Tenaga Nasional). 2020. *Malaysian Grid Code.* [Online]. [Accessed 01 May 2020]. Available from: <u>https://www.tnb.com.my/</u>

Togo, K. 2008. Japan's Historical Memory: Reconciliation with Asia. *The Asia-Pacific Journal.* 6(12), pp. 1-11.

Tominaga, K. 1991. A Theory of Modernization and Social Change of the Non-Western Societies: Toward a Generalization from Japan's Experiences. *International Review of Sociology.* 2(3), pp. 95-120.

Trafford, V. & Leshem, S. 2008. *Stepping Stones to Achieving Your Doctorate*. Maidenhead: McGraw-Hill.

Trebilcock, M.J. & Howse, R. 1995. *The Regulation of International Trade.* United Kingdom: Routledge.

UN, ADB, UNDP (United Nations Development Program). 2017. Asia-Pacific Sustainable Development Goals Outlook. Thailand: UN.

UN. 2006. Multi-Dimensional Issues in International Electric Power Grid Interconnections. New York: UN.

UN. 2009. Rethinking Poverty: Report on the World Social Situation 2010. Geneva: UN.

UN. 2015. Transforming our World: The 2030 Agenda for Sustainable Development. Geneva: UN.

UNCTAD. 2014. Trade Remedies: Targeting the Renewable Energy Sector. Geneva: UN.

UNCTAD. 2017. ASEAN AT 50: Achievements and Challenges in Regional Integration. Geneva: UN.

Underhill, G.R.D. & Zhang, X. 2005. The Changing State-Market Condominium in East Asia: Rethinking the Political Underpinnings of Development. *New Political Economy.* 10(1), pp. 1-24.

UNDSD. 2012. *A Guidebook to the Green Economy: Issue 1.* New York: UN Department for Economic and Social Affairs.

UNESCAP. 2016. Promoting Regional Energy Connectivity in Asia and the Pacific: Challenges and Opportunities for Aligning the Energy Sector with Sustainable Development. [Online]. [Accessed 03 January 2018]. Available from: <u>https://www.unescap.org/</u>

UNESCAP. 2018. Energy Interconnection in ASEAN for Sustainable and Resilient Societies. New York: UN.

UNESCAP. 2019. Energy & Development in the ASEAN Region: Statistical Perspectives. New York: UN.

UNFCCC. [No date]. *Kyoto Protocol*. [Online]. [Accessed 25 September 2014]. Available from: <u>http://unfccc.int/</u>

UNFCCC. 2014. *Timeline*. [Online]. [Accessed 28 May 2016]. Available from: <u>http://unfccc.int/</u>

UNFCCC. 2020. What is the United Nations Framework Convention on Climate Change? [Online]. [Accessed 13 January 2020] Available from: <u>https://unfccc.int/</u>

UNIPCC. 2007. Thematic Debate of the General Assembly: Climate Change as a Global Challenge. New York: IPCC.

UNIPCC. 2013. *Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change*. Cambridge: Cambridge University Press. [Accessed 25 November 2014]. Available from: <u>http://www.ipcc.ch/</u>

UNIPCC. 2019. IPCC Special Report on Climate Change and Land. New York: UNIPCC.

Ürge-Vorsatz, D. & Herrero, S.T. 2012. Building Synergies Between Climate Change Mitigation and Energy Poverty Alleviation. *Energy Policy*. 49, pp. 83-90.

Urpelainen, J. 2013. Promoting International Environmental Cooperation Through Unilateral Action: When Can Trade Sanctions Help? *Global Environmental Politics.* 13(2), pp. 26-45.

USAID-NREL. 2017. Grid-Scale Battery Storage. Boulder, CO: USAID-NREL Partnership.

Uttam, J. 2019. Beyond Neoliberalism: Analyzing the 'Post-Developmental State' in South Korea. *International Studies.* 56(4), pp. 236-254.

Vaismoradi, M., Turunen, H. & Bondas, T. 2013. Content Analysis and Thematic Analysis: Implications for Conducting a Qualitative Descriptive Study. *Nursing and Health Sciences.* 15, pp. 398-405.

Van Marrewijk, C. V. 2002. *International Trade & The World Economy*. Oxford: Oxford University Press.

Veenah, J. 2009. *Trade Flows Barriers, and Market Drivers in Renewable Energy Supply Goods*. Geneva: ICTSD.

Vezzoli, C., Ceschin, F., Osanjo, L., M'Rithaa, M. Moalosi, R., Nakazibwe, V. & Diehl, J.C. 2018. *Designing Sustainable Energy for All.* New York: Springer.

Victor, D. & Heller, T.C. 2007. The Political Economy of Power Sector Reform: The Experience of Five Major Developing Countries. Cambridge: Cambridge University Press.

Victor, D. 2009. The Politics of Fossil-Fuel Subsidies. Switzerland: International Institute for Sustainable Development.

Viner, J. 1950. *The Customs Union Issue.* Washington, D.C.: Carnegie Endowment for International Peace.

Vithayasrichareon, P., MacGill, I.F. & Nakawiro, T. 2012. Assessing the Sustainability Challenges for Electricity Industries in ASEAN Newly Industrialising Countries. *Renewable and Sustainable Energy Reviews.* 16, pp. 2217-2233.

Vogel, S.K. 1998. *Freer Markets, More Rules: Regulatory Reform in Advanced Industrial Countries.* Cornell Studies in Political Economy. New York: Cornell University Press.

Wade, R. & Mkandawire, T. 2017. *The Future of Developmental States*. Department Dialogue, 15 March 2017, London.

Wade, R. 1988. State Intervention: Taiwanese Practice. In: White, G. ed. *Developmental States in East Asia*. London: The Macmillan Press, pp. 30-67.

Wade, R. 1990. *Governing the Market: Economic Theory and the Role of Government in East Asian Industrialization.* Princeton: Princeton University Press.

Wade, R. 1992. East Asia's Economic Success: Conflicting Perspectives, Partial Insights, Shaky Evidence. *World Politics*. 44(2), pp. 270-320.

Wang, J. 2016. Revive China's Green GDP Programme. *Nature.* 534(37), pp. 37.

Wang, Y. 2014. 'Renqing' in Conducting Interviews with Chinese Business People: Insights from a Returning Researcher. 17 April 2014. *Field Research Method Lab at LSE*. [Online]. [21 April 2016]. Available from: <u>http://wp.me/</u>

Watcharejyothin, M. & Shrestha, R.M. 2009. Effects of Cross-Border Power Trade Between Laos and Thailand: Energy Security and Environmental Implications. *Energy Policy*. 37(11), pp. 4428-4441.

WCED. 1987. Our Common Future, From One Earth to One World: An Overview by the World Commission on Environment and Development (AKA Brundtland Commission). [Online]. Oslo: World Commission on Environment and Development. [Accessed 01 May 2015]. Available from: <u>http://www.un-documents.net/</u>

Weatherbee, D.E. *International Relations in Southeast Asia*. Lanham: Rowman & Littlefield.

Weiss, L. 1997. Globalization and the Myth of the Powerless State. NLR I-225, September-October 1997.

Weiss, L. 2003. Guiding Globalization in East Asia: New Roles for Old Developmental States. In: Weiss, L. ed. *States in the Global Economy*. Cambridge: Cambridge University Press, pp. 245-270.

Weiss, L. 2012. The Myth of the Neoliberal State. In: Chang, K-S., Fine, B., & Weiss, L. 2012. *Developmental Politics in Transition*. Hampshire: Palgrave Macmillan, pp. 274-298.

Weiss, T. G. 2013. *Global Governance: What? Why? Wither?* London: Polity Press.

Whally, J. 2011. What Role for Trade in a Post-2012 Global Climate PolicyRegime. The World Economy. [Online]. 34(11), pp. 1844-1862. [Accessed 10September2017].Availablefrom:<a href="http://o-onlinelibrary.wiley.com.wam.leeds.ac.uk/">http://o-</a>

Wharton Business Daily. 2019. China's Belt & Road Initiative: Why the Price is Too High. *Public Policy*. [Online]. [Accessed 12 September, 2020]. Available from: <u>https://knowledge.wharton.upenn.edu/</u>

Whelton, M. & Ballard, G. 2002. Wicked Problems in Project Definition. In Proceedings of the 10th Annual Conference on International Group for Lean Construction 10th Annual Conference, Porto Alegre, Brazil. Pp. 1–11.

White, G. & Wade, R. 1988. Developmental States and Markets in East Asia: An Introduction. In: White, G. ed. *Developmental States in East Asia*. London: The Macmillan Press, pp. 1-29.

White, P. 2009. *Developing Research Questions*. London: Palgrave Macmillan.

Whitley, S. 2013a. *At Cross-Purposes: Subsidies and Climate Compatible Investment.* London: Overseas Development Institute.

Whitley, S. 2013b. *Time to Change the Game: Fossil Fuel Subsidies and Climate.* London: Overseas Development Institute.

WHO. 2014. Meeting Report: International Experts Meeting on Air Pollution and Health in Asia. [Online]. World Health Organization Western Pacific Regional Office and Clean Air Initiative for Asian Cities. [Accessed 05 February 2016]. Available from: <u>http://www.wpro.who.int/</u>

WHO. 2018. [Online]. [Accessed 03 February 2019]. Available from: <u>https://www.who.int/</u>

Wiegratz, J. 2016. Neoliberal Moral Economy. London: Rowman & Littlefield.

Wolf, E. 1972. Ownership and Political Ecology. *Anthropological Quarterly*. [Online]. 45(3), pp. 201-205. [Accessed 01 May 2016]. Available from: <u>http://0-www.jstor.org.wam.leeds.ac.uk/</u>

Woliver, L. R. 2002. Ethical Dilemmas in Personal Interviewing. *Political Science and Politics.* 35(4), pp. 677-678.

Wong, J. 2004. The Adaptive Developmental State in East Asia. *Journal of East Asian Studies.* 4, pp. 345-362.

Woo-Cumings, M. 1999. Introduction: Chalmers Johnson and the Politics of Nationalism and Development. In: M. Woo-Cumings, ed. *The Developmental State*. Ithaca, New York: Cornell University Press, pp. 1–31.

Wood, A. & Mayer, J. 2010. Has China De-Industrialised Other Developing Countries? QEH Working Paper Series 175. [Accessed 27 February 2017]. Available from: <u>http://www3.qeh.ox.ac.uk/</u>

World Bank. 1992. *Governance and Development*. Washington, D.C.: World Bank.

World Bank. 1993. *The East Asian Miracle*. New York: Oxford University Press, Inc.

World Bank. 2008. Building Regional Power Pools: A Toolkit. Washington, D.C.: World Bank.

World Bank. 2012. Inclusive Green Growth: The Pathway to Sustainable Development. Washington, D.C.: The World Bank Group.

World Bank. 2018. Belt and Road Initiative. [Online]. [Accessed 15 May 2018]. Available from: <u>https://www.worldbank.org/</u>

World Bank. 2019. Belt and Road Economics: Opportunities and Risks of Transportation Corridors. New York: The World Bank.

World Meters. 2020. Electricity Generation. [Online]. [Accessed 12 December 2019]. Available from: <u>https://www.worldometers.info/</u>

WTO. 2009. *Energy in WTO Law and Policy*. Geneva: NCCR Trade Regulation.

WTO. 2011. World Trade Report 2011: The WTO and Preferential Trade Agreements. Geneva: WTO. [Accessed: June 28, 2017]. Available from: <u>https://www.wto.org/</u>

WTO. 2016a. The WTO in Brief: Part 1. [Online]. [Accessed 15 May 2016]. Available from: <u>https://www.wto.org/</u>

WTO. 2016b. WTO-Grantham Institute Panel Debate Trade's Contribution to Clean Energy Solutions. [Online]. [Accessed 01 July 2016]. Available from: <u>https://www.wto.org/</u>

WTO. 2017. The Uruguay Round. [Online]. [Accessed 01 July 2017]. Available from: <u>https://www.wto.org/</u>

Wu, Y. 2012. Electricity Market Integration: Global Trends and Implications for the EAS Region. In: Wu, Y., Shi, X. & Kimura, F. eds. *Energy Market Integration in East Asia: Theories, Electricity Sector and Subsidies*. Jakarta: Economic Research Institute for ASEAN & East Asia, PP. 138-145.

Wu, Y. 2016. Electricity Market Integration in ASEAN: Institutional and Political Barriers and Opportunities. In Li, Y. & Kimura, S. eds. *Achieving an Integrated Electricity Market in Southeast Asia: Addressing the Economic, Technical, Institutional, and Geo-political Barriers*. Jakarta: Economic Research Institute for ASEAN & East Asia, pp. 109-125.

Wu, Y. 2019. Electricity Market Integration: Global Trends and Implications for the EAS Region. The University of Western Australia, Discussion Paper. 12(19), pp. 1-23.

Wu, Y., Shi, X., and Kimura, F. 2012. The Electricity Sector Leads Energy Market Integration in East Asia: Introduction. In: Wu, Y., Shi, X. & Kimura, F. eds. *Energy Market Integration in East Asia: Theories, Electricity Sector and Subsidies*. Jakarta: ERIA. Yao, L.X., Shi, X.P. & Andrews-Speed, P. 2018. Conceptualization of Energy Security in Resource-Poor Economies: The Role of the Nature of Economy. *Energy Policy.* 114, pp. 394-402.

Yazan, B. 2015. Three Approaches to Case Study Methods in Education: Yin, Merriam, and Stake. *The Qualitative Report*. 20(2), pp. 134-152.

Yeung, H. W-C. 1997. Critical Realism and Realist Research in Human Geography: A Method or a Philosophy in Search of a Method? *Progress in Human Geography.* 21(1), pp. 51-74.

Yeung, H.W-C. 2003. Practicing New Economic Geographies: A Methodological Examination. *Annals of the Association of American Geographers*. 93(2), pp. 442-462

Yeung, H.W-C. 2018. The New Challenges of Asia's Latecomer Industrialisation. In: Bishop, M.L., Payne, A., Sen, K., Breslin, S., Öniş, Z., Muzaka, V., Booth, D., Lindsay, C. & Yeung, H.W-C. 2018. Revisiting the Developmental State. *Sheffield Political Economy Research Institute.* Sheffield: The University of Sheffield, pp. 20-22.

Yin, R.K. 1981a. The Case Study as Serious Research. *Knowledge: Creation, Diffusion, Utilization.* 3(1) pp. 97-114.

Yin, R.K. 1981b. The Case Study Crisis: Some Answers. *Administrative Science Quarterly.* 26(1), pp. 58-65.

Yin, R.K. 1983. *The Case Study Method: An annotated Bibliography.* Washington, D.C.: The COSMOS Corporation.

Yin, R.K. 1984. 2003. 2009. *Case Study Research: Design and Method*. London: Sage.

Yin, R.K. 2011. *Qualitative Research from Start to Finish.* New York: The Guilford Press.

Yokota, E. & Kutani, I. 2017. Study on Electricity Supply Mix and Role of Policy in ASEAN. Tokyo: ERIA.

Yokota, E. & Kutani, I. 2018. Comparative Analysis of Power Prices in the Philippines and selected ASEAN countries. Tokyo: ERIA.

York, R. & Rosa, E.A. 2003. Key Challenges to Ecological Modernization Theory. *Organization & Environment.* [Online]. 16(3), pp. 273-288. [Accessed 01 April 2016]. Available from: <u>http://www.skidmore.edu/</u>

Yosiyana, B. 2018. *ASEAN Energy Trends 2018* [PowerPoint Presentation]. ASEAN Centre for Energy Webinar, online.

Yu, X. 2003. Regional Cooperation and Energy Development in the Greater Mekong Sub-region. *Energy Policy*. 31, pp. 1121-1234.

Zaccai, E. 2012. Over Two Decades in Pursuit of Sustainable Development: Influence, Transformations, Limits. *Environmental Development.* 1, pp. 79–90.

Zachariadis, M. Scott, S. & Barrett, M. 2010. Exploring Critical Realism as the Theoretical Foundation of Mixed-Method Research: Evidence from the Economics of SI Innovations. *Working Paper Series 032010.* Cambridge: Cambridge University.

Zehra. F.A. 1998. Democracy and Economic Development: Modernization Theory Revisited. *Comparative Politics.* [Online]. 21(1), pp. 21-36. [Accessed 05 April 2016]. Available from: <u>http://www.jstor.org/</u>

Zeitoun, M. & Allan, J.A. 2008. Applying Hegemony and Power Theory to Transboundary Water Analysis. *Water Policy*. 10(2), pp. 3-12.

Zeng, K. & Liang, W. eds. *China and Global Trade Governance: China's First Decade in the World Trade Organization*. Oxon: Routledge.

Zha, D. 2015. China's Economic Diplomacy Focusing on the Asia-Pacific Region. *China Quarterly of International Strategic Studies*. 1(1), pp. 85–104.

Zhang, H.Z. 2020. China's 'Development Approach' to the Mekong Water Disputes. *The Diplomat.* [Online]. [Accessed 20 March 2020]. Available from: <u>https://thediplomat.com/</u>

Zhang, Y.F., Parker, D., & Kirkpatrick, C. 2008. Electricity Sector Reform in Developing Countries: An Econometric Assessment of the Effects of Privatization, Competition and Regulation. *Journal of Regulatory Economics*. 33(2), pp. 159-178.

Zhao, H. 2016. One Belt One Road and China-ASEAN Economic Cooperation. *IPP Review*. [Online]. [Accessed 21 September 2017]. Available from: <u>http://ippreview.com/</u>

Zuluaga, D. 2017. A Post-Brexit Framework for Electricity Markets. Institute of Economic Affairs. [Online]. [Accessed 15 December 2017]. Available from: <u>https://iea.org.uk/</u>

## Appendix A Interview Codes

			Meeting	
Code	Date(s)	Subject Location	type	Notes
Interview subjects				
	16/01/2018,			
	13/03/2018,			
	21/08/2018,			
Former Government	17/09/2018,			Background
Informant 1	09/10/2018	Washington, D.C., USA	Virtual	interview 3
Multilateral Organisation	19/07/2018,			Background
Informant 2	03/08/2018	Jakarta, Indonesia	Virtual	interview 1
Business 3	17/09/2018	Washington, D.C., USA	In person	-
Multilateral Organisation 4	06/06/2018	Manila, Philippines	In person	-
Think Tank 5	17/09/2018	Washington, D.C., USA	In person	-
Academic 6	18/10/2019	Leeds, United Kingdom	In person	-
	18/01/2018,			
Think Tank 7	03/02/2018	Washington, D.C., USA	In person	-
Development Bank 8	06/05/2018	Jakarta, Indonesia	In person	-
Business 9	19/09/2018	Washington, D.C., USA	In person	-
Academic 10	28/01/2018	Jakarta, Indonesia	Virtual	-
Think Tank 11	21/09/2018	Washington, D.C., USA	In person	-
Think Tank 12	10/03/2018	Washington, D.C., USA	In person	-
Academic 13	28/08/2019	London, United Kingdom	In person	-
Academic 14	07/03/2018	London, United Kingdom	In person	-
Think Tank 15	19/09/2018	Washington, D.C., USA	In person	-
Former Government 16	20/09/2018	Washington, D.C., USA	In person	-
Former Government 17	21/09/2018	Washington, D.C., USA	In person	-
Multilateral Organisation 18	04/11/2018	Jakarta, Indonesia	Virtual	-
Business 19	04/10/2018	Hong Kong, People's Republic of China	Virtual	-
				Background
Academic 20	17/09/2017	Leeds, United Kingdom	In person	interview 4
Academic 21	22/03/2018	Jakarta, Indonesia	Virtual	-
Business 22	17/09/2018	Washington, D.C., USA	In person	-
Multilateral Organisation 23	12/01/2018	Paris, France	In person	-
Academic 24	24/09/2018	Washington, D.C., USA	In person	-
	16/01/2018,			
Development Bank 25	23/05/2018	Washington, D.C., USA	Virtual	-
	12/01/2018,			Background
Academic 26	15/03/2018	Jakarta, Indonesia	Virtual	interview 2
Former Government 27	18/09/2018	Washington, D.C., USA	In person	-
Former Government 28	20/09/2018	Washington, D.C., USA	In person	-
Think Tank 29	17/09/2018	Washington, D.C., USA	In person	-
Journalist 30	06/05/2018	Tokyo, Japan	Virtual	-
Business 31	02/11/2019	Leeds, United Kingdom	In person	-
Academic 32	11/01/2019	Beijing, People's Republic of China	Virtual	-
<b>Conferences &amp; Workshops</b>				
Academic Conference 1	07/01/2016	Reading, United Kingdom	In person	-
Academic Conference 2	23/06/2016	Duisburg, Germany	In person	-
Academic Conference 3	30/09/2016	Tokyo, Japan	In person	-
Development Bank				
Conference 4	06/05/2018	Manila, Philippines	In person	-
Academic Conference 5	20/06/2018	York, United Kingdom	In person	-
Academic Workshop 6	09/07/2018	Leeds, United Kingdom	In person	-
Policy Conference 7	17/12/2018	Washington, D.C., USA	In person	-
Policy Conference 8	20/11/2018	Berlin, Germany	In person	-
Policy Conference 9	26/09/2018	Washington, D.C., USA	In person	-
Academic Workshop 10	22/03/2019	Oxford, United Kingdom	Virtual	-
Academic Conference 11	27/06/2019	Copenhagen, Denmark	In person	-
Government Conference 12	24/10/2019	Seoul, Republic of Korea	In person	-
Government Conference 13	25/10/2019	Seoul, Republic of Korea	In person	-
Government Workshop 14	25/10/2019	Seoul, Republic of Korea	In person	-
## Appendix B Ethical Approval

Performance, Governance and Operations Research & Innovation Service Charles Thackrah Building 101 Clarendon Road Leeds LS2 9LJ Tel: 0113 343 4873 Email: <u>ResearchEthics@leeds.ac.uk</u>

UNIVERSITY OF LEEDS

Clare Richardson-Barlow East Asian Studies School of Languages, Cultures and Societies University of Leeds Leeds, LS2 9JT

#### Faculty of Arts, Humanities and Cultures Research Ethics Committee University of Leeds

5 August 2016

Dear Clare

Title of studyTrade Governance of Renewable Energy in the Asia-PacificEthics referencePVAR 15-098

I am pleased to inform you that the above research application has been reviewed by the Arts and PVAC (PVAR) Faculty Research Ethics Committee and I can confirm a favourable ethical opinion as of the date of this letter. The following documentation was considered:

Document	Version	Date
PVAR 15-098 CGRB_NewEthicalReviewForm_Signed_v1_15072016.pdf	1	15/07/16
PVAR 15-098 CGRB_EthicsSupportingDoc1_InformationSheet_v1_15072016.pdf	1	15/07/16
PVAR 15-098 CGRB_EthicsSupportingDoc2_ConsentForm_v1_15072016.pdf	1	15/07/16
PVAR 15-098 CGRB_EthicsSupportingDoc3_RecruitmentMaterial_v1_15072016.pdf	1	15/07/16
PVAR 15-098 CGRB_EthicsSupportingDoc4_GatekeeperLetter_v1_15072016.pdf	1	15/07/16
PVAR 15-098 CGRB_EthicsSupportingDoc6_HealthandSafetyRiskAssessment_ lowrisk_v1_15072016.pdf	1	15/07/16
PVAR 15-098 CGRB_EthicsSupportingDoc7_HealthandSafetyRiskAssessment_ Appendix1_v1_15072016.pdf	1	15/07/16
PVAR 15-098 CGRB_EthicsSupportingDoc8_DataManagementPlan_v1_15072016.pdf	1	15/07/16
PVAR 15-098 CGRB_EthicsSupportingDocs5_InterviewQuestions_v1_15072016pdf	1	15/07/16

Committee members made the following comments about your application:

General comments			
This appears to be a very sound application, outlining a strong and well thought- through research project. Good luck with your fieldwork.			
Application section	Comment		
6.2	It is assumed that the gift will fall within the requirements of the <u>Protocol for reimbursement of research participants</u> . It is important to ensure that the gift will not interfere with the voluntariness of the participants' consent by acting as an undue inducement.		

	Risk assessment	Your fieldwork risk assessment will need to be signed off by one of your supervisors.	
Please notify the committee if you intend to make any amendments to the information in your ethics application as submitted at date of this approval as all changes must receive ethical approval prior to implementation. The amendment form is available at <u>http://ris.leeds.ac.uk/EthicsAmendment</u> .			
Please note: You are expected to keep a record of all your approved documentation, as well as documents such as sample consent forms, and other documents relating to the study. This should be kept in your study file, which should be readily available for audit purposes. You will be given a two week notice period if your project is to be audited. There is a checklist listing examples of documents to be kept which is available at <a href="http://ris.leeds.ac.uk/EthicsAudits">http://ris.leeds.ac.uk/EthicsAudits</a> .			
We welcome feedback on your experience of the ethical review process and suggestions for improvement. Please email any comments to <u>ResearchEthics@leeds.ac.uk</u> .			
Yours sincerely			
Jennifer Blaikie Senior Research Ethics Administrator, Research & Innovation Service On behalf of Dr Kevin Macnish, Chair, <u>PVAR FREC</u>			
	CC: Student's supervisor(s)		

### Appendix C Interview Protocol

The following interview protocol includes a research introduction and background provided to interview subjects, as well as sample interview questions. Questions varied based on interview subjects and progression of interview and not all prompts or questions were used.

### **Contextual Protocol: Research Objectives**

The variability of electricity produced from clean energy resources provides an opportunity for regional and international electricity market expansion and the efficient disposal and purchase of power capacity. Based on the resulting potential for cross border electricity trade, this research examines the expansion of these markets in Asia and opportunities for their further development. A number of issues play a role in this research: including national, sub-regional and regional electricity market development, sub-regional governance, interconnectors, and trade governance. A case study examination of 1 or 2 potential growth markets in East Asia will be conducted following informative expert interviews.

### **Introductory Protocol**

You have agreed to take part in a research study entitled 'The International Political Economy of Electricity Markets: Cross-Border Electricity Trade & Interconnections in East Asia.' The following information is for your convenience and understanding, but please do not hesitate to ask for clarifications or more details should you need to.

This interview will take between 30-60 minutes. During this time, I have several questions that I would like to cover. Follow up via email may be applicable should we run out of time or be interrupted. I may ask you to provide information, recount events, or describe your experiences and understanding about issues of importance to trade and economy policy, electricity markets or clean energy utilization. With your permission, the interview will be recorded in digital audio and subsequently transcribed. Once the transcript is finished, the voice recording will be disposed of.

Participants will remain anonymous unless they explicitly wish to be named in the research. If you prefer anonymity, the data will contain no personal information. With your permission, I may include information on your occupation but this requires your explicit approval and the interview does not depend on it. The data collected during this study may be used in presentation at conferences or in publications. However, all anonymity will be preserved.

You have the right to omit or refuse to respond to any question that I may ask. You will be given the right to withdraw at any point up to October 1, 2018. You also have the right to ask that any data you have supplied to me during the interview be withdrawn or destroyed. If you wish to withdraw from this study, let me know by phone or email at any time. My contact information is listed on the release form.

Please sign the attached release form to demonstrate agreement with these stipulations. This document states that: (1) all information will be held confidential, (2) your participation is voluntary and you may stop at any time. Thank you in advance for agreeing to participate.

### **Background Protocol**

- 1. How long have you been:
  - a. Working in your current positions?
  - b. At this institution?
  - c. Retired?
- 2. What is your:
  - a. Highest degree?
  - b. Area of expertise?
- 3. Probes:
  - a. Years of relevant experience?
  - b. Other relevant qualifications?
  - c. Briefly describe your role (office, committee, organization, etc.) as it relates clean energy development / electricity trade / electricity market integration (if appropriate).
- 4. Probes:
  - a. How are you involved in policy making/coordination/responses here?

- b. How did you get involved?
- c. What motivates your organization?
- 5. What are some of the current [research] projects you're conducting?

# Conversational or Transition Questions: (*Pick and choose based on situation*)

- 1. In recent years, we've seen numerous positive trends in policy towards the transition to clean energy. Is the speed of change fast enough to meet the climate and efficiency targets?
- 2. If we look at the East Asia/ European Union, what have been the most important legislative initiatives in favor of the energy transition?
  - a. Probe: Do you know of specific initiatives geared toward electricity market integration or electricity trade expansion?
- 3. What is the role of trade policy in climate change adaptation?
- 4. What trade policy and/or theoretical approaches can make the best case for optimising clean energy's contribution in addressing climate change and energy security challenges?

### **Key Questions: Electricity Markets:**

- 1. What is your understanding of cross-border electricity trade and/or electricity market integration in reaching global emissions reduction targets?
- 2. What is the role of trade policy in the growth of cross border electricity trade?
  - a. In what ways can governments incentivize or enable...
  - b. Cross border electricity trade?
  - c. Interconnector development/expansion?
- 3. What is changing about international and national policy initiatives in electricity market integration?
  - a. Probe: What is being accomplished through these initiatives?
- 4. What kinds of networks do you see developing around electricity market integration and cross border electricity trade?

- 5. How might current trade policies be put into effective practice in order to support wider cross border electricity trade?
- 6. Do certain global governance structures support cross border electricity trade better than others?
  - a. Why or why not?

### Key Questions: Electricity Markets & East Asia / Europe:

- 1. Where is this issue most alive in East Asia?
  - a. In Europe?
- 2. What is the relationship between interconnectors and cross border electricity trade in East Asia?
- 3. What resources are available for improving electricity market integration in East Asia?
  - a. *Probe*: Do you see a widening of the circle of participants in East Asia?
- 4. What is the political and physical capacity for increased cross border electricity trade in East Asia?
  - a. In European economies?
- 5. What are the policy obstacles to the increased exchange of electricity across borders in East Asia?
  - a. In European economies?
- 6. How can stakeholders in East Asia partner with their European counterparts in order to accelerate the trade of electricity across borders within their respective regions?
- 7. Are you aware of trade disputes over cross border electricity trade, and what discernible political economic patterns may be observed concerning these disputes?

### **Key Questions: Assessment**

1. How do you assess progress in electricity market integration or trade expansion?

- a. *Probe*: What kinds of assessment techniques tell you the most about progress?
- 2. What kinds of assessment most accurately capture what is happening internationally?
- 3. How is the assessment of market integration/expansion of trade used to improve policy responses in your department/organization/country?

### Key Questions: Institution/Department/Discipline

- 1. What are of the major challenges some your institution/department/discipline faces in attempting to encourage trade policy development in electricity sectors/clean energy development/market integration?
- 2. What are the major opportunities?
  - a. Probes:
    - i. How can barriers be overcome?
    - ii. How can opportunities be maximized?
- 3. Have you or your colleagues encountered resistance to reforms in your institution/department/discipline?
  - a. In international forums?
- 4. What is the strategy at your institution/department/discipline for clean energy/trade policy/electricity market integration/interconnector expansion?
  - a. Probes:
    - i. Is it working why or why not?
    - ii. What is the ideal timetable or strategy for success?

# Concluding Questions / Post Interview Comments/Considerations